

Sheng, Jie (2024) *An exploration of the influence of online parasocial interaction and parasocial relationship on consumer behaviour.* PhD thesis.

https://theses.gla.ac.uk/84401/

Copyright and moral rights for this work are retained by the author

A copy can be downloaded for personal non-commercial research or study, without prior permission or charge

This work cannot be reproduced or quoted extensively from without first obtaining permission from the author

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the author

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given

Enlighten: Theses <u>https://theses.gla.ac.uk/</u> research-enlighten@glasgow.ac.uk



## An exploration of the influence of online parasocial interaction and parasocial relationship on consumer behaviour

by

Jie Sheng

Submitted in fulfilment of the requirements for the Degree of Doctor of Philosophy

Adam Smith Business School College of Social Sciences University of Glasgow May 2024

#### Abstract

This thesis advances the understanding of parasocial Interaction (PSI) and parasocial relationship (PSR), which are critical for understanding consumer behaviour in media environments. PSI refers to media users' perceptions of one-to-one communication with media characters during media exposure, while PSR refers to an individual's psychological attachment to media characters. Despite their significance in marketing research, conceptual ambiguities and inappropriate PSR measurement persist, overlooking PSR's dynamic nature and development process. Therefore, this thesis aims to address these gaps by differentiating between them, improving their conceptualisation, introducing a new PSR measurement tool, and exploring the PSR development process and its outcomes.

Conducted in six phases, this research began with Phase 1: a systematic literature review of 172 articles. It identified seven prerequisites that distinguish PSI from PSR and laid the foundations for a PSR development model. A sequential mixed-method approach followed, starting with Phase 2, which comprised qualitative interviews with 25 participants to understand PSR and the PSR development process. This groundwork led to the operationalisation of PSR through Phase 3, an expert survey (n=11), and Phases 4 and 5, which were two consumer surveys (n=205 and n=206), resulting in a valid PSR measurement tool. A final survey (n=405) validated the PSR development model and tested the research hypotheses.

The findings reveal that PSR is a multidimensional construct characterised by intimacy, maintenance, and real-life relationship intention dimensions. A PSR development process is established, including exposure initiation, experimentation, intensification, and integration, highlighting the roles of PSI, perceived value, and trust. Subsequently, PSR fosters consumer intention to support and brand attitudes, positively affecting purchase intentions.

This thesis has theoretical contributions and practical implications. Theoretically, it advances the conceptualisation of PSR, elucidating its complex and dynamic nature, and advances the operationalisation of PSR by creating a reliable and valid measurement instrument for PSR in online environments. It also advances and validates the PSR development process. Practically, it guides marketers and influencers in developing long-term relationships step-by-step and generates more effective influencer marketing strategies.

# Contents

Abstract	I
Contents	II
List of Tables	IX
List of Figures	XI
Acknowledgement	XII
Author's Declaration	XIII
Abbreviations	XIV
Chapter 1: Introduction	15
1.1 Research background	15
1.2 Research rationale	17
1.3 Research aim and objective	20
1.4 Research methodology	21
1.5 Contributions	23
1.6 Thesis structure	24
Chapter 2: Systematic literature review	27
2.1 Chapter introduction	27
2.2 Rationale for conducting a systematic literature review	
2.3 Systematic literature review method	
2.3.1 Eligibility criteria	
2.3.2 Information sources and search strategy	32
2.3.3 Data selection process	
2.4 Debates on identifying the main domain characteristics of PSI and PSR in the of environment.	
2.4.1 The five views of the main domain characteristics of PSI in the offline environment.	35
2.4.2 The four views of the main domain characteristics of PSR in the offline environment.	
2.4.3 The interchangeable use of PSI and PSR in the offline environment	
<ul> <li>2.5 Prerequisites of PSI and PSR in the offline environment</li></ul>	41 43 44 45 46 47
2.6 Debates on identifying the main domain characteristics of PSI and PSR in the or environment.	

2.6.1 The four views of the main domain characteristics of PSI in the online	40
environment	48
environment	51
2.6.3 The interchangeable use of PSI and PSR in the online environment	
2.7 Prerequisites of PSI and PSR in the online environment	54
2.7.1 Prerequisite 1: One of the parties participating in PSI/PSR	
2.7.2 Prerequisite 2: The other parties participating in PSI/PSR	
2.7.3 Prerequisite 3: The nature of PSI/PSR	58
2.7.4 Prerequisite 4: The main domain characteristics of PSI/PSR	
2.7.5 Prerequisite 5: The process of PSI/PSR	
2.7.6 Prerequisite 6: The time when PSI/PSR can occur	
2.7.7 Prerequisite 7: The platforms where PSI/PSR can occur	61
2.8 The difference between PSI and PSR in offline and online environments	61
2.8.1 Frequency of potential PSI	
2.8.2 Speed	
2.8.3 Convenience	
2.8.4 Time (continuity)	
<ul><li>2.8.5 Available ways of communication</li><li>2.8.6 Opportunities to become two-sided interaction/relationship</li></ul>	
2.8.6 Opportunities to become two-sided interaction/relationship 2.8.7 Variety of entities/actors that can become the other parties in PSI/PSR	
2.8.8 The volume of information about the character (richness)	
2.8.9 Expression function	
2.8.10 Geographical barriers	
2.8.11 Media Accessibility	68
2.8.12 Opportunities for PSI/PSR	69
2.9 Research gap identification	69
2.9.1 The conceptualisation of PSR	
Theme 1: One party	71
Theme 2: The other party	
Theme 3: Connection	
Theme 4: Perception	
Theme 5: Real relationship	
Theme 6: Inequality Theme 7: Psychological closeness	
Theme 8: Time	
Theme 9: Develop	
Theme 10: Illusion	74
2.9.2 The need to explore the PSR development process	
2.10 Chapter Summary	76
Chapter 3: Analytical approach	
3.1 Chapter introduction	
3.2 Research paradigm	78
3.3 Overall research design	
3.3.1 Phase 1: Systematic literature review	
3.3.2 Phase 2: semi-structured interview	
3.3.3 Phase 3: expert survey	
3.3.4 Phase 4: online consumer survey	
<ul><li>3.3.5 Phase 5: online consumer survey</li><li>3.3.6 Phase 6: online consumer survey</li></ul>	
5.5.0 I hase 0. online consumer survey	

3.4 Chapter Summary	89
Chapter 4: Qualitative methodology for exploring PSR	91
4.1 Chapter introduction	91
4.2 Interview guide	91
4.3 Sampling approach	93
4.4 Interview modality	95
4.5 Data collection procedure	96
4.6 Characteristics of Interviewees	96
4.7 Data analysis process	97
4.8 Possible bias	
4.9 Chapter Summary	101
Chapter 5: Qualitative findings on consumer insights into PSR	
5.1 Chapter introduction	
5.2 Consumer's perception of PSR	
5.2.1 Deductive content analysis	
Category 1: Connection	
Category 2: Psychological nature	
Category 3: Value Category 4: Influence	
Category 5: Inequality	
Category 6: Long-term	
5.2.2 Inductive thematic analysis	109
Theme 1: Time	
Theme 2: Psychological closeness	
Theme 3: Value	
Theme 4: Inequality Theme 5: influence	
Theme 6: Connection	
5.3 PSR development stories	
5.3.1 Stage 1: Exposure initiation stage	
Post appeal	
Influencer attractiveness	123
Viewers' various motivations	124
5.3.2 Stage 2: Experimentation stage	
Perceived value	
PSI Other influencer characteristics	
5.3.3 Stage 3: Intensification stage Trust	
Perceived influencer authenticity	
PSR	
5.3.4 Stage 4: Integration stage	
Influencer stickiness	138
Intention to support	140

IV

People's well-being Brand advocacy	
Brand advocacyBrand trust	
Brand preference	
Positive brand attitude Purchase intention	
Actual purchase behaviour	
5.3.5 PSR vs real-life relationship	146
5.3.6 Parasocial breakup	147
5.4 Conceptualisation of PSR	147
5.4.1 The good definition criteria of PSR	148
5.4.2 PSR working definition	149
5.5 Implications of the qualitative study	150
5.6 Chapter summary	151
Chapter 6: Conceptualising PSR formation process	152
6.1 Chapter introduction	152
6.2 Theoretical framework	153
6.2.1 Tukachinsky and Stever (2018) PSR model	153
6.2.2 AIDA model	156
6.3 Proposed model of PSR development process	159
6.3.1 Stage 1: Exposure initiation	
Involvement Viewers' motivation	
Influencer attractiveness	
Visual appeal	167
6.3.2 Stage 2: Experimentation	
PSI Perceived value	
6.3.3 Stage 3: Intensification	
PSR	
Trust	
6.3.4 Stage 4: Integration Purchase intention	
Brand attitude	
Intention to support	
6.4 Moderators and moderated relationships	181
6.4.1 Perceived expertise	181
6.4.2 Perceived authenticity	
6.4.3 Interaction frequency	184
6.5 Chapter summary	
Chapter 7: Quantitative methodologies for PSR model validation	
7.1 Chapter introduction	186

V

	VI
7.2 The tested model	186
7.3 Conceptualisation and operationalisation	187
7.4 Questionnaire design	190
7.4.1 Questionnaire structure	191
7.4.2 Pilot test	193
7.5 Sampling approach	193
7.6 Execution of the quantitative study 3	196
7.6.1 Pre-screen study	197
7.6.2 Main study	198
7.7 Data analysis techniques	198
7.7.1 Data cleaning	198
7.7.2 Descriptive analysis	200
7.7.3 Scale reliability and validity testing	200
7.7.4 Research hypotheses testing	202
7.8 Possible bias and the efforts to minimise bias	203
7.9 Chapter conclusion	204
Chapter 8: Measurements for PSR formation model	205
8.1 Chapter introduction	205
8.2 Constructs' operationalisation	206
8.2.1 PSI	206
8.2.2 Perceived value of posts	209
8.2.3 Perceived influencer expertise	209
8.2.4 Trust	210
8.2.5 Perceived influencer authenticity	210
8.2.6 Brand attitude	210
8.2.7 Purchase intention	211
8.2.8 Intention to support	211
8.2.9 Interaction frequency	211
8.3 Rationale for developing a new PSR measurement	212
8.4 PSR measurement development process	213
8.4.1 Step 1: Specify the main domain characteristics of PSR	214
8.4.2 Step 2: Generate initial scale item pool	214
8.4.3 Step 3: Initial scale items purification	216
Research phase 3: expert survey	
Amending the PSR definition Research phase 4: online consumer survey – quantitative study 1	
8.4.4 Scale validation and reliability (Research phase 5: quantitative study 2)	
8.5 Chapter summary	

V	Π
Chapter 9: Empirical findings of the PSR formation model23	7
9.1 Chapter introduction	7
9.2 Study rationale	7
9.3 Respondents' information	9
9.4 Conceptual model testing	9
9.4.1 Scale reliability and validity	0
Reliability	
Convergent validity	
9.4.2 Correlation analysis	
9.4.3 Path analysis	.7
Model 1	
Alternative model – Model 2	
9.4.4 Mediation analysis	
PSI – Perceived value – PSR	
PSR – Brand Attitude – Purchase Intention25	
PSR – Intention to Support – Purchase Intention	
9.4.5 Moderation analysis	
9.5 Post-hoc analysis	6
9.6 Chapter summary	7
Chapter 10: Discussion25	8
10.1 Chapter introduction	8
10.2 Research objective 1: The difference between PSI and PSR in the offline and online environments	8
10.3 Research objective 2: The new definitions of PSI and PSR based on their unique	
prerequisites	2
10.4 PSR measurement	
10.5 Research objective 3: PSR development process and its outcome	7
10.5.1 PSI and perceived value (H1)	9
10.5.2 Perceived value and PSR (H2)27	0
10.5.3 Perceived value and trust (H3)27	'1
10.5.4 Trust and PSR (H4)27	2
10.5.5 PSR and purchase intention (H5)	'3
10.5.6 PSR and brand attitude (H6)	4
10.5.7 Brand attitude and purchase intention (H7)	4
10.5.8 PSR and intention to support (H8)	5
10.5.9 Intention to support and purchase intention (H9)	
10.5.10 The moderating role of perceived expertise (H10)	
10.5.11 The moderating role of perceived authenticity (H11)	
10.5.12 The moderating role of interaction frequency (H11)	

	VIII
10.6 Additional observation	
10.7 Chapter summary	281
Chapter 11: Contributions and future research	282
11.1 Chapter introduction	
11.2 Theoretical contributions	282
11.3 Practical implications	
11.4 Limitations and future research directions	
Appendices	291
Appendix 1: PRISMA 2020 guided systematic literature review	291
Appendix 2: The reviewed PSI definitions in the offline environment	292
Appendix 3: The reviewed PSR definitions in the offline environment	296
Appendix 4: The reviewed PSI definitions in the online environment	299
Appendix 5: The reviewed PSR definitions in the online environment	319
Appendix 6: Antecedents of PSR	329
Appendix 7: Consequences of PSR	332
Appendix 8: Interview guide	336
Appendix 9: Participant information sheet	338
Appendix 10: Consent form	341
Appendix 11: Interviewee profile	342
Appendix 12: Thematic analysis of the interview	346
Appendix 13: Questionnaire	350
Appendix 14: Pre-screen study questions	358
Appendix 15: PSR scale items initial pool	359
Appendix 16: Experts survey	361
Appendix 17: Experts comments about PSR working definition	
Appendix 18: The summary of PSR scale items	
References	366

# List of Tables

Table 2.1 Inclusion and exclusion criteria of the systematic literature review	31
Table 2.2 Search terms in Scopus and EBSCO	32
Table 2.3 Main domain characteristics of PSI in the offline environment	36
Table 2.4 Main domain characteristics of PSR in the offline environment	39
Table 2.5 The use of the terms PSI and PSR in the offline PSI studies	40
Table 2.6 The use of the terms PSI and PSR in the offline PSR studies	40
Table 2.7 Prerequisites for distinguishing between PSI and PSR in the offline enviro	
Table 2.8 Main domain characteristics of PSI definition in the online environment	
Table 2.9 Main domain characteristics of PSR definition in the online environment	52
Table 2.10 The use of the terms PSI and PSR in the online PSI studies	53
Table 2.11 The use of the terms PSI and PSR in the online PSR studies	54
Table 2.12 Prerequisites for distinguishing PSI and PSR in the online environment	55
Table 4.1 Semi-structured interview justification	100
Table 5.1 Content analysis of the interview	104
Table 5.2 The key factors in the exposure initiation stage of the 25 interviewees	120
Table 5.3 Key factors in the experimentation stage of the 25 interviewees	127
Table 5.4 The key factors in the intensification stage of the 25 interviewees	133
Table 5.5 The key factors in the integration stage of the 25 interviewees	139
Table 5.6 Good PSR definition criteria	149
Table 6.1 Research hypotheses summary	182
Table 7.1 Constructs and their definitions	189
Table 7.2 The summary of the model fit indices	201
Table 7.3 The Summary of the reliability and validity test criteria	202
Table 8.1 Operationalisation of study constructs	207
Table 8.2 PSR scale development procedure	215
Table 8.3 Criteria for factor retention in EFA.	224
Table 8.4 Quantitative study 1 – CFA results	225
Table 8.5 Quantitative study 1 – HTMT Matrix	226
Table 8.6 Quantitative study 2: PSR CFA results	229
Table 8.7 Quantitative study 2 – PSR HTMT Matrix	230
Table 8.8 Quantitative study 2: Nomological model CFA results	231
Table 8.9 Quantitative study 2: Nomological model HTMT Matrix	233
Table 8.10 Quantitative study 2: Nomological network – Path analysis	234
Table 8.11 Quantitative study 2: Test-retest 79 responses	234
Table 9.1 Quantitative study 3 CFA	241

	Х
Table 9.2 Quantitative study 3 - HTMT Matrix	246
Table 9.3 Quantitative study 3 – Correlation Analysis	247
Table 9.4 Quantitative study 3 – Model 1 – Path analysis results	249
Table 9.5 Quantitative study 3 – Model 2 – Path analysis results	251
Table 9.6 Structural and Moderation Test Results (Mixed Model Method) - H10&H	11254
Table 9.7 Structural and Moderation Test Results (Mixed Model Method) – H12	255
Table 10.1 The summary table of prerequisites to distinct PSI and PSR in the offline         online environments	
Table 10.2 The summary table of research hypothesis testing results	268

# List of Figures

Figure 1.1 The trend of publishing PSI and PSR research based on the ABDC journal ranking list	18
Figure 2.1 The process of selection of reviewed articles	33
Figure 3.1 Analytical framework	85
Figure 6.1 Conceptual framework	160
Figure 7.1 The empirically tested model	188
Figure 7.2 The A Priori Procedure Equation	195
Figure 8.1 Nomological network in quantitative study 2	.227
Figure 9.1 Model 1	238
Figure 9.2 Model 1 in AMOS	248
Figure 9.3 Model 2 in AMOS	250
Figure 9.4 The plot of the moderating effect of perceived expertise on the relationship between PSI and perceived value	.254
Figure 9.5 The plot of the moderating effect of perceived authenticity on the relationship between perceived value and trust	р .255
Figure 9.6 The plot of the moderating effect of interaction frequency on the relationship between perceived value and PSR	) .256

#### Acknowledgement

"A journey of a thousand miles begins with a single step" - Lao Tzu.

Reflecting on my doctoral journey, no ancient wisdom resonates more profoundly than this. This journey, enriched by the unwavering support, encouragement, and companionship of many, has culminated in the successful completion of this thesis.

I extend my deepest gratitude to my parents for their unconditional love. My mother's optimism has been a guiding light in moments of doubt, urging me to move forward. My father's trust and support have been my stronghold. I am also deeply thankful for the companionship, support, and encouragement from my boyfriend, Jiheng Wang, whose presence has been a source of immeasurable courage.

My supervisor, Dr. Alena Kostyk, deserves special acknowledgement. Her exceptional skill as a professor, mentor, and researcher, combined with her guidance, care, support, patience, and encouragement, has been instrumental in navigating the complexities of this journey. I consider myself extremely fortunate to have been supervised by her.

I am equally grateful to my supervisor Dr. Kalliopi Chatzipanagiotou, for her professional advice and support during key-decision making moments. Her expertise and calm guidance have been invaluable in guiding me toward the completion of my PhD thesis.

I am also grateful to Professor Cleopatra Veloutsou for her support and trust, along with insightful and forward-thinking advice at critical junctures. I am also thankful to Dr. Kirsten Cowan (University of Edinburgh) and Dr. Parichehr Riahi Pour for their care and support.

My thanks also go to my viva examiners, Professor Nina Michaelidou and Dr. Amy Goode, for their professional suggestions, and to the viva convener Dr. Jaylan Azer for her organisation. I am also thankful for my colleagues and my friends in China, who have been outstanding. A special thanks to influencers that I follow, such as Sulong Wang, a passionate source of inspiration for my research on the parasocial phenomenon.

Lastly, I must express gratitude to myself for enduring this challenging and extensive journey. With the support and encouragement from everyone around me, I have fulfilled my dream of becoming an independent researcher.

## **Author's Declaration**

I declare that, except where explicit reference is made to the contribution of others, this dissertation is the result of my own work and has not been submitted for any other degree at the University of Glasgow or any other institution.

Signature:

Print name: Jie Sheng

### Abbreviations

- ABDC Australian Business Deans Council
- AIDA Attention, Interest, Desire and Action
- AVE Average Variance Extracted
- BCa Bias-corrected and Accelerated
- CFA Confirmatory Factor Analysis
- CFI Comparative Fit Index
- CR Composite Reliability
- DF Degrees of Freedom
- E-WOM Electronic Word-of-Mouth
- EFA Exploratory Factor Analysis
- HARKing Hypothesising After the Results are Known
- HSM Heuristic-Systematic Model
- HTMT Heterotrait-Monotrait Ratio
- ICC Intraclass Correlation Coefficients
- IMC Instructional Manipulation Check
- KMO Kaiser-Meyer-Olkin
- NFI Incremental Fit Index
- PRISMA Preferred Reporting Items for Systematic Reviews and Meta-Analyses
- PSB Parasocial Bond
- PSI Parasocial Interaction
- PSR Parasocial Relationship
- r Pearson Correlation Coefficient
- RMSEA Root Mean Square Error of Approximation
- SDT Self-Determination Theory
- SEM Structural Equation Modelling
- TLI Tucker-Lewis Index
- U&G Uses and Gratifications Theory
- WOM Word-of-Mouth

### Chapter 1: Introduction

#### 1.1 Research background

Parasocial interaction (PSI) and parasocial relationship (PSR) have been integral to marketing research for nearly three decades. Initially introduced by Horton and Wohl in 1956 in the field of psychology, PSI and PSR aimed to understand the interaction and relationship between television viewers and hosts. Horton and Wohl (1956) observe that viewers often respond to a TV host's farewell as if engaged in a direct conversation and perceived a conversational give-and-take feeling, a phenomenon they termed PSI. PSR, on the other hand, refers to the illusion of face-to-face relationship viewers develop with television personalities, further extending the understanding of media consumption from the audience's viewpoint (Horton and Wohl, 1956).

These groundbreaking concepts provide a nuanced understanding of media consumption from the viewer's standpoint, enriching existing theories on social interaction and relationship within media-mediated environments (Liebers and Schramm, 2019; Tukachinsky et al., 2020). Subsequent research has extended the application of PSI and PSR across various research disciplines, including media science (Tsfati et al., 2022), psychology (Balaban et al., 2022), tourism (Ma et al., 2023), management (Hu et al., 2020), marketing (Xie et al., 2023) and interdisciplinary studies (Knupfer et al., 2023), exploring their implications in both traditional and digital media environments.

Initial studies within traditional media explored the impact of PSI and PSR on phenomena such as TV shopping experiences (Stephens et al., 1996), product placements in soap operas (Bernhold and Metzger, 2020), and consumer donation behaviour (Bentley, 2014). These findings indicate that PSI and PSR with media characters can influence consumer behaviour positively (Knoll et al., 2015; Bi et al., 2021). For instance, Lim and Kim (2011) discover that loneliness predisposes consumers to form PSI with TV shopping hosts, enhancing their shopping satisfaction. PSR with soap opera characters has been shown to enhance attitudes towards embedded products (Dias et al., 2017).

In the digital age, there are new kinds of influencers beyond TV hosts and soap opera characters, such as social media influencers. Social media influencers refer to people who actively gain other users' attention on online platforms and can influence their followers'

behaviour in a particular field by generating texts, images, videos, and live streaming (Sokolova and Perez, 2021). Influencers act as intermediaries between consumers and brands and significantly influence consumer behaviour and decision-making (Sakib et al., 2020). Previously, consumers relied on direct interaction and brand communication to get information, such as through advertising (de Vries et al., 2017). Consumers now turn to these influencers, perceived as friends and field experts, for authentic and credible information, moving away from traditional direct interactions and brand communications (Kapitan et al., 2021; Yousaf, 2022). Consequently, brands view these influencers as essential long-term partners, leading to complex relationships encompassing consumers, brands, and influencers (Kim and Kim, 2021).

Influencer marketing has become an integral part of digital marketing strategies (Kim, 2022). According to statistics, the global influencer marketing size reached \$21.1 billion in 2023 and projected to grow at a 33% CAGR (Dencheva, 2024; Grand View Research, 2023). Marketing managers regard influencer marketing as highly effective, significantly influencing multiple stages of the marketing funnel, from boosting brand awareness to driving sales with varied and engaging content (Traackr, 2024). Employing these strategies can result in extensive reach, enhanced customer engagement, and interactive communication (Traackr, 2024).

Academic research supports that collaboration with influencers helps brands achieve their marketing and marketing communication objectives, such as improving brand attitude (Choi et al., 2023), attitude towards advertising (Vrontis et al., 2021), and purchase intention (Ashraf et al., 2023). PSI and PSR have been identified as key factors in the success of influencer marketing, influencing consumers' perceptions and their purchasing decisions (Gong and Holiday, 2023; Masuda et al., 2022). For example, Gong and Holiday (2023) find that consumers high levels of PSI can significantly enhance their perceptions of authenticity and their intention to purchase. Furthermore, when viewers develop PSR with influencers, this tends to elevate their attitude towards the brand, as well as their trust in the brand and their intention to purchase the products recommended by the influencer (Reinikainen et al., 2021).

Understanding the roles of PSI and PSR and their impact on consumer behaviour and decision-making is crucial, as these phenomena have become as integral to daily media consumptions as traditional social and interpersonal activities (Fazli-Salehi et al., 2022; Leite and Baptista, 2022). Despite some academic contention over the prevalence of PSI and PSR

in everyday life, numerous studies have confirmed their significance as psychological states that commonly arise during and after media use (Noor et al., 2022; Lo et al., 2022). Consumers frequently experience PSI during media consumption, feeling engaged in a two-way communication (Oliver et al., 2019; Gong and Holiday, 2023). People have established PSR with social media influencers across a range of fields, such as lifestyle (Sheng et al., 2023), fitness (Kim et al., 2022) and gaming (Zhang et al., 2023), further highlighting its significance.

#### 1.2 Research rationale

When PSI and PSR were introduced to the marketing field, scholars initially treated these two concepts as well-established, overlooking the conceptual disputes between them. As a result, despite existing studies highlighting the significance of PSI and PSR in marketing, several conceptual issues remain unresolved. One primary issue is the lack of clarity regarding the main domain characteristics of PSI and PSR in both online and offline environments (Lu et al., 2023). A significant number of studies on PSI suggest that its main domain characteristic is "*relationship*", rendering it indistinguishable from PSR. For instance, Aw and Labrecque (2020) define PSI as "*one-sided relationships that individuals form with mediated personae, including celebrities*", emphasising "*relationship*" as the main domain characteristic. Furthermore, some scholars, such as So and Shen (2016) and Penttinen et al. (2022), argue that PSI and PSR represent the same concept, though this stance lacks robust evidence. Contrarily, a meta-analysis by Tukachinsky et al. (2020) indicates that PSI and PSR are related yet conceptually distinct. Supporting this differentiation, a recent empirical study by Lu et al. (2023) identifies PSI as an antecedent to PSR, further validating the distinction between the two concepts.

The inconsistency in definitions has led to the interchangeable use of PSI and PSR by many scholars (e.g., Liao et al., 2022; Conde and Casais, 2023), resulting in conceptual ambiguity. This lack of precision in defining constructs can lead to incorrect operationalisation and, consequently, biased research outcomes violating the criteria for a good definition (MacKenzie, 2003; Wacker, 2004). For instance, certain studies on PSI have employed measurement tools designed for PSR (e.g., Chen et al., 2022; Olfat et al., 2023), casting doubt on the validity of their empirical findings and the theories subsequently developed (MacKenzie, 2003; Cummins and Cui, 2014). Although attempts have been made to differentiate PSI from PSR, this debate remains underexplored, leaving the issues largely

unresolved (Hartmann and Goldhoorn, 2011; Lu et al., 2023). Several researchers have highlighted the differences between PSI and PSR to rationalise their choice of concept within particular study contexts, yet they have not fundamentally addressed this issue (e.g., Zafar et al., 2020; Utz et al., 2021).

Systematic literature reviews by Giles (2002) and Liebers and Schramm (2019) have provided valuable insights into PSI and PSR from psychological and media science perspectives, respectively. However, both reviews exhibit certain gaps and limitations in their coverage and analysis. Giles (2002) proposes four models of PSI formation and a conceptual framework for future research, but this review, being over two decades old, does not account for the dramatic shifts in media consumption towards social media and away from traditional media (Bruhn et al., 2012; Liu et al., 2018). Liebers and Schramm (2019) conduct a comprehensive literature review on PSI and PSR, encompassing 261 articles spanning from 1957 to 2015. Their systematic analysis covers the research contexts, methodologies, antecedents, and outcomes of PSI and PSR within the field of media science. Nevertheless, the focus of their review is primarily on the contemporary state of research on parasocial phenomena, without delving into the underlying conceptual challenges. Moreover, since 2015, the body of research on PSI and PSR has significantly expanded (see Figure 1.1). with over 150 studies published post-2016 not covered by the existing reviews. This surge in literature underscores the value of revisiting these topics through updated scientific frameworks (Paul et al., 2021; Paul et al., 2023).

# Figure 1.1 The trend of publishing PSI and PSR research based on the ABDC journal ranking list



Based on the discussion above, there is considerable research on PSI and PSR, yet efforts to clearly differentiate between the two are limited, leaving the issue largely unresolved (e.g., Utz et al., 2021; Zafar et al., 2020). This ambiguity underscores the need to delineate the relationship between PSI and PSR more clearly, and to establish distinct prerequisites for distinguishing between them. Consequently, the first research objective of this thesis is to identify the differences between PSI and PSR in both offline and online environments.

In addition, existing conceptualisations overlook the differences between offline and online environments. While PSI and PSR were originally conceptualised within the context of traditional media (Horton and Wohl, 1956), subsequent research has exemplified their presence in digital environments (Noor et al., 2022). The digital age extended PSI and PSR to include social media (Leite and Baptista, 2022; Burnasheva and Suh, 2022), AI (Dhiman et al., 2023), chatbots (Li and Wang, 2023; Youn and Cho, 2023), and more. This transition from offline to online platforms has not only broadened the conceptual scope of PSI and PSR but also introduced variations in their conditions, reflecting the fundamental differences between traditional and digital media landscapes. These changes mandate further investigations into PSI and PSR in both the online and offline environments.

Additionally, existing studies regard PSR as a static concept, implicitly assuming that the quality, strength, and stage of PSR with the mediated persona are consistent (Tukachinsky and Stever, 2018). However, in reality, individuals navigate through stages of forming, maintaining, and dissolving relationships, indicating that relationship development is a dynamic process with varying levels of strength. Thus, treating PSR as a static concept does not adequately capture its complexity and diversity (Tukachinsky and Stever, 2018). Hence, due to the above conceptual and contextual ambiguities, new and distinct conceptualisations and definitions of PSI and PSR are necessary (Gilliam and Voss, 2013; Murrow and Hyman, 1994) to facilitate new knowledge development on a more solid theoretical basis (Hempel, 1952; Gilliam and Voss, 2013). Therefore, the second research objective is to propose definitions of PSI and PSR based on their unique prerequisites.

The disagreement over conceptualisation between PSI and PSR has resulted in the use of inconsistent measurement scales that do not match their respective conceptual foundations. For example, Hudders and De Jans (2022) define PSI as "*one-sided mediated form of social interaction between audience and media characters*", highlighting its interactive essence. Yet, they employ PSR scales for measurement, which contradicts their conceptualisation. Similarly, Hsieh and Lee (2021) define PSI as a conversational give-and-take between users

and media performers but also resort to PSR scales for assessment. This misalignment is exacerbated by the existing scales' failure to clearly differentiate between PSI and PSR. The commonly utilised scales for PSI and PSR, developed by Rubin et al. (1985) and Rubin and Perse (1987), characterise the main domain feature of PSR as "*involvement*", a broad term that encompasses a range of psychological processes including transportation, PSI, identification, and worship (Brown, 2015). Therefore, these existing scales perpetuate the inconsistency between the conceptualisation of PSI and PSR and the scales chosen for their measurement.

Moreover, while the existing literature extensively explores the antecedents and consequences of PSR, the process of PSR development remains ambiguously outlined. For instance, Lim et al. (2020) demonstrate that individuals' wishful identification and emotional engagement significantly foster PSR, which in turn enhances consumers' behavioural loyalty. Leite and Baptista (2022) find that intimate self-disclosure heightens consumers' PSR with female beauty influencers, leading to increased brand trust and purchase intention. Additionally, influencers' attitudes, homophily, physical attractiveness, and social attractiveness are shown to positively influence consumers' PSR with YouTubers, subsequently boosting their purchase intention (Masuda et al., 2022). Despite these insights, the underlying mechanisms and the progression of PSR development remain inadequately understood.

A handful of scholars suggest that repeated PSI experiences with a persona are prerequisite for developing PSR (Bernhold and Metzger, 2020). This notion is in line with theories of social interaction and social relationships, yet it largely remains conceptual and is scantily supported by empirical evidence. In light of this, this thesis adopts the perspective of PSR as a multifaceted and dynamic process to explore the PSR development process. Based on the discussion above, an exploration of the mechanisms of PSR development is needed to advance the knowledge of PSR and the outcomes of PSR. Therefore, the third research objective is proposed to explore the formation process of PSR and test its outcomes.

#### 1.3 Research aim and objective

To address the research gaps highlighted above, this thesis aims to resolve the long-standing definitional debate and propose prerequisites to identify PSI and PSR in the offline and online environment, respectively. Moreover, this research aims to advance the understanding

of PSR as a dynamic process, scientifically measure PSR, and investigate the PSR development process and its relevant outcomes. To fulfil these aims, the thesis outlines the following three research objectives:

- 1. To identify differences between PSI and PSR in the offline and online environments
- 2. To propose definitions of PSR and PSI based on their unique prerequisites
- 3. To explore the formation process of PSR and test its outcomes

Following this, this thesis aims to answer the following research questions:

RQ1: Should PSI and PSR be treated as distinct concepts?

RQ2: How does the complexity of the online environment further inform the conceptualisation of PSI and PSR?

RQ3: How does PSR with influencers on social media platforms develop, and what are the outcomes of PSR?

#### 1.4 Research methodology

The philosophical stance of this research is rooted in post-positivism, guiding its methodological approach. Ontologically, the thesis adopts the realist view, acknowledging that while reality exists, it cannot be precisely captured (Zinkhan and Hirschheim, 1992). From an epistemology perspective, this thesis employs the view of modified objectivity and asserts that researchers should objectively capture and measure the phenomena (Hunt, 1993). Nevertheless, researchers can never be entirely objective because of the bias. However, it is crucial for researchers to aim for objectivity and minimise bias to approach the truth as closely as possible (Hunt, 1993). To enhance validity and reduce bias, the research methodology involves a meticulous process of hypothesis formulation, data collection, and rigorous analysis and review of findings (Ryan, 2006). Consistent with post-positivist, a mixed-methods approach, incorporating both qualitative and quantitative techniques, is employed to thoroughly investigate the PSR phenomenon (Johnson et al., 2007). The research commences with an exploratory qualitative phase to delve into PSR, followed by quantitative phases aimed at developing a new PSR measure, generalising qualitative insights, and testing the PSR development model (Harrison and Reilly, 2011).

First, Phase 1 of this thesis is a systematic literature review guided by the Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA) protocol to address the first research objective. A systematic literature review is a historical method that helps to review the debates about PSI and PSR from 1956 to 2022 in the fields of media science, psychology, management, marketing, social science and interdisciplinary fields to inspire further and solve the conceptual debates (Hunt, 2011; Torraco, 2005). Based on the systematic analysis and the views from existing literature, this thesis 1) argues that PSI and PSR are two similar but conceptually different concepts, 2) proposes seven prerequisites that can distinguish PSI and PSR in both offline and online environments, 3) identifies twelve aspects that PSI and PSR differ between offline and online environment and 4) propose PSI and PSR working definitions.

Then Phase 2 of this thesis is a qualitative study utilising semi-structured interviews to achieve Research Objectives 2 and 3. It is beneficial to understand the nature of PSR deeply through qualitative study, how people generate and develop PSR with the other party and confirm the conceptual model. Meanwhile, conducting qualitative research is an adequate approach to developing a new scale because it can contribute to the initial item creation (Hinkin, 1995; Carpenter, 2017). The semi-structured interview can generate a more relevant foundation for the initial item pool (Churchill, 1979). This study used snowball sampling and recruited 25 respondents to participate. The qualitative data is analysed by using content analysis and thematic analysis.

Next, Phase 3 comprises an expert survey, and Phase 4 involves a quantitative study (Quantitative Study 1). These two phases can address Research Objectives 2 and 3, aiming to reduce scale items, explore the dimensionality of the PSR scale, and assess the PSR scale's reliability and validity. In Phase 4, an online survey method is employed, and the questionnaire is designed on Qualtrics to collect data via Prolific. This study collected 220 responses and obtained 205 valid responses for data analysis. Utilising exploratory factor analysis through SPSS (28.0) and confirmatory factor analysis through SPSS AMOS (28.0), the study confirms that PSR comprises three dimensions and retained eight scale items.

Subsequently, Phase 5, Quantitative Study 2, is undertaken to achieve Research Objectives 2 and 3, aiming to refine and test the reliability and validity of the newly developed PSR scale. Phase 5 also used the online survey method by developing a questionnaire on Qualtrics to collect data via Prolific. This study collected 220 responses, out of which 206 were deemed valid for data analysis. The reliability and validity of the PSR scale were established through the use of confirmatory factor analysis, path analysis, and test-retest, all conducted via SPSS (28.0) and SPSS AMOS (28.0).

Finally, Phase 6 of this thesis, Quantitative Study 3, aims to generalise the findings from the qualitative research and empirically test the PSR development process model, addressing Research Objective 3. This study involves designing a questionnaire on Qualtrics to gather data through Prolific, resulting in 454 responses, with 405 considered valid for data analysis. The PSR development model is tested using confirmatory factor analysis, path analysis, mediation analysis, and moderation analysis, conducted through SPSS (28.0) and SPSS AMOS (28.0).

#### **1.5 Contributions**

This thesis aims to make significant contributions in two areas: academic contributions and practical implications. From an academic perspective, first, this thesis advances the conceptualisation of PSR by emphasising its complex and dynamic nature. It broadens the application of PSR, contributing significantly to the literature across fields such as media, psychology, management, marketing, social science, and interdisciplinary studies. Second, this thesis distinguishes between PSI and PSR, addressing the need for studies on this distinction as highlighted by scholars like Hartmann and Goldhoorn (2011) and Lu et al. (2023). It proposes seven prerequisites for differentiating PSI and PSR in both offline and online contexts. These prerequisites are instrumental in conceptually separating PSI from PSR, thereby resolving the confusion and interchangeable use of these terms among academics.

Third, by analysing the transition of PSI and PSR from traditional to online environments, this thesis enhances our understanding of the conditions under which these concepts operate in various settings. This offers a systematic understanding of the key attributes of PSI and PSR across multiple media environments. Fourth, this thesis advances the operationalisation of PSR in the online environment. A new PSR measurement instrument, developed based on the new conceptualisation of PSR, comprises eight items across three dimensions, capturing the main domain characteristics of PSR in the online environment. This initiative addresses the need for PSR scales in the online context, as existing scales are primarily grounded in the offline environment (Kim, 2022). The development of this new, valid, and reliable PSR measurement instrument promises to support future empirical studies requiring a refined measure of PSR in the online environment.

Fifth, this thesis enriches the existing literature by exploring the complex process of PSR development, assessing its formation process and key outcomes. Previous research has investigated the antecedents and consequences of PSR in different contexts, such as social media marketing (Labrecque, 2014) and AI marketing (Lee and Cho, 2020). Focusing on PSR within influencer marketing, this thesis modifies the conceptual model of PSR development stages proposed by Tukachinsky and Stever (2018), identifying crucial concepts at each stage of PSR development and thereby enriching our understanding of consumer behaviours in influencer marketing. Sixth, by employing a quantitative research method, this thesis validates the conceptual framework of PSR and generalises the PSR development process across various industries. It also validates the positive outcomes of PSR, shedding light on the effectiveness of influencer marketing.

From a practical perspective, this thesis will benefit practitioners by providing a stronger conceptual foundation, leading to improved operationalisation of PSR (MacKenzie, 2003), useful in industry-driven research for a deeper understanding of consumer behaviour. This knowledge enables brands to forge long-lasting customer relationships and execute successful online marketing campaigns (Leite and Baptista, 2022; Wahab et al., 2022). It can guide companies to pursue collaborations with influencers strategically, enhancing brand attitude and consumer purchase intention (Hudders and De Jans, 2022; Gomes et al., 2022). The effectiveness of influencer marketing is significantly influenced by the strength of PSR between consumers and influencers (Reinikainen et al., 2021). Additionally, this thesis offers marketers and policy makers a detailed roadmap for developing and maintaining consumer relationships, from the initial encounter to ongoing engagement, underlining the importance of PSI, perceived value, and trust in these dynamics.

#### **1.6 Thesis structure**

This thesis includes eleven chapters. Chapter 1 introduces the thesis, providing a comprehensive overview, including the background of PSI and PSR, the thesis rationale, research aims and objectives, the mixed-method research methodology adopted, and the expected academic contributions and practical implications.

Chapter 2 presents a systematic literature review focusing on PSI and PSR. It critically evaluates existing literature from 1956 to 2022, especially debates around these concepts, their conceptualisation, and main domain characteristics. It proposes seven prerequisites for

distinguishing PSI and PSR across environments and discusses their differing dynamics offline and online. This chapter concludes by identifying research gaps and the necessity of treating PSR as a developmental process.

Chapter 3 delves into the analytical framework, starting with a discussion on post-positivism as the guiding research paradigm, emphasising its ontological and epistemological foundations. It then outlines the exploratory sequential mixed-methods design, detailing the qualitative and subsequent quantitative studies. Lastly, it describes the data collection approaches, including semi-structured interview, quantitative study 1 consumer survey, quantitative study 2 consumer survey and quantitative study 3 consumer survey.

Chapter 4 details the methodology and design of the qualitative phase, including the interview guide design, sampling approach, data collection process, interviewee characteristics, and data analysis techniques. It also addresses potential biases and measures to enhance the rigour of the qualitative research.

Chapter 5 presents the findings from the semi-structured interviews, utilising content analysis and thematic analysis. It offers an in-depth examination of how interviewees develop PSR with their favourite influencers and discusses key factors influencing PSR development through the four stages, including exposure initiation, experimentation, intensification, and integration stages.

Chapter 6 proposes a conceptual framework for the PSR development process, identifying four key stages based on literature and qualitative data. It examines critical concepts within each stage, especially focusing on involvement, perceived value, PSR, and purchase intention, leading to the formulation of twelve research hypotheses.

Chapter 7 discusses the methodology and design of the quantitative phase, covering the operationalisation of constructs, questionnaire design, sampling approach, data analysis techniques, and a pilot test. Efforts to minimise bias are also summarised.

Chapter 8 is dedicated to the amendment of the PSR definition and the development of a new PSR measurement instrument. It outlines the rationale for the new scale, followed by the steps involved in its development, including item generation, purification, dimension exploration, and testing for reliability and validity.

Chapter 9 presents the testing and findings related to the research hypotheses. Utilising confirmatory factor analysis, path analysis, mediation analysis, and moderation analysis, this chapter offers insights into the conceptual framework empirically testing outcomes.

Chapter 10 systematically reviews the research objectives and the thesis's approach to addressing them. It discusses additional findings, including the PSR scale development and empirical studies' additional relationships.

Chapter 11 concludes the thesis by summarising the academic contributions, practical implications, and reviewing the study's limitations. It also suggests directions for future research, offering a roadmap for continued exploration within the field.

#### Chapter 2: Systematic literature review

#### 2.1 Chapter introduction

In marketing, the concepts of PSI and PSR are important in understanding and explaining consumer behaviour in media environments (e.g., consumer behaviour in influencer marketing). Despite their importance, there remains a lack of consensus regarding the main domain characteristics of PSI and PSR across both online and offline contexts. Efforts to distinguish between the two have been made, yet the debate has garnered limited attention, leaving the core issues unresolved. These issues have been identified as the **first research gap**. Therefore, this chapter focuses on resolving the long-standing definitional debate by proposing seven prerequisites for their identification in both the offline and online environments. Recognising the evaluation of historical research as a valuable method for addressing existing literature gaps (Hunt, 2011), this chapter conducts a systematic literature review by reviewing 172 published works on PSI and PSR from 1956 to 2022 in the field of management, marketing, media science, psychology, tourism and information systems to achieve the first research objective of this thesis: to identify the differences between PSI and PSR in offline and online environments.

The structure of this chapter is organised as follows. Firstly, the chapter introduces the concept and methodology of a systematic literature review. Subsequently, the chapter discusses the debates surrounding PSI and PSR within offline environments, proposing seven prerequisites to differentiate between the two. Following this, the discussion extends to PSI and PSR in online contexts, presenting seven prerequisites that facilitate their identification in digital environments. Furthermore, this chapter conducts a comparative analysis of the potential conditions affecting PSI and PSR in offline versus online settings. Subsequently, a content analysis of the existing PSR definitions highlights the necessity to propose a new definition of PSR that captures its complex and dynamic nature within online environments, marking this as the **second research gap**. Lastly, the chapter highlights the importance to further investigate the development process of PSR, identified as the **third research gap**.

#### 2.2 Rationale for conducting a systematic literature review

PSI and PSR are two critical concepts in marketing that help understand consumer behaviour in media environments. Despite the wealth of research on PSI and PSR, certain conceptual ambiguities persist. Current literature does not adequately delineate the main domain characteristics of PSI and PSR within online and offline contexts (Lu et al., 2023). Some scholars contend that "*relationship*" should be the main domain characteristic of PSI, thereby entangling it with PSR (e.g., Zhu et al., 2023). While there are claims that PSI and PSR represent the same concept (e.g., So and Shen, 2016; Penttinen et al., 2022), these lack substantial support. This inconsistency in definitions has resulted in the terms PSI and PSR being used interchangeably by some scholars (e.g., Liao et al., 2022; Conde and Casais, 2023), thereby exacerbating the conceptual ambiguity surrounding the two constructs. It needs to be noted that the conceptual ambiguity might lead to operationalisation issues and violate the criteria for a good definition (MacKenzie, 2003; Wacker, 2004).

PSI and PSR are two different concepts that can be proved based on the differentiation of social interactions and social relationships. Social interaction and social relationship are two distinct concepts (Bales, 1950), PSI and PSR act as media-mediated phenomenon, supplementing the understanding of how people interact and develop relationships in media environments, thereby substantiating their distinctiveness (Perse and Rubin, 1989; Lim and Kim, 2011). Social interaction emphasises that people do some activities that affect people based on their automatic psychological activities, which can be reflected through verbal or non-verbal communication (Argyle, 2017). Such social interaction can promote or change social relationships and only occurs when people encounter each other face-to-face or through media (Miell and Dallos, 1996; Hall, 2016). However, social relationship emphasises the connection between people and needs to be established through multiple social interactions, and both parties agree to develop the relationship (Goffman, 1972). The duration of the relationship is from the time the relationship is formed to the time the relationship is broken up.

Although some scholars have tried to distinguish between PSI and PSR, the debate has not received much attention to date. Some researchers identify the differences between PSI and PSR to explain why one of the two related concepts is chosen in the context of their study (e.g., Zafar et al., 2020; Utz et al., 2021). Other researchers distinguish the difference between PSI and PSR to discuss the problem of using them interchangeably in the current literature (Yuan and Lou, 2020). Two systematic literature reviews of PSI and PSR, Giles

(2002) and Liebers and Schramm (2019) provide detailed research progress and future research directions from the perspective of psychology and media science, respectively, but do not focus on the conceptual issues. Although some scholars have tried to explain the differences between PSI and PSR (e.g., Hartmann and Goldhoorn, 2011), key issues remain unsolved. Therefore, this is identified as the first research gap, and aims to address it in this chapter.

To solve the key issues, this thesis employs a systematic literature review. A systematic literature review is an evidence-based research method that allows researchers to systematically identify and review relevant existing research to achieve specific research objectives (Siddaway et al., 2019). Although systematic literature review is not widely used in marketing research, it has been regarded as a useful approach to evaluating the existing literature to solve the remaining issues in the literature and reconceptualise them in a historical way (Snyder, 2019). Different from traditional narrative reviews, the systematic literature review employs explicit and scientific methodology, which can generate comprehensive, transparent, and replicable findings (Tranfield et al., 2013; Siddaway et al., 2019).

The systematic literature review approach is well-suited for this thesis for three main reasons. Firstly, it provides a historical perspective to address longstanding issues in the literature (Hunt, 2011), facilitating the differentiation between PSI and PSR based on contemporary research. Systematically reviewing the current literature will benefit from state-of-the-art insight to advance the understanding of PSI and PSR (Paul et al., 2020). Compared to traditional narrative reviews, a systematic literature review produces a more comprehensive understanding of the nature of PSI and PSR and elucidates their distinctions, significantly enriching marketing knowledge. The adoption of a clear and reliable procedure in this systematic literature review can minimise bias, decrease subjectivity, and yield more practical and beneficial outcomes for implications (Snyder, 2019).

Therefore, this systematic literature review aims to address the first research gap to resolve the long-standing definitional debate, propose prerequisites to identify PSI and PSR in the offline and online environments, respectively, and address this thesis's first research objective. Therefore, this systematic literature review aims to 1) propose prerequisites that can distinguish PSI and PSR in offline and online environments 2) explore how PSI and PSR change from offline to online environments.

#### 2.3 Systematic literature review method

This systematic literature review is guided by the Preferred Reporting Items for Systematic Reviews and Meta-Analysis (PRISMA) 2020 statement developed by Page et al. (2021), which aims to increase the transparency and rigour of systematic literature reviews (Liberati, 2009; Pahlevan-Sharif et al., 2019) and increase the quality, reliability of the review findings (Tranfield et al., 2003, Siddaway et al., 2019). The full PRISMA 2020 is presented in Appendix 1.

#### 2.3.1 Eligibility criteria

This systematic literature review has two sets of eligibility criteria for selecting reviewed papers: inclusion criteria and exclusion criteria, to increase the transparency of the review (Siddaway et al., 2019; Snyder, 2019). This review proposes six inclusion and two exclusion criteria (as listed in Table 2.1) based on the research objective and journal quality. First, PSI and PSR originally rooted in psychology, have found application across a spectrum of disciplines. This review, therefore, includes pertinent studies from a variety of fields, such as media, psychology, marketing, management, tourism, and information systems, to ensure a comprehensive examination of these concepts. Conversely, studies from unrelated disciplines where PSI or PSR are applied with divergent meanings, such as in zoology or chemistry, are excluded.

To ensure the high quality and rigour of the reviewed literature, the reviewed paper should be published in peer-reviewed academic journals. Conference papers are excluded since conference papers only show the most essential part of the research and do not present the whole paper, which could lead to bias in reviewing the paper. Moreover, the paper reviewed should be published in English to minimise bias that could be caused by misunderstanding the translation.

In addition, to guarantee the quality of sources, the Australian Business Deans Council (ABDC) Journal List has been chosen as the criterion for screening high-quality articles. The ABDC list is widely regarded as a highly dependable resource capable of identifying premier academic research on a global scale, free from regional biases (Grossmann et al., 2019). The ABDC list is extensively utilised in systematic literature reviews for the selection of articles, establishing its reputation as a trusted instrument for identifying high-quality research (e.g.,

Kashif and Udunuwara, 2020; Mondal and Chakrabarti, 2019). It encompasses a wide array of fields and disciplines, covering all areas specified in the inclusion criteria for this review. Consequently, to facilitate solid theoretical development, the journals under review are required to be ranked as A\*, A, or B within the ABDC journal quality list.

Table 2.1 Inclusion and exclusion criteria of the systematic literature review

Inclusion Criteria
<b>Fields of publication</b> : Academic journals in the areas of media, psychology, marketing, management, tourism, and information systems
Publication type: Published in peer-reviewed academic journals
Language: English
Journal Ranking: Journal classified as A*, A, or B in the ABDC journal list
<b>Time period</b> : 1956–2022
Research focus: empirical research on PSI and PSR in the offline or online context
Exclusion Criteria
Irrelevant fields: Fields unrelated (e.g., zoology and chemistry)
Paper types: Conference paper and meta-analysis

Furthermore, to ensure the replicability of this review, the period for this review is restricted to published and in-print journal articles between 1956 and 31<sup>st</sup> December 2022. The first PSI and PSR paper is published by Horton and Wohl in 1956. The reason starting from 1956 is that reviewing historical research helps identify controversial issues and potential solutions (Hunt, 2011). Articles published up to 31<sup>st</sup> December 2022 are incorporated in this review to cover the latest developments in understanding PSI and PSR.

Furthermore, the articles selected for review must be empirical in nature and situated within either an offline or online context, enabling the review to comprehensively explore the evolution of PSI and PSR from traditional to digital environments. However, meta-analysis papers are excluded from this review as they do not align with its scope and purpose, which is to offer an in-depth understanding of the parasocial phenomenon based on individual empirical studies rather than a synthesis of findings across various studies as seen in meta-analyses (Paul et al., 2021; Paul and Criado, 2020). Such analyses aggregate results from multiple studies, which could result in redundant information.

#### 2.3.2 Information sources and search strategy

For the search engine, Scopus and EBSCO are two databases used for paper selection in this systematic literature review. In business management, Scopus is a world-leading database with broader publication coverage than other data databases, such as the Web of Science (Mongeon and Paul-Hus, 2015; Zhu and Liu, 2020). Scopus is considered the primary database for searching papers for systematic literature reviews in marketing (Chatterjee et al., 2021; Mariani et al., 2021). EBSCO is another professional and high-quality research platform covering various peer-reviewed journals from multiple research fields (EBSCO, 2023). EBSCO is also widely used in systematic literature reviews in marketing (e.g., Hassan and Rahman, 2021; Mandler et al., 2021). Scopus and EBSCO cover academic journals, especially English papers, which are appropriate databases for this systematic literature review.

This review adopts keyword searches and sets restrictions on the selection of articles. This review set up several search terms because there are many ways to express the concepts of PSI and PSR. The entire search terms in this review are presented in Table 2.2. In terms of study record, according to search terms and restrictions, the author's name, title, year of publication, DOI, abstract and keywords of all literature are exported to a Microsoft Excel spreadsheet.

Table 2.2 Search terms in Scopus and EBSCO

Search terms in Scopus and EBSCO
parasocial interaction/para social interaction/para-social interaction/PSI parasocial relationship/para social relationship/ para-social relationship/PSR

#### 2.3.3 Data selection process

To ensure the effectiveness of the data selection and article review process, the researchers conduct a pilot test on data selection and management for the online PSR articles. Every article is carefully read during the pilot test, and an Excel sheet is used for data management. After reading each article carefully, this review records the information accordingly in Excel sheets: research aim, the definition of PSI/PSR, media type, research method, sample region and size, platform, characters, character type/field, research findings, antecedents of

PSI/PSR, consequences of PSI/PSR and moderator. The formal systematic literature review is subsequently conducted after the successful completion of the pilot test.

The data selection process contains four stages, and the whole process is shown in Figure 2.1. The first step is searching articles in Scopus and EBSCO based on the search terms and criteria discussed above. In total, 815 articles are identified. Afterwards, the researcher checks whether the journal articles qualify for the A\*, A, and B standards in the ABDC journal list. At this stage, 501 articles have been removed due to being duplicates or not meeting the required standards, leaving 314 articles retained. The third stage is checking whether the retained articles are eligible for other criteria. The researcher screens the remaining 314 articles based on their titles, abstracts, and eligibility criteria. Then, the researcher carefully read the full text of each paper to make the final decision. In the third stage, 142 articles are removed since these papers do not qualify for the criteria, and 172 papers are retained.



Figure 2.1 The process of selection of reviewed articles

Finally, to thoroughly understand PSI and PSR, the 172 articles are divided into four groups: offline PSI, offline PSR, offline PSI and online PSR. The rationale for reviewing PSI and PSR in offline and online environments separately is that they have distinct characteristics in offline and online environments. PSI and PSR are first investigated in offline environments, such as TV (Papa et al., 2000) and then studied in online environments (Farivar et al., 2022). The offline and online environments have different characteristics, such as convenience and media consumption patterns, which affect the scope of PSI and PSR. Additionally, the media characters that people can have PSI and PSR are broader in the online environment compared with offline environments. Therefore, the definition scope and research contexts differ in offline and online environments. Reviewing these papers

separately allows for a more robust understanding of PSI and PSR and contributes to understanding how PSI and PSR transit from offline to online environments. Based on this, 22 offline PSI papers,12 offline PSR papers, 87 online PSI papers and 51 online PSR papers are identified for review.

# 2.4 Debates on identifying the main domain characteristics of PSI and PSR in the offline environment

The very first definitions of PSI and PSR were proposed by Horton and Wohl in 1956, providing the initial theoretical underpinnings for these phenomena. In their work, they defined PSR as:

"We propose to call this seeming face-to-face relationship between spectator and performer a para-social relationship".

They posited that PSR represents an illusory form of relationship where viewers feel a sense of connection with media characters. Horton and Wohl further elaborated on the concept of PSI, suggesting it as the conversational exchange that occurs between the audience and media figures. They described PSI as:

"This simulation of conversational give and take may be called para-social interaction".

The proposal of these two concepts was ground-breaking. On the one hand, it is the first time describing two different processes between viewers and media figures when they participate in media activities from the audience's psychological perspective, and thereby, more theories related to PSI and PSR are developed and tested in various fields. For example, TV shopping viewers' PSI can increase viewers' satisfaction with TV shopping (Lim and Kim, 2011). Consumers' PSR with soap opera characters enhances consumers' attitudes toward the placed products (Dias et al., 2017).

On the other hand, PSI and PSR can be regarded as a supplement to relevant theories of interpersonal interaction, social interaction, interpersonal relationship and social relationship, which fills in people's daily interactions and relationships mediated by the media (Perse and Rubin, 1989; Lim and Kim, 2011). Interaction and building relationships are important
activities in people's daily lives, and the PSI and PSR are beneficial for understanding people's virtual interaction and relationship-building with media characters. Although reallife interactions and relationships differ from PSI and PSR in the virtual environment, they have similar elements that can lead people to have the same feelings and experiences (Giles, 2002). In other words, PSI and PSR capture the two critical people's psychological activities related to media activities and help to understand people's behaviour in the media environment (Rubin and Perse, 1987).

While Horton and Wohl's (1956) introduction of PSI and PSR marked a significant advancement in understanding individual's behaviour in media environment, their work has been critiqued for its lack of precise definitions and clear differentiation between these two similar concepts, leading to potential ambiguities (Dibble et al., 2016). This vagueness is evident in their discussion of PSR, for instance, where they describe it as having an asymmetrical "*interaction relationship*" during television viewing, a characterisation that blurs the lines between PSI and PSR.

# 2.4.1 The five views of the main domain characteristics of PSI in the offline environment

To understand the viewpoints of PSI definition in the offline PSI studies, this thesis critically reviews the 22 offline PSI studies and focuses on the various conceptualisations of PSI. The 22 reviewed PSI definitions in the offline environment are presented in Appendix 2.

Scholars mainly hold five different views on the main domain characteristics of PSI in the offline environment, including users' interaction/experience/perception, users' relationship, user's process, users' involvement, and users' communication (Table 2.3). Specifically, more than half of the papers conceptualise PSI as a *"users' relationship*", which does not align with the original definition of PSI as the conversational give-and-take interaction proposed by Horton and Wohl (1956). Only three out of 22 papers followed the original definition of PSI as *"users' interaction*".

The first view of the main domain characteristic of PSI is "*users' interaction*", aligning with Horton and Wohl's (1956) original definition. This viewpoint emphasises the viewer's direct and personal interaction experience with media characters during media exposure (Hartmann and Goldhoorn, 2011). PSI is characterised by a perceived conversational exchange with media entities, a form of virtual interaction that persists only when engaged in media

environments (Giles, 2002). Furthermore, this view of conceptualisation of PSI can be seen as a supplement to social interaction literature, with viewers often equating these media interactions to their everyday social interactions (Giles, 2002).

Main Domain Characteristics of PSI Definition in Offline Environment		
Main domain characteristics	Definition Example	Sources
Users' interaction - experience/perceptions	The <i>experience</i> of TV viewers to be engaged in an immediate, personal encounter with a TV performer.	Cummins and Cui (2014); Oliver et al. (2019); Hartmann and Goldhoorn (2011)
Users' relationship	An imaginary face-to-face <i>relationship</i> of audience members with media personalities.	Rubin and Perse (1987); Perse and Rubin (1989); Grant et al. (1991); Conway and Rubin (1991); Kim and Rubin (1997); Papa et al. (2000); Greenwood (2008); Greenwood and Long (2009); Moyer-Guse and Nabi (2010); Sun (2010); Lim and Kim (2011); Tian and Yoo (2015); So and Shen (2016); Bi et al. (2021)
Users' involvement	Interpersonal <i>involvement</i> of the media user with what he or she consumes.	Rubin et al. (1985); Skumanich and Kintsfather (1998); Bentley (2014)
Users' process	Viewer different cognitive, emotional, and/or behavioural <i>processes</i> when responses to media characters.	Knoll et al. (2015)
Users' communication	Users' <i>communication</i> phenomenon	So and Nabi (2013)

Table 2.3 Main domain characteristics of PSI in the offline environment

Although "*users' interaction*" accurately reflects the main domain of PSI, it has gained limited support among scholars. Among the 22 papers reviewed on PSI in offline contexts, only three, specifically those by Hartmann and Goldhoorn (2011), Cummins and Cui (2014), and Oliver et al. (2019), adopt this viewpoint. Hartmann and Goldhoorn's (2011) advocacy for a return to PSI's original definition and their provision of a precise definition that aligns with the "interaction" perspective may have been a crucial factor in this resurgence. As a result, the notion of "users' interaction" as the main domain characteristic is gradually gaining broader acceptance among scholars.

The second view of the main domain characteristics of PSI is the "*users' relationship*", which is the most controversial view in the PSI definition. As is shown in Table 2.3. 13 out of 22 papers reviewed endorse this viewpoint, positing "users' relationship" as PSI's main domain characteristic and framing PSI as an illusory connection between audiences and media personas (e.g., Tian and Yoo, 2015; Bi et al., 2021). Moreover, some researchers support that PSI and PSR are the same concepts (e.g., Rubin and Perse, 1987), leading to a blurring of conceptual boundaries and making it challenging to distinguish between the two. Such ambiguities could further complicate the operationalisation of these concepts and violate the criteria for a good definition (MacKenzie, 2003; Wacker, 2004).

While some scholars advocate for "*users' relationship*" as the main domain characteristic of PSI, the evidence for this viewpoint is not robust. According to Horton and Wohl (1956), PSI and PSR as related, each capturing separate psychological processes. Thus, PSI and PSR possess unique definitions and main domain characteristics (Dibble et al., 2016). Conceptualising PSI as a type of relationship markedly deviates from Horton and Wohl's (1956) original conceptualisation of PSI, signifying a shift away from their foundational intent.

The third perspective on PSI's main domain characteristics is termed "*users' involvement*", a concept introduced by Rubin et al. (1985) that builds upon Horton and Wohl's (1956) foundational work by integrating PSI and PSR. This viewpoint encompasses a wider range of interactions than "users' interaction", including "seeking guidance from a media persona, seeing media persona, imagining being part of a favourite program's social world, and desiring to meet media performers" (Rubin et al., 1985). Some papers adopt this view, such as Skumanich and Kintsfather (1998) and Bentley (2014).

However, this perspective fails to consider that PSI should be restricted to the duration of media consumption, as the notion of "users' involvement" implies a virtual friendship between the audience and media characters that extends beyond the act of media consumption (Oliver et al., 2019). Furthermore, "users' involvement" encompasses a wide array of psychological processes, including transportation, PSI, identification, and worship, thereby expanding the PSI concept to include diverse psychological phenomena and extending its scope beyond the original scope (Brown, 2015; Nordlund, 1978).

Another view on PSI's main domain characteristic is the "users' process," a view exclusively adopted by Knoll et al. (2015). They argue that PSI encompasses the cognitive, emotional,

and behavioural processes of viewers. While PSI could involve these psychological phases, defining PSI solely in terms of such processes is problematic. This is because many other constructs, like consumer engagement, exhibit similar psychological dynamics (Claffey and Brady, 2014; Claffey and Brady, 2019). Moreover, the psychological experiences of viewers during PSI are not uniform (Giles, 2002), indicating that the "users' process" may not precisely capture the main domain characteristic of PSI.

The final view of PSI is "*users' communication*", proposed by So and Nabi (2013), which highlights the exchange of information, both verbal and non-verbal, between the parties involved (Belch and Belch, 2021). This perspective suggests that during PSI experiences, audiences engage in a communicative process, receiving both verbal and non-verbal cues from media characters. Verbal interactions might include the content of the message, fostering a sense of dialogue with the media figures (Horton and Wohl, 1956), while non-verbal interactions could involve gestures or expressions conveyed by the characters (Cummins and Cui, 2014).

In conclusion, drawing from Horton and Wohl's (1956) initial conceptualisation of PSI, five distinct interpretations have emerged within the offline context. This suggests that Horton and Wohl (1956) did not clearly define PSI and PSR and adequately differentiate between these two closely related constructs. Their definitions were ambiguous, which has caused confusion for some researchers.

# 2.4.2 The four views of the main domain characteristics of PSR in the offline environment

To understand the viewpoints of PSR definition in offline PSR studies, this thesis critically reviews the 12 offline PSR research studies and focuses on the various conceptualisations of PSR. The reviewed PSR definitions in the offline environment are presented in Appendix 3.

Unlike the discourse surrounding the conceptualisation of PSI, the literature on PSR consistently treats it as a distinct "relationship", clearly separating it from PSI concepts. However, their focus on the defining characteristics varies, with terms such as "*relationship*", "*attachment*", "*bond*", and "*psychological connection*" being employed to describe these nuances. Detailed insights into these varied interpretations are provided in Table 2.4. Three studies specifically describe PSR as a "people's bond" (e.g., Bond, 2021), suggesting it

represents a more intensified form of relationship established between parties (Arantola, 2002). Meanwhile, another study posits PSR as "people's attachment" to another entity, implying a deeper level of relationship (e.g., Sun and Wu, 2012). Although there is debate regarding the depth of PSR relationships, the consensus among scholars is that PSR among researchers is that PSR signifies the connection established between individuals and another party.

Main Domain Characteristics of PSR Definition in Offline Environment		
Main domain characteristics	Definition Example	Sources
Media users' relationship	Viewer has the illusion of a face- to-face <i>relationship</i> with a fictional or real media character.	Stephens et al. (1996); Hoffner and Cohen (2014); Behm-Morawitz et al. (2017); Bernhold and Metzger (2020); Dias et al. (2017)
Media users' attachment	The emotional <i>attachment</i> to characters	Sun and Wu (2012)
Media users' bond	Viewers durable and continuing <b><i>bonds</i></b> with television characters.	Gabriel et al. (2018); Bernhold (2019); Bond (2021);
Media users' psychological connections	Viewers psychological connections with media figures	Young et al. (2013)

Table 2.4 Main domain characteristics of PSR in the offline environment

#### 2.4.3 The interchangeable use of PSI and PSR in the offline environment

The ambiguity in defining PSI and PSR has led some scholars to use the terms interchangeably. For example, some authors use both PSI and PSR to describe the relationship between viewers and media characters (e.g., Lim and Kim, 2011). Table 2.5 shows the terms PSI and PSR are used in PSI in the offline environment. Of the 22 articles, nine papers use PSI and PSR interchangeably, and only four explicitly differentiate between the two concepts.

The same issue uses PSI and PSR interchangeably, also found in PSR studies in offline environments. Table 2.6 presents the terms PSI and PSR used in PSR studies offline. Two papers use PSI and PSR interchangeably, and one uses PSR and PSB (parasocial bond)

interchangeably. Some scholars identify PSI and PSR as distinct concepts (e.g., Bernhold and Metzger, 2020).

The use of the terms PSI and PSR in the offline PSI studies		
Use PSI and PSR interchangeably	Rubin et al. (1985); Perse and Rubin (1989), Grant et al. (1991); Papa et al. (2000); Greenwood (2008); Greenwood and Long (2009); Moyer-Guse and Nabi (2010); Sun (2010); Lim and Kim (2011)	
Identify PSI and PSR as different concepts	So and Nabi (2013); Oliver et al. (2019); Cummins and Cui (2014); Hartmann and Goldhoorn (2011)	
Use PSI (EPSI)	Rubin and Perse (1987); Conway and Rubin (1991); Kim and Rubin (1997); Skumanich and Kintsfather (1998); Bentley (2014); Tian and Yoo (2015); Knoll et al., (2015); So and Shen (2016); Bi et al. (2021)	

Table 2.5 The use of the terms PSI and PSR in the offline PSI studies

Table 2.6 The use of the terms PSI and PSR in the offline PSR studies

The use of the terms PSI and PSR in the offline PSR studies		
Use PSI and PSR interchangeably	Stephens et al. (1996); Dias et al. (2017)	
Identify PSI and PSR	Sun and Wu (2012); Behm-Morawitz et al. (2017); Bernhold and Metzger (2020)	
Use PSR	Bernhold (2019); Derrick et al. (2009); Young et al. (2013); Bond et al. (2021); Bonus et al. (2021); Hoffner and Cohen (2014)	
Use PSR and PSB (parasocial bond)	Gabriel et al. (2018)	

The practice of using PSI and PSR interchangeably undermines academic rigour, potentially leading to confusion and misinterpretation (Hartmann and Goldhoorn, 2011). Ambiguous definitions can result in improper operationalisation and fail to meet the good definition criteria (MacKenzie, 2003; Wacker, 2004; Stern et al., 2001). For instance, some studies on PSI employ PSR measurement tools, casting doubt on the validity of their findings and the

theories derived from them (Bi et al., 2021; Tian and Yoo, 2015; MacKenzie, 2003; Cummins and Cui, 2014). Furthermore, this lack of clarity hampers effective communication of research outcomes to the academic community (Stern et al., 2001).

#### 2.5 Prerequisites of PSI and PSR in the offline environment

Although efforts have been made to distinguish PSI from PSR and to define PSI more precisely (e.g., Dibble et al., 2016), the issue remains unresolved. For example, Hartmann and Goldhoorn (2011) revisit the concept of PSI, focusing on its interactive nature but not clarifying its distinction from PSR. Identifying clear prerequisites for PSI and PSR could effectively address the ongoing debate that has persisted for nearly seven decades. To date, no scholars have outlined such prerequisites to aid in differentiating these closely related concepts. The subsequent section discusses the prerequisites for PSI and PSR in traditional media settings and examines their similarities and differences. The seven prerequisites are presented in Table 2.7.

# 2.5.1 Prerequisite 1: One of the parties participating in PSI/PSR

The first prerequisite for distinguishing PSI and PSR is the identification of the parties involved. Typically, these phenomena require two distinct entities: a human participant and a media character. For instance, in the context of television viewership, the human viewer constitutes one party, while the media character, such as a TV show host, represents the other.

Scholarly works have employed various terminologies to denote the human participant in PSI and PSR. Regarding PSI, the terms "TV viewers" (Oliver et al., 2019) and "audience members" (So and Nabi, 2013) are commonly used to describe the human element. Additionally, terms like "users" (Cummins and Cui, 2014) and "media users" (Bentley, 2014) are also noted, underscoring the role of the individual engaging with the media. Conversely, the literature on PSR less frequently specifies the human participant, though some references use "viewer" (Bernhold, 2019) or "audience" (Bond, 2021) in this context as well.

Hence, the first prerequisite emphasises the importance of identifying the specific 'one party' involved in PSI and PSR, despite the overlap in the terminologies used. To delineate between PSI and PSR, it is crucial to decide whether 'one party' is engaged with the media. If 'one party' is a traditional media user, the interaction qualifies as PSI. In contrast, PSR encompasses a broader range of individuals and not limited to media users.

	Offline PSI	Offline PSR
1. One of the parties participating in PSI/PSR	media users (Oliver et al., 2019)	individual (Bernhold and Metzger, 2020)
2. The other parties participating in PSI/PSR	media characters, includes real and fictional characters (Cummins and Cui, 2014)	media characters, includes real and fictional characters (Cummins and Cui, 2014)
3. The nature of PSI/PSR	one-way unequal interaction	an individual's connection (Dias et al., 2017)
4. The main domain characteristics of PSI/PSR	perception of direct and reciprocal communication (Cummins and Cui, 2014; Oliver et al., 2019)	the connection (Dias et al., 2017)
5. The process of PSI/PSR (from the viewer's perspective)	triggered by mutual awareness, attention, and adjustment (Hartmann and Goldhoorn, 2011)	triggered by experiencing at least one-time parasocial interaction or a cross- situational, stable, and schematic cognitive pattern of images and interaction scripts that includes affective aspects (Bernhold and Metzger, 2020)
6. The time when PSI/PSR can occur	occurs during the media exposure period (Cummins and Cui, 2014; Oliver et al., 2019)	occurs from the moment the relationship is established to the breakup
7. The platforms where PSI/PSR can occur	traditional media (Bentley, 2014)	

# Table 2.7 Prerequisites for distinguishing between PSI and PSR in the offline environment

The one party in PSI specifically refers to individuals engaging with traditional forms of media, such as television, radio, movies, and books. PSI occurs when individuals interact with media characters, facilitated by the consumption of traditional media (Skumanich and Kintsfather, 1998; Oliver et al., 2019). This interaction often mirrors a conversational exchange, where the audience feels a sense of give-and-take with the media character (Horton and Wohl, 1956; Oliver et al., 2019). During their interaction with media characters, individuals can perceive verbal and non-verbal cues from media characters, which in turn, can initiate PSI (Cummins and Cui, 2014). Media performers employ various professional techniques, such as body language, warm tones, and eye contact, to enhance the audience's

PSI (Cummins and Cui, 2014; Oliver et al., 2019). If individuals are not media users, they would not experience these cues, and consequently, PSI would not occur. Thus, PSI is exclusive to traditional media users (Stephens et al., 1996).

Contrastingly, the one party of PSR is broader and not confined to media users alone. PSR can exist among individuals regardless of their engagement with traditional media. Once established, PSR can persist beyond the duration of media exposure, indicating that PSR can last without media consumption (Bond, 2021). This enduring nature means that individuals can form and maintain PSR whether they consume traditional media or not. Certain traits are common among individuals who form PSR with media personalities. For instance, individuals with materialistic tendencies and low self-esteem are more inclined to develop PSR (Sun and Wu, 2012). Similarly, those with fewer personal connections and lower quality friendships might seek PSR with media figures for companionship (Hoffner and Cohen, 2014). The tendency to form PSR is closely linked to personal traits and attachment styles (Bernhold and Metzger, 2020).

# 2.5.2 Prerequisite 2: The other parties participating in PSI/PSR

The other party in both PSI and PSR encompasses a range of media characters within the traditional media landscape, including both real and fictional entities. This aspect is the only common prerequisite for PSI and PSR. In the literature, researchers employ a variety of terms to describe the other party. In PSI research, terms such as "media personalities", "performer", and "media character/character" are prevalent. Conversely, in PSR studies, "personae" and "characters" are frequently used to refer to the other party of PSR.

Real characters constitute a significant portion of the other party in PSI and PSR. This group includes media characters like TV program hosts (Hartmann and Goldhoorn, 2011), TV shopping hosts (Grant et al., 1991), athletes (Sun, 2010), and political figures, such as Donald Trump (Gabriel et al., 2018). These media characters, often celebrated or professional figures, have the platform to appear in traditional media and become recognised by the public. They strive to align with their character portrayal, enhancing PSI and strengthening PSR through effective performance and affective empathy (Horton and Wohl, 1956; Cummins and Cui, 2014).

Fictional characters also serve as the other party in PSI and PSR. These are characters crafted within narratives, who, despite their fictitious nature, possess distinct personalities and are perceived as real within their contexts (NFI, 2023). They commonly appear in various media formats, such as TV series, movies and video games (Young et al., 2013; Brown, 2015). The attractiveness and representation of these fictional characters often influence the likelihood of forming PSR (Knoll et al., 2015). Typically, audiences develop PSR with central characters in television series. However, it is also possible for viewers to establish PSR with antagonistic characters, driven by a mix of antipathy and hostility towards these negative personas (Dias et al., 2017; Bernhold, 2019).

#### 2.5.3 Prerequisite 3: The nature of PSI/PSR

Within the scholarly domain, there exists a limited discussion concerning the real definition of PSI and PSR. However, it is essential to differentiate PSI and PSR by identifying their real definition because it can capture the nature of the concepts even though the real definitions are not often used in research (Gilliam and Voss, 2013). Based on the analysis, this paper argues that the nature of PSI and PSR are different.

PSI is characterised by a one-way, asymmetrical interaction between media viewers and media characters. When individuals engage with media characters through viewing or listening, it mimics a social interaction (Hartmann and Goldhoorn, 2011). However, the nature of traditional media as a channel imposes limitations, resulting in a lack of reciprocity. The absence of reciprocity means that media characters emerge solely as initiators of social interaction, unable to discern the viewers' focus or receive any form of feedback (Ellis and Holmes, 1982; Masse et al., 2018). Consequently, media consumers are relegated to passive recipients of the interaction, unable to reciprocate or establish a real two-way social exchange, which underscores the lack of reciprocity and the unequal nature of PSI.

Conversely, PSR embodies a connection between individuals and media characters. From the media characters' standpoint, there is an intentional design for them to seem open and appealing for sustained relationships with the audience, serving as a fundamental prerequisite for relationship formation (Miell and Dallas, 1996). Real media figures might view their connection with fans collectively, rather than on an individual basis. Fictional characters, by their nature, lack the capacity to acknowledge such relationships. From the audience's perspective, the selection of media characters based on attractiveness and appeal facilitates the formation of PSR, fostering a sense of connection with the characters (Young et al., 2013). However, this connection is not acknowledged by the media characters and thus cannot reciprocate the feelings or acknowledge the relationships formed by the audience.

### 2.5.4 Prerequisite 4: The main domain characteristics of PSI/PSR

Given the prevalence of PSI and PSR among media audiences and the rich insights gained from examining these phenomena from the viewers' standpoint, this study aligns with current literature by adopting the perspective of media viewers to explore PSI and PSR. Some scholars suggest that the main domain characteristic of PSI is the "relationship", deeming it indistinguishable from PSR (e.g., So and Shen, 2016; Bi et al., 2021). This paper posits that PSI and PSR are distinct constructs, each with unique main domain characteristics.

The main domain characteristic of PSI is the media viewers' perception of engaging in direct and reciprocal communication with media characters (Oliver et al., 2019). The transportation effect plays a pivotal role, enabling viewers to perceive media personas as though they share the same physical space, facilitating direct communication as if they were co-located (Brown, 2015). Furthermore, the attention and focus viewers dedicate to media characters, coupled with the reception of feedback, cultivate a virtual experience of conversational exchange, akin to real-life interactions. The verbal and non-verbal cues emitted by media characters serve as mechanisms for viewers to feel as though they are engaged in genuine social exchanges, fostering a sense of reciprocity in the interaction (Cummins and Cui, 2014).

Contrastingly, the main domain characteristic of PSR is the formation of a personal connection with media characters, mirroring the dynamics of real-life social relationships (Bernhold and Metzger, 2020). PSR encapsulates the connection individuals establish with media entities, which can engender feelings of intimacy and the perception of a real social tie (Hoffner and Cohen, 2014). For some, this connection may manifest as a friendship with the media character, indicative of a deeper, more enduring relationship (Bernhold, 2019; Bernhold and Metzger, 2020). Thus, PSR represents a long-standing relationship sustained through ongoing interactions with the media characters.

### 2.5.5 Prerequisite 5: The process of PSI/PSR

The differentiation between the mechanisms of PSI and PSR is not extensively explored, yet it is fundamental for distinguishing between the two concepts. PSI and PSR are initiated by different stimuli, leading to distinct processes. One perspective on PSI suggests that it develops gradually over time, with Greenwood and Long (2009) treating PSI as a long-term relationship between media users and media figures, which contrasts with the fundamental nature of PSI as a more immediate, interaction-based phenomenon, rather than an enduring relationship. This paper contends that such a view is more applicable to PSR, as it conflates the gradual development characteristic of relationships, thereby misrepresenting the PSI process.

PSI is triggered by media viewers' awareness, attention, and adjustment to media characters, mirroring real-life social interactions where individuals actively perceive and adapt their behaviour towards others (Malle, 2005; Hartmann and Goldhoorn, 2011). Techniques employed by media characters, such as eye contact and body language, foster an illusion of reciprocity, making viewers feel as though there is a mutual adjustment, as media characters' responses are pre-planned and not directly reactive to viewers' inputs (Cummins and Cui, 2014; Hartmann and Goldhoorn, 2011).

On the other hand, PSR formation needs at least one time PSI, requiring time to develop, similar to the formation of real-life social relationships (Bernhold and Metzger, 2020). The passage of time is a critical precursor to PSR, with prolonged and frequent media exposure increasing the likelihood of PSR formation (Tian and Yoo, 2015; Greenwood, 2008). Regular viewers, through extended viewing time, tend to establish deeper connections with media characters than occasional viewers, with increasing PSI leading to a stronger connection due to growing confidence in the media characters (Perse and Rubin, 1989; Greenwood, 2008).

Furthermore, PSR can sometimes emerge from atypical situations, where individuals develop cross-situational, stable cognitive patterns and affective interaction scripts that transcend the conventional 'fourth wall' of media interaction (Hartmann et al., 2008; Dibble et al., 2016). However, PSR formed under such unusual circumstances deviates from the long-term, stable relationships typically discussed in PSR literature and represents an unconventional trigger of PSR (Tukachinsky and Stever, 2018). Therefore, PSR arising not directly from PSI signifies an exception rather than the norm in PSR formation.

#### 2.5.6 Prerequisite 6: The time when PSI/PSR can occur

Understanding the timing of PSI and PSR is crucial for distinguishing these two similar concepts. In the context of PSI within offline environments, only a few studies explicitly state that PSI occurs during media exposure (e.g., Cummins and Cui, 2014; Oliver et al., 2019). Conversely, the literature on PSR does not commonly specify the exact timing of its occurrence.

PSI takes place when individuals engage with traditional media, such as watching television, reading books, or listening to the radio (Cummins and Cui, 2014). This suggests that PSI occurs during the media consumption period, suggesting that PSI is ephemeral, existing only during the time of media exposure (Oliver et al., 2019). The exposure to traditional media allows individuals to perceive and engage in a semblance of reciprocal communication with media characters, undergoing a psychological process of awareness, attention, and behavioural adjustment essential for PSI (Behm-Morawitz et al., 2017; Hartmann and Goldhoorn, 2011). This interaction fosters a sense of direct communication with the characters (Cummins and Cui, 2014).

Conversely, while PSR and PSI may occur concurrently, their durations differ significantly due to the nature of PSR as a connection. The duration of PSR starts from the establishment of the relationship to its ends, implying that PSR can persist beyond direct media exposure, which is the same with real-life relationships (Rubin et al., 1985; Bond, 2021). Individuals can continue to maintain PSR with media characters beyond their media consumption sessions. However, when individuals decide to relinquish their connection with media figures, the PSR ceases to exist and transitions into parasocial breakup (Hu et al., 2016).

# 2.5.7 Prerequisite 7: The platforms where PSI/PSR can occur

The platforms on which PSI and PSR can occur serve as another key differentiator between the two concepts. Analysis of PSI literature reveals that PSI typically necessitates a media platform for its occurrence, with television (Hartmann and Goldhoorn, 2011) and radio (Lim and Kim, 2011) being common mediums. In contrast, PSR's existence is not bound to a specific medium or platform.

PSI can manifest across a variety of media forms within the traditional media landscape. Viewers can engage in PSI with media or fictional characters through television (Grant et al., 1991; Tian and Yoo, 2015; Bi et al., 2021), movies and books (Ingram and Luckett, 2019; Bonus et al., 2021), and radio (Lim and Kim, 2011). Conversely, the occurrence of PSR transcends the need for a media platform. Once established, PSR embodies a more enduring psychological process that remains relatively stable over time, independent of continuous media exposure (Tukachinsky and Stever, 2018). This distinction highlights the media-dependent nature of PSI, in contrast to the more enduring and media-independent nature of PSR.

# 2.6 Debates on identifying the main domain characteristics of PSI and PSR in the online environment

The scope of PSI and PSR has broadened from traditional to online media, enabling these phenomena across various platforms. Traditionally, PSI and PSR have been observed through such as television (Bi et al., 2021) and radio (Lim and Kim, 2011). The digital age extends these phenomena to include social media (Leite and Baptista, 2022), among other online mediums. Despite this expansion, there remains ongoing debate around the conceptualisation of PSI and PSR in digital spaces. The distinctions and overlaps between PSI and PSR, previously contested in traditional media contexts, persist online, with some scholars using the terms interchangeably (e.g., Wei et al., 2022) or treating them as the same concepts (e.g., Zhang et al., 2022), without a consensus or clear resolution.

# 2.6.1 The four views of the main domain characteristics of PSI in the online environment

This thesis critically reviews the 87 online PSI research, particularly their various conceptualisations. The 87 reviewed PSI definitions in the offline environment can be found in Appendix 4. Of the reviewed 87 articles, ten do not define PSI.

There are mainly four different views of PSI's main domain characteristics in the online PSI studies, including "users' interaction", "users' relationship", "users' involvement", and "users' intention and relationship" (Table 2.8).

Main domain characteristics of PSI definition in the online environment		
Main domain characteristics	Definition example	Sources
Users' interaction - experience	Parasocial interaction refers to the one-sided mediated form of social <i>interaction</i> between audience and media characters	Lee (2013); Labrecque (2014); Kim and Kim (2017); Hu et al. (2017); Gong and Li (2017); Daniel et al. (2018); Zhou and Jia (2018); Breves et al. (2019); Sakib et al. (2020); Kelton and Pennington (2020); Yuksel and Labrecque (2016); Choi et al., (2019); Nanda and Banerjee (2020) ; Hsieh and Lee (2021); Lee et al. (2021); Lo et al. (2022); Hudders and De Jans (2022)
Users' relationship (intimacy/ connection)	Parasocial interactions can be defined as a viewer's perceived <i>relationship</i> with a media personality.	<ul> <li>Keng et al. (2011); Men and Tsai (2015); Chiu and Huang (2015); Lueck (2012);</li> <li>Colliander and Erlandsson (2013); Kim and Song (2016); Xiang et al. (2016); De</li> <li>Jans et al. (2018); Lee (2018); Phua, Lin and Lim (2018); Jin (2018); Jin and</li> <li>Muqaddam (2019); Liu et al. (2019); Aw and Labrecque (2020); Shan et al. (2019);</li> <li>Jin and Ryu (2020); Sokolova and Kefi (2020); Boerman (2020); Kim (2020);</li> <li>Ledbetter and Meisner (2021); Yuksel and Labrecque (2016); Lee and Lee (2017);</li> <li>Handarkho (2020); Kang et al. (2021); Yılmazdoğan et al. (2021); Handarkho</li> <li>(2021); Hoabin Ye et al. (2021); Jin et al. (2021); Penttinen et al. (2022); Yousaf</li> <li>(2022); AW et al. (2022); Chen et al. (2022); Wang and Hu (2022); Chang and Kim</li> <li>(2022); Tseng et al. (2022); Kim (2022); Masuda et al. (2022); Zhang et al.</li> <li>(2022); Kim (2022); Lee and Lee (2022); Wei et al. (2022); Manchanda et al. (2022);</li> <li>Liao et al. (2022); Deng et al. (2022)</li> </ul>
Users' involvement	Interpersonal <i>involvement</i> of the media user with what he or she consumes.	Thorson and Rodgers (2006); Goldberg and Allen (2008); Men and Tsai (2013); Tsai and Men (2013); Zheng et al. (2020); Kim et al. (2020); Tsai et al. (2021)
Users' interaction and relationship	The <i>intimacy</i> between media users and media characters can be regarded as a one-sided <i>interaction</i> .	Xu et al. (2021); Youn and Jin (2021); Kim and Kim (2021); Chen et al. (2021)

# Table 2.8 Main domain characteristics of PSI definition in the online environment

The views of PSI's main domain characteristics are reduced from five in the traditional media environment to four views in the online environment. In the online environment, the *"users' process"* and *"users' communication"* have disappeared, but a new view argues that PSI is *"users' interaction and relationship*" is emerged.

Despite some scholars, such as Hartmann and Goldhoorn (2011), distinguishing between PSI and PSR in traditional media, similar disputes endure in online studies without resolution. Over half of the papers (46 out of 77) conceptualise PSI as "users' relationship", viewing it as the connection users form with media entities (e.g., Liao et al., 2022). This interpretation, however, conflicts with the definition of PSR. Conversely, 17 out of 77 papers adopt PSI's main domain characteristic is "users' interaction", which in line with PSI's interactive nature (e.g., Hudders and De Jans, 2022).

Some studies advocate for PSI as "users' involvement" (e.g., Tsai et al., 2021), as discussed above, a term encompassing a range of engagements with media personas as outlined by Rubin et al. (1985). However, this broader interpretation, suggesting a continuous connection beyond mere media exposure, diverges from the immediate nature of PSI (Oliver et al., 2019).

A novel perspective proposes that PSI embodies both interaction and connection, suggesting an intimacy akin to a unilateral interaction (e.g., Xue et al., 2021). For example, Xue et al. (2021) define PSI as "the intimacy between media users and media characters can be regarded as a one-sided interaction". This view essentially merges the concepts of PSI and PSR but lacks clarity in differentiating the two, failing to adequately define PSI's conceptual scope.

In conclusion, the definition of PSI in online studies remains a subject of varied interpretations. Despite calls from scholars to clarify the distinction between PSI and PSR (e.g., Cummins and Cui, 2014), a significant portion of online PSI research continues to conceptualise PSI as a form of "relationship", thereby blurring the lines between PSI and PSR. Moreover, an emerging perspective attempts to combine "interaction" and "relationship" as a unified definition for PSI. However, this approach does not align with the criteria for good definition criteria as outlined by MacKenzie (2003), indicating a need for more precise and distinct conceptualisations and effective prerequisites to distinguish the two concepts.

# 2.6.2 The three views of the main domain characteristics of PSR in the online environment

The reviewed 51 PSR definitions are presented in Appendix 5, reveals three primary perspectives on PSR's main domain characteristics within the online context. Information about the three views is presented in Table 2.9. The three views of PSR's main domain characteristics are in line with the offline studies, including "*people's relationship*", "*people's bond*", and "*people's attachment*". Notably, the "people's psychological process", a view previously considered in offline PSR studies, is absent, as it fails to encapsulate the essence of PSR. This is because psychological activities related to media, such as psychological ownership, differ from PSR and involve distinct psychological processes (Kwon, 2020).

The majority of scholars underscore "relationship" as the main domain characteristic of PSR (e.g., Jacobson et al., 2022; Burnasheva and Suh, 2022), viewing it as the connection between media users and media entities, resonating with the foundational work of Horton and Wohl (1956). Few scholars define PSR as "people's bond" (e.g., Hu et al., 2020;), interpreting it as an enhanced relationship fostered between individuals and media figures (Arantola, 2002). Additionally, one study frames PSR as "people's attachment", indicating a deeper form of relationship (e.g., Farivar et al., 2022).

# 2.6.3 The interchangeable use of PSI and PSR in the online environment

The interchangeable use of PSI and PSR persists in online studies, mirroring a problem initially identified in offline contexts. Table 2.10 illustrates this issue within online PSI research, where, among 87 reviewed papers, 23 employ PSI and PSR interchangeably (e.g., Wei et al., 2022; Manchanda et al., 2022). Conversely, a minority of these papers (8 out of 87) acknowledge and highlight the distinction between PSI and PSR (e.g., Deng et al., 2022; Hudders and De Jans, 2022), drawing attention to this critical issue.

Main domain characteristics of PSR definition in the online environment			
Main domain characteristics	Definition example	Sources	
People's relationship (connection)	Parasocial relationships can be defined as unilateral <i>relationships</i> that a media audience develops with a media character.	Richards and Calvert (2015); Tsiotsou (2016); Yuan et al. (2016); Escalas and Bettman (2017); Chung and Cho (2017); Hwang and Zhang (2018); Lim and Kim (2018); Han and Yang (2018); de Bérail et al. (2019); Munnukka et al. (2019); Lee and Cho (2020); Zafar et al. (2020); Lim et al. (2020); Reinikainen et al. (2020); Agnihotri and Bhattacharya (2020); Sokolova and Perez (2021); Tsiotsou (2015); Iannone et al. (2018); Feder (2020); Hwang et al. (2020); Farivar et al. (2021); Utz et al. (2021); Chen et al. (2021); Breves et al. (2021); Reinikainen et al. (2021); Aw and Chuah (2021); Yuan et al. (2021); Whang and Im (2021); Breves et al. (2021); Yang and Ha (2021); Tsfati et al. (2022); Balaban et al. (2022); Wahab et al. (2022); Sundermann and Munnukka (2022); Liao et al. (2022); Cheung et al. (2022); Leite and Baptista (2022); Burnasheva and Suh (2022); Jacobson et al. (2022)	
People's bond	The emotional <i>bonds</i> that audiences form with media personas.	Hu et al. (2020); Yuan and Lou (2020); Noor et al. (2022)	
People's attachment	Parasocial relationship is a bond-based <i>attachment</i> , in which followers consider that they are in a relationship with an influencer.	Farivar et al. (2022)	

# Table 2.9 Main domain characteristics of PSR definition in the online environment

The use of the terms PSI and PSR in the online PSI studies		
Use PSI and PSR interchangeably	Thorson and Rodgers (2006); Chiu and Huang (2015); Kim et al. (2016); Kim and Song (2016); Xiang et al. (2016); Kim and Kim (2017); Gong and Li (2017); Sanz-Blas et al. (2017); Jin (2018); Fu et al. (2017); Kim and Kim (2020); Kang et al. (2021); Yılmazdoğan et al. (2021); Chen et al. (2021); Tsai et al. (2021); Penttinen et al. (2022); Chang and Kim (2022); Tseng et al. (2022); Kim (2022); Masuda et al. (2022); Zhang et al. (2022); Wei et al. (2022); Manchanda et al. (2022)	
Identify PSI and PSR	Boehmer (2016); Zhou and Jia (2018); Breves et al. (2019); Yuksel and Labrecque (2016); Daniel et al. (2018); Lee et al. (2021); Deng et al. (2022); Hudders and De Jans (2022)	
Use PSI	Goldberg and Allen (2008); Keng et al. (2011); Men and Tsai (2013); Lee (2013); Men and Tsai (2013); Lee and Jang (2011); Tsai and Men (2013); Labrecque (2014); Men and Tsai (2015); Lueck (2012; Colliander and Erlandsson (2013); Kassing and Sanderson (2012); Lee and Watkins (2016); Hu et al. (2017); Tsai and Men (2017); De Jans et al. (2018); Lee (2018); Esteban-Santos et al. (2018); Phua et al. (2018); Jin and Muqaddam (2019); Liu et al. (2019); Aw and Labrecque (2020); Shan et al. (2019); Jin and Ryu (2020); Zheng et al. (2020); Sokolova and Kefi (2020); Boerman (2020); Sakib et al. (2020); Kim (2020); Kelton and Pennington (2020); Kim et al. (2020); Kim (2020); Kelton and Pennington (2020); Xim et al. (2020); Ledbetter and Meisner (2021); Lee and Lee (2017); Choi et al. (2019); Handarkho (2020); Nanda and Banerjee (2020); Xu et al. (2020); Xu et al. (2021); Youn and Jin (2021); Kim et al. (2021); Handarkho (2021); Kim and Kim (2021); Tsai et al. (2021); Hsieh and Lee (2021); Hoabin Ye et al. (2021); Jin et al. (2021); Yousaf (2022); Lo et al. (2022); AW et al. (2022); Chen et al. (2022); Wang and Hu (2022); Kim (2022); Lee and Lee (2022); Liao et al. (2022); Fazli-Salehi et al. (2022); Shen et al. (2022)	

Table 2.10 The use of the terms PSI and PSR in the online PSI studies

In online PSR studies, the interchangeable use of terms is also evident, as shown in Table 2.11. Out of 51 reviewed PSR studies, 11 papers use PSI and PSR interchangeably (e.g., Leite and Baptista, 2022), while a larger proportion (14 out of 85) differentiate between the two, recognising them as distinct constructs. This lack of clarity and consistency undermines the conceptual rigor, potentially leading to misinterpretations between the intended message by the researchers and the understanding by the readers (Peter, 1981; MacKenzie, 2003).

Although some scholars attempt to distinguish between PSI and PSR, their efforts are often confined to the context of their individual studies, aiming to justify the selection of either PSI or PSR based on the research context (e.g., Utz et al., 2021) or to provide conceptual clarity specific to their study's framework (e.g., Sokolova and Perez, 2021). Additionally, comparisons between PSI and PSR are sometimes made to provide some discussions regarding the interchangeability issue that exists in existing literature (Yuan and Lou, 2020).

The use of the terms PSI and PSR in the online PSR studies		
Use PSI and PSR interchangeably	Tsiotsou (2016); Pressrove and Pardun (2016); Berkelaar (2017); Lee and Cho (2020); Hu et al, (2020); Lim et al. (2020); Quelhas- Brito et al. (2020); Agnihotri and Bhattacharya (2020); Yang and Ha (2021); Liao et al. (2022); Jacobson et al. (2022)	
Identify PSI and PSR	Zafar et al. (2020); Reinikainen et al. (2020); Yuan and Lou (2020); Sokolova and Perez (2021); Utz et al. (2021); Farivar et al. (2021); Whang and Im (2021); Tsai et al. (2021); Zhong et al. (2021); Breves et al. (2021); Balaban et al. (2022); Sundermann and Munnukka (2022); Noor et al. (2022); Leite and Baptista (2022)	
Use PSR	Richards and Calvert (2015); Yuan et al. (2016); Finsterwalder et al. (2017); Escalas and Bettman (2017); Chung and Cho (2017); Hwang and Zhang (2018); Lim and Kim (2018); Han and Yang (2018); de Bérail et al. (2019); Munnukka et al. (2019); Yuan et al. (2021); Tsiotsou (2015); Iannone et al. (2018); Feder (2020); Hwang et al. (2020); Breves et al. (2021); Reinikainen et al. (2021); Aw and Chuah (2021); Yuan et al. (2021); Yuan et al. (2021); Tsfati et al. (2022); Wahab et al. (2022); Farivar et al. (2022); Cheung et al. (2022); Burnasheva and Suh (2022)	
Use PSR and parasocial identification	Chen et al. (2021)	

#### Table 2.11 The use of the terms PSI and PSR in the online PSR studies

# 2.7 Prerequisites of PSI and PSR in the online environment

The above analysis underscores the persistent confusion surrounding the definitions and the main domain characteristics of PSI and PSR, along with their interchangeable usage in the current literature. Addressing this issue is essential, as precise and clear construct definitions are crucial for theory development (MacKenzie, 2003) and can significantly impact the theoretical advancement process (Gilliam and Voss, 2013). To navigate these challenges, the subsequent discussion focuses on applying seven prerequisites in the offline environment to the online environment. The seven prerequisites for identifying online PSI and PSR are presented in Table 2.12.

# 2.7.1 Prerequisite 1: One of the parties participating in PSI/PSR

As discussed in the previous section, PSI and PSR necessitate the involvement of two entities: individuals and media characteristics. In the context of influencer marketing, for instance, the individual is represented by a social media user, while the media characteristic is embodied by a social media influencer.

Prerequisites for distinguishing between PSI and PSR in online environment		
	Online PSI	Online PSR
1. One of the parties participating in PSI/PSR	people who are exposed in the online environment, e.g., new media or technology product users (Kelton and Pennington, 2020)	individual (Burnasheva and Suh, 2022)
2. The other parties participating in PSI/PSR	real person, media characters, and anthropomorphic objects (Lee, 2018; Kim et al., 2020)	real person, media characters, and anthropomorphic objects (Han and Yang, 2018; Hwang et al., 2020)
3. The nature of PSI/PSR	one-way unequal interaction	An individual's connection (Farivar et al., 2022)
4. The main domain characteristics of PSI/PSR	perception of direct and reciprocal communication (Choi et al., 2019; Kelton and Pennington, 2020)	the connection (Wang and Hu, 2022)
5. The process of PSI/PSR (from viewer perspective)	triggered by mutual awareness, attention, and adjustment (Hartmann and Goldhoorn, 2011)	triggered by experiencing at least one-time parasocial interaction or a cross- situational, stable, and schematic cognitive pattern of images and interaction scripts that includes affective aspects (Bernhold and Metzger, 2020)
6. The time when PSI/PSR can occur	occurs during the exposure period (Yuksel and Labrecque, 2016; Kelton and Pennington, 2020)	occurs from the moment the relationship is established to the breakup
7. The platforms where PSI/PSR can occur	through new media (Tseng et al., 2022)	

# Table 2.12 Prerequisites for distinguishing PSI and PSR in the online environment

An examination of the terminology used in existing literature on online PSI and PSR reveals a variety of terms employed to describe the individual involved in PSI and PSR. In online PSI studies, researchers have referred to the individual as "customer" (Lee and Lee, 2017) and "viewer" (Kim, 2022), indicating the diverse roles individuals assume in PSI contexts. Conversely, some studies have adopted broader terms such as "individuals" (Handarkho, 2021) not specifying the group of people in PSI. For online PSR studies, researchers have used terms, such as "consumers" (Cheung et al., 2022) and "followers" (Farivar et al., 2022) to describe the other party in PSR.

The distinction between the individual's role in PSI and PSR is particularly important in the online environment. Kelton and Pennington (2020) argue that recognising the presence of the individual in the online environment is an essential step in identifying PSI and PSR. For PSI to occur, individuals must be engaged with the media, thereby facilitating interaction with the media entity (Chiu and Huang, 2015). The absence of online exposure prevents the possibility of experiencing PSI.

The advent of Web 1.0 and Web 2.0 has expanded the scope of PSI in online environments compared to offline environments. Traditional media interactions, such as eye contact or verbal exchanges, have been supplemented by digital platforms, enabling interactions with a wider array of entities (Men and Tsai, 2013). This technological evolution has broadened the opportunities for PSI through media with characteristics such as smart speakers (Lee and Cho, 2020) and virtual assistants (Noor et al., 2022).

In contrast, PSR remains consistent across online and offline contexts, with individuals as one party (Burnasheva and Suh, 2022). The formation of PSR does not necessitate media or technological engagement as is the case with PSI (Zafar et al., 2020). However, the chance to develop PSR varies among individuals and is influenced by factors such as personal motivations, psychological conditions, and personality traits (As and Chuah, 2021). For instance, individuals using social media for entertainment and relationship-building purposes are more inclined towards PSR (Yuan et al., 2016). Similarly, individuals with high social anxiety or a strong need to belong are more likely to form PSR (De Berail et al., 2019).

# 2.7.2 Prerequisite 2: The other parties participating in PSI/PSR

In the context of both offline and online environments, the second prerequisite for PSI and PSR remains consistent: they share the same other parties. The range of entities with which individuals can form PSI and PSR in the online environment is considerably more extensive than in traditional media contexts. Within traditional settings, the scope is primarily limited to media characters, such as television hosts (Bernhold and Metzger, 2020). Conversely, the online environment facilitates PSI and PSR with a wider array of entities, encompassing

media characters like influencers (Farivar et al., 2022) and technological products (Lee and Park, 2022).

The diversity of entities for PSI and PSR online mirrors the second prerequisite identified in offline contexts but on a broader scale, including real individuals, media characters, and anthropomorphic objects (Han and Yang, 2018; Lee and Park, 2022). Real individuals, such as celebrities, sports stars, and politicians (Escalas and Bettman, 2017; Ledbetter and Meisner, 2021), influencers with significant online followings (Lou and Yuan, 2019), members of online communities (Choi et al., 2019), serve as potential focal points for PSI and PSR. Notably, this extends to animal influencers as well (Jacobson et al., 2022).

Media characters continue to play a significant role as the other party in PSI and PSR in the online environment. With the advent of digital streaming platforms, such as Netflix, viewers have the opportunity to engage with series and programs in an on-demand manner (Nanda and Banerjee, 2020). This accessibility allows audiences to foster PSI and subsequently develop PSR with these media characters (Nanda and Banerjee, 2020).

Anthropomorphic objects have emerged as a novel category of entities involved in PSI and PSR, which refer to non-human objects endowed with human-like characteristics, (Noor et al., 2021). Websites represent the initial non-human objects identified by researchers as capable of eliciting PSI and PSR among users (Thorson and Rodgers, 2006). This offers a valuable opportunity for businesses, as it enables them to disseminate comprehensive information and enhance engagement with their target audience (Goldberg and Allen, 2008).

The evolution of social media and applications has further expanded the scope to include applications (Lee, 2018) and platforms themselves, such as Facebook (Chiu and Huang, 2015). The relationships formed with these platforms are influenced by users' motivations and dependency relations, promoting continued engagement with the technology (Lee, 2018). Companies and brands have become the other parties for the generation of PSI an PSR, as individuals anthropomorphise brands, attributing them with human characteristics, including gender and distinct brand personalities (Wang and Hu, 2021). Consumers often follow brands' official social media accounts to stay updated, mirroring the way they connect with friends (Ohlwein and Bruno, 2021). In response, businesses and organisations actively cultivate personas on social media, aiming to have informal, conversational, and enduring interaction with their audience, thereby facilitating the development of PSR with brands (Yuan et al., 2021).

Furthermore, advanced technological products have also become significant to the other parties in the formation of PSI and PSR. These products and services are designed with human-like interaction capabilities, such as smart speakers (Kim and Choudhury, 2021), augmented reality (AR) try-on features (Yuan et al., 2021), and intelligent personal assistants like Siri (Han and Yang, 2018) and Amazon Echo (Whang and Im, 2021). Individuals can establish and sustain PSR through repeated interactions with these anthropomorphised electronic devices, fostering a sense of relationship (Lee and Cho, 2020).

# 2.7.3 Prerequisite 3: The nature of PSI/PSR

The nature of PSI in digital contexts is the same as that of traditional media, characterised by an asymmetrical interaction between the entities involved. The online environment, however, enhances perceived interactivity, leading individuals to believe in two-way communication. For instance, engaging with an influencer's vlog through likes and comments may simulate a two-way interaction (Reinikainen et al., 2020). Yet, the lack of direct response from the influencer highlights the unequal nature of PSI. Moreover, the bridging functions in the online environments allow viewers to interact socially rather than PSI (Kassing and Sanderson, 2012).

Similarly, the nature of PSR lies in the connection between two entities, consistent with offline environments. Individuals are inclined to form and maintain relationships with entities, including real individuals, media characters, and anthropomorphised objects, who are perceived as appealing, trustworthy, and relatable (Sokolova and Perez, 2021). However, this relationship is predicated on the individual's wistfulness and imagination, given the unawareness and non-consentness of the other party. Thus, the relationship is fundamentally one-sided, rooted in illusion and imagination (Reinikainen et al., 2020).

# 2.7.4 Prerequisite 4: The main domain characteristics of PSI/PSR

The debates on the main domain characteristics of PSI and PSR transcend from traditional to digital environments, which remain unresolved. For example, scholars define PSI as "relationship" (Manchanda et al., 2022) or "interaction and relationship" (Kim and Kim, 2021).

This paper posits that the main domain characteristic of PSI is the "perception of direct and reciprocal communication" (Kelton and Pennington, 2020), which aligns with traditional media. The digital environments, enhanced by cutting-edge technology, foster a more realistic and immersive interaction, leading individuals to feel actively involved in mutual interactions with the other parties. Content creators in PSI often adopt a direct conversational style, as exemplified by the famous Chinese influencer Jiaqi Li, who engages viewers with direct appeals during live-streaming commerce, which makes viewers have high PSI. This perceived immediacy of interaction fosters a sense of mutual recognition (Hartmann and Goldhoorn, 2011). Moreover, the possibility of sending messages to celebrities, influencers, or brands via social media reinforces the illusion of "direct communication" (Lee and Jang, 2013).

However, despite the perceived two-way interaction, the nature of this interaction remains unreciprocated. Primarily, reciprocity implies an exchange of information, yet PSI often lacks real two-way communication. The communication is predominantly controlled by the other party, with selective or non-existent responses to viewer engagement (Stever and Lawson, 2013). Consequently, PSI should be considered a one-directional interaction rather than a reciprocal communication. Furthermore, interactions with virtual assistants may simulate a bidirectional exchange of information, but these are devoid of emotional reciprocity, being based on pre-programmed responses.

Consistent with traditional media, the main domain characteristic of PSR is the connection formed by individuals. Despite attempts by the other parties to cultivate relationships, the development and maintenance of PSR are unilateral, with the viewers fully aware of the connection while the other parties remain unknown. Thus, the relationship is essentially a figment of the audience's imagination (Reinikainen et al., 2020). Notably, the strength of PSR may be stronger online compared to offline, where diverse media and technological platforms bolster confidence in the relationship. The online environment also enhances the potential for companionship, with constant access via devices, such as mobile phones and computers, fostering habitual activity. Through sustained interaction and exposure, individuals may form deeper PSR (Farivar et al., 2022).

# 2.7.5 Prerequisite 5: The process of PSI/PSR

In line with the offline environment, PSI and PSR are distinct psychological phenomena, each with unique trigger mechanisms. PSI is triggered by a sense of mutual awareness,

attention, and adjustment when individuals interact with media personas (Hartmann and Goldhoorn, 2011). Conversely, the development of PSR is a relatively long process, necessitating prolonged exposure and experiencing multiple interactions with the media figures (Reinikainen et al., 2020). The formation of PSR begins with dedicating time to the other parties, leading to repeated encounters with them (Lim et al., 2020). Essentially, the establishment of PSR requires at least one time of PSI experience (Dibble et al., 2016). Moreover, PSR is characterised by a consistent and enduring cognitive pattern of perceptions, encompassing emotional elements, which further differentiates it from the more immediate and interactive nature of PSI (Lim et al., 2020; Dibble et al., 2016).

#### 2.7.6 Prerequisite 6: The time when PSI/PSR can occur

The differentiation between PSI and PSR can be significantly informed by understanding the timing of their occurrence, an aspect often overlooked in studies. Concerning PSI, there are primarily three perspectives on its timing: during media exposure (Lee et al., 2021), in the absence of direct interactions (Kelton and Pennington, 2020), and through frequent interactions (Aw et al., 2022). This paper posits that PSI occurs during the exposure period, supporting the view of Lee et al. (2021).

In both online and offline settings, PSI necessitates the user's interaction within the media environment (Balaban et al., 2022), confined to the duration of media consumption (Lee et al., 2021). PSI is activated when users interact with the other party and perceive communication from them, fostering mutual awareness, attention, and adjustment (Hartmann and Goldhoorn, 2011). Once viewers stop media consumption, PSI also stops due to the absence of media stimulus. Hence, the notions that PSI can occur without direct interaction (Kelton and Pennington, 2020) or through frequent interactions (Aw et al., 2022) do not accurately capture the time when PSI can occur.

Regarding the timing of PSR occurrence, Balaban et al. (2022) suggest that PSR results from repeated media exposures and can extend beyond the immediate exposure period. This paper argues that the timing of PSR in the digital landscape aligns with that in offline environments, beginning with the relationship established and concluding with the ending of the relationship. In online environments, individuals are inclined to form connections with favoured entities and away from content they dislike. Consequently, users are more likely to

develop PSR with the other parties they find appealing and with whom they share similarities (Xiang et al., 2016; Lee and Watkins, 2016; Bernhold, 2019).

# 2.7.7 Prerequisite 7: The platforms where PSI/PSR can occur

In digital environments, PSI can transpire across a multitude of platforms, extending beyond the limitations of traditional media. Research has identified a range of online mediums that can trigger PSI, including websites (Kim and Kim, 2017), social media platforms (Gong and Li, 2017), e-commerce sites (Liao et al., 2022), mobile applications (Lee, 2018), and chatbots (Hsieh and Lee, 2021). Unlike the unidirectional communication characteristic of traditional media, the online environment offers a broader and more dynamic platform for interaction, facilitating more immediate and effective communication. This enhanced interactivity affords a level of personalisation in exchanges that closely resembles face-to-face communication (Zeng et al., 2009; Men and Tsai, 2015).

Moreover, the diverse functionalities of these platforms cater to a global audience, delivering both entertainment and information across various cultures and regions. Localised social media platforms further tailor the user experience by incorporating elements that resonate with specific cultural contexts (Men and Tsai, 2013).

Aligns with the offline prerequisite, the formation of PSR in online environments does not necessarily need the use of media or technological platforms. Once established, PSR tends to exhibit a degree of stability, persisting until the time when PSR ends (Tukachinsky and Stever, 2018), underscoring the enduring nature of PSR.

# 2.8 The difference between PSI and PSR in offline and online environments

PSI and PSR were originally conceptualised within the context of traditional media (Horton and Wohl, 1956), but subsequent research has validated their presence in digital environments (Noor et al., 2022). The transition from offline to online platforms has not only broadened the conceptual scope of PSI and PSR but also introduced variations in their conditions, reflecting the fundamental differences between traditional and digital media landscapes. This part aims to compare PSI and PSR across offline and online settings through a typological analysis, shedding light on the distinct conditions that facilitate these phenomena in each environment.

### 2.8.1 Frequency of potential PSI

The frequency of PSI in the online environment significantly surpasses that within traditional media contexts, primarily due to the unrestricted access and consumption afforded by online platforms. In contrast to the offline environment, where PSI primarily occurs through television with real and fictional characters (Rubin et al., 1985; Knoll et al., 2015), the digital environment removes the constraints of fixed locations and scheduled consumption. Traditional media consumption, often confined to specific settings like home viewing, is characterised by planned behaviour, limiting the opportunities for PSI to designated times and places (Rubin and Perse, 1987; Cummins and Cui, 2014).

Conversely, the online environment enables PSI across diverse platforms, such as social media and virtual assistants. This accessibility has led to the integration of media usage into daily routines, with the habitual use of mobile phones unconscious and repetitive interaction with media (Chiu and Huang, 2015). For instance, the act of following influencers and regularly checking their updates on social media platforms has become a common practice (Men and Tsai, 2013; Leite and Baptista, 2022), heightening the frequency of PSI opportunities online. Consequently, the pervasive nature of digital media consumption significantly increases the likelihood of experiencing PSI.

#### 2.8.2 Speed

For PSI, the immediacy of the experience remains consistent across both online and offline environments. PSI represents an automatic, immediate mental response that media users experience upon exposure to a media environment (Hartmann and Goldhoorn, 2011). Consequently, individuals undergo PSI instantly and reflexively when they interact with the other party when perceive the feeling of having a conversational give-and-take during media exposure, ensuring that the speed at which PSI occurs is uniform across different media environments.

Regarding PSR, the formation process accelerates in digital environments compared to traditional environments. The development of PSR typically necessitates at least one time of PSI, followed by repeated interactions with the media figure (Lim et al., 2020). In traditional contexts, the sequential dissemination of media content means that users must adhere to predetermined schedules to interact with specific media personalities, leading to infrequent

opportunities for interaction (Rubin et al., 1985). For instance, an individual might develop PSR with a news host by consistently watching a scheduled broadcast. Conversely, online media facilitates more frequent and diverse interactions through various digital platforms (Burnasheva and Suh, 2022). This heightened frequency of interaction expedites the development of PSR, enabling users to establish relationships with media figures more swiftly in the digital environment than is possible with traditional media.

### 2.8.3 Convenience

The digital environment significantly enhances the ease with which individuals can have PSI and develop PSR compared to traditional media environments. The accessibility of digital media allows individuals to immerse themselves in a media environment via electronic devices at any moment, fostering PSI (Tsiotsou, 2015). This constant connectivity also facilitates more frequent interactions with media personas, thereby streamlining the process of developing and sustaining PSR (Bernhold and Metzger, 2020). Conversely, traditional media confines users to specific locations and schedules, primarily through stationary devices like TVs and radios (Rubin and Perse, 1987), limiting the opportunities for PSI and PSR.

Furthermore, the online environment offers customised content, enabling users to access upto-date information effortlessly, such as browsing the latest social media posts, engaging in live streams, or interacting with smart speakers (Aw et al., 2021). This personalised approach not only increases the frequency of PSI occurrences but also aids in the maintenance of PSR. Additionally, the active search capabilities inherent to digital platforms empower users to seek out interactions with media figures at their convenience, moving beyond the passive consumption model typical of traditional media.

# 2.8.4 Time (continuity)

The 24-hour accessibility of the Internet significantly extends the duration of PSI compared to traditional media contexts. In traditional media environments, the episodic and scheduled broadcasting of content inherently limits the length of PSI, as engagement typically concludes with the end of a program (Rubin and Perse, 1987). Conversely, the digital environment, with its vast array of content, facilitates prolonged exposure to media, enabling extended periods of PSI. For instance, individuals can interact in lengthy live streaming

(Shen et al., 2022) or watch entire series online without the constraints of scheduled broadcasts.

However, the potential for extended PSI in the digital environment is not uniform. The rise of short-form content platforms like TikTok introduces PSI experiences of significantly shorter duration, sometimes lasting no more than 15 seconds (Ledbetter and Meisner, 2021). Additionally, various levels of attention among users can influence the duration of PSI, with reduced attention to media environments potentially reducing PSI opportunities (Lueck, 2012). Hence, the duration of PSI online can vary widely, offering both longer and shorter experiences relative to traditional media.

PSR in the online environment also tend to be more enduring. PSR, as a virtual connection between viewers and media figures, strengthens based on repeated interactions (Escalas and Bettman, 2017). The limited information channels and content availability in offline environments can hinder the continuous development and maintenance of PSR. In contrast, the digital environment offers an abundance of channels and a wealth of information, enabling individuals to engage frequently with media figures and actively seek out information.

# 2.8.5 Available ways of communication

Online environments significantly enhance the ways for PSI and PSR compared to traditional media. In offline contexts, interactions between media viewers and media characters are primarily one-way, with limited opportunities for viewers to respond to interactive cues, especially verbal ones, from media figures (Lee, 2013). For instance, in TV shopping programs, audience interaction with TV shopping hosts is typically confined to telephone communication, offering minimal interaction (Grant et al., 1991). Moreover, information about PSR entities is often disseminated unilaterally by media journalists, precluding reciprocal communication (Kim and Song, 2016).

Conversely, digital platforms break away from the one-directional communication model of traditional media, offering interactive opportunities for two-way communications (Men and Tsai, 2015; Shen et al., 2022). Viewers can actively interact with the other party through comments, likes, reposts, and private messaging (Gong and Li, 2017). Live streaming platforms facilitate real-time interactions between viewers and streamers, further enriching

the scope for interaction (Lim et al., 2020). Additionally, advancements in technology enable direct conversations with smart speakers and virtual assistants, providing an opportunity for interpersonal communication (Whang and Im, 2021).

It is crucial, however, to acknowledge that despite the multiplicity of interaction channels in online environments, the nature of communication often remains primarily one-sided. The control of communication typically lies with the media figures or platforms, with actual responses to user interactions being infrequent (Stever and Lawson, 2013; Kim and Song, 2016). Thus, while digital environments offer expanded platforms for interaction, the essence of these exchanges frequently retains a unilateral characteristic, mirroring traditional media's communication dynamics, albeit with greater interactivity.

#### 2.8.6 Opportunities to become two-sided interaction/relationship

The digital environment offers the potential to transform the traditionally one-sided nature of PSI and PSR into more reciprocal interaction and relationship. Social media platforms, in particular are designed to facilitate two-way communication, enhancing perceived interactivity and enabling users to engage directly with the other party of PSI and PSR (Labrecque, 2014). This two-way interaction and communication can foster a sense of authenticity and similarity, reminiscent of real-life interactions (Thorson and Rodgers, 2006).

Moreover, automated responses from celebrities, influencers, and social commerce entities can simulate real-life conversations, further blurring the lines between parasocial and social interactions and relationships. Advanced technologies like AI-driven smart speakers offer real-time, interactive communications (Whang and Im, 2021). However, the perception of automated responses can decrease the expectation of real two-way communication (Labrecque, 2014).

Despite these advancements, the control of communication largely remains with the media figures, who may not engage with every message, thereby maintaining the one-way nature of PSI and PSR. Nonetheless, social media's bridging functions offer a sense of reciprocity; some celebrities and influencers may actively respond to their follower's messages, potentially transforming PSI into actual social interactions (Kassing and Sanderson, 2012).

Certain online contexts provide the opportunity for evolving PSI and PSR in a two-way manner. Live streaming, for example, allows live streamers to interact with viewers in realtime, often responding to comments and fostering personal connections, especially when incentivised by viewers' virtual gifts (Zimmer et al., 2018). Similarly, online communities facilitate direct interactions among members, making the transition from parasocial to social relationships more feasible (Choi et al., 2019). These dynamics underscore the internet's role in making interactions more accessible and conducive to forming social relationships (Sanz-Blas et al., 2017).

# 2.8.7 Variety of entities/actors that can become the other parties in PSI/PSR

The online environment significantly expands the diversity of entities with which individuals can develop PSI and PSR. Unlike the traditional media environment, where interactions were mainly with professional or fictional characters within limited media forms, the digital environment provides a broader scope of potential the other parties.

In online environment, individuals not only continue to engage with traditional media figures like TV hosts, celebrities, and fictional characters but also with a wider array of entities, including ordinary people and technology products (Masuda et al., 2022; Yuan et al., 2021). Celebrities transitioning from traditional to digital platforms, influencers who command significant online followings (Farivar et al., 2022).

Moreover, brands and digital assets have become the other parties of PSI and PSR. Consumers, for instance, have been found to develop PSR with brands, for example, Huawei (Liao et al., 2022). The advent of AI-based assistants such as chatbots and smart devices like Amazon Echo further diversifies the entities with which PSR can be established (Youn and Jin, 2021; Zhong et al., 2021). In addition, some advanced technology, such as AR try-on, has been confirmed that people can develop PSR (Yuan et al., 2021).

### 2.8.8 The volume of information about the character (richness)

The advent of new media in the online environment has significantly enhanced the capacity for media characters and companies to enrich the information they provide, fostering deeper connections and facilitating the maintenance of long-term relationships (Leite and Baptista, 2022). Brands, in particular, leverage the creation of distinct corporate personas to engage

the public, using social platforms to deliver information about brand values, new product launches, and promotional activities, thereby developing PSR with consumers (Men and Tsai, 2015).

Media characters benefit from the expansive reach and immediacy of social media to increase their self-disclosure, sharing personal and professional information that draws followers closer (Kim and Song, 2016; Leite and Baptista, 2022). The level of openness fosters an intimacy similar to that experienced with real-life friends, contrasting with the more scripted and rehearsed nature of traditional media appearances (Lee and Jang, 2011). Such direct insights into the lives of media characteristics enhance not only PSI but also PSR by offering a glimpse into the authenticity of these media figures (Kim and Song, 2016).

Moreover, professional self-disclosure, which includes sharing details about work or upcoming projects, can bolster followers' interest and support, reinforcing their interaction with the media character (Kim and Song, 2016). Both personal and professional self-disclosure on social media serve as effective marketing strategies for the other parties, enabling them to develop and maintain longer-term relationships with their followers.

# 2.8.9 Expression function

The mode of expression in the online environment tends to be more relaxed and authentic compared to traditional media. In conventional environments, even when hosts adopt a conversational style, the delivery of content is often formal, bound by the structured nature of traditional media formats. This formality necessitates the use of carefully scripted language and a professional demeanour by television hosts and other media figures (Horton and Wohl, 1956; Grant et al., 1991).

In contrast, the rise of social media has transformed how messages are conveyed by media personalities, allowing for a more casual and interactive communication style (Tsai and Men, 2017). This shift enables individuals, companies, and even technology products to employ more straightforward, interpersonal communication techniques. By adopting tones and styles that mimic everyday conversation, they can foster greater interaction and develop stronger relationships with their audience (Zeng et al., 2009).

#### 2.8.10 Geographical barriers

The advent of online technologies and social media has transcended the geographical limitations traditionally associated with PSI and PSR. Digital platforms decrease physical barriers, facilitating connections across the globe and allowing individuals to interact with media figures from any location (Chiu and Huang, 2015; Yuan et al., 2021). Social networking sites (SNS) have become global communication hubs where celebrities and influencers, irrespective of their geographical origins, establish official accounts to interact with a worldwide audience (Men and Tsai, 2013).

This global accessibility enables media users to form PSI and PSR with entities across diverse geographical landscapes, effectively creating a boundary-less platform for interaction. Such widespread connectivity contrasts sharply with traditional media, which often prioritises local content and is constrained by regional broadcasting limitations. The online environment, therefore, not only broadens the scope for PSI and PSR but also fosters a more inclusive and diverse media environment, unbounded by geographical constraints.

### 2.8.11 Media Accessibility

The online media environment, with its broader accessibility features, opens up opportunities for a more inclusive range of individuals to experience PSI and PSR, transcending some of the limitations inherent in traditional media. The level of accessibility in traditional media often imposes constraints on the ability of individuals to have PSI and develop PSR, particularly affecting those with disabilities. Traditional media, such as television, primarily rely on verbal and visual cues to facilitate PSI (Cummins and Cui, 2014). However, for individuals with hearing or visual impairments, these modes of communication may not be fully accessible, hindering their ability to receive interactive signals from media characters.

In contrast, digital media offers enhanced accessibility features that can accommodate a wider range of needs, thereby increasing the potential for PSI and PSR. Online platforms provide functionalities such as videos with subtitles, catering to individuals with hearing difficulties by translating auditory information into text. While some traditional media, like news programs, might offer sign language interpretation, this is not a widespread practice across all content. Moreover, digital audio content, such as podcasts, can serve as an

alternative for individuals with visual impairments, allowing them to engage with content and form connections with media characters.

#### 2.8.12 Opportunities for PSI/PSR

The analysis shows that the digital environment affords enhanced opportunities for the formation of PSI and PSR surpassing traditional media due to various facilitative factors inherent to online platforms. Despite this, both traditional and digital media serve as vital medium for the generation of PSI and PSR. While the scope for PSI and PSR is considerably broader online, it is improbable that digital platforms will entirely supplant traditional media (Boehmer, 2016). Boehmer (2016) highlights that, in the context of sports consumption, traditional media remains predominant, with platforms such as Twitter ranking lower in terms of usage. This enduring preference for traditional over digital media is partly due to ingrained habits and familiarity with traditional information sources. Furthermore, it is essential to acknowledge that cultural diversity may significantly influence individuals' media consumption preferences and information-seeking behaviours (Men and Tsai, 2011).

### 2.9 Research gap identification

The above analysis aims to bridge the first research gap by delineating PSI and PSR as fundamentally distinct concepts. Through a systematic literature review, this gap is addressed by introducing seven prerequisites that facilitate the clear identification of these two concepts. Furthermore, based on the insights gained from the systematic literature review, this chapter brings to light another two critical research gaps.

The second research gap identified concerns the conceptualisation of PSR. Despite the prior review proposing seven prerequisites that could differentiate PSI and PSR in both offline and online contexts, it becomes evident that the definition of PSR requires further refinement. The third research gap highlighted pertains to the existing perception of PSR as a static concept, thereby overlooking its dynamic nature (Tukachinsky and Stever, 2018). Current literature explores the antecedents and consequences of PSR, yet there exists a notable deficiency in understanding the PSR development process.

#### 2.9.1 The conceptualisation of PSR

While there are disputes regarding the main domain characteristics of PSI, the existing literature has advanced the conceptualisation of PSI and has developed an appropriate measurement instrument to measure it. Hartmann and Goldhoorn (2011) systematically improve the definition of PSI, aligning it with the "interaction" view of PSI and precisely capturing the interactive nature of viewers' experiences when exposed to the media environment. According to Hartmann and Goldhoorn (2011), PSI is conceptualised as "the experience of TV viewers engaging in an immediate, personal, and reciprocal encounter with a TV performer". Their work significantly contributes to understanding PSI and capturing viewers' psychological processes during PSI experiences.

Although scholars generally agree that PSR describes the connection between individuals and another party, debates persist regarding the types of connections and strengths of the connections within PSR. For example, some scholars only point out that PSR is a "relationship" (e.g., Leite and Baptista, 2022), while other scholars argue that PSR is a deeper "bonding" (e.g., Yuan et al., 2021). Apart from the disagreement of the main domain characteristics, there are still some issues regarding the conceptualisation of PSR.

To analyse PSR definitions and the components deeply, this section evaluates the existing definitions of PSR in both offline and online environments. It is a practical and helpful procedure to critically review the existing definitions, which can provide insights into the critical components of the concept (Murrow and Hyman, 1994). The technique uses in this section is content analysis, which helps to identify the keywords and main components of PSR definitions in the existing literature (Bos and Tarnai, 1999). 12 PSR definitions in the offline environment and 51 PSR definitions in the online environment are reviewed and evaluated.

The researcher first read the definitions and became familiar with them definitions. Excel is used for the subsequent data analysis. The unit of content is words, including phrases and single words. One hundred-seven keywords are identified in total. After identifying the keywords, the researcher manually groups them to find the categories. PSR definitions provide valuable information to capture the main domain of PSR. 10 themes are identified, including "connection", "inequity", "psychological closeness", "dynamic", "develop", "viewers", "characters", "perception", "interpersonal", and "illusion".
### Theme 1: One party

One party is an essential part of PSR, which is the subject of PSR. Researchers use a variety of terms to describe the one party, and there are 18 different words and terms identified in the previous definitions, such as "*one party*" (Lim and Kim, 2018) and "*one*" (Bernhold and Metzger, 2020). Among these, some of the studies limit one party of PSR as the specific group based on their research context, for example, "consumers" (Cheung et al., 2022), which are appropriate in their research. However, this practice limits the definition to the specific research group and research context, and a broader group of people is excluded from the scope of PSR. Most definitions discuss one party of PSR, but some definitions neglect to point out the one party in the definition (e.g., Sun and Wu, 2012; Gabriel et al., 2018). In other words, the existing definition of one party in PSR is inaccurate and has parsimonious issues (Gilliam and Voss, 2013; Murrow and Hyman, 1994). Therefore, this thesis argues that a good definition of PSR should include one party of PSR.

### Theme 2: The other party

The other party is also a significant part of the definition of PSR, similar to one party. Twenty-six terms are identified in the reviewed definitions to refer to the other party, such as "media performer" (Lim et al., 2020) and "celebrities" (Hwang et al., 2020). These terms are limited since they are included only in the research context and neglect another group of objects. While this approach is appropriate, it fails to provide readers with a full picture of who the other party of PSR is. According to this review, people can develop PSR with a broader range of targets, including media characters, community members, organisations, social media platforms, websites, and technology products (Lee and Watkins, 2016; Choi et al., 2019; Labrecque, 2014; Kim and Kim, 2017; Lee and Cho, 2020). Therefore, in the existing PSR definition, the other party is inaccurate and has parsimonious issues (Gilliam and Voss, 2013; Murrow and Hyman, 1994). Therefore, this thesis proposes that a good definition of PSR should include the other party of PSR.

## Theme 3: Connection

Connection is the main domain characteristic of PSR (Feder, 2020). Researchers who use the connection as PSR's main domain characteristic accurately capture that PSR is a connection between one party and the other party. However, researchers use inconstant terminology to express this relationship, such as "*relationship*" (Iannone et al., 2018) and "*bond*" (Yuan et al., 2021). The disagreement of the terminology reflects the conceptual theme of PSR, which is unclear (Gilliam and Voss, 2013). Also, using unambiguous terms, such as "*link*", does not align with the good definition standard (MacKenzie, 2003). Therefore, the above analysis suggests the need to explore the main domain of PSR further and contribute to the conceptualisation of PSR (Murrow and Hyman, 1994).

### Theme 4: Perception

Perception is identified as a theme of PSR, which is another view of PSR's main domain characteristics. Scholars who hold this view believe that PSR is people's perception rather than that there is a connection between one party and the other party. For example, Stephen et al. (1996) define PSR as "*Feelings of friendship and intelligence on the viewer's part with remote" personalities such as soap opera rules, news anchors, and talk-show hosts*". When researchers hold the view that PSR is a perception, they usually use terms including "*feelings*", "*feel*", "*perception*" and "*a sense of*" as main domain characteristics. Scholars hold this view mainly because they define PSR from one party's perspective and focus on how one party perceives this connection.

### Theme 5: Real relationship

Real relationship is identified as one of the themes, but it raises ambiguous issues because it could lead readers to fail to distinguish between PSR and real-life relationships. Some researchers use ambiguous terms such as *"interpersonal"* (Tsiotsou, 2016) to help readers understand this similar connection developed in the media environment, but this is not a good definition standard (Gilliam and Voss, 2013). Therefore, a good PSR definition should avoid these terms and clearly distinguish between PSR and real-life relationships (MacKenzie, 2003).

## Theme 6: Inequality

Inequality as a theme clearly illustrates the unequal nature of PSR. In the reviewed PSR definitions, scholars use "*unilaterally*" (Yuan et al., 2016), "*one-sided*" (Lim et al., 2020) and "*asymmetrical*" (Utz et al., 2021) to express the unequal nature of PSR. Although PSR has two parties, one party and the other party, the connection is developed based on one

party's wish because the other party of PSR does not have consent to establish the relationship. In other words, people unilaterally develop this connection with the other party (Hwang and Zhang, 2018). Some scholars may argue that one of the aims of the other party is to establish a relationship with more people to gain more followers, and they know their fans and the relationship exists, so the connection is not one-sided. The other party knows the relationship exists with their fans, but it is a relationship with their fans as a group of people. Still, they do not know specific individuals in the relationship. Therefore, this connection is inequality, which means only one party know the exact other party, but the other party do not know who one party are (Lim et al., 2020; Feder, 2020). Therefore, this thesis proposes that a good PSR definition should imply the unequal nature of PSR.

### Theme 7: Psychological closeness

Psychological closeness is identified as a theme. Based on the analysis, scholars use 15 different words in the existing PSR definition to define what kind of connection is, such as *"psychological"* (Bernhold, 2019) and *"intimate"* (Dias et al., 2017). Dynamic emerges from this theme since previous research used various words and terms to reflect this psychological connection's various strengths and status. For example, some researchers assert that this connection is imitated, so they use *"close"* and *"intimate"* (Dias et al., 2017). However, *"passive"* as a keyword is also used by Lim and Kim (2018) to reflect another strength and status of PSR. Therefore, this thesis argues that a good PSR definition should imply the dynamic nature of PSR.

#### Theme 8: Time

Time is another theme identified in the existing literature. Some of the research argues that the development of PSR needs time, and some scholars regard PSR as a long-term connection. According to the analysis, "*continuing*" (Bernhold, 2019), "*durable*" (Bernhold, 2019), "*over time*" (Whang and Im, 2021), "*long-lasting*" (Balaban et al., 2022) and "*long-term*" (Leite and Baptista, 2022) as keywords, and scholars use them to describe PSR as a long-term connection. Indeed, PSR development needs time and repeated exposure and interaction with another party (Lim et al., 2020). Therefore, this thesis argues that a good PSR definition should imply time.

#### Theme 9: Develop

"Develop" is a theme identified in defining PSR. Seven words are used in the PSR definition, such as "*develop*" (Lee and Cho, 2020) and "*form*" (Hu et al., 2020). Although these words do not imply a specific time, such as "long-term", they suggest PSR may need time or over time. Therefore, this thesis argues that a good PSR definition should imply time.

### Theme 10: Illusion

Illusion as a theme represents one of the characteristics of PSR. In the reviewed PSR definition, five words are identified, such as "*imaginary*" (Han and Yang, 2018) and "*illusionary*" (Munnukka et al., 2019). The illusion shows that the connection between one party and the other party is not a real-life relationship imagined by one party. However, including illusion in the PSR definition is non-parsimonious since inequality also reflects the illusion characteristics of PSR (Murrow and Hyman, 1994).

In addition, the existing literature implies that PSR development needs time but neglects to include how the connection developed. Therefore, this thesis argues that a good PSR definition should also include how the connection developed.

Based on the review, the study proposes the working definition of PSR as **people unilaterally establishing dynamic connections with the other party (real people, media characters, and anthropomorphic objects) through media**. The new definition adds value to the existing PSR definitions in terms of clearly stating the dynamic nature of PSR. However, this working definition needs further validation. To focus on this issue is essential since the conceptualisation of a construct is the foundation of science (MacKenzie, 2003). Marketing researchers need to focus on the conceptualisation issues at the first stage of research, which will benefit the subsequent theory testing and development (Gilliam and Voss, 2013).

### 2.9.2 The need to explore the PSR development process

Reflecting on the analysis from the previous section, it is clear that PSR should be regarded as a dynamic and complex construct. Nevertheless, the current literature treats PSR as a static and monolithic construct, failing to capture its complex and dynamic nature (Tukachinsky and Stever, 2018). Conceptually, some studies highlight the temporal aspect of PSR development, underscoring that it is not a static construct. For instance, the formation of PSR necessitates viewers' repeated encounters with a media figure (Lim et al., 2020; Feder, 2020). This indicates that the relationship development with a media figure is a gradual process, unfolding over time (Whang and Im, 2021). Additionally, a handful of studies provide empirical evidence supporting the temporal element of PSR development (Pressrove and Pardun, 2016). Farivar et al. (2022) find that the duration of Instagram users' followings of influencers positively impacts PSR, further corroborating the dynamic aspect of PSR development.

While the existing literature recognises the complexity of PSR, empirical research often regards it as a more simplistic construct, primarily focusing on examining its antecedents and consequences, as detailed in Appendices 6 and 7. A notable oversight in these studies is their assumption that the quality, strength, and stages of PSR are uniform (e.g., Leite and Baptista, 2022; Brevers et al., 2021). While such studies contribute significantly to the development of PSR-related theory, they do not adequately capture and reflect the dynamic and complex nature of PSR.

The limited exploration of the PSR development process has been underscored in certain studies. Joshi et al. (2023) stresses the importance for future research to delve into the PSR development process and strategies for maintaining relationships with consumers. Tukachinsky and Stever (2018) have theoretically conceptualised the PSR process from a media standpoint. Drawing on Knapp's model of relationship development, they outline four stages in the PSR development process: initiation, experimentation, intensification, and integration. This framework significantly aids in understanding the stages of PSR development. Nonetheless, further investigation is needed for three key reasons.

Firstly, the PSR model proposed by Tukachinsky and Stever (2018) is mainly rooted in the traditional media landscape (e.g., TV). This thesis shifts the focus towards influencer marketing, examining how consumers develop PSR with influencers within the social media context. Consequently, there is a necessity to modify the model to suit the influencer marketing context. The traditional media environment differs markedly from influencer marketing, particularly in terms of the interactive and personalised content prevalent on social media platforms. This leads to a shift in consumer behaviour, from passive consumption to active engagement. Moreover, as discussed in Section 2.8, the conditions underpinning PSI and PSR differ between offline and online environments, highlighted by

a greater potential for PSR development and a quicker pace of PSR formation in the digital environment. Thus, it is important to modify the PSR model that can capture consumers PSR development with social media influencers.

Secondly, while Tukachinsky and Stever (2018) identify key predictors in the PSR development process, they do not hypothesise about the relationships among these predictors, nor do they empirically test the model. Their work lays a theoretical foundation for understanding the PSR development process, serving as a valuable starting point that offers initial insights. However, the absence of empirical testing leaves the model's validity and reliability in question, and its practical applicability remains unclear.

Thirdly, recent advancements in PSR research suggest that the model proposed by Tukachinsky and Stever (2018) could be refined to achieve a more profound understanding of how individuals develop PSR and the outcomes of these relationships. Over 40 high-quality journal articles have contributed further insights into the antecedents and consequences of PSR. Notably, attributes such as trustworthiness (Yuan and Lou, 2020) and perceived value (Zhong et al., 2021) have been identified as significant predictors in the development of PSR. Similarly, positive outcomes like brand attitude (Balaban et al., 2022) and purchase intention (Wahab et al., 2022) have been linked to PSR in the context of social media. Given the evolving landscape of PSR research, there is a compelling need to refine the model in light of the latest literature.

## 2.10 Chapter Summary

This chapter reviews the existing literature on PSI and PSR, identifies the current arguments in PSI and PSR, and identifies the research gaps. By adopting the historic systematic literature review approach, this chapter reviews 172 PSI and PSR studies and finds that existing research fails to clarify the main domain characteristics of PSI and PSR in online and offline environments. Following an examination of existing definitions and the criteria for good definition as proposed by Murrow and Hyman (1994), this review formulates seven prerequisites for differentiating PSI and PSR in traditional settings, along with an additional seven prerequisites in the online environment.

This chapter also discusses the changes in PSI and PSR in online environments (compared to traditional environments) and the expansion of their conceptual scope. As technology

continues to evolve, the definition scope of PSI and PSR can be expected to expand. In addition, using a typology approach, potential conditions for PSI and PSR in the online environment are discussed.

Moreover, employing content analysis techniques, this chapter conducts a critical evaluation of the existing definitions of PSR, identifying ten key components. However, given the discovery of several issues in the conceptualisation of PSR and theoretical disputes concerning its main domain characteristics, the second research gap has been identified as the necessity to reconceptualise PSR. To confirm the key components of PSR and to gain an in-depth understanding of PSR, a qualitative study is designed using semi-structured interviews. The qualitative study design is presented in Chapter 4, and the research findings are discussed in Chapter 5. The refined conceptualisation of PSR is subsequently evaluated by experts, with the final PSR definition presented in Chapter 8.

Additionally, this chapter highlights the complex and dynamic nature of PSR and underscores the importance of investigating the PSR development process, delineating this as the third research gap. A qualitative study using semi-structured interviews has been designed to gain novelty and an in-depth understanding of the PSR development process, which is presented in Chapter 5. Based on existing literature, Chapter 6 involves theoretically constructing the PSR development process. Empirically validating the PSR development model is detailed in Chapters 7 and 9. To systematically address the above-mentioned two research gaps, the next chapter outlines the analytical approach adopted in this thesis, which includes six research phases, laying the philosophical foundation for the research design.

## Chapter 3: Analytical approach

## 3.1 Chapter introduction

The last chapter critically reviewed offline and online PSI and PSR literature, identifying pivotal research gaps to be addressed in this thesis. This chapter presents the methodological decisions made in this thesis to address the research gaps and their underlying rationale. These choices are crafted to fill the research gaps scientifically underscored in Chapter 2, with a comprehensive discourse on the research process and its justifications. Notably, these methodological decisions are congruent with one another and align seamlessly with the research objectives.

The structure of this chapter is outlined as follows. Initially, it elaborates on post-positivism as the guiding research paradigm for this investigation, with a particular emphasis on two fundamental underpinnings: ontology and epistemology. Subsequently, it examines the analytical framework and the exploratory sequential mixed-methods design. The thesis comprises six phases, each with its detailed rationale and justification thoroughly discussed.

### 3.2 Research paradigm

The whole research is guided by the research paradigm, which is the foundation of research (Ugwu et al., 2021). A research paradigm embodies a set of assumptions, beliefs and principles guiding the researchers to formulate the research focus, the theory development process, and the research design (Marsden and Littler, 1996). Researchers need to consider these assumptions and develop coherent, consistent and well-thought-out assumptions to design credible and scientific research (Easterby-Smith et al., 2018). Two fundamental assumptions are critical for researchers to consider, the assumptions concerning the nature of reality (ontology) and the assumptions regarding the nature of knowledge (epistemology) (Gannon et al., 2022). Given the diverse backgrounds and experiences of researchers, they possess varying assumptions about the nature of reality and knowledge, which subsequently influence their choice of methodology and interpretation of data (Goulding, 1999). The author of this thesis has adopted the post-positivism stance towards these assumptions, guiding the entire research design.

Ontology pertains to how researchers perceive reality (Bryman and Bell, 2011). The researcher adopts a realist perspective, positing that reality exists but cannot be precisely captured (Zinkhan and Hirschheim, 1992). It is assumed that reality is a complex system that exists independently of human perception, rather than being contingent upon the existence of social actors (Hunt, 1993). This research is centred on influencer marketing, examining the PSR between consumers and influencers on social media, and posits that this phenomenon objectively exists. Social media facilitates communication between individuals. As long as social media continues to provide this capability, users will form relationships with influencers they favour and imitate the behaviours of these influencers. For example, people engage in influencer marketing activities (Leite and Baptista, 2022). On one hand, people watch influencers' posts, videos and live streaming on social media and like and comment on these contents to represent their interactions (Lim et al., 2020). These behaviours exist globally on social media in the real world. On the other hand, individuals also contemplate influencers and process information regarding them (Shen et al., 2022), and these cognitive processes also hold objective existence.

While reality indeed exists, capturing it in its entirety remains elusive for researchers. The diversity of researchers' backgrounds and cultures significantly influences their perceptions of reality (Zinkhan and Hirschheim, 1992). As a result, researchers can comprehend reality through developed models and interpret it according to their beliefs, knowledge, and existing theories. Nevertheless, creating law-like generalisations and uncovering the truth, as advocated by positivism, proves challenging (Zinkhan and Hirschheim, 1992). The critique of positivism has intensified as researchers acknowledge their limitations in fully and objectively explaining reality or acquiring definitive insights into complex social phenomena (Marsden and Littler, 1996; Hunt, 1993). These limitations stem from the constraints imposed by their language and culture, making it impossible to transcend these boundaries (Zinkhan and Hirschheim, 1992).

Consequently, achieving objective and absolute perspectives is unattainable. The subject of this thesis, influencer marketing, is a continuously evolving phenomenon shaped by the variety of social media platforms, the expanding spectrum of influencers, and the intricacies of influencer mechanisms, underscoring the complexity of influencer marketing. Additionally, the dynamic nature of PSR is influenced by individuals' emotional and psychological processes, their unique needs, and also by social and cultural factors. Hence, while researchers can strive to approach reality as closely as possible, attaining a precise understanding of the truth remains unfeasible (Hunt, 1993).

Epistemology concerns the nature of knowledge, perceived as "a way of looking at the world and making sense of it" (Al-Ababneh, 2020), constructed through a scientific process that builds upon new observations and is informed by existing theories (Bryman and Bell, 2011). Given the tentative nature of knowledge, theories are refined and expanded based on new observations and evidence. Researchers should employ scientific methods to test and develop theories, thereby enhancing our understanding and explanation of reality (Baker, 1995). The author of this thesis adopts the view of modified objectivity, advocating that researchers should aim to capture and measure phenomena objectively. Nevertheless, complete objectivity is unattainable due to potential biases. Despite this, researchers should aim to be objective and try to reduce bias as much as possible to strive to get as close as possible to the truth (Hunt, 1993).

The bias can stem from multiple sources. First, context significantly influences knowledge, especially since marketing is a context-driven discipline, making it challenging to formulate law-like generalisations without context (Gamble et al., 2011; Sheth and Sisodia, 1999). For the focal construct of this thesis, PSR, individuals can generate PSR with various parties through various media, such as people developing PSR with influencers (Sokolova and Perez, 2021), brands (Liao et al., 2022) and virtual assistants (Yuan et al., 2021). Therefore, it is impracticable to investigate PSR without a specific research context.

Secondly, the research method possesses inherent limitations that may introduce bias (Podsakoff et al., 2012). For instance, qualitative research approaches are limited in their generalisability, whereas quantitative research methods may lack depth in understanding and explanation (Hair et al., 2021). Furthermore, the selection of research method and the interpretation of data are influenced by the researcher's prior knowledge and experience, which can also contribute to bias.

To endeavour to approach reality more closely and attain objectivity, the researcher ought to contemplate the potential bias and its impact on the research process (Hunt, 1993). Reflecting on the biases discussed earlier, the focus of this thesis is specifically on PSR within the context of influencer marketing, aiming to comprehend how individuals establish PSR with influencers and the subsequent influence of PSR on consumer decision-making. Secondly, the researcher employs triangulation of methods to mitigate the limitations inherent in any single method (Oppermann, 2000). This approach helps to address the weaknesses of a specific method, thereby offering a more rounded perspective. A mixedmethods strategy is adopted, balancing the shortcomings of both qualitative and quantitative research, and providing the flexibility needed to tackle more complex problems (Harrison and Reilly, 2011). Moreover, methodological decisions in this thesis are made with clear rationale, ensuring adherence to the scientific research process.

### 3.3 Overall research design

Guided by the assumptions discussed above, the overall research design incorporates triangulation of methods to overcome the limitations inherent in any single approach and aims to reduce bias associated with a singular research methodology (Oppermann, 2000). Method triangulation involves employing various methods and techniques to fill research gaps, thereby enhancing the credibility and validity of the research findings (Archibald, 2015). Concurrently, diverse approaches facilitate a more comprehensive and in-depth exploration of the research phenomenon (Gibson, 2016). In general, there are three types of method triangulation: multiple quantitative approaches, multiple qualitative approaches, and mixed-methods approaches (Harrison and Reilly, 2011). This study employs a mixed-methods approach.

The mixed methods design combines both quantitative and qualitative methods (Johnson et al., 2007). Although not traditionally common in marketing, an increasing number of scholars are adopting this approach (Harrison and Reilly, 2011). Aligned with postpositivism, which typically utilises quantitative methods to test theories and ensure objectivity, the researcher also aims for objectivity through quantitative methods. However, given the complexity of reality, employing qualitative methods is advantageous for providing new and in-depth insights and perspectives from consumers. This contributes to theory development and facilitates a closer approximation to reality (Healy and Perry, 2000). Moreover, both quantitative and qualitative approaches have their own sets of strengths and weaknesses; thus, their integration can mitigate the biases inherent in a single-method approach, compensate for individual methodological weaknesses, and yield more robust conclusions (Harrison and Reilly, 2011; Gunbayi, 2020; Dawadi et al., 2021). Another benefit of the mixed methods approach is its capacity to offer diverse viewpoints by combining the perspectives of existing literature and participants, thereby enriching and deepening the overall understanding of the PSR phenomenon (Bryman, 2006).

Some scholars may criticise the mixed methods approach for potentially producing conflicting findings, underscoring the importance of cross-validating results to confirm

research accuracy (Freshwater, 2007). Additionally, the mixed methods approach suffers from a lack of standardised procedures and guidelines for data analysis, leading some studies to neglect thorough analysis and presentation of either quantitative or qualitative findings (Harrison and Reilly, 2011). For instance, Harrison and Reilly (2011) observe that in marketing, many exploratory mixed-methods designs lack detailed development and data analysis for the qualitative component. To mitigate this potential issue, this thesis commits to conducting comprehensive analyses of both quantitative and qualitative data, ensuring that findings and discussions for each are thoroughly presented.

The researcher has chosen an exploratory sequential mixed methods design, wherein the sequence of employing quantitative and qualitative approaches significantly influences the research design and its emphasis, with different design types facilitating the achievement of various research objectives (Harrison and Reilly, 2011). This thesis is guided by three research objectives, each of which necessitates distinct consideration. The first research objective aims to identify the differences between PSI and PSR in online and offline contexts. Given that this objective primarily engages with the ongoing debates within the existing literature rather than aiming to uncover new aspects of reality, it necessitates an initial understanding of the current body of knowledge. This approach aligns with the postpositivist perspective, which holds that new knowledge can be informed by existing theories (Synder, 2019). Therefore, the attainment of the first research objective is best approached through a systematic literature review, thus excluding it from the exploratory sequential mixed-methods design's purview.

The exploratory sequential mixed methods design is deemed appropriate and optimal for addressing research objectives 2 and 3, given their nature. Research objective 2 is aimed at understanding the nature and conceptualisation of PSR and PSI, which serves as a necessary foundation for tackling research objective 3: exploring the process of PSR development. In essence, a comprehensive understanding of PSR is required before the process of its development can be effectively examined. The current literature's failure to fully encapsulate the dynamic aspects of PSR necessitates a qualitative approach for an in-depth exploration, which aids in conceptualising PSR and delineating its key components (Hair et al., 2021). Concurrently, the existing literature's inadequacy in explaining the PSR development process, coupled with the employment of a qualitative approach, sets the stage for subsequent quantitative research (Crick, 2020). Insights derived from qualitative studies enable quantitative research to test emerging theories and generalise findings from an objective standpoint (Crick, 2020). This ongoing back-and-forth process, employing both

inductive and deductive reasoning, significantly contributes to the development of theory (Zaltman et al., 1982).

Furthermore, the necessity to develop a new PSR measurement instrument for measuring online PSR further advocates for the application of the exploratory sequential mixed methods design. Current PSR scales fail to capture the main domain characteristic of PSR as proposed in this thesis, thereby underscoring the imperative to develop a new PSR measurement to enhance construct validity (MacKenzie, 2003; Peter, 1981). While qualitative research is not an essential step in the scale development process, it provides significant advantages by facilitating the generation of more relevant scale items and laying a solid foundation for the initial development of the scale item pool (Churchill, 1979). As indicated by Harrison and Reilly (2011), the exploratory sequential mixed methods design proves to be effective in the creation of new scales items, which is evident in the current literature. Notable instances of employing the exploratory sequential mixed methods design in developing new measurement tools include the online brand community engagement scale (Baldus et al., 2015) and the luxury brand attachment scale (Shimul et al., 2019).

The analytical framework delineated in Figure 3.1 outlines six distinct phases within this research, each contributing to the comprehensive exploration of parasocial phenomenon. Phase 1 initiates the study with a systematic literature review aimed at distinguishing between PSI and PSR and have an in-depth understanding of parasocial phenomenon. This phase sets the foundational understanding necessary for subsequent research stages. In Phase 2, a qualitative study is employed through semi-structured interviews. This approach is designed to deepen the conceptualisation of PSI and PSR, and PSR development process. Following this, Phase 3 involves conducting an expert survey. This phase is critical for evaluating the conceptualisation of PSR and the new developed PSR scale items, offering an opportunity to gather feedback from experts with specialised knowledge in the field.

Phase 4 progresses to quantitative research method, utilising online survey to validate the PSR scale. This phase is pivotal in ensuring the reliability and validity of the PSR measurement instrument. Similarly, Phase 5 continues with the use of an online survey, further refining and validating the PSR scale. This ongoing process is essential for refining and validating the PSR scale. This ongoing process is essential for refining and validating the PSR scale. This ongoing process is essential for refining and validating the PSR scale. Finally, Phase 6 conducts online consumer survey aimed at examining and confirming the PSR development process. This phase seeks to empirically test the theoretical model developed to understand how consumers develop PSR with social media influencers. The rationale for using these methods is discussed below.

### 3.3.1 Phase 1: Systematic literature review

A systematic literature review is an evidence-based research methodology that enables researchers to methodically identify and review relevant existing research to fulfil specific research objectives (Siddaway et al., 2019). Researchers undertake distinct steps to rigorously conduct the systematic literature review, aiming to yield comprehensive, transparent, and reproducible findings within a particular domain (Tranfield et al., 2013; Siddaway et al., 2019). Consequently, employing a systematic literature review as the initial phase of this study facilitates a profound comprehension of PSI and PSR and aids in distinguishing between these two concepts. This approach addresses ongoing debates in the extant literature (Hunt, 2011; Paul et al., 2020). Utilising this method can efficiently address Research Objective 1: to identify differences between PSI and PSR in the offline and online environments.

Hence, this systematic literature review targets Research Objective 1 to resolve the persistent definitional debate and propose prerequisites to identify PSI and PSR in both offline and online contexts. Specifically, this systematic literature review seeks to:

- propose prerequisites that can distinguish PSI and PSR in offline and online environments
- explore how PSI and PSR change from offline to online environments

Adopting a precise and reliable procedure in this systematic literature review helps to diminish bias, reduce subjectivity, and generate more feasible and beneficial solutions for implications (Snyder, 2019). Neglecting the scientific review process could adversely affect the credibility of research outcomes. Moreover, a systematic literature review entails evaluating the quality of the papers reviewed, which might lead to the exclusion of low-quality papers, possibly restricting insights (Xiao and Watson, 2017). Nonetheless, in this instance, the systematic literature review's goal is to enhance the understanding and conceptualisation of PSR, emphasising the importance of reviewing high-quality papers.

## **Figure 3.1 Analytical framework**



Research objective 1: To identify differences between PSI and PSR in the offline and online environments Research objective 2: To propose new definitions of PSR and PSI based on their unique prerequisites Research objective 3: To explore the formation process of PSR and test its outcomes

RQ1: Should PSI and PSR be treated as distinct concepts?

RQ2: How does the complexity of the online environment further inform the conceptualisation of PSI and PSR?

RQ3: How des PSR with influencers on social media platforms develop, and what are the outcomes of PSR?

## 3.3.2 Phase 2: semi-structured interview

This study employs a qualitative approach, utilising semi-structured interviews to achieve Research Objective 2: To propose new definitions of PSR and PSI based on their unique prerequisites and Research Objective 3: To explore the formation process of PSR and test its outcomes. This approach enables a profound understanding of PSR from the consumer's viewpoint, significantly contributing to the conceptualisation of PSR. Furthermore, it facilitates a comprehensive exploration of how consumers develop PSR with influencers, thus aiding in investigating the formation process of PSR.

Among various qualitative methods, such as interviews and focus groups, interviews are deemed more suitable for this research. Although focus groups are recommended for initial item pool generation (Churchill, 1979), interviews provide deeper insight into individual attitudes and independent data, surpassing what is typically achieved in focus group settings (Stokes and Bergin, 2006). Focus groups, characterised by their group discussion format, can sometimes lead to ideas being influenced by the group's dominant views, a limitation acknowledged by some respondents (Stokes and Bergin, 2006). In contrast, interviews, being conversational in nature, offer a more focused environment where individuals can express their thoughts and feelings about their relationship with influencers more freely and comprehensively. Hence, the interview method is preferred for its ability to elicit more depth and personal responses.

The semi-structured interview, a widely utilised interview format, refers to the technique in which the interviewer asks questions and communicates with the interviewees based on a drafted question but also has the flexibility to ask probing questions to get in-depth responses (Bryman and Bell, 2011). During the interview process, the interviewee has a one-on-one conversation with the interviewer, which can limit their scope of thought and prevent them from being inspired by others (Stokes and Bergin, 2006; Hoover et al., 2018). To overcome this limitation, the interviewer needs to utilise professional techniques to create a detailed interview guide that can elicit deeper insights from the interviewee (Bryman and Bell, 2011).

Thus, semi-structured interviews are instrumental in achieving Research Objectives 2 and 3 by:

- obtain an in-depth understanding of PSR from the consumers' perspective.
- contribute to the conceptualisation of PSR.

- gain insights into the PSR development process.
- generate more relevant PSR scale items for PSR scale development.

## 3.3.3 Phase 3: expert survey

The expert survey is a crucial phase in the development of measurement tools, enhancing face validity and contributing to the refinement of PSR definitions (Hardesty and Bearden, 2004; Rossiter, 2002). It involves a process wherein experts assess proposed scale items, ensuring they accurately measure the construct in question (Hardesty and Bearden, 2004). Although face validity and content validity are often used interchangeably, face validity specifically refers "*the extent to which the "objects" (people) that were measured with the instrument agreed that the instrument was valid*" (Knapp, 2002, p.102). The significance of face validity lies in its fundamental role in ensuring construct validity (Churchill, 1979; Hardesty and Bearden, 2004).

The selection of experts is critical, as their familiarity with PSR-related research could significantly influence the survey's outcomes. Experts are likely to offer meaningful suggestions if they have a comprehensive understanding of PSR studies. Therefore, this study identifies academic experts based on their prior research experience and publication records in the field of PSR, guaranteeing their capacity to provide constructive feedback. Given that researchers may offer varying perspectives and recommendations, the complexity and depth of these suggestions necessitate careful consideration. Thus, it is important to establish a clear criterion for item retention prior to the analysis of data (Hardesty and Bearden, 2004).

In this research, the expert survey effectively addresses Research Objective 2: to propose new definitions of PSR and PSI based on their distinct prerequisites, and Research Objective 3: to explore the formation process of PSR and test its outcomes. The experts not only offer valuable recommendations for amending the PSR definition to meet Research Objective 2 but also contribute to the development of PSR measurement tools. This, in turn, facilitates the precise measurement of PSR and the empirical testing of the PSR model. Therefore, the objectives of the expert survey are to:

- increase face validity.
- improve the conceptualisation of PSR.

• reduce scale items of PSR scale.

### 3.3.4 Phase 4: online consumer survey

In Phase 4, online survey is used for testing new developed PSR scale validity and reliability. This is the first step in testing the newly developed PSR scale validity and reliability, which can effectively help to address Research Objective 3: to explore the formation process of PSR and test its outcomes.

Surveys are among the most prevalent methods in quantitative research, involving data collection through standardised questions (Bryman and Bell, 2011). They enable the documentation of people's attitudes and opinions via self-report (Saunders et al., 2023). Surveys can be administered in various formats, including online, email, and telephone. This study opts for the online survey due to its ability to reach participants without geographical limitations (Bryman and Bell, 2011). Online surveys offer considerable advantages over other formats, such as being cost-effective and straightforward to implement (Claycomb et al., 2000; Couper, 2011). Nonetheless, given that surveys rely on self-reporting, they are susceptible to potential biases. For instance, differences in respondents' response styles may introduce response bias (Baumgartner and Steenkamp, 2001). Additionally, some participants may not fully engage with online surveys, which could decrease the validity of the data collected (Kostyk et al., 2019). To mitigate these issues, this study employs various data cleaning techniques to identify and exclude inattentive responses, such as analysis of response times (Leys et al., 2013; Berger and Kiefer, 2021) and checks for extreme responses (Hyman and Sierra, 2012).

This study aims to address research objective 3 by developing an appropriate PSR scale. In particular, the consumer survey helps to:

- reduce PSR scale items (Gerbing and Anderson, 1988).
- explore PSR dimensions (Gerbing and Anderson, 1988).
- check the PSR scale's internal reliability and validity.

#### 3.3.5 Phase 5: online consumer survey

Phase 5 of the thesis involves the use of an online survey method, serving as the subsequent step in evaluating the reliability and validity of the newly developed scale. Validation is an ongoing procedure that generally requires multiple phases of testing to ensure the accuracy and reliability of a measurement tool. During this phase, the PSR scale is subject to additional tests to assess its internal consistency, test-retest reliability, and construct validity. This step is imperative for the refinement of the PSR measure, guaranteeing its applicability for the subsequent study and future study use. Thus, the objective of this phase is to contribute to Research Objective 3: to explore the formation process of PSR and test its outcomes. The specific aim for this phase includes:

• to refine and valid the PSR scale based on Phase 4.

## 3.3.6 Phase 6: online consumer survey

In Phase 6, the study employs an online consumer survey as a strategic tool for testing the conceptual framework, with a specific focus on Research Objective 3: to explore the formation process of PSR and assess its outcomes. This phase is critical for empirically examining the PSR development process model, providing significant insights into the formation of PSR and the impact of PSR on consumers subsequent behaviour. The aims of this phase include:

- to empirically test PSR development process model.
- to confirm PSR dimensions.

## 3.4 Chapter Summary

This chapter focuses on the research philosophy and the overarching research design. Initially, it introduces the guiding post-positivist view of the entire research design. From an ontological standpoint, the researcher embraces a realist perspective, positing that reality exists, albeit its precise capture remains elusive (Zinkhan and Hirschheim, 1992). Epistemologically, the thesis author advocates for a stance of modified objectivity, arguing that researchers should endeavour to capture and measure phenomena objectively. Yet, it is acknowledged that absolute objectivity is unattainable due to inherent biases. Nonetheless, researchers are urged to strive for objectivity and minimise bias to approximate the truth as closely as possible (Hunt, 1993).

Moreover, this chapter outlines the comprehensive research design, informed by an exploratory sequential mixed methods approach, prioritising qualitative research followed by quantitative research. The initial phase involves systematic literature review to thoroughly grasp the concepts of PSI and PSR, detailed in Chapter 2. The second phase consists of semi-structured interviews to gain deeper insights into PSR and to create an initial pool of items for the PSR scale development. The intricate design of these semi-structured interviews is discussed in Chapter 4. The third phase utilises an expert survey to collect feedback on the face validity of PSR scales and to refine the definition of PSR. Subsequently, phases four and five encompass two online consumer surveys aimed at assessing the reliability and validity of the newly developed PSR scale. Finally, phase six involves an online consumer survey to empirically test the conceptual framework associated with the PSR development process. The meticulous quantitative research design is elaborated in Chapter 7. The next chapter will elaborate on the detailed design of the qualitative phase (Phase 2).

## Chapter 4: Qualitative methodology for exploring PSR

### 4.1 Chapter introduction

The previous chapter introduced the overall exploratory sequential mixed-methods design, underpinned by post-positivism. This chapter concentrates on the qualitative research design, specifically employing semi-structured interviews as the approach to address the second research objective: proposing new definitions of PSR and PSI based on their distinct prerequisites, and the third research objective: exploring the formation process of PSR development and testing its outcomes. The selection of semi-structured interviews with consumers who have cultivated long-term relationships with influencers allows for:

- Obtain an in-depth understanding of PSR from the consumers' perspective.
- Contribute to the conceptualisation of PSR.
- Gain insights into the PSR development process.
- Generate more relevant PSR scale items for PSR scale development.

The structure of this chapter is outlined as follows. Firstly, this chapter introduces the design of the interview guide and the sampling approach. Subsequently, it provides a detailed description of the data collection process, the characteristics of the interviewees, and the techniques employed for data analysis. Additionally, this chapter addresses potential biases and discusses measures undertaken to enhance the rigour of qualitative research.

### 4.2 Interview guide

An interview guide consists of a pre-prepared list of questions that are posed during an interview (Bryman and Bell, 2011). Interview guides can take various forms, such as visual prompts; however, this study utilises a detailed list of questions to enhance the dependability of the research (Bryman and Bell, 2011). In order to create an interview guide that is both clear and logically structured, this study follows the structure proposed by Kallio et al. (2016). The interview guide is divided into five sections, detailed in Appendix 8.

The first section of the interview guide is an introduction outlining essential information to be communicated to the interviewees. This includes reminders of the interview's research topic and the ethical considerations involved. Particularly, it emphasises the importance of seeking the interviewee's consent to record the conversation. Recording is crucial for ensuring transcription accuracy and for generating objective data for later analysis (Saunders et al., 2023). This introductory part of the guide is designed to put interviewees at ease and boost their confidence, encouraging them to share their experiences and thoughts openly (Magnusson and Marecek, 2015).

The second part of the interview guide consists of warm-up questions designed to help interviewees become comfortable with the interview context and the topic of influencer marketing, thus facilitating engagement in the subsequent discussion (Arsel, 2017). This study's warm-up questions are centred around the interviewees' use of social media and their general online behaviour. For instance, the question, "*Can you tell me about the video-based social media platforms you generally use*?" prompts respondents to reflect on their everyday social media activities, easing them into the topic of influencer marketing.

The third section delves into the interviewees' PSR with their favourite influencers on social media. Initially, the interviewer encourages the interviewees to think of one or two favourite influencers they follow, focusing their attention solely on these individuals for the subsequent questions regarding their relationships with influencers. The questions are formulated to gain insights into the interviewees' general perceptions of and interactions with these influencers, including interactions beyond the online environment.

The fourth section is aimed at uncovering how interviewees establish connections with influencers. It begins by inquiring about the first time the interviewees noticed the influencer and proceeds to investigate the evolution of their relationship. This part also seeks to understand the influencers' impact on the interviewees, including any influence-driven purchasing behaviours.

The fifth and final section of the interview guide focuses on the interviewees' perceptions of their relationships with influencers, asking, for example, "*In your own words, how would you describe your relationship with this influencer?*" This line of questioning is intended to shed light on the dynamic nature of PSR. The interview concludes by requesting demographic information, with the interviewees' consent, and expressing gratitude for their participation (Magnusson and Marecek, 2015).

The question design within the interview employs several techniques to explore the interviewees' experiences with their favourite influencers effectively. The core of the

interview utilises open-ended questions, to delve into the participants' experiences (Magnusson and Marecek, 2015; Saunders et al., 2023). Open-ended questions are chosen for their ability to allow respondents the flexibility to express their thoughts and opinions in their own words, enabling researchers to obtain developmental and nuanced answers (Saunders et al., 2023). Probing questions are strategically prepared to extract rich and indepth insights from the interviewees, based on the responses they provide (Bryman and Bell, 2011; Robinson, 2023).

To minimise bias and enhance the reliability of the collected data, the researcher carefully ensures that none of the questions are leading or loaded. Avoiding such questions is crucial as they could potentially influence the interviewees' responses, hindering the capture of their genuine feelings and perceptions (Granot et al., 2012; Cairns-Lee et al., 2021). This attention to question formulation aids in reducing bias and increasing the reliability of the findings (Cairns-Lee et al., 2021). Furthermore, the structuring of interview questions follows a progression from general to specific. This ordering is designed to initially encourage broader responses, gradually focusing on more detailed aspects of the interviewees' experiences (Brinkmann, 2022).

In addition, the interview questions are crafted to encourage interviewees to reconstruct their experiences rather than merely recall them. This approach aims to circumvent potential memory biases, allowing for a more accurate and detailed recounting of experiences (Granot et al., 2012). Through these meticulous question design techniques, the study aims to achieve a thorough and insightful exploration of the participants' interactions with their favourite influencers.

## 4.3 Sampling approach

The target population for this semi-structured interview should meet three criteria: being aged 18 years and over, being users of online and video-based social media, and having already established long-term relationships with influencers. Given that this study seeks to gain a deep understanding of PSR and how consumers develop PSR with influencers, it is essential to interview participants who are active on social media. Additionally, participants must have already established PSR with influencers to provide insights into their perceptions, feelings, and evaluations of their relationships with these influencers. Consumers with experience can offer more in-depth insights regarding the research topic (Churchill, 1979).

The existing literature supports selecting participants who have already developed a PSR with another party. For instance, it may require participants to identify their favourite influencers or celebrities (Lim et al., 2020). Accordingly, this study verifies the names and brief details of participants' favourite influencers and the duration of their acquaintance with these influencers.

Purposive sampling represents an optimal approach for identifying interviewees who meet the above criteria. As a form of non-probability sampling, purposive sampling necessitates that the researcher selects participants based on specific characteristics and criteria (Saunders et al., 2023). Since the objective of this study is to explore the PSR between consumers and social media influencers in depth, choosing participants through purposive sampling is crucial for selecting qualified participants. This method enables researchers to gain profound insights into PSR and the development of such relationships by consumers with social media influencers (Hair et al., 2021).

Regarding participant recruitment, the snowball sampling technique was employed (Robinson, 2013). Snowball sampling, a non-probability sampling technique, involves researchers starting with a few qualified participants and then expanding their sample through recommendations (Bryman and Bell, 2011). This technique proves efficient when the target participants are challenging to identify within the population (Hair et al., 2021). In this study, identifying participants who are long-term social media users and have followed an influencer for a long period was particularly challenging. Therefore, the research began with finding qualified participants known to the researcher and then expanded the participant group through recommendations.

One major advantage of snowball sampling is its time and cost efficiency, as all information about potential participants comes directly from the participants themselves, making it suitable for this semi-structured interview (Noy, 2008; Hair et al., 2021). Although snowball sampling may be criticised for its lack of representativeness, such as producing a homogeneous sample, this study aims to delve deeper into understanding PSR rather than to generalise the findings (Easterby-Smith et al., 2021). Furthermore, to minimise bias, the researcher sought participants from diverse backgrounds who followed influencers in various fields (Robinson, 2013).

This study aims to gather at least 20 respondents for the sample. Given that the sampling frame comprises individuals who have established long-term relationships with influencers,

regarded as a homogeneous population, a minimum sample size of ten is considered appropriate for such populations (Francis et al., 2010). Adopting the viewpoint of Marshall et al. (2013), to ensure research rigour and gather sufficient information from participants, a sample size of 20 to 30 respondents is deemed suitable. Moreover, this interview focuses on collecting in-depth information on how consumers evaluate their relationships with influencers, obviating the need for a larger sample size (Boddy, 2016). Additionally, this study employs data saturation to determine the adequacy of the sample size. If data analysis yields no new ideas, it indicates that the current sample size is sufficient and there is no need for further data collection (Guest et al., 2006).

### 4.4 Interview modality

The semi-structured interviews were conducted virtually using Zoom. Zoom is a professional collaboration platform that offers real-time video conferencing capabilities and has become widely adopted for interviews, particularly in the aftermath of COVID-19 (Oliffe et al., 2021). It is recognised as an easy-to-use tool for efficient data management during interviews (Archibald et al., 2019). Conducting interviews via Zoom from the comfort of their own homes likely makes candidates feel more at ease and more inclined to speak openly, thereby yielding richer data (Oliffe et al., 2021). An additional clear benefit of this method is its cost-effectiveness (Maulana, 2022). The flexibility in scheduling afforded by Zoom allows both interviewers and interviewees to choose convenient dates and times for the interviews, thus saving on potential expenses such as transportation and accommodation (Archibald et al., 2019). Furthermore, Zoom interviews are not constrained by geographical boundaries, enabling the researcher to engage qualified participants globally, which helps in reducing sampling bias.

Despite some scholars pointing out the limitations of online qualitative research, including a perceived lack of interaction and the inability to read body language effectively (Bruggen and Willems, 2009), current technologies have made it possible to conduct online "face-to-face" interviews. This study employed Zoom to facilitate interviews, enabling real-time verbal and non-verbal communication, as well as written discussions, through video (Zoom, 2023). Another potential challenge of online interviews is the dependency on the quality of the internet connection. Poor internet connectivity can adversely impact the quality of the interview; therefore, the interviewer ensures a stable and high-quality internet connection before commencing the interview.

#### 4.5 Data collection procedure

This study follows the ethical guidelines, having first received ethical approval from the College Research Ethics Committee before starting data collection (see Appendices 9 and 10). Before the formal semi-structured interviews commenced, the researcher undertook pilot tests with three qualified participants. These preliminary sessions aimed to validate the clarity and comprehensibility of the interview questions, assess the length of the interviews, and refine the data analysis process (Majid et al., 2017). Such pilot studies are invaluable for novice researchers, providing an opportunity to build confidence. By examining the procedures and outcomes of these pilot tests, the researcher can identify and rectify any weaknesses, thereby enhancing the robustness of the study (Malmqvist et al., 2019). Following these preparatory steps, the researcher proceeded with the formal semi-structured interviews.

The data collection for the semi-structured interviews was in Glasgow, UK, employing Zoom as the medium of communication. This phase spanned three months, from November 2022 to January 2023. Utilising a snowball sampling technique for participant recruitment, the initial point of contact was a group of acquaintances (Robinson, 2013). From this starting point, six respondents met the criteria and agreed to participate. These initial participants then recommended an additional nineteen respondents. Formal invitations to participate were extended via email (Sappleton and Lourenço, 2015). Upon confirming the eligibility and willingness of these prospective respondents, the researcher forwarded the Participant Information Sheet (Appendix 9) and Consent Form (Appendix 10) to them, subsequently arranging the interviews. Ensuring ethical considerations, the interviews were conducted only after receiving signed consent forms from the participants. In total, twenty-five interviews were carried out via Zoom, with the discussions transcribed for subsequent data analysis.

### 4.6 Characteristics of Interviewees

The profile of the 25 interviewees is presented in Appendix 11. Regarding the gender distribution, there are more females (n=16) than males (n=9). Initially, the researcher approached potential male and female interviewees. However, only a few male potential participants had followed influencers for a long time. In other words, it appears that females are more inclined to establish and maintain long-term relationships with influencers, and

PSR are more commonly associated with female users than male users. This finding aligns with previous research indicating a significant difference in the way male and female viewers interact and build relationships, emphasising that female viewers are more likely to engage in PSR with influencers than male viewers (Sokolova and Perez, 2021).

The age range of the interviewees spans from 24 to 45, with a mean age of 29. This is consistent with the global age distribution of social media users, where users aged 20-29 constitute the largest group at 32.2%, followed by those aged 30-39 at 22.2% (Newberry, 2023). The occupations of the interviewees vary, including students (40%), those working part-time (13.33%), and full-time employees (33.33%). The amount of time interviewees spends on social media ranges from 45 minutes to five hours. They utilise a variety of social media platforms, such as YouTube, TikTok and Instagram. The duration of the interviewes ranged from 15 to 45 minutes, with an average of 32 minutes. The influencers followed by the interviewees span a broad array of fields, such as lifestyle (e.g., Xenia Adonts), beauty and makeup (e.g., Lisa Eldridge), fitness (e.g., Greg Doucette) and entertainment (e.g., Sam Chui).

## 4.7 Data analysis process

This study utilised the Voice Memos application to record conversations, with the consent of the interviewees. The audio files were securely stored for transcription purposes. For transcription, this study employed Otter.ai, which offers a professional automatic transcription service capable of converting audio into editable text (Otter, 2023). Upon completion of the transcription process, the audio files were deleted to maintain confidentiality. NVivo software was utilised for the qualitative data analysis, serving as a professional tool for coding and analysing qualitative data (Woods et al., 2016).

The research incorporates two methods for data analysis: content analysis and thematic analysis (Bryman and Bell, 2011). Content analysis was conducted following a deductive approach, guided by the concept of PSR (Mayring, 2014). This method aids in gaining a comprehensive understanding of PSR and its conceptualisation (Razzaq et al., 2023). Content analysis is selected for its objective, systematic, and quantitative analysis of content, facilitating the identification of keywords and their frequency counts (Kassarjian, 1977; Bos and Tarnai, 1999).

Given the concept-driven nature of content analysis in this study (focusing on PSR), the researcher ensured familiarity with the definition of PSR through extensive literature reviews. This preparation allowed for systematic coding of keywords related to PSR while avoiding confusion with other constructs. Objectivity was maintained by considering words, including phrases and single words reflecting PSR, as the units of content.

Thematic analysis was employed as an additional method aligned with qualitative data analysis principles. This approach provides flexibility in data analysis and effectively identifies patterns and themes (Braun and Clarke, 2020), including both explicit and implicit ideas (Braun and Clarke, 2006). The study conducted two thematic analyses to achieve its objectives: the first aimed at deepening the understanding of PSR and generating relevant PSR scale items, while the second focused on exploring the PSR development process.

Following the guidelines by Braun and Clarke (2006), the analysis process involved four steps: transcription of data, thorough data familiarisation, inductive coding for insights into PSR through NVivo, and the transfer of initial codes to Excel for sub-theme and theme categorisation (Braun and Clarke, 2022). The identification of themes was theoretically grounded, with a continuous review of literature to ensure relevance and coherence. Some codes were eliminated due to redundancy or irrelevance, ensuring that the interpretation process was iterative, aiming to identify the most suitable themes. Finally, each theme was defined, and comparisons were drawn with earlier analyses of interview content and PSR definitions (Nowell et al., 2017).

## 4.8 Possible bias

To enhance the rigour and reduce potential bias in this study, peer debriefing sessions with supervisors were integrated. Conducted regularly, these sessions enabled a critical review of the research design, the development of interview guides, the conduct of the interview process, and the techniques used in data collection and analysis. This iterative consultative approach played a crucial role in refining the study's methods and interpretations, thereby safeguarding the integrity and credibility of the research findings and minimising bias.

Reflecting on and minimising bias is crucial for approximating reality more closely (Hunt, 1993). Consequently, the researcher identifies several biases that could impact the study's rigour, as detailed in Table 4.1. Despite previous discussions on enhancing rigour, a

comprehensive summary of potential biases and efforts to mitigate them is beneficial. In the context of this semi-structured interview study, biases may arise from five areas: interviewer bias, question bias, interviewee bias, sampling bias, and data analysis bias.

Firstly, interviewer bias is a significant concern. The interviewer plays a pivotal role in organising and conducting interviews (Saunders et al., 2023). Essential skills for effective interviewing include procedural, interpersonal, and reflexive capabilities (Hoover et al., 2018). Inappropriate or unprofessional conduct by the interviewer can adversely affect the interview process. To reduce this bias, the researcher follows Hoover et al.'s (2018) recommendation for training, systematically preparing through relevant courses to conduct interviews, design questions, and analyse qualitative data professionally.

Question bias can also affect outcomes. As interviews are fundamentally communicative processes, the construction and phrasing of questions can influence interviewees' responses. Non-scientific question types, such as leading and loaded questions, may not accurately capture interviewees' true feelings, introducing bias (Granot et al., 2012). To mitigate this, the study meticulously develops an interview guide devoid of leading and loading questions and employs scientific question design principles, such as using open-ended questions and a progression from general to specific queries (Saunders et al., 2023; Brinkmann, 2022). A pilot test was also conducted to ensure question comprehensibility before the formal study.

Interviewee bias represents another potential source of bias. The research aims for interviewees to provide accurate and detailed accounts of their experiences and feelings. However, factors such as negative emotions or discomfort can impact their willingness to share (Oliffe et al., 2021). Ensuring a comfortable and familiar interview setting can enhance the accuracy of the data provided (Oliffe et al., 2021). Therefore, this study implements several strategies to foster a positive environment for interviewees, including informed consent procedures and the use of comfortable online Zoom interviews. Moreover, to counter memory-related limitations, interview questions are crafted to facilitate the reconstruction of experiences rather than direct recall (Granot et al., 2012).

Sampling bias is another concern. Inadequate participant qualification can introduce bias. This study employs a purposive sampling strategy, targeting participants who meet three

Semi-structured interview justification						
Timeline	Objective	Rationale	Possible bias	Efforts to reduce bias		
November 2022 - February 2023	To address research objective 2: To propose a new definition of PSR (and PSI) based on their unique prerequisites. To address research objective 3: To explore the formation process of PSR development and test its outcome.	<ol> <li>Obtain an in-depth understanding of PSR from the consumers' perspective.</li> <li>Contribute to the conceptualisation of PSR.</li> <li>Gain insights into the PSR development process.</li> <li>Generate more relevant PSR scale items for PSR scale development.</li> </ol>	<ul> <li>Interviewer bias: Inappropriate or unprofessional behaviour and manner.</li> <li>Question bias: Includes leading, unclear, or lengthy questions.</li> <li>Interviewee bias: social desirability bias, emotional states, and discomfort.</li> <li>Sampling bias: Not qualified or non- representative participant selection.</li> <li>Data analysis bias</li> </ul>	<ul> <li>Interviewer training: Systematic qualitative research training, unbiased question design and data analysis techniques.</li> <li>Interview guide: Detailed interview guide with clear questions.</li> <li>Pilot testing: Practice the interview with three participants to check questions clarity and data analysis.</li> <li>Informed consent: Provide participant information sheets and collect signed consent forms.</li> <li>Purposive and snowball sampling: Ensure geographic diversity of interviewees.</li> <li>Transparency: Record all processes the using professional transcription and data analysis tools.</li> <li>Multiple data analysis approach: Employ both deductive content analysis.</li> </ul>		

# Table 4.1 Semi-structured interview justification

specific criteria outlined in the research design. This approach ensures only qualified individuals are interviewed.

However, a homogeneous sample may limit the generalisability of findings (Easterby-Smith et al., 2021). To address this, the study aims to include participants from diverse backgrounds and those following various influencer fields to minimise bias. Additionally, to counteract potential biases from small sample sizes, the study adheres to scientific sampling size guidelines, aiming for 20-30 participants and considering data saturation as a criterion for sample adequacy (Marshall et al., 2013; Guest et al., 2006).

Finally, the data analysis process itself can introduce bias. To counter this, the study utilises both content analysis and thematic analysis as data analysis techniques, thereby aiming to reduce bias (Archibald, 2015). Pre-defining data analysis techniques and adhering to scientific guidelines, such as those proposed by Braun and Clarke (2006) for thematic analysis, further ensure the reduction of bias.

## 4.9 Chapter Summary

This chapter discusses the methodology of Phase 2 of the thesis, specifically the semistructured interview, effectively addressing research objectives 2 and 3. It focuses on the qualitative research design, explaining and justifying the use of semi-structured interviews. Initially, the chapter introduces the design of the interview guide and the techniques employed to develop questions. Subsequently, the purposive and snowball sampling approaches are examined. Utilising these two sampling methods, a total of 25 interviewees were recruited for this study. The chapter then details the online interview processes and the profiles of the 25 interviewees. Furthermore, it discusses the data analysis process, particularly the content analysis and thematic analysis approaches. Lastly, the chapter summarises this study's potential biases and the efforts made to mitigate them. The following chapter will present the findings from the semi-structured interviews, including the results of content analysis and thematic analysis to understand the PSR, and the process by which consumers establish PSR with their favourite influencers. The evaluation of a new PSR definition and the development of a PSR measurement instrument in the online environment are presented in Chapter 8.

## Chapter 5: Qualitative findings on consumer insights into PSR

## 5.1 Chapter introduction

The previous chapter delineated the design of the semi-structured interviews. This chapter is dedicated to presenting the findings from interviews with 25 participants, exploring their perceptions of the relationships they have developed with their favourite influencers, alongside the processes they have developed with them. These qualitative study findings are pivotal for an in-depth understanding of PSR and contribute significantly to the conceptualisation of PSR. This qualitative aspect of the findings aids in investigating the development process of PSR and in formulating the conceptual framework presented in Chapter 6.

The structure of this chapter is organised as follows. Initially, it presents the findings related to consumers' perceptions of PSR, utilising content and thematic analysis. Subsequently, criteria for a good definition of PSR are proposed. Then, through thematic analysis, the four stages of the PSR development process are examined, along with the critical factors influencing each stage, including exposure initiation, experimentation, intensification, and integration. Finally, the chapter concludes with a discussion of the implications of the qualitative research findings.

## 5.2 Consumer's perception of PSR

This study garnered extensive data on consumer perceptions of PSR through interviews with 25 participants who have established PSR with social media influencers. The interviews were conducted with the interviewees' consent and recorded using the Voice Memos app. Subsequently, the audio files were transcribed into editable text by Otter.ai. Following a verification process to ensure the text's accuracy against the audio files, the transcribed text was employed for data analysis. The study utilises two data analysis methodologies: content analysis and thematic analysis (Bos and Tarnai, 1999; Braun and Clarke, 2020). The adoption of these two approaches is strategic, as it allows for the achievement of research objectives and the mitigation of bias (Braun and Clarke, 2020; Mayring, 2014).

### 5.2.1 Deductive content analysis

The content analysis method adopts deductive logic, anchored in the PSR concept (Mayring, 2014). Through this method, it facilitates a profound comprehension of PSR and aids in the conceptualisation of PSR (Razzaq et al., 2023). The selection of content analysis is attributed to its being an objective, systematic, and quantitative scientific technique for content examination (Kassarjian, 1977). Furthermore, content analysis is instrumental in identifying keywords and quantifying their occurrences (Bos and Tarnai, 1999).

Employing a deductive reasoning, the study discerns 35 keywords via content analysis. The researcher ensures that each keyword aptly reflects PSR to affirm face validity. It is imperative to categorise these keywords for coherent analysis. Consequently, the 35 keywords are classified into seven sub-categories and six categories, detailed in Table 5.1.

Four categories are validated from the analysis of PSR definitions in Chapter 2, comprising "connection", "psychological nature", "inequality", and "long-term". Additionally, two novel categories have emerged: "value" and "influence".

## **Category 1: Connection**

Consistent with the analysis of PSR definitions provided earlier, the category "connection" is recognised as encapsulating the main domain of PSR as the linkage between one party and another. As indicated by the existing literature, PSR reflects the connection between viewers and media characters (Yuan et al., 2016; Feder, 2020). Within this category, three key words are identified: "connection", "relate", and "bond". "Connection" emerges as the term most frequently used by interviewees, followed by "relate" and "bond". These keywords all reflect the connection nature of PSR. Some interviewee sheds light on their connections with influencers:

'Well, I would say we get the connection with these influencers.' Interviewee 8

' I do think it is quite a personal connection, like I am quite invested in them. Like, I get value from them each day, I feel like they bring me inspiration. And it is also quite like a personal thing.' Interviewee 18

This "connection" category confirms the findings of the systematic literature review, underscoring that connection is the main domain characteristic of PSR, reflecting the relational nature of PSR. The frequent use of "connection" by interviewees highlights its significance in describing their relationships with media characters and influencers.

Content analysis of interview - PSR						
No	Category	Sub-category	Key words			
1	Connection		Connection			
		Connection	Bonds			
			Relate			
2	Psychological nature		Psychological			
			Companion			
			Private			
			Deep			
		Psychological nature	Loyal			
		r sychological flature	Distance			
			Dynamic			
			Close			
			Intimacy			
			Emotional			
		Belongingness	Fans			
			Followers			
			Big sister			
			Friends			
			Mentors			
			Big brother			
	Value		Purposeful			
3		Value	Encouragement			
			Important			
4	Influence	Influence	Influence			
5	Inequality	Inequality	One-way			
6	Long-term	Long-term	Long-term			

 Table 5.1 Content analysis of the interview

## Category 2: Psychological nature

The category "*psychological nature*" mirrors the nature of PSR, aligning with the prior analysis of PSR definitions. As indicated by the current literature, PSR is individual's psychological process that can generate the intimate feelings (Bernhold, 2019; Dias et al., 2017). Participants develop diverse psychological connections with the other party (Lee and Park, 2022; Noor et al., 2022). This group contains twelve keywords, with the majority

corroborating the previous PSR definition analysis in Chapter 2.9. For instance, several interviewees describe the connections as "*intimate*", indicating:

### 'I think, from my perspective, it is a very intimate relationship.' Interviewee 2

New keywords have been introduced to further elucidate the dynamic psychological aspects of the connection, such as "distance" and "dynamic", which suggest the psychological connection is not static as argued by the existing literature (e.g., Yang and Ha, 2021), overlooking the complex and dynamic nature of PSR. These keywords emphasising the need for a more comprehensive understand of PSR's multifaceted dynamics. As discussed by the interviewees:

'I do not know anything about their life. So I would say that it is kind of a more distant relationship for me.' Interviewee 11

## 'So my relationship with this influencer is something dynamic.' Interviewee 5

The sub-category "*belongingness*" emerges from the interviews, signifying a crucial aspect of PSR. Generally, belongingness refers to the state or sensation of being part of a specific group. Individuals perceive themselves as integral members of an understanding community (Tsiotsou, 2016). This deep sense of belonging fosters their emotional connections and their strong feelings (Escalas and Bettman, 2017; Iannone et al., 2018). Respondents express a sense of belonging to the community of the influencer's fans and followers:

'I am a big fan of Jiaqi. Actually, some of my friends are also Jiaqi's fans. We love watching his videos, he cares about those ladies, and he respects those ladies, so that is why we love him' Interviewee 7

Beyond the general sense of belongingness among ordinary followers, more intimate forms of belonging are observed, exemplified by terms such as "big brother" and "big sister". This sense of belongingness has been observed in the current literature, which found that viewers regard their favourite influencers as "big sisters" (Reinikainen et al., 2020). As some of the interviewees mentioned that:

'Yeah, but this influencer is different because he is just like our big brother.' Interviewee 5

### 'Emma, she is just like, like a big sister.' Interviewee 3

The novel insight into the nature of PSR underscores the argument that the psychological connection it fosters is dynamic rather than static (Tukachinsky and Stever, 2018). This perspective highlights the variability in how different individuals perceive their connection with others. It reflects the multifaceted and evolving nature of PSR, illustrating that these relationships are subject to change and are influenced by personal interpretations and experiences.

### Category 3: Value

The emergence of "*value*" as a new category from the qualitative data. In general, value refers to the importance, worth, or usefulness of something (Lou and Yuan, 2019). The qualitative analysis reveals that such relationships endow one party with a variety of values, encompassing significant values. Existing literature also indicates individuals also perceived various value from the other party, which benefit for generating long-term relationships with the other party (Zhong et al., 2021).

Several keywords encapsulate these significant values, including "*purposeful*", "*encouragement*", and "*important*". The establishment of PSR fulfils individual needs, and there is an acknowledgment that the other party, in turn, seeks benefits from their followers, highlighting a mutual exchange of value:

'They provide useful information. And they like followers because they want to be more popular. So we both have our purpose.' Interviewee 12

Additional positive values associated with PSR are identified, such as "*encouragement*" and the opportunity to "*learn*". These new insights increase the understanding the various values that the other party can provides. An interviewee's reflection captures this aspect of value:

'I would say they are the other people I have much to learn from. So yeah, I will keep improving and learning from them in the good things.' Interviewee 25
These insights reveal the multifaceted nature of value within PSR, illustrating not only the direct benefits derived from these relationships but also the reciprocal expectations and contributions between the involved parties (Whang and Im, 2021).

#### Category 4: Influence

The category "*influence*" has been newly identified through interviews, with the term "influence" mentioned 13 times. Generally, influential relationships are often seen within familial contexts, for instance, the support of family members and coaches can have a positive impact on the careers of elite athletes, as demonstrated through their ongoing support (Côté, 1999; Storm et al., 2014). This type of influential relationship is also evident in the responses from the interviews, with most respondents noting that the influence exerted by the other party is predominantly positive:

'We are greatly influenced by them...I am a loyal fan and I watch their videos, so my thinking is influenced by them. For example, if my favourite influencer recommends a perfume, I would try or buy it even if I do not like it.' Interviewee 9

These observations suggest that the dynamics of influence within PSR mirror those found in close personal relationships, highlighting the significant impact influencers can have on their followers. Although the existing literature mainly focuses on the influential power on consumers' decision-making process, such as their purchase decisions (Masuda et al., 2022; Leite and Baptista, 2022), the qualitative data also suggest influence on other aspects, such as people's attitude towards life. As interviewee 3 discusses:

'They influence my attitudes towards life. Like, maybe some bad things happen, but you will think positively' Interviewee 3

These insights reinforce the notion that the "influence" category is a critical component of PSR, reflecting how influencers can shape their followers' thoughts, behaviours, and decision-making process. This new "influence" category provides new insight into the components of PSR in online environments.

#### Category 5: Inequality

The qualitative data indicate that inequality characterises the nature of PSR, corroborating the analysis of the PSR definition presented in Chapter 2. The term "*one-way*" is mentioned seven times, signifying a situation where only one party is aware of the relationship's existence. Some scholars also point out PSR is not a reciprocity process; rather it is a one-way experience (Lim and Kim, 2018; Chen et al., 2021). The other party might acknowledge that fan groups form relationships with them, yet they do not engage in "one-to-one" relationships. Respondents acknowledge that influencers do not recognise their connections:

'Because we do not have a two-way interaction relationship, it is more like a oneway.' Interviewee 14

'I totally see DYS as my friends, even though I get it is just a one-way thing. I am just one of their fans, I mean, they have got like over 2 million followers, so they obviously do not know me.' Interviewee 21

These insights confirm that inequality is a key component of PSR, reflecting the unequal nature of PSR. This one-way relationship reflects the distinctive nature of PSR, where emotional investment and perceived closeness exist predominantly on the follower's side, while influencers remain largely unaware of individual followers.

#### Category 6: Long-term

Consistent with the earlier analysis of the PSR definition in Chapter 2, PSR is often perceived as a long-term connection (Balaban et al., 2022). The term "*long-term*" is highlighted 15 times, underlining the enduring nature of such relationships. In other words, if consumers develop PSR, people will keep the relationship in a long run rather than for a short period (Balaban et al., 2022; Leite and Baptista, 2022). This perception of a long-term connection is supported by the interview data:

'I have been following him for about six to seven years now. So I would say it is a long-term friendship.' Interviewee 14

These insights confirm that PSR is often perceived as a long-term commitment rather than a short-term relationship. When consumers develop PSR with influencers, they tend to maintain this relationship for an extended period.

#### 5.2.2 Inductive thematic analysis

The second technique employed to explore consumer perceptions of their relationships with influencers is thematic analysis. Adhering to the four-step process of thematic analysis outlined by Braun and Clarke (2006) and employing inductive reasoning, this study delineated 103 initial codes, 20 sub-themes, and six themes. These themes comprise "*time*", "*psychological closeness*", "*value*", "*inequality*", "*influence*", and "*connection*". The comprehensive table of thematic analysis is available in Appendix 12.

#### Theme 1: Time

Time is a pivotal factor in the development of relationships. Within the context of PSR, time is acknowledged as a crucial component closely associated with PSR (Whang and Im, 2021; Reinikainen et al., 2020). The likelihood of establishing PSR increases with the amount of time individuals spend on social media platforms (Pressrove and Pardun, 2016). Additionally, the consequences of PSR further underscore the significance of time. For instance, viewers who have formed a PSR with live streamers tend to dedicate more time to watching and re-watching the live streamer's content (Lim et al., 2020). Certain respondents highlighted that the formation of this psychological connection is time-dependent:

'I think my loyalty to her has accumulated over time... Every time I read her posts, and almost in those three years, I found myself actually having basically the same thoughts towards every social issue with her.' Interviewee 4

This psychological connection, once established, often persists in the long term (Yuan et al., 2021). Both the analysis of PSR definitions and content from prior studies support this perspective. As expressed by one of the interviewees:

'It has already been a very long time. I have developed such strong psychological bonds with them over these years. It is a special feeling that grows stronger as time passes 'Interviewee 9

Moreover, for some individuals, engaging with the influencer becomes habitual, and they come to view visiting the influencer's profile as an integral part of their daily routine. One interviewee noted:

'It is a kind of habit. I will check their updates every two to three days if they add new videos.' Interviewee 9

Another noteworthy aspect emerging from the qualitative data is the frequency of interaction. Individuals who have established PSR with influencers often visit their social media profiles regularly. Some respondents reported checking the influencer's social media account daily:

'I use Weibo every day to see whether he has some updates.' Interviewee 2

Conversely, other participants mentioned visiting the influencer's account two to three times a week or nearly every week:

'I will visit on a weekly basis to check what is new on her channel, and this is one of the influencers I followed.' Interviewee 11

'... frequency, I think, is an average of two to three days.' Interviewee 9

Time, as a central theme, emerges across all three analyses, underscoring the critical role it plays in PSR (Whang and Im, 2021). It is imperative to highlight that the emphasis on time reflects the sustained interaction and gradual accumulation of experiences in the development and maintenance of PSR, broadening the conventional understanding of "long-term" relationships.

# Theme 2: Psychological closeness

Psychological closeness embodies the nature and main domain characteristic of PSR, highlighting the conscious and motivational origin of the connection (Richards and Calvert, 2015; Tsfati et al., 2022). On one hand, evidence from prior analyses of PSR definitions underlines that PSR is characterised by an evolving consciousness within the relationship rather than an unconscious state (Tsfati et al., 2022). For instance, earlier definitions underscore the importance of a conscious state of "*intimacy*" or "*familiarity*" within the

relationship, contingent upon the psychological state of one party (Richards and Calvert, 2015; Chung and Cho, 2017). Individuals possess the awareness to unilaterally develop a connection with another entity (Yuan et al., 2016; Lim and Kim, 2018). This intentional engagement in the relationship was articulated by an interviewee:

'So I view her as a friend, rather than just a KOL or a blogger who shares daily life as our entertainment.' Interviewee 4

On the other hand, the deliberate establishment of this relationship mirrors individuals' motivations and objectives. Research indicates that motivational factors precede the development of PSR (Yuan et al., 2016). Individuals experiencing social anxiety who seek virtual interaction as a form of escapism or relaxation are inclined to form PSR with virtual entities (de Bérail et al., 2019; Lee and Cho, 2020). Likewise, Zafar et al. (2020) find that motivations such as leisure, enjoyment, and information seeking contribute to the formation of PSR. Although not previously highlighted in PSR definitions or content analyses, interviewees voiced their information-seeking motivation that trigger their connections with influencers:

'I would say that learning more, learning different influencers' opinions, and accepting this diversity is good. Because you can always learn something different from the different people... And you will find that you feel close to them' Interviewee 1

Furthermore, entertainment serves as another motivation for individuals to develop PSR with influencers, as noted by Interviewee 11:

'I prefer to see some bloggers, like they will do talk shows, upload their travel experience, just for entertaining... so gradually it feels like I am experiencing these things with my friends' Interviewee 11

The motivation to build relationships also underpins the formation of PSR, as remarked by an interviewee:

'YouTubers around the world ... especially American YouTubers and Canadian YouTubers... So Richard can make me have a close connection with the world.' Interviewee 8 Individuals seeking benefits, such as discounts, may also form PSR with influencers on social media platforms, as discussed by Interviewee 7:

'At that time, I was experiencing a difficult financial period, so I always bought discounted products...Jiaqi often gives a lot of products' discounts...you know, he is like my friend in my real life helping me to get through those tough times.' Interviewee 7

The psychological connection encompasses a range of emotions, defined as "mental states of readiness stemming from cognitive evaluations of events or thoughts" (Bagozzi et al., 1999, p. 184). Emotions are transient, predominantly short-term (Houben et al., 2015), and can exhibit polarity, such as pleasure or displeasure (Houben et al., 2015; Ruiz-Mafe et al., 2018). Previous PSR definitions highlight the emotional aspect of PSR, such as Agnihotri and Bhattacharya (2020) describing PSR as a mutual relationship marked by intimacy and affection. Empirical evidence also suggests that emotional engagement significantly influences PSR (Lim et al., 2020), a sentiment echoed in interviews:

'I know him, and I like him.' Interviewee 2

'He creates many short videos on motivation, teamwork, and entrepreneurship. I really enjoy his content, so I quite like him. He is usually the first one I see on my newsfeed.' Interviewee 6

In PSR, individuals experience various emotional states and feelings towards the counterpart. Terms like "*emotional attachment*" (Sun and Wu, 2012) or "*emotional intimacy*" (Lee and Park, 2022) are employed in defining PSR. Interviewees have also expressed diverse emotional responses:

'And I really admired her courage and also her views towards this policy...I deeply relate to her' Interviewee 4

Belongingness forms another dimension of psychological closeness as discussed in the above content analysis. It relates to an individual's psychological well-being and stems from the inherent need to belong (Moeller et al., 2020). A sense of belonging can engender positive emotions such as happiness (Braxton and Lau-Gesk, 2020), whereas its absence may adversely affect well-being, leading to feelings of depression (Moeller et al., 2020). The

desire for belongingness, particularly among individuals with low self-esteem, is an antecedent of PSR (Escalas and Bettman, 2017; Hwang and Zhang, 2018). Shared characteristics further enhance the sense of belonging and likelihood of establishing PSR, such as identity similarity (Hu et al., 2020). This was reflected in an interviewee's comment:

'Because I am a vegetarian. So, I follow some vegetarian influencers or some vegan influences. We have some similarities; we are both into fitness or vegan diets.' Interviewee 14

This sense of belonging was also evident in how respondents perceive influencers as "*big brothers*" and "*big sisters*", fostering a sense of belonging within PSR:

*`...he comes from, you know, Indian background and Indian culture is really close to me.' Interviewee 6* 

*`...she has a Greek background. I also have a Greek background. So it was nice to her and connect with her' Interviewee 15* 

The dynamic psychological nature of PSR, illuminated through previous analyses and thematic investigation, demonstrates the multifaceted psychological states, strengths, emotions, and feelings individuals harbour towards their parasocial counterparts.

# Theme 3: Value

Interview data elucidates "*value*" as a central theme in PSR, underscoring perceived value as a pivotal element in the development and sustenance of PSR. Despite earlier PSR definitions overlooking the role of value, subsequent research has highlighted that perceived value can significantly impact PSR (Quelhas-Brito et al., 2020). One form of value derived from PSR is hedonic value, which encompasses the enjoyment and playfulness individuals experience in relationships (Hughes et al., 2019). This form of value primarily arises when individuals engage with content generated by influencers. Within this context, interactive methods are employed to evoke emotional pleasure, promote continuous viewing, and enhance customer engagement (Wongkitrungrueng and Assarut, 2020; Guo et al., 2021). The provision of hedonic value through PSR is validated by interviewees: 'So when you know you have a bad day, and then someone gives you this sort of happiness push that you needed for the day, and I think this is what I have seen them.' Interviewee 6

'I think he can really relax my mood. Especially when I wash my clothes or do other things, I am really relaxed.' Interviewee 8

Beyond hedonic value, PSR also imparts utilitarian value, which pertains to the practical benefits a media user gains from a relationship (Overby and Lee, 2006; Meske et al., 2019). Respondents often mention deriving valuable information from influencers:

'When I want to buy something, like a new skincare product or some new makeup, I will watch their videos to see their recommendations or if they have any views about this makeup to see the advantages and disadvantages of this new thing.' Interviewee 9

'I follow these influencers because they can give me some knowledge I do not know.' Interviewee 11

Additionally, symbolic value has emerged as a novel sub-theme in PSR. Users may engage in various symbolic actions on social media to fulfil their needs (Kane, 2012). Viewers often find aspects of their self-identity and social identity reflected through engagement with influencers. Interviews reveal that participants find symbolic value in PSR, particularly in terms of the ideal self, self-reflection, and resonance:

*'When I watch Pamela's videos, I always think that, oh, I can have the same shape as Pamela.' Interviewee 1* 

'Well, I think because sometimes when we follow these inferences, because they somehow represent us, so if, like, extend yourself, you know, because they all love Vendors. So that is kind of me... So, I would say that means a part of me because we share the same values.' Interviewee 14

'Because most of the time, it's quite hard to express our feelings. But his songs expressed exact meaning about my feelings. So, I know that he understands me, and I understand him.' Interviewee 2 Value, as an overarching theme, indicates that PSR is capable of providing individuals with hedonic, utilitarian, and symbolic values, each contributing to the multifaceted benefits derived from these relationships.

#### Theme 4: Inequality

Inequality is identified as a fundamental characteristic of PSR, underscored by the relationship's development and maintenance predominantly relying on one party, with the other party engaging minimally. This disparity manifests in two distinct aspects: the unilateral formation of PSR by one party and the one-sided efforts to maintain PSR by the same party. As highlighted by some studies, PSR inherently features an asymmetrical relationship where one party independently fosters a connection without the reciprocal awareness or involvement of the other party (Yuan et al., 2016; Utz et al., 2021). While influencers may recognise a general connection with their audience, they lack personal recognition of individual followers, leading to a predominantly one-sided or one-way relationship (Lim et al., 2020; Cheung et al., 2022; Gabriel et al., 2018).

The unequal nature of PSR also evident in qualitative data, where respondents acknowledge the predominantly one-way dynamic of their connections:

'Because we do not have a two-way interaction relationship, it is more like a oneway.' Interviewee 14

The initial act of following an influencer, driven by a desire to learn more about them, illustrates the unilateral effort in establishing a connection:

*'When I have my social media accounts, since I want to know about this person, I follow him on social media.' Interviewee 2* 

Furthermore, maintaining the relationship often involves unilateral efforts from the followers, with minimal reciprocal engagement from influencers:

'If I leave a comment, they will not reply to me. Mostly, you know. And they have been living sending us a thing, uploading their videos, but they have so many fans that they will not notice us. So, it is not like a mutual relationship.' Interviewee 3 However, the absence of mutual engagement does not imply that influencers are passive in PSR. Influencers actively seek to expand their follower base and enhance their social influence, which can lead to more brand collaborations and increased profits (Jin and Phua, 2014; Taillon et al., 2020). A key strategy for influencers is the sharing of self-disclosure content, which includes both professional and personal information (Chung and Cho, 2017; Kim and Song, 2016; Leite and Baptista, 2022). Such disclosures help reduce followers' uncertainties about the media character, fostering attributional confidence (Rubin and Perse, 1987; Perse and Rubin, 1989):

'She shares her lifestyle and some things happening around her.' Interviewee 10

'He sometimes posts information related to his work, which I think is also a way for his team to do publicity. For example, he is recording a variety show somewhere.' Interviewee 2

Influencers' frequent content updates aim to sustain engagement with their audience:

'She creates artwork very frequently, almost three or four pieces per day. She updates her account very frequently.' Interviewee 10

Despite the recognition of the relationship's unidirectional nature, some followers perceive a level of concern from the influencers towards their audience:

'I think he cares about those ladies. He often emphasises that if you cannot afford it, do not buy it. So I think, yeah, he stands by our side and really listens to the voice of consumers.' Interviewee 7

In summary, the intrinsic inequity within PSR, as corroborated by both literature and qualitative insights, underscores a relationship primarily characterised by the followers' unilateral initiation and maintenance efforts, with influencers contributing indirectly through content creation and engagement strategies.

#### Theme 5: influence

"Influence" emerges as a novel theme from the qualitative data analysis, highlighting how one party can be significantly impacted by another within PSR. Despite existing research validating that PSR can shape perceptions, such as enhancing source trustworthiness (Chung and Cho, 2017), the notion of influence has not been explicitly incorporated into the definition of PSR. This analysis delineates two primary dimensions of influence: influence-related and brand-related.

The literature documents that PSR can influence one party's view of another, contingent upon the presence of a PSR. For instance, Finsterwalder et al. (2017) demonstrate that individuals are more inclined to forgive celebrities' misdemeanours when they have PSR with them. This increased tolerance was echoed in interviews:

'I found myself becoming those who have double standards. In most cases, I will unfollow some bloggers who hold different views against me. But I have a double standard towards the influencer and other bloggers after I become a loyal fan. When they have different opinions, I will think that it is quite normal because people are not always the same.' Interviewee 4

PSR are also influential in fostering real-life interactions and cultivating social bonds (Tsiotsou, 2016). For instance, within consumption communities, PSR can lead to genuine engagement and social connections (Yuan et al., 2021). This effect was observed in interview narratives:

'For my best friend. I first noticed her...I just saw the article she wrote, and I felt it was really warm. I liked the way that the girl records her life... We are in the same city, and we are in the same university, so I wanted to be a friend with her... We become the real-life friends.' Interviewee 9

Furthermore, brand-related outcomes linked to PSR have been extensively documented. Engaging in PSR with smart speakers, for example, can enhance advertising effectiveness, including brand and advertising attitudes (Lee and Cho, 2020). Establishing PSR with digital celebrities has been shown to boost consumers' purchase intentions and electronic word-ofmouth (e-WOM) (Hwang and Zhang, 2018). PSR has a positive impact on consumers' repeat purchase intentions (Yuan et al., 2021). Such influences were discussed by respondents: 'And then Jillian, she mainly makes clothing and home products now, so we had heard like lines come out. I always definitely check them out. I do not always buy from them. We have a very small home, so I do not. I always check the value. And I always like to consider some of her stuff for gifts because it is really nice gifts for people.' Interviewee 13

'I have purchased some products that Claire recommended.' Interviewee 15

Through qualitative data analysis, the influential capacity of PSR is underscored, with respondents acknowledging the positive effects PSR has on their perceptions and actions.

# Theme 6: Connection

The theme of "*connection*" has been recognised in both the analysis of previous definitions of PSR and in content analysis, underscoring the relational essence of PSR (Feder, 2020; Yuan et al., 2016). This connection signifies the perceived bond or relation between individuals and influencers or media figures, highlighting the intrinsic nature of PSR (Farivar et al., 2022). Several respondents have articulated their sense of connection:

'It makes me feel there is a connection between me and this influencer.' Interviewee 10

'I find I relate to them. I always feel the intimate feelings with them.' Interviewee 13

The above discussions also confirm that connection is a fundamental component of PSR, reflecting consumers develop the relationship with social media influencers.

#### 5.3 PSR development stories

PSR is a developed process, as evidenced by the qualitative findings. Previous studies have regarded PSR as a static concept; however, Tukachinsky and Stever (2018) have theoretically conceptualised PSR as a process from a media perspective, contributing significantly to the understanding of the stages in PSR development. The qualitative findings corroborate the four stages of the PSR development process proposed by Tukachinsky and Stever (2018), namely: exposure initiation, experimentation, intensification, and integration.

Furthermore, this study offers a deep understanding of how consumers develop PSR with social media influencers, identifying key concepts as contributing factors at each stage.

# 5.3.1 Stage 1: Exposure initiation stage

The exposure initiative stage, the initial phase where viewers virtually meet the other party, is detailed through the critical factors presented in Table 5.2, based on interviews with 25 participants. Most respondents recall the moment they first became aware of the other party, although some memories are vague, possibly due to the passage of more than ten years.

This stage is characterised by three distinct approaches to acquaintance with the other party, as revealed through the qualitative data: (1) consumption of traditional media, (2) engagement with social media, and (3) word-of-mouth recommendations.

Some respondents indicated their first acquaintance with the other party through traditional media. Traditional media, such as television, remains a prevalent medium through which consumers foster PSR with media figures (Bernhold, 2019; Bonus et al., 2021). While viewing TV programmes, audiences initially notice individuals such as hosts, forming their initial impressions during the process (Stephens et al., 1996). One interviewee remarked:

'Abbey, I found her through a TV segment she did. She was on the local news one morning when I was getting ready.' Interviewee 13

Books, another form of traditional media, serve as a means for viewers to become acquainted with the other party. However, unlike prior empirical studies, respondents do not report forming PSR with fictional characters from books but with the authors themselves. For example, Interviewee 16 discusses their first encounter with an author:

'So, the first time I know him is because he is the author of, like, a novel book. It is not an academic book. It is more into a comedy book, and I know him from the introduction part of the book.' Interviewee 16.

	Key factors in the PSR development stage 1- exposure initiation					
Factors	Construct	Theme	Sub-theme	Interviewee	Frequency	
Post-related factor	Post appeal	Content attractiveness – background music		2		
		Appealing cover page/headline/title		1; 3;7;12	8	
		Post appeal		4		
		Topic attractiveness		17		
		Content attractiveness		19	]	
			Influencer popularity	1; 8; 25	7	
	I. G	Social attractiveness	Perceived similarity	8;19		
T. (1	Influencer attractiveness	Physical attractiveness		9;14		
Influencer-related		Vocal attractiveness		7		
factor	Other perceived characteristics of the influencer	Perceived differentiation		7	3	
		Perceived popularity		25		
		Perceived influencer bravery		17		
Platform-related	Personalised recommendation systems	Social media recommendation		3; 21; 24; 25	7	
factor		Website		23		
		Big data		10;11	1	
	Motivation	Searching with keywords		3;5;14;24	10	
		Relationship building	Meeting new people	19		
		Information seeking Sea	Searching for products	3; 25		
Consumer-related factors			Searching for specific brands	18		
			Searching for information	21		
		Entertainment		8	1	
	Involvement	Involvement/attention		1;11;15;21	4	
	Consuming traditional media	Watch TV show		6;13	4	
Traditional media		Watch Reality show		20		
		Read book		16		
WOM	WOM	WOM		6;12;21;22	4	

 Table 5.2 The key factors in the exposure initiation stage of the 25 interviewees

Online environments are the most common means by which most respondents initially encounter the other party. The vast amount of information online means that viewers might not recognise all content creators. It was noted by the respondents that there are primarily two methods through which they notice the other party: (1) personalised recommendation systems and (2) keyword searching.

A personalised recommendation system is an intelligent platform or algorithm that, through extensive data mining, identifies users' preferences and activities online to recommend content tailored to their needs (Buder and Schwind, 2012; Behera et al., 2020). Social media platforms utilise these systems to suggest content (Behera et al., 2020). One respondent mentioned encountering the other party through such a system on social media, as evidenced by Interviewee 25:

'I just got a YouTube account. So, I got into the YouTube and the front page. And because my profile was proper, I guessed because my profile was in Chinese. So, they just read the YouTube and recommended me some of the Chinese content YouTube videos, and Toress was one of them because he has Chinese subtitles.' Interviewee 25

Additionally, some websites recommend content through personalised systems, which is how another interviewee first met the other party. As Interviewee 23 described:

'During that time, we did not have a cell phone, but there's a very famous website, and everyone visits this website... I'm interested in makeup, and the website recommended me a popular content creator in makeup...and that is the first time I know her.' Interviewee 23

Besides personalised recommendations, viewers also encounter the other party through keyword searches, a prevalent method for finding relevant information (Wang et al., 2016). Interviewee 24 shared their experience:

'Back to the time when I was having serious anxiety. I have just tapped some keywords, like anxiety and anxiety disorder. And I think Andaxiong just popped out in the first one, and I clicked and watched it.' Interviewee 24

While personalised recommendation systems and keyword searches facilitate exposure to the other party, this does not guarantee viewers' attention due to their limited focus, multitasking habits, and selective engagement with information (Duff and Sar, 2014; Santoso et al., 2020). Viewers' psychological processes play a critical role in whether they notice and engage with the other party, influenced by post-related factors, influencer-related factors, and viewers' motivation.

# Post appeal

Post appeal significantly contributes to viewers noticing the other party. Post appeal is defined as the extent to which viewers find an influencer's post visually or aesthetically appealing (Ki and Kim, 2019). Viewers are drawn to posts they find attractive, with appealing posts often characterised by enticing headlines and cover pages. A headline succinctly summarises the post or video, and compelling headlines are crucial for capturing the limited attention of users (Jeganathan and Szymkowiak, 2020). As expressed by one interviewee:

'The headline is really eye-catching, and that is why I want to see the content.' Interviewee 3

The cover of a post typically includes images and text, with engaging covers effectively drawing viewer interest. For instance, one respondent remarked on their initial notice of an influencer:

'I first noticed him on Weibo because the video cover page is attractive. The cover page showed that a man who was trying cosmetics. It is very uncommon since, at that time, women promoted cosmetics, and it is like a stereotype that women need to do that. But Jiaqi, as a man, tried a lot of lipsticks on his face, so that's why I noticed him.' Interviewee 7

Content appeal plays a crucial role in leading viewers to notice the other party. This is particularly true on social media platforms like TikTok, where users scroll through brief videos, gravitating towards those they find appealing (Yang and Ha, 2021). As stated by Interviewee 19:

'So, with those, as I am scrolling through, and I find things that are intriguing or thought-provoking, that is piqued my interest, and I will watch them.' Interviewee 19

Therefore, post appeal, such as an attractive cover page and headline, has been identified as one of the key factors that increase consumer focus. These appealing elements can capture consumers' attention and enhance their interest.

#### Influencer attractiveness

Influencer attractiveness is a significant factor in drawing viewers' attention. It refers to the audience's perception of the extent to which an influencer's features and qualities are appealing (Gunes, 2011; Kim and Part, 2023). Analysis of qualitative data reveals three dimensions of influencer attractiveness: social, physical, and voice attractiveness.

Social attractiveness pertains to the extent to which an influencer is perceived as likable (Liu et al., 2019). Influencer popularity and perceived similarity contribute to their social attractiveness, prompting viewers to engage with their content (Yuan and Lou, 2020). For instance:

'I was searching for the iPhone-related content. And I think because he is one of the most famous, he was like showing up as the first one. So, I just clicked on his video. And he was doing like a review of the newest iPhone, was iPhone X.' Interviewee 25

'This influencer creates content related to Business Class flights on YouTube. The first video I saw was about Sichuan Airlines, a Chinese airline company. And this influencer, he looks like, I think he is Asian. I am from China. So, these characteristics bring us closer.' Interviewee 8

Physical attractiveness is defined as the degree to which an influencer's physical attributes are considered aesthetically pleasing (Patzer, 1983; Wiedmann and von Mettenheim, 2020). Existing literature has found influencers' physical attractiveness positively influence PSR (Han and Yang, 2018). Respondents indicated that they are drawn to influencers whom they find visually appealing:

'She looks cute, and I want to see more about her daily life.' Interviewee 23

Voice attractiveness, as mentioned by interviewees, is another compelling reason for focusing on influencers. This novel insight suggests voice attractiveness could be seen as one of the aspects of influencer attractiveness, which is overlooked by the existing literature. The appeal of an influencer's voice can significantly capture viewers' interest:

'When I first listened to his songs, I was attracted by his euphonious voice.' Interviewee 2

In sum, influencers' various forms of attractiveness play an important role in increasing consumers' attention and interest. Existing literature identifies social and physical attractiveness as two aspects of influencer appeal. These qualitative findings provide new insights, revealing that voice attractiveness is also an important aspect of influencer attractiveness.

#### Viewers' various motivations

Moreover, viewer motivation plays a crucial role in determining attention towards influencers. Motivation is the desire to fulfil personal needs through media or technology (Park and Lennon, 2004; Kim, 2020). Current literature identifies various people's motivations that can drive individuals to use social media, and they are all related to PSR (Lee and Cho, 2020; Zafar et al., 2020). When an influencer's content aligns with viewers' motivations, it significantly increases their attention:

'Hearthstone is a card game I started playing two years ago. When I first came into this game, I had no idea about the function of each card and how to use it, so I searched Hearthstone as the keyword on Bilibili. There's a video explaining the functions of each card, and this video is made by Valila. So, I watched this video and found it very helpful.' Interviewee 21

As discussed above, current literature has found various motivations people have for using social media and relate to PSR (Yuan et al., 2016). Based on the qualitative data, viewers primarily have four types of motivation, including information seeking, entertainment, relationship building, and economic reward. Information seeking involves actively acquiring information to satisfy personal goals (Asghar, 2015; Zafar et al., 2020). Respondents mention using social media and watching videos to obtain diverse, current, and useful

knowledge, learn new things, and search for information. This motivation for information seeking is illustrated by an interviewee's statement:

'I use social media to find people from different countries and regions; they have various lifestyle and cultural backgrounds, and I am interested in knowing these kinds of things.' Interviewee 1

Entertainment motivation entails seeking temporary spiritual relaxation, pleasure, and fun (Leung and Wei, 1998; Park et al., 2009). Respondents often use social media for fun, relaxation, and discovering exciting content (Esteban-Santos et al., 2018). The abundance of entertainment content on social media makes it a popular choice for this purpose (Lee and Cho, 2020). An interviewee remarked:

'I think what I like most about these social media is it is just a break. And I like watching funny videos that are pretty short. Just a break from reality.' Interviewee 20

Interviewees also indicate using social media for making connections. Relationship building is defined as utilising social media to establish and maintain relationships (Yuan et al., 2016). When people seek to connect with others, they use social media to communicate with friends and develop PSR with influencers (Liu et al., 2019). An interviewee notes:

'I love people who share their views on social media. I love to read their thoughts and views towards some social issues, which makes me feel connected.' Interviewee 4

Regarding economic rewards, some influencers share discounts on social media, leading people to search for these deals. Economic rewards involve monetary promotions used by influencers, celebrities, and brands to attract viewers on social media (Coker et al., 2015), typically through discounts and coupons. This motivation is highlighted by an interviewee:

'Jiaqi Li shares a lot of useful information about cosmetics for young ladies. And he often gives a lot of discounts, so I often check his posts to find product discounts.' Interviewee 7

Several interviewees first learn about influencers or celebrities through word-of-mouth (WOM) from friends and colleagues. People often recommend influencers or celebrities they know or follow to others, which facilitates initial awareness. As one interviewee describes:

'And the first one is because of my work. I was going to work with a brand planner for an event, and they mentioned that Sophie is their ambassador in the UK. So, then I opened the Instagram and saw her posts.' Interviewee 22

Involvement is identified as a critical factor at the exposure initiation stage. Involvement refers to the degree to which viewers are engrossed in the activity they are engaging in on social media (Vazquez et al., 2017). As discussed, there are primarily three ways viewers initially become aware of influencers or brands, with awareness indicating whether viewers are attentive or focused on the subject. Viewers' notice of influencers or brands signifies their focus and involvement. An interviewee mentions:

'Some influencers, their video headlines are attractive, and it captures my attention to see the content.' Interviewee 1

Therefore, the above discussion provides new insights into how consumers develop PSR at the initial stage, identifying several key factors that enhance consumer involvement, including post appeal, influencer attractiveness, and viewers' various motivations. The next section will discuss the second stage of PSR development.

#### 5.3.2 Stage 2: Experimentation stage

Experimentation constitutes the second phase of PSR development. This stage is characterised by the curiosity viewers have towards the other party, as evidenced by the responses from 25 interviewees detailed in Table 5.3.

Viewers are not merely content with superficial encounters; their interest deepens, compelling them to seek a more comprehensive understanding of the other party. This desire for a more profound connection is highlighted in the interview excerpts:

	Key factors in	the PSR development stage 2 - experi	mentation	
Factors	Construct	Theme	Interviewee	Frequency
Perceived influencer characteristics	Perceived expertise	Perceived expertise	1;5;8;9;10;11;14;15;21; 25	10
	Perceived trustworthiness	Perceived trustworthiness	7	1
	Perceived ambition	Perceived ambition	4	1
	Perceived similarity	Perceived similarity	1;2;4;5;7;9;10;13;14;15;18;19;22;25	14
	Perceived authenticity	Perceived authenticity	13;18	2
	Perceived interestingness Perceived interesting		15;16;19	3
	Perceived humility of the influencer Perceived humility of the influencer		14	1
	Perceived attractiveness	Perceived attractiveness 2;23		2
	Perceived influencer critical thinking	Perceived influencer critical thinking	17	1
	Knowledge of influencer's personality	Knowledge of influencer's personality	5;6	2
		Perceived utilitarian value	1;2;7;11;12;13;14;15;17;18;21;22;23;25	14
	Perceived value	Perceived hedonic value	3;8;9;10;11;13;18;20;21;22	10
		Perceived symbolic value	2;14;17;20	4
	Following an influencer	Following social media account	2;4;5;8;10;11;12;16;17;19;20;21;22	13
	Consumer engagement	Consumer engagement	1;4;11;12;13;17;19;22	8
Consumer engagementConsumer ofViewer- influencerTrustTrLiking the influencerLiking the	Trust	5	1	
	Liking the influencer	Liking the influencer	6	1
related factors		Interest in this influencer	2;6;12;13;16	5
	Interest in the influencer	Want to know more about the influencer	3;20	2
		Want to see updates	3	1
	Connecting with the influencer	Connecting with the influencer	3	1
	Parasocial interaction	Parasocial interaction	3;4;6;7;9;11;14;16;17;18;23;25	12
Repeated exposure	Repeated exposure	Repeated exposure 11		1

# Table 5.3 Key factors in the experimentation stage of the 25 interviewees

'I want to know more about him, not only from the Bilibili, not only from those funny videos, but I also want to know what he was like in daily life, like in real life.' Interviewee 3

'And this is why I guess I started exploring more about him. I also figured out that he has got a lot of online classes available, you can join, etc.' Interviewee 6

The exploration for more information about the other party takes various forms, ranging from conducting searches to consuming more content created by them. The online environment facilitates this engagement, enabling viewers to interact with the content creators' videos directly. The act of engagement is a recurrent theme in the interviews, as illustrated by the following comments:

'When I find the video is very interesting and I like the way that the influencer makes the video, I will go to that influencer's page and watch other videos to make sure I truly like the content or the influencer, and then I follow some of them.' Interviewee 11

'If it was a topic I was reading about in the news, but I saw a different angle by these two accounts, I would have clicked on the post to read more. And I would have liked or found the content relevant and followed them.' Interviewee 17

Therefore, interaction and engagement are crucial for viewers to assess their affinity for the content or the creator and to navigate their subsequent relationships with them.

# Perceived value

Perceived value plays a pivotal role in the experimentation phase, encapsulating the comprehensive assessment of relevant factors when interacting with another party (Zeithaml, 1988; Wongkitrungrueng and Assarut, 2020). It influences the decision-making process regarding whether viewers wish to initiate a relationship with another entity. This is substantiated by responses indicating that viewers are inclined to follow and develop a relationship with the other party if they discern value or benefits. As expressed by one interviewee:

'Claire, she has really nice ideas. And the content of her videos is quite nice, easy to watch, and very useful. So that is why I follow her.' Interviewee 15

The interviews reveal three facets of perceived value: utilitarian, hedonic, and symbolic value. Utilitarian value, the primary aspect, refers to the practical benefits viewers gain from the content, encapsulated as the degree of usefulness assessed by media consumers from engaging with content or services (Overby and Lee, 2006; Meske et al., 2019). Interviewees frequently mention gaining information, insights, and new knowledge from the content:

'I would say, because those influencers create useful videos, I can apply them to my daily life. For example, vegan fitness influencers will post some recipes for their diet. Also, they will make a short video of how they cook it, so I can just use it in my life.' Interviewee 14

'The business influencers often share new knowledge for me, which could be useful or inspiring.' Interviewee 12

Such utilitarian value fosters a willingness to establish a relationship with the content creator, evidenced by followership on social media platforms, as mentioned by one interviewee:

'I just have this passion for Europe, Paris, Italy, and all those things. So all these beautiful places. So, I think I was also following her because she would show me this different world that I had not been able to enjoy myself. So it is probably that, again, just the inspiration thing.' Interviewee 18

Hedonic value, the second aspect, pertains to the entertainment and enjoyment derived from the content, defined as the extent of fun and playfulness experienced by the viewer (Hughes et al., 2019). This value is reflected in respondents' experiences of happiness and relaxation:

'The first reason is because their videos are interesting and really funny. So I could release my stress when watching the videos.' Interviewee 9

Viewers perceiving hedonic value are more inclined to develop relationships with the content creator, as highlighted by another interviewee:

'I felt that he was very interesting and also the videos and live streaming, which I can get more fun compared with other video game influencers, so I follow him.' Interviewee 21

Lastly, symbolic value emerges as a critical aspect, representing the extent to which individuals identify with or see their social identity reflected in the content (De Toni et al., 2021). Some respondents recognise this value in influencers, seeing a reflection of themselves and their values:

'I think sometimes when we follow these inferences, because they somehow represent us, so if, like, extend yourself, you know, because they all love the same thing. So that is kind of me. ... so I will say that it means part of me because we share the same values.' Interviewee 14

This perceived symbolic value is echoed by a respondent from the LGBT community, highlighting the influential role of certain figures:

'I think they are drag queens. But they are also very influential in, like, the LGBT community. COURTNEY ACT is big; I guess she has done a lot of reality shows. But she also does a lot of work where she helps communities and does things that are positive in the community...it resonates with me; it still represents me as a person.' Interviewee 20

These insights demonstrate the multifaceted nature of perceived value and its significant impact on viewers' decision to follow content creators, driven by utilitarian, hedonic, and symbolic benefits.

# PSI

In the experimentation stage, PSI also plays a significant role in the development of PSR. As discussed in previous chapters, PSI pertains to the viewer's perception of direct, reciprocal, and one-to-one communication with influencers during media exposure (Oliver et al., 2019). One participant mentioned feeling as though influencers were speaking directly to them, thereby enhancing their enjoyment of this highly interactive experience. Interviewee 25 described their experience as follows:

'When I watch Torres's video, he is holding a camera and recording as he climbs and walks around. Suddenly, he turns the camera to his face, speaks to the camera, and shares his feelings. I really enjoy this kind of interaction.' Interviewee 25

Influencers employ various techniques to augment viewers' feelings of PSI, such as addressing the camera directly, employing first and second person pronouns ('I' and 'you'), and posing questions. These techniques are specifically used to create a sense of familiarity, making viewers feel as though the influencer is a friend speaking directly to them. This approach was evident in an interview with another participant:

'I remember that when I started to follow his videos to work out, I really liked it because he is like down to earth, like your friend. During the workout, he said something like, oh, you can do it at home with your own water bottle. So when you hear about that, it is just like a friend talking to you.' Interviewee 14

# Other influencer characteristics

In the second stage, viewers also perceive some of the influencers' characteristics that benefit relationship development, such as the perceived humility and attractiveness of the influencer. Influencer expertise is frequently discussed, often cited by respondents. Perceived influencer expertise is the extent to which viewers recognise an influencer's skills, knowledge, and competence in specific areas (Farrington-Darby and Wilson, 2006; Xiang et al., 2016). There are two aspects of influencer expertise that viewers discuss. One is the influencer's expertise in their field, as one interviewee notes:

'She competed in the fitness field. She was a player in the fitness field. So, because she is very professional and showed a sense of professionalism in their fields, and that is the reason why I follow her.' Interviewee 14

The other aspect is the influencer's video production skills, as the same interviewee states:

'Really good content quality either in the video, filming or the background editing, even the background music.' Interviewee 14

Perceived similarity is another critical element that plays a significant role in the second stage. It refers to the degree to which people perceive similarities in beliefs, education, social status, and the like with the influencer (Rogers and Bhowmik, 1970; Naderer et al., 2021). Respondents frequently discuss their perceived similarity with influencers:

'We have similar lifestyles, stages, so like, getting married with kids at, like, roughly the same time and that kind of stuff. That's another reason why I'm following her.' Interviewee 13

'Sophie focuses on cleaning and organising the home, the stuff, and I like everything that is related to home decoration and home organisation because I like this.' Interviewee 22

Therefore, in the second stage, consumers exhibit a high level of interest in influencers and seek to learn more about them. At this stage, it is crucial for consumers to perceive value from influencers' posts, making them feel they can benefit from following these influencers. In particular, influencers' various characteristics and PSI are important in enhancing this stage of development.

# 5.3.3 Stage 3: Intensification stage

Intensification is identified as the third stage of PSR. Table 5.4 presents the key factors associated with the intensification stage, as derived from interviews with 25 participants. During this phase, individuals persist in recognising the value offered by the other party, gradually adopting the practice of following them as a habitual activity. Essentially, individuals show a readiness to initiate and sustain a relationship with the other party (Tukachinsky and Stever, 2018). An interviewee illustrated this notion by stating:

'The reason why I keep following them is that I think they make great videos. The quality is really good. And from their videos, I can always find the newest information, and there is information about that area. And sometimes, if I need to decide whether to purchase a new phone or choose a place for travelling, their videos are like really good sources for me to take a look and make my decision.' Interviewee 25

Key factors in the PSR development stage 3 - intensification				
Factors	Construct	Theme	Interviewee	Frequency
	Perceived respect for the influencer	Perceived respect for the influencer	1	1
	Perceived self-discipline	Perceived self-discipline	1	1
	Perceived similarity	Perceived similarity	4	1
	Perceived authenticity	Perceived authenticity	4; 9; 13	3
Viewer-	Perceived influencer credibility Perceived influencer credibility		17	1
	Parasocial interaction Parasocial interaction		3	1
	Liking the influencer	Liking the influencer	2; 19	2
influencer	Trust	Trust	3; 5;7;9;10; 11; 12; 13; 14; 15; 16; 23	12
related	Commitment	Commitment	2	2
factors		Keep following	3; 4; 6; 14; 16; 17; 18; 19; 20; 21; 23	11
		Want to keep relationship	5; 9	2
		Attachment to influencer	3	1
		Connect with influencer	5; 10; 11; 18; 19	5
		Feel close	6; 7; 8	3
		Keep watching	7; 21	2
	Consumer engagement	Keep engagement	9; 18; 21	3
Value	Perceived value	Perceived utilitarian value	23; 25	2
		Time accumulation	1;3;4; 5;7; 8; 10; 12; 13; 14; 16; 22	12
Time	Time	Frequency of visits	2; 5; 10; 19; 21	5
		Habituation	3; 7; 8; 16; 24	5

# Table 5.4 The key factors in the intensification stage of the 25 interviewees

Time emerges as a pivotal element in the intensification stage. Similar to interpersonal or social relationships, individuals invest time to cultivate relationships with the media entity (Pressrove and Pardun, 2016; Whang and Im, 2021). A majority of respondents revealed that following and engaging with the media figure has become a part of their routine:

'They have become a very important part of my life. Although they cannot update their videos very often, like twice a week or even once a week, I will watch their past videos very often, many times. For example, when I have to kill time, I will watch their videos again and again. This has become my lifestyle.' Interviewee 8

Several respondents highlighted the gradual development of a connection with the media figure over time:

'I think this kind of connection is accumulated through time. You know, every time I read her posts, and almost in those three years, I found myself actually having basically the same thoughts towards every social issue with her. And we all share the same views ... I think it is sort of time concrete.' Interviewee 4

The data indicates that the shortest duration for forming a connection is under three months, as one interviewee mentioned:

'I think it is just three months, and I become a loyal fan of him because I need guidance to deal with my size of the disorder. I mean, I actually have a psychological consult, but it is expensive. So, I need to find a new way to receive guidance to guide me to deal with my situation, and Andaxiong really helps.' Interviewee 24

Conversely, it took several years for some respondents to establish a connection, reflecting a prolonged engagement over time:

'Just as I said, I started listening to his songs when I was in primary school ... until I went to college, I started to use social media and follow him, and at that time, I found I must become his loyal fan.' Interviewee 2

Viewing frequency, associated with time, is identified as a crucial aspect in the intensification stage. Regular virtual interactions with the media figure are efforts made by individuals to maintain their connection. Through frequent exposure, individuals stay

updated and continue to derive value. Most respondents report engaging with the media figure's content daily or weekly.

'So I would say I definitely see at least one picture of this each day with a story mainly on their stories, I suppose. And then less on the feed, but definitely look at the feed, but it probably is mainly the story, and I'll see that at least once a day.' Interviewee 18

'Sometimes I might get the notification that they have a new video, or when I have some free time, I would say once per week.' Interviewee 15

The above findings confirm that time plays an important role in the PSR development process, particularly in the intensification stage. Followers interact with social media influencers at varying frequencies. As the frequency of interactions increases, consumers are likely to develop stronger connections with these influencers.

# Trust

During this stage, individuals' trust in the other party increases. Trust is defined as an individual's belief in the other party's ability to deliver consistent and competent quality (Jun and Yi, 2020; Chaudhuri and Holbrook, 2001). In virtual or media environments, fostering trust is challenging. As one interviewee noted, scepticism towards social media influencers is common:

'At the beginning, I think I do not trust these kinds of bloggers or influencers. Because I think they just want to sell their products.' Interviewee 7

However, familiarity over time, particularly during the intensification stage, leads to a stronger trust in the other party, recognising them as trustworthy (Yuan and Lou, 2020). This sentiment was echoed by interviewees:

'I keep following them, and it's because of established trust. So over, it has been many years, and they both establish, like this trust in our community, and they have built brand communities that are authentically theirs.' Interviewee 15 'I really trust him because I have that really good experience with the previous products I bought from his product.' Interviewee 11

Nonetheless, trusting an individual does not automatically extend to the same level of trust in their advertisements. This observation was highlighted by the interviewees:

'She will always currently in these two years have some videos for advertisements. So, I will not watch that video. Because I always trust those influencers. They are real experiences. They are really used to this product, and I will buy them. But I will not trust such advertisements. Or sometimes I feel there is an advertisement; I met half trust about this.' Interviewee 9

As discussed above, trust is a key factor in the PSR development process. At the intensification stage, consumers develop strong trust in influencers, and this trust level increases over time. Consumers regard influencers as trustworthy and believe in their recommendations.

# Perceived influencer authenticity

Perceived influencer authenticity emerges as a pivotal factor discussed by respondents in the intensification stage. Perceived influencer authenticity refers to the extent to which an individual regards an influencer as genuine and driven by intrinsic motivations for content creation (Shoenberger and Kim, 2022; Jun and Yi, 2020). Evaluations of an influencer's motivations—whether a post stems from genuine personal interest and the desire to share or from brand-driven incentives—are crucial. Respondents shared their perceptions of what makes an influencer appear authentic:

'And I really just think it comes down to the personalities that are humble, authentic and not pretentious. And you can tell when an influencer is just trying to say your piece of clothing versus they love their life and want to share it with your approach.' Interviewee 18

Similar to trust, the perception of authenticity is something that develops over time. Interviewee 15 remarks: 'She's, like, slowly built her business by being authentic. And I think that is why myself and a lot of our other audience are kind of very dedicated. Because there is that like level of authenticity, you do not see often on Instagram.' Interviewee 15

Influencer authenticity is one of the key characteristics that consumers focus on. Perceived influencer authenticity is driven by the consistent behaviour of influencers, which helps evaluate their motivations, whether they are driven by personal interest or intrinsic motivations.

# PSR

The outcome of the intensification stage is a willingness among individuals to sustain PSR with the other party. This implies that individuals have developed dynamic attachments to the other party. Reflecting on this relationship, one of the respondents shares their experience:

'I always have such a psychological relationship with them. I have already followed them for a long time .... I know they do not know me, but I have watched these videos for a long time and know them well. I always have such psychological bones with them. Yeah, such special feelings.' Interviewee 9

At this stage, the emotional attachment to the other party signifies an eagerness to keep the relationship alive. It is a stage where individuals feel a heightened sense of closeness to the other party. Several interviewees express their connection with influencers:

'I felt like I got quite closer to him because I was spending most of my days with, right in my sort of further work and life.' Interviewee 6

A dedication to influencers is also evident, with some respondents showing their commitment by continuing to follow them. As one interviewee notes:

'I want to continue following them. Because I want to continue having even my opinions challenged and for me to see all sides of an issue.' Interviewee 17

Moreover, respondents share their desires to maintain their relationship with the other party. For instance, Interviewee 5 states: 'I will keep this kind of encouragement and keep this relationship with this influencer. Because I still have curiosity about what he will talk about in the future.' Interviewee 5

Therefore, by the end of the intensification stage, consumers have developed a stable and long-term relationship with social media influencers. At the same time, consumers show a high willingness to maintain these long-term connections with influencers. The next section will discuss the outcomes of PSR.

#### 5.3.4 Stage 4: Integration stage

Integration represents the culmination of the development of PSR, serving as both the final stage and the outcome phase of the relationship. Key elements identified in the integration stage from the discussions with 25 interviewees are detailed in Table 5.5. From the qualitative data, three primary themes have emerged: outcomes related to the influencer, outcomes related to the viewer, and outcomes associated with the brand. At this stage, individuals are at a crossroads where they may either continue their PSR, experience parasocial breakups, or transition their PSR into real-life connections.

# Influencer stickiness

The initial facet of PSR outcomes pertains to those associated with influencers. At this stage, individuals develop a profound connection with the other party, characterised by attributes such as stickiness and loyalty towards the other party (Hu et al., 2020). Over time, this stickiness can transform into loyalty, where consumers exhibit a high level of commitment to following and supporting the influencer. This was elaborated upon by one of the interviewees, who noted:

'I think, for me, I stick with them. And I think, you know, it is nice to have that constant in my life still. I think that kind of returns into a nostalgia kind of thing.' Interviewee 20

		Key factors in the PSR develop	ment stage 4 - integration	
Factors	Construct	Theme	Interviewee	Frequency
	Influencer preference	Influencer preference	1;25	1
	Stickiness to influencer	Stickiness to influencer	20	1
	Support for the influencer	Support for the influencer	1; 2; 4; 16; 17; 18; 21; 23; 25	9
	Liking the influencer	Liking the influencer	2	1
Influencer -related outcomes	Trust	Trust	2	1
	Positive attitude to the influencer	Positive attitude to the influencer	2	1
	Intention to support	Intention to support	3; 5; 6; 7; 8; 9; 10; 11; 12; 13; 14; 15; 18; 19; 23; 24	16
	Loyalty to the Influencer	Loyalty to the Influencer	4	1
	Social relationship/interaction	Forming real-life friendships	9	1
		Attending offline event	10; 20	2
	Donation	Donating to influencer	17; 21	2
Viewer-		Feel happy	3	1
related outcomes	Well-being	Feeling encouraged	24	1
		Change in attitude towards Life	3	1
	Advocacy for brand	Advocacy for brand	18; 25	2
	Brand preference	Brand preference	2; 8; 25	2
Brand-	Brand trust	Brand trust	12; 18	2
related outcomes	Purchase product	Purchase product	1; 2; 3; 4; 6; 7; 8; 9; 10; 11; 12; 13; 14; 15; 16; 17; 18; 19; 20; 23	20
	Positive brand attitude	Positive brand attitude	2	1
	Purchase intention	Purchase intention	2; 3; 6; 8; 9; 12; 13; 14; 16; 21; 23; 24; 25	13

# Table 5.5 The key factors in the integration stage of the 25 interviewees

Therefore, consumers develop profound connections with influencers, marked by attributes like stickiness and loyalty. Such loyalty is indicative of the strong bonds formed during the integration stage of PSR development.

#### Intention to support

Most respondents talk about whether they intend to support the other party or have supported the other part. Intention to support refers to the extent to which the individual is willing to help the other party maintain and increase popularity in the long run (Kim and Drumwright, 2016). The intention to support has been observed from the respondents:

'So now, since I definitely want to help her because, like I mentioned, she's a photographer. And I know recently she decided to open a print shop, which is like printing her images. So I mean, if I was nearby in Europe, I could purchase something, I totally would because they're so beautiful.' Interviewee 18

Many respondents not only express a desire to support influencers but have actively taken steps to do so, such as consistently commenting and sharing posts to increase the influencer's popularity. Interviewee 2 elaborates:

'I want him to become more popular. Because in China, there are a lot of top influencers, and I want him to be one of them. So, whenever he posts something, I will like and comment to support him. I also bought his album to support him.' Interviewee 2

Donations and the provision of virtual money are also recognised as methods of supporting influencers, viewed as investments in the relationship (Cao et al., 2022; Hsieh et al., 2022). As one interviewee explains:

'The article was really good and relevant to me. For example, it was about arts, humanities education in India ... and after I read that article, I donated a small amount to support him.' Interviewee 17

Additionally, some respondents support influencers by purchasing products they have created or endorsed. For instance, one interviewee recounts their experience of buying a product to show their support:

'So I saw through his social media, and I saw one of the shirts he was wearing; it had a kind of dinosaur print. I had actually ordered one to support him.' Interviewee 19

The underlying reason why individuals choose to support or have supported an influencer is their belief in such actions as a form of relationship investment, likely to encourage influencers to produce high-quality content, as highlighted by Interviewee 25:

'Because I think helping them is a way to support them in continuing to create good quality videos for other people, and I will benefit individually. So, it is kind of like an investment instead of spending money.' Interviewee 25

The qualitative findings provide new insights into consumers' intentions to support, or actual support for, their favorite influencers. Consumers actively engage in various ways to show their support for their favorite influencers, which should be considered one of the key outcomes of PSR.

# People's well-being

The second aspect of PSR involves outcomes related to the viewers. Interviewees report feeling more positive through their interactions and relationships with influencers on social media. Reflecting on their personal PSR with influencers, respondents describe feeling happier and receiving encouragement. As one interviewee puts it:

'I feel happier than before because I can always feel happy from their videos. Meanwhile, some influencers will share their daily life. For example, Emma uploaded something about her daughter. I have a nephew who is the same age as Emma's daughter. I can relate to them, which makes me feel happy.' Interviewee 3

Respondents derive emotional satisfaction and support from their engagements with influencers (Han and Yang, 2018). Real-life feelings of loneliness and stress can be mitigated

as influencers provide companionship and share positive outlooks with their viewers, thereby contributing to their well-being (Bernhold, 2019; Bernhold and Metzger, 2020). Another interviewee further illustrates this aspect:

'I think it is more like mental guidance, and you can always find some hotness, some encouragement from their video content. They really helped me through the most difficult time.' Interviewee 24

In conclusion, the qualitative findings highlight the significant impact of PSR on people's well-being. Through interactions and relationships with influencers on social media, respondents report increased positivity, happiness, and emotional support. Influencers, by sharing their personal lives and positive outlooks, foster a sense of connection and encouragement among their viewers, thereby enhancing their overall well-being.

# Brand advocacy

A significant outcome of PSR is brand advocacy. Brand advocacy entails consumers endorsing a brand through word-of-mouth (Wallace et al., 2014; Sahin and Baloglu, 2014). Upon viewing influencers' videos that introduce a brand or product, consumers become acquainted with the product and brand. Subsequently, they are likely to recommend the brand to their friends or generate content about it (Hwang and Zhang, 2018). Particularly, the likelihood of consumers advocating for a brand increase if they purchase the product and perceive the brand positively. This was corroborated in an interview:

Amanda Dupont started her own skincare range called the Lip. So, this lady, she is a South African influencer like supporting local brands. So, as I saw her, she's an actress in South Africa. She was using the skincare range, and then I found out she actually created the skincare range. And since then, I have been buying it and using it. So I have also advocated it to my friends, and I have a few friends using it.' Interviewee 18

Therefore, brand advocacy emerges as a brand related outcome of PSR. This underscores the influential power of social media influencers in shaping consumer purchasing behaviour and perceptions.
#### Brand trust

Brand trust represents another significant consequence of PSR. Interviewees suggested that the endorsement of products or brands by influencers often leads consumers to develop trust in those brands. This trust stems from the consumers' confidence in the influencers; hence, they extend this trust to the brands endorsed by them. As articulated by Interviewee 12:

'I trust the brand because I trust her, and then I start to trust the brand.' Interviewee 12

This process, whereby consumer trust is transferred from the influencer to the brand, is further elucidated by another interviewee:

'So definitely made me trust brands, I would say, a lot more than I would have in the past mainly, I suppose, because I am so overwhelmed by choice half the time. And you do not really know what the best thing is, and there are so many brands out there, like the sustainable and the ethical. So you actually have no idea. But there is someone like the influencers who I have learned the values and what they stand for over the years. And then they promote something I definitely, without a doubt, will be trusted more.' Interviewee 18

In sum, brand trust is identified as one of the key outcomes of PSR. In the third stage of PSR, consumers develop trust in social media influencers. This trust is then transferred to the brand in the fourth stage, benefiting brands by facilitating the development of stronger relationships with consumers.

## Brand preference

Brand preference constitutes an additional outcome deriving from the PSR benefits. It is defined as the tendency of consumers to favour one brand over its competitors (DelVecchio et al., 2006). Influencers play a crucial role in shaping consumer purchase decisions, with consumers often opting for brands used or endorsed by influencers (Zafar et al., 2020). This influence is demonstrated through the experiences shared by interviewees:

'Sam Chui he really likes the Empire Airlines. He was giving a very high comment on Empire Airlines. So when I travelled from my hometown to the UK, I chose to take Empire Airlines.' Interviewee 8

Similarly, another interviewee recounted their brand preference influenced by a celebrity endorsement:

'For Messi, because he was signed by Adidas, so whenever I choose my football boots, I will go for Adidas instead of Nike or Puma.' Interviewee 25

Therefore, brand preference is identified as another brand-related outcome in which consumers' preference for a brand is influenced by their favourite influencers.

# Positive brand attitude

Positive brand attitude constitutes another favourable outcome resulting from PSR advantages. Brand attitude is understood as the consumer's evaluation of a specific brand (Mitchell and Olson, 1981). Respondents indicate that endorsements or shares by influencers lead them to develop a more favourable attitude towards these brands. This influence is exemplified by the reflections of interviewee 2:

'I like to notice what he wears. For example, last time, he carried a Prada bag and wore an Essential jacket, which looked really cool. So, I thought that Prada and Essential are very fashionable brands. So, If I want to buy bags and jackets, I will first choose Prada and Essential.' Interviewee 2

In sum, a positive brand attitude is a notable outcome of the advantages associated with PSR. As evidenced by interviewees, influencer recommendations can significantly enhance a consumer's perception of a brand, making them more likely to view it as fashionable and desirable.

## Purchase intention

Many respondents revealed that influencer endorsements often lead to an increase in their purchase intentions. Purchase intention refers to the consumer's likelihood of buying a product or service, influenced by influencers' endorsements, which is based on their evaluations (Hwang and Zhang, 2018; Lu and Chen, 2021). This effect is illustrated by the experience of Interviewee 3:

*'When they recommend something in their videos, sometimes it makes me want to buy the products.' Interviewee 3* 

Influencers not only endorse products but also present them comprehensively in their videos, aiming to enhance consumers' intention to purchase. This effect is particularly pronounced in highly interactive platforms such as live streaming (Xin et al., 2023). Interviewee 21 provided insight into why live streaming increases his purchase intention:

'He always wears the clothes that he sells in the live streaming. It looks good on him, and I really want to buy these clothes. Because he usually uses some words, such as high quality, most reasonable price, and you must buy it, these words always make me want to buy it.' Interviewee 21

In sum, the qualitative findings indicate that influencer endorsements significantly boost their followers' purchase intentions. As illustrated by interviewees, influencers not only endorse products but also showcase them effectively in their content. This detailed presentation, combined with persuasive language and visual appeal, enhances consumers' desire to purchase the endorsed products.

#### Actual purchase behaviour

A significant number of respondents reported engaging in actual purchases prompted by influencer endorsements. This reflects the substantial influence that influencers wield in guiding consumer purchasing decisions. An illustrative example of this phenomenon is provided by the experience of Interviewee 14:

'I just bought the products they endorsed. For example, I bought the protein vegan food because my favourite influence just endorsed it, and it is healthy and does not harm animals.' Interviewee 14 Therefore, the qualitative findings highlight that influencer endorsements can not only increase their follower's purchase intention but also drive their follower's actual purchase behaviour.

# 5.3.5 PSR vs real-life relationship

Respondents indicate diverse experiences with influencers, ranging from virtual connections to real-life interactions. Among 25 interviewees, only one has transitioned a PSR into a real-life friendship, suggesting that converting virtual relationships into real-life connections is relatively uncommon. Interviewee 9 shared their unique experience:

'Her videos make me feel warm, and I want to be friends with her ... I followed her, and she followed me back, and we communicated on the app. We feel that we are very passionate, and we are keen to connect with each other to be friends ... So, now she is my best friend in real life.' Interviewee 9

Contrastingly, other respondents expressed reluctance to form real-life friendships with their favourite influencers. A prevalent sentiment is that knowing influencers in person might be odd, with a preference for maintaining virtual interactions. Interviewee 4 articulated this perspective:

'I do not think we meet each other in person is a good experience. Actually, sometimes I think I have to keep the distance between me and my favourite influencers. Because, you know, people are not always perfect, I am afraid.' Interviewee 4

Nonetheless, some interviewees are open to attending offline events or campaigns, valuing opportunities to see influencers in real-life settings without pursuing personal interactions. Interviewee 2 reflects this view:

'Some crazy fans will go to the airport to see their favourite celebrities or influencers, but I am not that person. But I will go to the concert if I have the opportunity.' Interviewee 2 Although it is uncommon for consumers to develop real-life relationships with social media influencers, the qualitative findings provide new insights and opportunities, suggesting that consumers can indeed develop social relationships with these influencers.

#### 5.3.6 Parasocial breakup

The phenomenon of parasocial breakups is observed among the respondents as well. A key reason individuals may choose to unfollow someone and undergo a parasocial breakup is due to the other party's engagement in actions or behaviours that are discordant with the individual's own values.

Interviewee 19 elucidates this by sharing their experience with JK Rowling, an author who has garnered significant influence on social media. Despite being primarily known for her literary works, JK Rowling is regarded as an influencer by many, including Interviewee 19, who has followed her on social media since getting a social media account. However, Interviewee 19 experienced a parasocial breakup with JK Rowling due to conflicting values:

'I stopped following JK Rowing because, as I said, I'm from the LGBT community, and she has a different stance on certain perspectives. So now, I do not relate with her as much as I thought I had.' Interviewee 19

In conclusion, although consumers primarily discuss their PSR with social media influencers, the phenomenon of parasocial breakups highlights that these relationships can end. While influencers can build strong, one-sided connections with their followers, these relationships are vulnerable to shifts in perception and values. When influencers act in ways that conflict with the personal values of their followers, it can lead to a parasocial breakup.

### 5.4 Conceptualisation of PSR

The qualitative findings offer significant insights into conceptualising PSR. To propose a good definition of PSR, this study adheres to the criteria for a well-constructed definition as recommended by Murrow and Hyman (1994) and employs the methodology for formulating a definition in marketing as suggested by Gilliam and Voss (2013).

## 5.4.1 The good definition criteria of PSR

Building on the preceding analysis of PSR from the consumers' perspective, the examination of the existing PSR definition in Chapter 2, and the criteria for a well-crafted definition proposed by Murrow and Hyman (1994), this part introduces twelve criteria for defining PSR. These encompass eight domain-specific criteria and four general criteria. The twelve definition criteria are detailed in Table 5.6.

Firstly, a good PSR definition should encompass two entities: one party and the other parties. The one party denotes individuals who unilaterally initiate the relationship, while the other parties represent the counterpart in the relationship. For instance, in this study, the other parties are identified as social media influencers.

Secondly, it is crucial for a PSR definition to acknowledge the conscious and motivational basis of the psychological connection. The relationship exists objectively, beyond mere individual perception. PSR signifies individuals' conscious engagement in the relationship rather than an unconscious state. Reflecting on qualitative data, the deliberate establishment of this relationship by people is evident in their motivations and objectives. Although prior definitions have emphasised emotions, it's important to recognise that emotions, being potentially transient, constitute only a segment of the psychological bond (Houben et al., 2015).

Furthermore, a well-crafted definition of PSR must acknowledge its inherently unequal nature. PSR represents a unilateral connection, where knowledge is possessed by one party but not by the others (Lim and Kim, 2018; Jacobson et al., 2022). This asymmetry is also echoed by some interviewees' reflections on the nature of PSR. Additionally, the definition should capture the dynamic nature of PSR, recognising the varied perceptions and intensities with which individuals experience these relationships. Contrary to previous definitions that confined PSR to intimate or close connections (e.g., Dias et al., 2017), this qualitative data reveal that individuals describe their relationship with influencers in terms of varying degrees of closeness, such as "distant", "close", and "private", and indicate that the strength of PSR can range from weak to strong, influenced by numerous factors.

Definition Criteria	
Domain-specific criteria	Includes one party
	Includes other parties
	Defines main domain rather than perception
	Implies inequality
	Implies dynamic nature of PSR
	Implies influence and value
	Implies time
	Includes how the connection developed
General Criteria	Parsimonious
	Unambiguous
	Useful to practitioners
	Useful to academics

Table 5.6 Good PSR definition criteria

Moreover, akin to social relationship, PSR is valued for their influential capacity, providing both utilitarian and hedonic benefits. Individuals may become more tolerant of differing viewpoints or alter their attitudes towards products or brands endorsed by the counterpart. The temporal dimension is also crucial in defining PSR. As supported by existing literature and qualitative evidence, PSR is a prolonged engagement (Bernhold, 2019), necessitating time for initiation and sustained interaction for maintenance. The development of PSR involves multiple interactions with the counterpart (Lim et al., 2020; Feder, 2020; Reinikainen et al., 2020), and its persistence requires frequent engagement. Furthermore, the role of media, such as social media platforms, in fostering PSR underscores the need to incorporate this element into the definition. Beyond these domain-specific criteria, a good PSR definition should strive for parsimony, clarity, and applicability, serving both academic and practical purposes (Murrow and Hyman, 1994).

## 5.4.2 PSR working definition

Drawing upon the twelve criteria for an effective definition of PSR, this study formulates a working definition guided by the methodology outlined by Gilliam and Voss (2013). As a preliminary step, Gilliam and Voss recommend crafting an initial definition. This research adheres to the criteria for a good definition set forth by Murrow and Hyman (1994), to discern what elements should be included or excluded. Based on an analysis of existing PSR definitions, alongside a thorough examination of qualitative data through content and thematic analysis, this study posits the following working definition: *an individual's* 

# dynamic perception of a valuable psychological connection, unilaterally developed with the other party based on repeated exposure through media or technology.

This proposed working definition necessitates an assessment of its added value, as recommended by Gilliam and Voss (2013). It is formulated on the foundation of existing PSR literature and qualitative insights (MacKenzie, 2003; Churchill, 1979), enhancing the previous definitions by reducing vagueness and precisely delineating the roles of 'one party' and 'other parties' within PSR, thereby narrowing the scope of other parties.

Furthermore, the revised definition encapsulates the main domain characteristics of PSR, such as its inherent inequality, dynamism, and influence, while elucidating the mechanisms of PSR development and incorporating the temporal aspect. It also minimises ambiguity by specifying that PSR constitutes a psychological connection, rather than merely reflecting individuals' feelings, and by implicitly addressing both the antecedents and consequences that previous definitions have overlooked. For instance, the mention of a 'valuable other party' hints at the antecedents of PSR.

Subsequently, the researcher evaluates the definition for ambiguity and vagueness, concluding that it is both parsimonious and unambiguous. Moreover, the researcher examines the boundaries of the new definition, compares them with similar concepts, and deems the definition suitable. This provisional definition will be subject to further refinement based on feedback from experts. Comments from experts and the definitive PSR definition are to be presented in Chapter 9.

# 5.5 Implications of the qualitative study

This qualitative research offers several implications. Firstly, the qualitative data facilitate a comprehensive understanding of PSR, elucidating how individuals perceive the relationship and the nature of PSR. The study affirms that connection, inequality, and psychological closeness are central attributes of PSR, unveils new insights into its dynamic nature, and accentuates the significance of value in comprehending PSR. Consequently, eight domain-specific criteria for a good PSR definition have been identified to conceptualise PSR. A working definition of PSR is proposed, which will undergo revision in Chapter 8 following an expert survey.

Secondly, the research verifies that PSR is a process, not a static condition. By characterising PSR as a process, it captures its evolving nature and underscores the importance of time in the development of PSR. The four stages of PSR development delineate the progression from the initial virtual encounter with the other party (exposure initiation stage), through growing interest (experimentation stage), to a demonstrated desire to sustain the relationship (intensification stage) and culminating in the outcomes of the relationship (integration stage).

Thirdly, the study identifies key factors influencing the PSR development process, thus enriching our understanding of consumers' psychological decisions in cultivating a relationship with another party at each stage. Drawing on qualitative data, distinct factors influencing PSR development at each stage are highlighted. Given the scant research on the PSR development process, this study lays the groundwork for comprehending this intricate process. Furthermore, the qualitative data contributes to the conceptual framework's development, resulting in a more precise model of the PSR development process by integrating qualitative insights into the conceptual framework outlined in Chapter 6.

Fourthly, three categories of potential PSR outcomes are discerned: influencer-related, individual-related, and brand-related outcomes. The qualitative insights shed light on the reasons behind individuals' preferences for purchasing products endorsed by influencers on social media (e.g., the intention to support the influencers). Moreover, the qualitative data disclose various outcomes of PSR, such as its evolution into real-life relationship or parasocial breakup.

## 5.6 Chapter summary

This chapter outlines the insights gathered from 25 semi-structured interviews with consumers who have been following social media influencers for an extended period. Employing content and thematic analysis methods, the chapter explores the keywords and themes associated with the PSR phenomenon, facilitating a thorough comprehension of PSR. Based on the findings, the chapter puts forward several criteria for an effective definition of PSR, aiding in the conceptualisation of PSR. Additionally, through thematic analysis, the chapter narrates the journey of the PSR development process and identifies pivotal factors at each of the four stages. These revelations about the PSR development process are instrumental in advancing the conceptual framework, details of which will be presented in the subsequent chapter.

# Chapter 6: Conceptualising PSR formation process

## 6.1 Chapter introduction

The previous chapter discussed the qualitative data findings and elucidated the conceptualisation of PSR and the process of PSR development. This chapter pivots towards theorising the PSR development process, drawing upon the insights gleaned from qualitative data and extant literature. As corroborated by the qualitative findings (refer to Chapter 5.3), PSR development including four stages: exposure initiation, experimentation, intensification, and integration.

Exposure initiation, the initial stage, marks the point where individuals first encounter or become acquainted with their favoured influencers through various media channels or WOM recommendations. This pivotal stage in PSR relationship building is chiefly propelled by individuals' diverse motivations and their attraction to the influencers' characteristics and content. Proceeding to the experimentation stage, individuals develop an interest in the influencers and seek further knowledge about them. At this stage, individuals assess the influencers' characteristics and values, aiming to forge a connection. Following this, during the intensification stage, individuals form a stronger bond with the influencers, aspiring to sustain this relationship in the long term. The culmination of PSR development is the integration stage, representing the outcome stage where individuals have intention to support to the influencers. Influencers significantly impact their followers' decision-making and behaviour in this stage.

The structure of this chapter is outlined as follows. Initially, two models are examined as the theoretical underpinning for the development of the PSR model—Tukachinsky and Stever's (2018) PSR model and the Attention, Interest, Desire and Action (AIDA) model. Subsequently, the four stages of the PSR development process are discussed in detail: exposure initiation, experimentation, intensification, and integration. The key concepts within each stage are identified and explored. Research hypotheses are posited, culminating in the presentation of the PSR development process model in Figure 6.1.

#### 6.2 Theoretical framework

#### 6.2.1 Tukachinsky and Stever (2018) PSR model

Tukachinsky and Stever (2018) focus on the development process of PSR and propose a model comprising four stages: exposure initiation, experimentation, intensification, and integration. This model, conceptualised from a media science perspective, addresses the scant investigation of PSR as a dynamic process. It outlines the key concepts inherent in each stage, laying a foundation for understanding how individuals develop PSR with media figures.

Grounded in Knapp's relationship development model, Tukachinsky and Stever (2018) argue that the progression of PSR mirrors the establishment of real-life interpersonal and social relationships. According to their model, advancing through each PSR development stage from initiation to integration is pivotal for someone to become a fan. However, it is crucial to acknowledge that not every individual will traverse all four stages, as they possess full control over this process and can choose to stop at any stage or proceed in fostering the relationship.

This thesis integrates Tukachinsky and Stever's (2018) PSR model as a theoretical framework for several reasons. Firstly, it represents the only model theorising PSR development that captures its dynamic nature and aligns with this thesis's conceptualisation of PSR. Whereas most literature views PSR as a static concept, Tukachinsky and Stever (2018) argue for its consideration as a dynamic process. Secondly, their four-stage model, underpinned by the social relationship development paradigm, posits PSR as an alternate form of interpersonal and social relationships, albeit differing in the lack of reciprocity (Hartmann and Goldhoorn, 2001; Kim and Kim, 2017). Lastly, qualitative findings (see Chapter 5.3) affirm the existence of these four stages in PSR development with influencers, thus validating the model's application within this thesis.

Despite the importance of Tukachinsky and Stever's (2018) PSR model, there is a compelling case for an amended version. This PSR model primarily draws from studies of offline PSR, focusing largely on traditional media contexts such as television, soap operas, and films. These contexts differ markedly from online environments. As outlined in Chapter 2.8, the transition from traditional to online media represents significant shifts, particularly in terms

of geography and accessibility. Online media eliminates geographic barriers, enabling global content access and information acquisition. The interactive and personalised nature of social media content results in changes to consumer behaviour, notably a shift from passive consumption to active engagement (Kaplan and Haenlein, 2010). This interactive dynamic allows for two-way interactions, as opposed to the one-way interaction characteristic of traditional media environments (Swani and Labrecque, 2020). For instance, social media users can like, comment on, and share posts, fostering a highly interactive experience (Kabadayi and Price, 2014; Santos et al., 2022). Conversely, traditional media consumption is limited to content provisioned by the media outlets.

Due to the differences between offline and online environments, the conditions of PSI and PSR are different in the offline and online environments. Chapter 2.8 highlights 12 aspects in which PSR diverge in online contexts as opposed to offline ones. For instance, the online environment provides enhanced opportunities for the development of PSR and facilitates a more rapid formation of these relationships. The convenience and heightened accessibility of online media contribute to increased PSI frequencies, thus expediting the PSR development process and altering its pattern (Leite and Baptista, 2022; Burnasheva and Suh, 2022). This implies that PSR in the online contexts have a wider scope and may exhibit distinct critical conditions and contributions at each stage of PSR development. Consequently, it is imperative to revise Tukachinsky and Stever's (2018) PSR model to reflect the unique dynamics of the online environment.

Given that this thesis is centred on influencer marketing, a prominent digital strategy and widespread phenomenon, it is essential to adapt Tukachinsky and Stever's (2018) PSR model specifically for the online influencer marketing context. This adaptation should incorporate the distinctive attributes of influencer marketing, insights from qualitative findings, and current PSR research. The focus on influencer marketing necessitates considering only influencers as the other parties in PSR. A key attribute of influencer marketing is the deep connection audiences form with influencers and their reliance on these influencers' recommendations for making purchase decisions (Yuan and Lou, 2020; Whang and Im, 2021). This profound relational connection effectively reduces the perception of commercial persuasion exerted by influencers (Breves et al., 2021) and positively impacts the perception of products and brands endorsed by influencers, leading to favourable brand attitudes (Reinikainen et al., 2021; Balaban et al., 2022) and heightened purchase intentions (Masuda et al., 2022; Leite and Baptista, 2022).

The qualitative findings further inform the amendment of Tukachinsky and Stever's (2018) PSR model. Insights from twenty-five interviewees shed light on the nuanced process of developing PSR with favoured influencers, offering deeper understanding into this phenomenon. Contrary to Tukachinsky and Stever's (2018) suggestion that the initiation stage of PSR begins with an individual's first interaction with the media figure, the qualitative data reveals a broader range of initial encounters. As detailed in Chapter 5.3.1, individuals often first become aware of influencers through media consumption, such as television and social media. However, it also emerges that initial awareness and impressions can be formed through WOM encounters, as illustrated by an interviewee's experience with discovering an influencer recommended during professional collaboration: "And the first one is because of my work. I was going to work with a brand planner for an event, and they mentioned that Sophie is their ambassador in the UK. So, then I opened Instagram and saw her posts." (Interviewee 22). This finding suggests that the initiation of PSR may occur through a variety of channels, including indirect recommendations or mentions, broadening the conventional understanding of how these relationships begin.

At this stage, individuals form a preliminary impression of the other party, accompanied by a significant degree of uncertainty regarding them (Tukachinsky and Stever, 2018; Yuan and Lou, 2020). The attraction to the other party, such as a media figure's attractiveness, prompts individuals to seek further exposure to them (Tukachinsky and Stever, 2018; Reinikainen et al., 2021). This pursuit of additional encounters is driven by the initial allure, whether it stems from the media figure's physical appearance, charisma, perceived similarities, or the content they produce. As individuals engage more deeply with the media figure, their initial uncertainties begin to diminish, laying the groundwork for a stronger parasocial relationship.

Furthermore, the current PSR studies also suggest the PSR model proposed by Tukachinsky and Stever (2018) can be improved. More than 40 high-quality journal articles have provided more insights into the antecedents and consequences of PSR. For example, in the experimentation stage, Tukachinsky and Stever (2018) highlight the role of perceived similarity. Still, current literature also suggests that perceived influencer expertise is essential in developing PSR (Yuan and Lou, 2020). In the final stage, the theorising of PSR outcomes proposed by Tukachinsky and Stever (2018) focuses on self-concept and mood. However, the current literature suggests PSR can generate more positive outcomes, such as influencer stickiness (Hu et al., 2020) and brand engagement (Zhong et al., 2021).

Tukachinsky and Stever (2018) identify the key predictors in the PSR development process but do not propose assumptions regarding the relationships among these predictors, nor do they empirically test the model. While they provide a theoretical foundation for the PSR development model, serving as a valuable starting point that offers insights, the absence of empirical testing leaves the model's validity and reliability in question. Moreover, the practical application of the model remains ambiguous. Hence, it is crucial to consult current literature and qualitative findings to refine Tukachinsky and Stever's (2018) PSR model, aiming to gain a deeper understanding of how people form PSR with influencers within the influencer marketing context.

#### 6.2.2 AIDA model

The AIDA model, introduced by E. St. Elmo Lewis in 1898, was initially proposed as a strategy for personal selling within the life insurance industry. It has since gained widespread application in the field of marketing, particularly concerning research on online communications and social media (Hassan et al., 2015). The model is employed extensively to comprehend the psychological processes of consumers and their behaviour in media environments (Song et al., 2021). AIDA posits that consumers need to go through a sequence of stages in their journey as customers, starting with gaining attention, developing interest, cultivating desire, and ultimately taking action (Kang, 2022).

In the initial attention stage, securing consumers' attention at the cognitive level represents the foremost step in the communication process (Song et al., 2021). Various factors can enhance consumer attention, including marketing strategies like advertising and search engine optimisation, as well as the internal motivation of consumers (Hassan et al., 2015). For instance, research indicates that within the context of social media, the creativity and authenticity of short videos play pivotal roles in elevating consumer attention and transitioning them into the interest stage (Wu and Lai, 2023).

Interest, delineated as the second stage, embodies consumers' emotional responses towards objects, prompting them to seek additional information at the affective level (Lee and Hoffman, 2015). When consumers find content intriguing, they are inclined to pursue further details and expand their knowledge (Kang, 2022). For instance, online consumers demonstrate their interest in products by actively seeking information from social media

influencers (Wong et al., 2023). Throughout this engagement, individuals are more likely to experience positive emotions, such as enjoyment (Wu and Lai, 2023).

Once individuals have garnered attention and cultivated interest, they may subsequently develop a desire for something (Kang, 2022). Desire encapsulates the motivational emotions and aspirations consumers harbour towards possessing the offering (Weng et al., 2021; Wong et al., 2023). The perceived value of the offering significantly influences consumer desires (Weng et al., 2021). Consequently, companies prioritise articulating a compelling value proposition and fostering consumer trust as strategies to evoke consumer desires (Prathapan et al., 2018). When consumers recognise the value of the objects, their desire to acquire these items emerges, guided by their needs and preferences (Pashootanizadeh and Khalilian, 2018).

The final stage involves taking action, which corresponds to the behavioural level (Wu and Lai, 2023). In this stage, companies endeavour to motivate consumers towards actionable outcomes. This stage is characterised by consumers exhibiting positive behaviours, such as manifesting the intention to purchase and proceeding to actual buying (Song et al., 2021). Nevertheless, it is not guaranteed that consumers will always progress to taking action; rather, they might instead develop a more positive attitude towards the product or service without immediate action (Wu and Lai, 2023).

The AIDA model is deemed suitable for this thesis in identifying the key concepts at each stage of PSR development for two primary reasons. Firstly, the AIDA model has received extensive application in research pertaining to online and social media contexts. For instance, previous studies have employed the AIDA framework to explore strategies for promoting brands online effectively, thereby enhancing communication effectiveness and yielding positive outcomes with brands (Gupta et al., 2016; Song et al., 2021). Secondly, AIDA encapsulates the individual psychological process encountered when engaging with the online environment. This aspect is particularly pertinent to the current research's objective of pinpointing essential psychological conditions at the individual level (Wei et al., 2022). Consequently, the AIDA model is regarded as an instrumental framework that facilitates the identification of crucial psychological cues within the process of relationship development. By adopting the AIDA model, this study delineates involvement, perceived value, PSR, and purchase intention as the pivotal concepts at each stage of PSR development.

According to the AIDA model, the initial stage involves capturing and enhancing consumer attention (Kang, 2022). This study suggests that involvement plays a pivotal role at this stage, as it influences the extent of consumers' focus on their behaviour. Involvement is characterised as the degree to which viewers concentrate on the act they are experiencing when using social media (Vazquez et al., 2017). Given the vast array of information present within the social media environment, coupled with consumer distraction, individuals may not attend to every influencer they encounter (Chang and Thorson, 2023). Hence, as the first stage in the development of PSR, it is essential for consumers to concentrate on the content they are viewing, particularly that which is produced by specific influencers.

In the second stage, marked by the influence of interest, this thesis posits that perceived value is a crucial determinant in consumers' overall assessment when sourcing information from influencers in whom they have developed an interest (Song et al., 2021). Perceived value is delineated as the extent to which a media user derives usefulness and enjoyment from engaging with influencers' posts (Overby and Lee, 2006; Lou and Yuan, 2019). As outlined by the AIDA model, during the interest phase, consumers actively process the information related to the target to form their perception, such as assessing the ease of use and the usefulness (Song et al., 2021). Consequently, consumers are able to perceive the value offered by influencers during the interest stage through consumption with the content they produce.

In accordance with the AIDA model, during the desire stage, consumers' objectives are predominantly driven by emotional factors (Wen et al., 2021). This thesis advocates that, under the influence of desire, consumers aspire to develop and sustain a stable, long-term connection with influencers, which is PSR. As elaborated in Chapter 5.4.2, PSR refers the psychological connection that develops between consumers and influencers. This connection encapsulates consumers' distinct aspiration to cultivate a relationship with the influencer, which evolves into an actual psychological connection that has been established.

The final stage of the AIDA model is action, characteristically manifested through actual behaviour, behavioural intentions, or attitudes (Song et al., 2021; Wu and Lai, 2023). In light of this, the thesis argues that purchase intention plays a pivotal role in the ultimate stage of PSR development. Purchase intention is defined as consumers' intention to purchase a product or service under the influence of influencers based on their evaluation (Hwang and Zhang, 2018; Lu and Chen, 2021). It is considered one of the most significant antecedents in forecasting consumers' actual buying actions and a vital factor for assessing the efficiency

of influencer marketing (Silva et al., 2019). Furthermore, many studies in influencer marketing have highlighted that consumers' purchase intentions are significantly influenced by their preferred influencers (Breves et al., 2021; Balaban et al., 2022).

#### 6.3 Proposed model of PSR development process

## 6.3.1 Stage 1: Exposure initiation

In the first stage of PSR, individuals encounter the other party for the first time through various media channels. This exposure initiation stage is typically short, as individuals critically evaluate the other party and gather basic information (Tukachinsky and Stever, 2018). For instance, when individuals engage with social media, they encounter numerous posts or videos produced by unknown creators. This marks the beginning of the exposure initiation phase (Sokolova and Perez, 2021). At this point, individuals decide whether they wish to continue engaging with and receiving information from this creator. Some viewers opt not to proceed, remaining in the initial stage of exposure (Tukachinsky and Stever, 2018).

Consuming media does not necessarily lead to initial impressions about others due to generally low levels of attention and ineffective communication. This lack of communication effectiveness stems from individuals filtering out information (Duff and Sar, 2014; Wolfe, 2021). On one hand, the tendency to multitask results in a failure to engage with consumed information fully (Santoso et al., 2020). On the other hand, viewers engage in processing information from multiple sources simultaneously, leading to intuitive feelings and limited understanding of the other party (Hartmann and Goldhoorn, 2011). Hence, capturing attention and focusing on the other party represents critical initial steps at the beginning of PSR. The AIDA framework supports this, highlighting the importance of attracting attention to raise awareness, thereby providing an opportunity to progress to the next stage (Lagrosen, 2005; Daugherty and Hoffman, 2013). Subsequent sections will explore key factors further. The complete conceptual framework is depicted in Figure 6.1.

## Involvement

As is discussed in Chapter 6.2.2, involvement is identified as a pivotal concept in the exposure initiation stage, dictating whether individuals will pursue further knowledge about another party and deepen their PSR. The definition of involvement varies across disciplines,

# Figure 6.1 Conceptual framework



such as marketing and media, depending on the research context and objectives. From a marketing perspective, involvement pertains to the consumer's engagement with a product and decision-making processes regarding purchases (Zaichkowsky, 1986). In contrast, the media perspective on involvement emphasises the relationship between media consumers and the information presented, along with the locus of the consumers' psychological activities (Sun et al., 2008). Social media involvement, specifically, relates to users' perceptions of their concentration level while engaging with social media practices (Vazquez et al., 2017). Given the focus of this thesis on users' involvement as the extent to which viewers concentrate on the act they are experiencing when using social media (Vazquez et al., 2017).

Drawing on existing literature, visual attention theory, and qualitative findings, this thesis delineates three pivotal conditions that bolster consumer involvement on social media platforms. Primarily, the motivations of viewers play an instrumental role in augmenting consumer involvement (Leung and Bai, 2013). When individuals perceive that their needs are being met, their likelihood of involvement more deeply increases (Leung and Bai, 2013). The literature indicates that the extent of online users' involvement is, in part, influenced by their diverse motivations (Sun et al., 2008). Furthermore, interviewees have highlighted various motivations, including the pursuit of information and the desire to forge relationships, as catalysts for their involvement with social media content (refer to Chapter 5.3.1). Consequently, this thesis asserts the pivotal role of viewers' motivations in fostering increased consumer involvement in social media.

Secondly, the theory of visual attention suggests that bottom-up factors can account for the extent of a viewer's focus within a complex environment (Wilson et al., 2015). Bottom-up characteristics pertain to the intrinsic properties of objects that may involuntarily heighten attention (Katsuki and Constantinidis, 2013). Fundamental aspects of an object or content, such as colour and size, are shown to significantly enhance consumers' focus (Wilson et al., 2015). The qualitative findings further indicate that essential features of social media posts can boost consumer engagement. As explored in Chapter 5.3.1, the appeal of social media posts, including background music and the design of the post's cover page, plays a role in drawing consumers' attention and engagement with influencers. Consequently, this thesis posits that the appeal of a post constitutes a vital factor in augmenting consumer involvement in the social media context.

Thirdly, researchers in social relationships propose that individuals' facial features and qualities contribute significantly to their focus and attention, particularly the eyes (Richards et al., 2015). In influencer marketing, influencers' attributes, such as their physical traits, have been identified as critical factors in the initial stage of PSR development (Yuan and Lou, 2020; Yuan et al., 2021). The qualitative findings further indicate that the key characteristics of influencers are closely linked to consumer involvement, predominantly influencers' attractiveness. Interviewees have noted that influencers' social, physical, and vocal appeal are principal factors drawing them into the media consumption experience (refer to Chapter 5.3.1). Consequently, this thesis advocates that influencer attractiveness is a vital condition for augmenting consumers' involvement within the social media context. Next, this chapter will provide a critical review of the three key factors.

## Viewers' motivation

Although the definition of motivation has yet to be universally agreed upon, it generally refers to the internal mechanisms that arouse individuals, ultimately driving them to take specific actions (Kleinginna and Kleinginna, 1981; Oh and Syn, 2015). This thesis defines **motivation** as the drive to satisfy individual needs through social media (Park and Lennon, 2004; Kim, 2020). Adopting self-determination theory (SDT), it argues that individuals are capable of making their own choices and decisions (Ryan and Deci, 2000). Drawing on SDT and qualitative findings, it identifies two primary motivations for consuming social media content: intrinsic and extrinsic (Kim and Drumwright, 2016).

Intrinsic motivation refers to the natural tendency of individuals towards activities that are inherently satisfying, such as enjoyment (Ryan and Deci, 2000). The intrinsic motivations for using social media predominantly include entertainment, seeking information, and building relationships (Kim, 2020; Zafar, Qiu, and Shahzad, 2020; Chiu and Huang, 2015). However, the motivations for using different social media platforms vary (Kim and Lee, 2016). Oh and Syn (2015) conduct a comparative study on users' motivations across Facebook, Twitter, Delicious, YouTube, and Flickr, finding that motivations are not uniform across these platforms due to the differing functions and information each provides.

**Entertainment** refers to individuals seeking temporary spiritual relaxation, enjoyment, and fun through social media (Park et al., 2009; Yang and Ha, 2021). Researchers argue that the motivation for entertainment is the primary driver for the use of social media (Chiu and

Huang, 2015; Liu et al., 2019; Kim, 2020). Individuals increasingly turn to the internet for daily entertainment, particularly via social media platforms like Instagram and YouTube, which offer a wealth of entertainment options for consumption (Heinonen, 2011; Noguti and Waller, 2020). People are more inclined to seek relaxation and distraction by engaging with content on social media in a low-cognitive manner (Noguti and Waller, 2020). When individuals engage with entertaining content, and viewers are presented with this interactive amusement, they experience relaxation and joy, perceiving the characters within the media as sharing something compelling (Liu et al., 2019; Croes and Bartels, 2021).

**Information seeking** is defined as the proactive endeavour to gather information to fulfil personal objectives via social media (Asghar, 2015; Zafar et al., 2020). The acquisition of information is deemed crucial for media users (Esteban-Santos et al., 2018). In traditional media, individuals access news through specific television programmes (Rubin et al., 1985). Within the online sphere, social media is considered a vital channel for acquiring information about friends (Frampton and Fox, 2021), celebrities and influencers (Kim and Kim, 2020; Jin and Ryu, 2020), products (Leonhardt et al., 2020), brands (Colliander and Erlandsson, 2013; Qin, 2020), and for keeping abreast of news and solving problems (Chiu and Huang, 2015). Influencers, in particular, are known to furnish detailed information about products, brand insights, and product usage experiences, thereby catering to consumers' information-seeking motivations (Wongkitrungrueng et al., 2020).

This thesis defines **relationship building** as the use of social media by individuals to establish and sustain connections (Yuan et al., 2016). The fundamental social need among people is the desire to connect with others (Oleson, 2003). Due to its convenience and effectiveness, individuals frequently turn to social media to forge and enhance relationships. On one hand, social media facilitates ongoing interaction and the maintenance of relationships with family and friends (Williams and Merten, 2011). On the other hand, it enables individuals to interact and cultivate connections with celebrities, influencers, and community members (Sokolova and Perez, 2021). Consequently, individuals seeking to build and uphold relationships are more inclined to engage with social media platforms.

Extrinsic motivation, as opposed to the intrinsic motivations previously discussed, pertains to engaging in activities for the sake of achieving a separable outcome, such as evading punishment or securing monetary rewards (Ryan and Deci, 2000). Although extrinsic motivation plays a crucial role in determining viewers' behaviour on social media, the lack of viewers' self-reports means that extrinsic motivations, particularly economic incentives,

are often overlooked by scholars (Vilnai-Yavetz and Levina, 2018). Typically, these economic incentives include discounts and coupons. The qualitative findings further highlight consumers' extrinsic motivation when engaging with influencers' content on social media. For example, one interviewee mentioned, "*Jiaqi Li shares a lot of useful information about cosmetics for young women. And he often offers many discounts, so I frequently check his posts to find product discounts*" (Interviewee 7).

**Economic rewards** refer to monetary incentives used by influencers, celebrities, and brands to draw viewers, potentially offering utilitarian value to consumers within the social media context (Coker et al., 2015). Enticing promotional content on social media can impact consumers' purchasing decisions and motivate followers to stay updated with the latest promotional offers (Yasa et al., 2020; López et al., 2021). Research indicates that economic rewards are effective in augmenting consumers' website browsing and online shopping activities (Joines et al., 2003), fostering brand participation on social media (Kim and Drumwright, 2016), enhancing consumer engagement on social media platforms, improving the company's reputation through word-of-mouth, and boosting the sharing of e-commerce content (Ryu and Feick, 2007; Vilnai-Yavetz and Levina, 2018).

The existing literature and qualitative findings indicate that the motivations of social media users are crucial factors influencing their involvement. In the digital context, the involvement of users is partially determined by the variety of their motivations (Sun et al., 2008). Sun et al. (2008) demonstrate that online users' motivations for entertainment, information seeking, social interaction, and control have a positive impact on their involvement, subsequently affecting their reliance on media. Further studies corroborate that the motivations behind using social media, such as information acquisition, relationship building, socialising, and entertainment, are positively associated with the level of user involvement with brands' social media pages (Leung and Bai, 2013). The qualitative evidence also supports these findings. For example, one interviewee shared, "*I remember trying to improve at playing League of Legends, but I wasn't very skilled. So, I decided to look up some video tutorials from top game bloggers. I just entered some keywords and found a video that seemed to fit my needs. I clicked on it and began to engage with the content" (Interviewee 21).* 

#### Influencer attractiveness

Attractiveness is regarded as a quality or an attribute that plays a vital role in daily life and the online environment (Walther et al., 2008; Yuan and Lou, 2020). In social relationships, the attractiveness of an individual can influence the impression of other people and the relationship between themselves (Krebs and Adinolfi, 1975). For instance, people will regard attractive persons with more positive qualities, such as honesty (Cloutier et al., 2008; Paunonen, 2006). In a romantic relationship, the perceived attractiveness of a partner is positively related to love and higher satisfaction (Sangrador and Yela, 2000).

In the social media environment, attractiveness is one of the critical influencer characteristics (Gupta et al., 2022; Wiedmann and von Mettenheim, 2020). Influencer attractiveness can lead to several positive outcomes and determine the marketing campaign's success (Kim and Park, 2023; Vrontis et al., 2021). Viewers are more willing to establish relationships with attractive influencers. For example, according to the existing research, viewers think that LeBron James is attractive, which is positively related to fans establishing PSR with LeBron James (Yuan et al., 2016). Other studies also confirm that people are more willing to develop long-term relationships and become fans if the influencer is attractive, it will increase PSI (Lee and Watkins, 2016), increase consumer engagement on social media (Fang et al., 2020) and increase people's purchase intention (Chen et al., 2021).

This thesis proposes influencer attractiveness, which is presented in three aspects: physical, social, and voice attractiveness, based on current studies and qualitative findings (Knoll et al., 2015; Kim and Kim, 2021). Physical, social, and vocal attractiveness contribute to the overall attractiveness of influencers, but people perceive them differently based on their different sensory perceptions (Groyecka et al., 2017). **Physical attractiveness** refers to viewers' perception of the degree to which the influencer's physical features are pleasing to observe, such as the influencer's facial or body appearance (Wiedmann and von Mettenheim, 2020; Kim and Park, 2023). Influencers' physical attractiveness is based on viewers' visual perception (e.g., *'She looks cute, and I want to see more about her daily life*' (Interviewee 23)). People are likelier to watch the content and videos generated by more attractive influencers because it leads to visual aesthetics (Ki et al., 2020).

**Social attractiveness** can be defined as viewers' perception of a person's familiarity and likeability, indicating social media users' intention to communicate and build relationships

with the other party (Liu et al., 2019; Han and Yang, 2018). In most cases, people tend to regard influencers who are similar to themselves as socially attractive (Lee and Watkins, 2016). In the social media environment, viewers easily perceive the internal similarities (e.g., age and background) and external similarities (e.g., value and attitude) with the influencers (Fu et al., 2017). Further, suppose the influencer has a large number of followers. In that case, social media users are more likely to perceive the influencer's popularity (De Veirman et al., 2017) and regard the influencer as having high social attractiveness (e.g., '*And I think because he is one of the most famous, he was like showing up as the first one*' (Interviewee 25)).

Voice attractiveness can be regarded as the third aspect of influencer attractiveness, assessed auditorily (Groyecka et al., 2017). In advertising research, scholars have found that voice quality and prosodic can influence consumers' perceptions and attitudes (Gelinas-Chebat and Chebat, 1992; Casado-Aranda et al., 2018). Interviewees also mention that they are attracted by influencers' voices (e.g., '*When I first listened to his songs, I was attracted by his euphonious voice*' (Interviewee 2)). Therefore, this thesis proposes **voice attractiveness** is the third dimension of influencer attractiveness and can be defined as the viewer's perception of the degree to which the influencer's voice characteristics are pleasing to hear.

Existing literature and qualitative findings suggest that social media users' attractiveness influences their involvement. Psychology scholars find that individual facial beauty can appeal to people's spatial attention (Sui and Liu, 2009). As discussed above, users' involvement focuses on to what extent people are concentrated and pay attention to the act when they consume social media (Vazquez et al., 2017). People are attracted to attractive influencers in the online environment (Lee and Watkins, 2016). People pay more attention to Facebook users who are highly physically attractive than unattractive users (Seidman and Miller, 2013). Influencers pay great attention to their appearances, such as delicate makeup and nice clothes, to increase their attractiveness. High attractive people can increase other people's attention, especially in unconscious conditions (Hung et al., 2016). Meanwhile, influencers also employ performance techniques (e.g., smiling) to increase social attractiveness. If people perceive attractive faces with rotated faces and smiles, it can significantly attract viewers' visual focus (Kim et al., 2018). The qualitative findings also support this (e.g., 'When I watch beauty videos, I noticed that she is a very interesting person who looks very down to earth. I also noticed that she also has a Greek background, as I mentioned. So, I watched her videos and listened to her suggestions.' (Interviewee 15)).

#### Visual appeal

In this thesis, **visual appeal** is defined as the extent to which viewers perceive an influencer's post as visually or aesthetically pleasing (Ki and Kim, 2019). Within advertising and social media contexts, maintaining a high aesthetic standard in images and videos can elicit strong emotional reactions and convey specific messages to the audience (Teng et al., 2014). In the social media environments, users show a preference for high-quality posts and images that are visually attractive (Al-Emadi and Ben Yahia, 2020). For instance, posts characterised by high informational quality, design excellence, creativity, and technological sophistication are deemed visually appealing (Cheung et al., 2022). Employing photos with aesthetic appeal on social media, such as those with high image contrast, can enhance processing fluency and aesthetic response, subsequently leading to more positive behaviours (Kostyk and Huhmann, 2021). Moreover, influencers often utilise professional video equipment and image post-processing techniques to boost visual appeal and attract a larger following (Al-Emadi and Ben Yahia, 2020).

Images and videos of high visual appeal in online contexts exert significant persuasive effects and influence user behaviour. Firstly, visual appeal can enhance customers' perceptions of product qualities, positively affect their attitude towards the product, and lead to positive behavioural outcomes, such as purchase intention (Huettl and Gierl, 2012) and word-of-mouth recommendations (Ki and Kim, 2019). Additionally, visual appeal contributes to the enhancement of customers' hedonic value (Amanah and Harahap, 2020), enriches the customer experience (Zhang et al., 2015), and fosters customer satisfaction (Chi, 2018). Furthermore, the high visual appeal of social media content has a positive impact on online customer engagement (Rietveld et al., 2020).

Both existing literature and qualitative findings underscore the significant influence of visual appeal on viewer involvement. Visual attractiveness serves as a powerful stimulus, capable of making an instant impact on viewers, encouraging them to pay greater attention and devote more time to the content (Lindgaard et al., 2006; Gilani et al., 2013). Content creators on social media often employ design techniques to craft attractive posts, notably through the use of aesthetically pleasing photographs (Kostyk and Huhmann, 2021; Al-Emadi and Ben Yahia, 2020). Research indicates that content with visual appeal on platforms like YouTube can heighten viewers' focus and interest, thereby increasing the number of views and likes (Gupta et al., 2016). Qualitative evidence supports these findings, as exemplified by an

interviewee's remark: " *Some influencers, their video headlines are attractive, and it captures my attention to see the content*" (Interviewee 1).

In conclusion, exposure initiation represents the first stage, in which consumers are introduced to influencers via social media. As delineated by the AIDA model, alongside qualitative findings and existing literature, three pivotal elements are instrumental in determining consumer involvement: consumer motivations, influencer attractiveness, and visual appeal. The level of consumer involvement serves as a predictor for whether individuals will pursue further information about the influencer and establish PSR. High consumer involvement facilitates progression to the experimentation phase, during which consumers begin to form a PSR with the influencer, setting the stage for their ongoing journey of PSR development. The subsequent section will elaborate on the theorisation of the experimentation stage.

## 6.3.2 Stage 2: Experimentation

During the experimentation stage, media users form their impressions of the other party and express interest in gaining further knowledge about them (Tukachinsky and Stever, 2018). As posited by the AIDA model, cultivating interest is crucial for acquiring additional information and motivating individuals to progress to the subsequent stage (Lee and Hoffman, 2015). At this stage, media users continue to engage with the media content, forming perceptions of the other party and reducing uncertainty regarding them (Tukachinsky and Stever, 2018). Individuals interpret the other party through both auditory and visual cues, facilitating the experience of PSI by consuming the other party's posts or live streaming on social media platforms (Boerman, 2020; Kim et al., 2020).

Within the social media environment, media users readily access auditory information and interactive signals, fostering a belief in conversational communication with the other party (Cummins and Cui, 2014). Additionally, non-verbal cues, including physical gestures, are communicated to the audience, enriching their experience (Cummins and Cui, 2014). Such non-verbal cues, like eye contact, have been identified as crucial components that lead viewers to feel a sense of physical interaction with the other party (Horton and Wohl, 1956; Oliver et al., 2019). The conveyance of non-verbal information varies with the medium; while images and videos effectively transmit these cues visually, audio or textual formats

may struggle to do so. Through this interaction, viewers assess various attributes of the other party, perceiving their characteristics and values.

PSI

Drawing from Tukachinsky and Stever's (2018) PSR model, and supported by qualitative findings, PSI is identified as a critical concept in the experimentation stage. PSI is broadly understood as the conversational communication experience by viewers with another party through media (Tukachinsky et al., 2020). It accentuates the digital interaction between individuals and the sensation of mutual awareness, attention, and adaptation with another party during media consumption (Lo et al., 2022; Hartmann and Goldhoorn, 2011). While some researchers perceive PSI as the virtual connection between viewers and media attributes, this thesis delineates PSI and PSR as distinct constructs, with seven criteria differentiating them as outlined in Chapter 2. Specifically, this research concentrates on **PSI** within the context of influencer marketing, and defines it as individuals' perception of direct, reciprocal and one-to-one communication with influencers through social media (Hartmann and Goldhoorn, 2011).

Verbal and non-verbal cues from communicators can significantly enhance viewers' experiences of PSI. Direct cues, in comparison to indirect ones, are more effective in fostering stronger PSI (Oliver et al., 2019). The manner in which information is presented verbally plays a crucial role in both its delivery and the recipient's perception (Parhankangas and Renko, 2017). For instance, employing interactive verbal strategies, such as posing questions, can positively influence viewers' perceptions of engaging in a direct dialogue, even if the questions are not personally addressed to them (Cummins and Cui, 2014). Similarly, body language, a prevalent form of non-verbal communication, impacts PSI significantly. Horton and Wohl (1956) note that TV presenters' expressive body language, like making eye contact with the camera, can intensify viewers' sense of PSI.

PSI can lead to a range of outcomes. For instance, Kim and Kim (2017) find that individuals' PSI with websites could boost well-being and promote WOM recommendations. Moreover, when consumers experience PSI with an endorser, this can enhance their perception of the source's credibility, improve their attitude towards both the advertisement and the product, and heighten their purchase intentions (Gong and Li, 2017). Similarly, users' PSI with a smart speaker is linked to increased perceptions of the device's usefulness and ease of use,

which, in turn, influences their attitude, intentions to continue using the product, and online purchase intentions (Hsieh and Lee, 2021). Additionally, users' PSI with live streamers can amplify consumers' affective responses, leading to an increase in impulsive buying urges and behaviour (Lo et al., 2022).

### Perceived value

As explored in section 6.2.2, perceived value emerges as a key concept within the experimentation stage, crucial for determining whether individuals will advance to the subsequent stages of PSR development. Echoing the AIDA model, consumers actively assimilate information about the target during the interest stage to form their perceptions, such as perceived ease of use and perceived usefulness (Song et al., 2021). Contrary to certain existing studies that view perceived value as a unidimensional construct, this thesis adopts a multidimensional perspective. It distinguishes between hedonic value and utilitarian value, aligning with contemporary consumer behaviour research (Chang and Tseng, 2013; Lou and Yuan, 2019). Consequently, this thesis conceptualises **perceived value** as the overall evaluation of relevant factors when engaging with influencers on social media (Zeithaml, 1988; Lou and Yuan, 2019).

In the online environment, utilitarian value is closely linked to consumer purchasing behaviour, as consumers can effortlessly discover products they desire and proceed to payment (Bridges and Florsheim, 2008). Additionally, the efficiency and utility of e-services further enhance the online purchasing experience for consumers (Etemad-Sajadi and Ghachem, 2015). In the context of online shopping, utilitarian value manifests in consumers' pre-purchase considerations and decisions. This encompasses evaluating product characteristics and quality, the ease of searching for products, transaction convenience, and the availability of discounts (Wu et al., 2014; Fang et al., 2016; Sharma and Klein, 2020).

The **utilitarian value** of an influencer's social media post can be defined as the degree of usefulness a media user assesses from reading or watching influencer posts (Overby and Lee, 2006; Meske et al., 2019). Viewers may perceive utilitarian value through recommendations from influencers that aid in assessing products or services (Guo et al., 2021). It is common for consumers to discover valuable information within social media accounts (Zafar et al., 2020). When consumers recognise the utilitarian value in social media posts, it leads to an

enhancement of their PSR with the influencer, which in turn enhance consumer brand engagement and fosters brand loyalty (Zhong et al., 2021).

Beyond utilitarian value, consumers can perceive hedonic value from influencers' posts. Hedonic value encapsulates the emotional and entertainment aspects of online experiences, including recreational, emotional, and experiential benefits (Overby and Lee, 2006; Wongkitrungrueng and Assarut, 2020). In this study, the **hedonic value** of an influencer's social media content refers to the fun and playfulness that a media user experiences from reading or watching influencers' posts (Hughes et al., 2019). Such experiences can significantly influence consumer behaviour on social media platforms (Mathwick et al., 2001; Tasci and Ko, 2015; Zhong et al., 2021). The enjoyment derived from using social media often leads to continued consumption and increases in impulse buying tendencies (Hou et al., 2019). Furthermore, playfulness in consumers can elevate their likelihood of making advanced purchases as well as enhance their attitudes and intentions to buy (Bun et al., 2017; Yang et al., 2018).

This thesis argues that PSI will enhance the perceived value of influencer posts. On one hand, viewers are likely to experience positive emotions when engaged in what feels like a reciprocal conversation (Oliver et al., 2019). Previous research has shown that users' PSI can heighten their gratification and overall satisfaction (Chiu and Huang, 2015; Choi et al., 2019). For instance, if television viewers have PSI with TV shopping hosts, this experience is found to boost their satisfaction levels (Lim and Kim, 2011). Similarly, when users form PSI with mobile shopping applications, it not only increases their perceived value of the service but also their satisfaction (Tseng et al., 2022). On the other hand, the experience of engaging in two-way communication can assist individuals in assessing the utilitarian value (Hsieh and Lee, 2021).

Existing research has demonstrated a positive relationship between PSI and utilitarian value (Goldberg and Allen, 2008). For instance, users' PSI with a website can positively affect their perceived ease of use and the site's usefulness (Goldberg and Allen, 2008). Similarly, when users form PSI with smart speakers, it enhances their perception of the devices' usefulness and ease of use (Hsieh and Lee, 2021). This positive correlation between PSI and perceived value is also evident in qualitative findings. For example, one interviewee expressed, "When I watch videos, sometimes I feel like they are directly talking to me. They suggest whether or not to buy certain products and share their experiences with those products, which helps me decide whether to buy them or not. You know, it's like having a

*personal recommendation from those influencers who have already tried the product, which is very helpful.*" (Interviewee 9). Despite the lack of empirical research examining this relationship within the context of influencer marketing, this thesis anticipates the continuation of the positive relationship in this context. Formally,

H1: PSI positively influences perceived value on social media.

In summary, during the experimentation stage, consumers seek out additional information about the influencers they find intriguing. The AIDA model, supported by qualitative findings and contemporary literature, identifies PSI as a critical element in enriching the consumer experience and thereby augmenting the perceived value of influencer posts (Hsieh and Lee, 2021). As consumers attribute high value to these posts, they go into the intensification stage, wherein their connections with the influencers strengthen. The subsequent section will delve into theorising about the intensification stage.

## 6.3.3 Stage 3: Intensification

The intensification stage represents the third stage in the development of PSR. During this stage, individuals aspire to develop a relationship, strive to sustain it, and ultimately form a strong attachment to the other party. The AIDA model posits that this aspiration stems from individuals' inherent needs and desires (Pashootanizadeh and Khalilian, 2018). For instance, individuals may come to view the other party as friends to fulfil their need for social connection (Tukachinsky and Stever, 2018; Esteban-Santos et al., 2018). Throughout this stage, the level of trust towards the other party intensifies, accompanied by an increase in the frequency of interactions.

## PSR

This thesis proposes that people establish a strong PSR with influencers at the end of the intensification stage. In this thesis, **PSR** refers to an individual's dynamic psychological attachment to influencers, developed unilaterally through media and perceived as personally meaningful. Existing research has found various antecedents and outcomes of PSR (Appendics 6&7). For antecedents of PSR, influencers' intimate self-disclosure (Leite and Baptista, 2022), influencer attractiveness (Farivar et al., 2022), and influencer expertise (Aw and Chuah, 2021) are all positively related to PSR. For outcomes of PSR, existing research

has found that if people have PSR with an influencer, it can increase consumer purchase intention (Wahab et al., 2022; Masuda et al., 2022), brand trust (Reinikainen et al., 2020) and brand attitude (Agnihotri and Bhattacharya, 2020).

This thesis proposes that perceived value positively influences PSR. By forming PSR with influencers, individuals can continue deriving both hedonic and utilitarian benefits from their content, thereby satisfying a range of needs, such as entertainment (Lee and Cho, 2020; Zafar et al., 2020). From a hedonic perspective, such value fulfils individuals' desires for relaxation and entertainment, facilitating the development of connections. For instance, one interviewee remarked, "*I felt that he was very interesting and also the videos and live streaming, which I can get more fun compared with other video game influencers, so I follow him*" (Interviewee 21). Additionally, enjoyment from using virtual assistants has been linked to stronger PSR with these systems (Noor et al., 2022). Moreover, in the social media environment, the hedonic enjoyment derived from posts has been positively associated with consumers' PSR with brands (Zhong et al., 2021).

From a utilitarian perspective, value is positively associated with PSR. For instance, the perceived usability of a blog has been shown to enhance PSR (Quelhas-Brito et al., 2020). Similarly, Zhong et al. (2021) observe that the utilitarian benefits characteristic of social media posts from hospitality brands significantly influence PSR. Qualitative evidence further supports a positive correlation between utilitarian value and PSR. An example from the findings illustrates this: "*I keep following them…I would say, because those influencers create useful videos, I can apply them to my daily life. For example, vegan fitness influencers will post some recipes for their diet. Also, they will make a short video of how they cook it, so I can just use it in my life*" (Interviewee 14). Despite the empirical research gap in testing this relationship within the influencer marketing context, this thesis anticipates that the positive association between utilitarian value and PSR will persist in this domain. Formally,

H2: Perceived value of influencer's social media content positively influences PSR on social media

# Trust

Trust is a critical concept within relationship marketing, as articulated by Morgan and Hunt (1994), who define trust as "confidence in an exchange partner's reliability and integrity". It

serves as a metric for gauging the quality of relationships between two entities (Kim and Kim, 2021). Trust is known to precipitate a range of positive outcomes; notably, a direct correlation exists between brand trust and brand loyalty (Delgado-Ballester and Luis Munuera-Alemán, 2005; Sung and Kim, 2010). In the context of this study, **trust** is conceptualised as an individual's belief that an influencer will provide consistent and competent quality (Jun and Yi, 2020; Chaudhuri and Holbrook, 2001).

In online environments, particularly within influencer marketing, establishing trust between consumers and influencers presents a notable challenge. Often, influencers collaborate with brands to endorse and recommend products, directing their content more towards product and brand promotion. This focus aims to provide beneficial information about the products and brands rather than reflecting the influencers' genuine personal opinions (Hudders et al., 2021). Digital celebrities seamlessly integrate advertisements and promotions into their daily life updates, obscuring the distinction between genuine endorsements and commercial intent for consumers (Lueck, 2012). However, when viewers discern brand influence over the influencers (Martínez-López et al., 2020). Consequently, to bolster consumer trust, influencers must enhance their trustworthiness (Lou and Yuan, 2019). Research indicates that consumers are discerning in their engagement with influencers, tending to trust and follow the recommendations of those they perceive as trustworthy and favour (Sokolova and Perez, 2021; Agnihotri and Bhattacharya, 2020).

This thesis proposes that perceived value positively influences trust, although empirical research presents mixed findings on this relationship. The discrepancy arises in different contexts; for instance, while the hedonic value of shoe products is shown to enhance brand trust in a traditional retail environment, their utilitarian value does not have a significant impact on brand trust (Achmad et al., 2020). Conversely, in the digital environment, Albayrak et al. (2019) observe that the hedonic value of a website does not influence brand trust. Yet, Dastan and Gecti (2014) find that both the utilitarian and hedonic values of a smartphone positively affect brand trust.

In influencer marketing, Lou and Yuan (2019) demonstrate that the informative value of influencer posts significantly enhances users' trust in the advertised content, whereas the entertainment value does not affect trust in the same manner. Conversely, Sharma and Klein (2020) find evidence suggesting that when consumers perceive value in an online buying group, their trust in the group increases. Qualitative findings support this, indicating a

positive relationship between perceived value and trust. For instance, one respondent stated, *"I really trust him because I have that really good experience with the previous products I bought from his product. He's suggestion is really helpful and trustful."* (Interviewee 11). Echoing Sharma and Klein (2020), this thesis posits that a similar effect exists between social media users and their preferred influencers, thereby suggesting that perceived value plays a crucial role in fostering trust. Therefore,

H3: Perceived value of influencer's social media content positively influences trust in influencer on social media

This thesis posits that viewers' trust in influencers will have a positive impact on PSR. Research in the domain of influencer marketing has consistently demonstrated that individuals are more inclined to develop PSR with influencers whom they perceive as trustworthy and credible (Yuan et al., 2016; Yuan and Lou, 2020). Furthermore, Sokolova and Perez (2021) corroborate that influencers' credibility significantly enhances PSR between viewers and influencers. Despite the existing gap in empirical studies specifically exploring the relationship between trust and PSR within influencer marketing, trust is acknowledged as a critical determinant in the evolution of consumer relationships with brands. For instance, Zhang et al. (2020) finds that trust serves as an antecedent to long-term consumer-brand relationships, such as brand love. Similarly, Li et al. (2020) identify a positive correlation between consumer trust and PSR, with one participant noting, "*I keep following them, and it's because of established trust*." (Interviewee 15). Consequently, this thesis anticipates a positive relationship between trust and PSR in the influencer marketing context. Formally,

H4: Trust in influencer positively influences PSR on social media

In conclusion, during the intensification stage, consumers develop stronger connections with influencers. Guided by the AIDA model, alongside qualitative findings and current literature, it is evident that consumers desire to cultivate long-term relationships with influencers, culminating in the development of psychological attachment by the end of this stage. Trust emerges as a critical factor in this process, significantly contributing to the development of PSR. Upon establishing a PSR with influencers, consumers are motivated to take action, transitioning into the outcome stage, integration. The subsequent section will delve into theorising about the integration stage.

### 6.3.4 Stage 4: Integration

Based on the discussion in Section 6.2.2, the final stage of the PSR development process involves the individual developing loyalty and responding to the other party. The AIDA model suggests that individuals may exhibit behaviours such as engaging in actual behaviour, forming behavioural intentions, or adopting attitudes (Song et al., 2021; Wu and Lai, 2023). At this stage, consumers are more susceptible to influence from influencers. For instance, followers who establish PSR with digital celebrities, under their influence, may experience urges to make purchases impulsively (Zafar et al., 2020). Drawing upon the AIDA model, qualitative findings, and existing literature, this thesis posits that individuals are likely to show support for influencers and have positive brand attitudes and purchase intentions.

Consumers are aware that influencers collaborate with brands to endorse products and earn commissions (Coates et al., 2020; Lee et al., 2021). Nonetheless, they trust their favourite influencers to provide authentic product reviews rather than misleading advertisements (Audrezet et al., 2020). Influencer recommendations prompt consumers to make purchases, especially if the influencer is favoured by the consumer (Yang and Ha, 2021). Some consumers are drawn to influencers who sell products through live streaming, attracted by the promise of low prices and high-quality service (Ye et al., 2024). Interviews with 25 consumers revealed a tendency to react positively to their favoured influencers' recommendations or endorsements, showing intentions to purchase or actual buying behaviour. For example, one interviewee stated, "*I just bought the products they endorsed*. *For example, I bought the protein vegan food because my favourite influence just endorsed it, and it is healthy and does not harm animals.*" (Interviewee 14). Further, existing studies highlight that establishing PSR with influencers can lead consumers to enhance purchase intentions towards the recommended products (Wahab et al., 2022; Masuda et al., 2022).

# **Purchase intention**

In this thesis, **purchase intention** is defined as a consumer's intention to purchase a product or service under influencers' influence based on their evaluation (Hwang and Zhang, 2018; Lu and Chen, 2021). While a gap exists between purchase intention and actual buying behaviour, purchase intention serves as a reliable predictor of future purchasing actions (Silva et al., 2019). Researchers have explored various antecedents of purchase intention, such as WOM (Yusuf et al., 2018; Park et al., 2021) and brand awareness (Dabbous and Barakat, 2020). Influencers have been shown to boost consumers' purchase intentions significantly. For instance, Leite and Baptista (2022) discover that influencers' intimate self-disclosures positively affect the PSR and, in turn, enhance consumers' intention to purchase. Furthermore, another study confirms that influencers' homophily, physical attractiveness, and social attractiveness positively correlate with PSR, subsequently enhancing purchase intentions (Masuda et al., 2022).

This thesis argues that PSR will positively affect purchase intention. Social media users often seek information about products or brands from their preferred influencers, viewing them as key opinion leaders (Casaló et al., 2020). Influencers' detailed product descriptions can enhance followers' perceptions of the product's value and foster a more favourable attitude toward it (Lee and Watkins, 2016; Lee, 2018). Moreover, influencers' persuasive language plays a role in stimulating consumers' desire to purchase (Hwang and Zhang, 2018). Empirical research has consistently shown a positive link between PSR and purchase intention (Yuan et al., 2021; Hwang and Zhang, 2018; Farivar et al., 2021; Agnihotri and Bhattacharya, 2020). For instance, followers' PSR with influencers significantly influences their intention to purchase (Farivar et al., 2021). Recent studies, such as that by Balaban et al. (2022), suggest that PSR enhances perceptions of an influencer's credibility, which in turn affects purchase intention. Qualitative data also support this positive association, as highlighted by a respondent's comment, "When they recommend something in their videos, sometimes it makes me want to buy the products." (Interviewee 3). Therefore, the hypothesis is formally presented as:

## H5: PSR positively influences purchase intention on social media

This thesis outlines the mechanism through which PSR is formed between consumers and influencers, leading to purchase intention at this stage. It further discusses two important concepts in this process: brand attitude and the intention to support.

### Brand attitude

Brand attitude reflects the comprehensive evaluation of a brand as perceived by individuals (Mitchell and Olson, 1981). This thesis adopts this view and proposes that **brand attitude** can be defined as the summary evaluation of people's perception of a brand that an influencer endorsed or recommended on social media. As a key concept in marketing, brand attitude is

instrumental in explaining and predicting consumer behaviour (Kudeshia and Kumar, 2017; Chu and Chen, 2019). For instance, a positive brand attitude among consumers correlates with an increase in purchase intention (Tseng and Wang, 2023). The literature identifies several antecedents of brand attitude, including brand storytelling (Hong et al., 2021) and personalisation (de Groot, 2022). Within influencer marketing, influencers play a significant role in enhancing consumers' attitudes towards endorsed brands (Agnihotri and Bhattacharya, 2020). For example, audiences who watch brands recommended by YouTube vloggers tend to develop a more favourable attitude towards those brands (Lee and Watkins, 2016). Additionally, Reinikainen et al. (2021) find that the perceived 'coolness' of an influencer can positively affect consumers' brand attitudes.

This thesis proposes that a viewer's PSR with influencers positively impacts their brand attitude. Influencers frequently recommend products and endorse brands on social media, which can influence their followers and alter their behaviours (Shen et al., 2022). Research exploring the link between PSR and brand attitude has consistently found a positive relationship (Agnihotri and Bhattacharya, 2020). For instance, Lee and Cho (2020) discover that individuals PSR with smart speakers exhibit a more positive attitude towards the brands mentioned by the smart speakers.

A strong PSR with influencers enhances individuals' attitudes towards the endorsed brand (Reinikainen et al., 2021). Balaban et al. (2022) also find that PSR with influencers increases the perception of the influencer's credibility, which, in turn, improves consumers' attitudes towards the endorsed brand. Qualitative data further corroborate this relationship, as one interviewee remarked, "*I like to notice what he wears. For example, last time, he carried a Prada bag and wore an Essential jacket, which looked really cool. So, I thought that Prada and Essential are very fashionable brands. So, If I want to buy bags and jackets, I will first choose Prada and Essential.*" (Interviewee 2). Based on the analysis, this thesis proposes the following hypothesis:

H6: PSR positively influences brand attitude on social media

Additionally, this thesis posits that brand attitude will positively affect purchase intention. Brand attitude's role in marketing is important for understanding and predicting consumer behaviour (Kudeshia and Kumar, 2017; Chu and Chen, 2019). The existing body of research demonstrates a positive correlation between brand attitude and purchase intention (Medina-Molina et al., 2021). For example, enhanced attitudes towards green energy brands have
been shown to increase purchase intention (Hartmann and Apaolaza-Ibanez, 2012). Social media activities related to corporate social responsibility can elevate consumers' brand attitudes and, subsequently, their purchase intentions (Chu and Chen, 2019). If consumers hold a positive attitude towards a brand, their purchase intention is likely to increase (Tseng and Wang, 2023). This effect has been observed in the context of influencer marketing as well, where Gong and Li (2017) demonstrate that a more positive attitude towards products recommended by influencers leads to higher purchase intention. Reinikainen et al. (2021) further establish that a positive brand attitude can enhance brand trust, which in turn boosts purchase intention. Consequently, this thesis proposes that:

H7: Brand attitude positively influences purchase intention on social media

#### Intention to support

In this thesis, **intention to support** is conceptualised as the extent to which the individual is willing to help the influencers maintain and increase popularity in the long run. From the perspective of relationship investment, the intention to support is viewed as a crucial element where followers invest in their relationship with influencers. Relationship investment theory suggests that within the consumer-brand relationship, one party may engage in both tangible and intangible contributions to sustain and enhance the relationship (Luo et al., 2009). For instance, fans may support their favourite celebrities by dedicating time, money, and gifts and participating in various marketing and promotional campaigns to ensure their favourite celebrities' success (Donabedian, 2021). Similarly, fans are likely to purchase products and generate positive WOM about their favourite celebrities in online environments (Kim and Kim, 2020). In influencer marketing, followers actively engage with their preferred influencers' endorsements and related marketing initiatives (e.g., "I want him to become more popular. Because in China, there are a lot of top influencers, and I want him to be one of them. So, whenever he posts something, I will like and comment to support him. I also *bought his album to support him*" (Interviewee 2)). Thus, this thesis contends that individuals will support their favourite influencers.

This thesis asserts that a PSR positively influences individuals' intention to support influencers. Communal relationship theory posits that individuals are inclined to attend to others' needs and benefits in close relationships, such as friendships (Clark and Waddell, 1985). Followers who establish PSR with influencers often feel a sense of intimacy,

regarding them as friends in some cases (Chung and Cho, 2017; Agnihotri and Bhattacharya, 2020). Such followers perceive the influencers as attentive to their needs, deriving satisfaction and well-being from the relationship, which in turn fosters a willingness to support and help the influencer gain fame, for example, through sharing information on social media (Labrecque, 2014; Kim and Kim, 2020; Chen et al., 2021). Hwang and Zhang (2018) discover that consumers who establish PSR with influencers tend to generate positive E-WOM on social media, a form of support for the influencers. Although empirical studies directly testing the relationship between PSR and intention to support are lacking, qualitative evidence suggests a positive correlation (e.g., *"Because I think helping them is a way to support them in continuing to create good quality videos for other people, and I will benefit individually. So, it is kind of like an investment instead of spending money"* (Interviewee 25)). Therefore, this study hypothesises a positive relationship between PSR and the intention to support:

H8: PSR positively influences intention to support on social media

Moreover, this thesis proposes that the intention to support influencers positively impacts consumers' purchase intention. If followers are committed to supporting influencers to enhance their popularity, they may engage in various activities such as increasing views, likes, and comments and giving virtual gifts (Donabedian, 2021; Chatzopoulou et al., 2010; Lu et al., 2018). Importantly, purchasing promoted products is a keyway to support influencers, especially since many influencer campaigns aim to promote products and brands and boost sales. Influencers often enter into sales-based contracts with brands, requiring follower purchases to support them (Shen et al., 2022). Thus, followers intending to support their favourite influencers will likely show increased purchase intention. This relationship is evidenced in the qualitative findings (e.g., "So now, since I definitely want to help her because, like I mentioned, she's a photographer. And I know recently she decided to open a print shop, which is like printing her images. So I mean, if I was nearby in Europe, I could purchase something." (Interviewee 18)). Consequently, this study proposes:

H9: Intention to support positively influences purchase intention on social media

In summary, at the PSR outcome stage, consumers are predisposed to purchase products endorsed by influencers. Drawing from the AIDA model, qualitative insights, and existing literature, this study introduces purchase intention, brand attitude, and intention to support as key positive outcomes of PSR. The forthcoming section will explore the moderators and moderation relationships within the PSR development process. A summary of the research hypotheses is presented in Table 6.1.

# 6.4 Moderators and moderated relationships

#### 6.4.1 Perceived expertise

**Perceived expertise** can be defined as the degree to which viewers perceive the number of skills, knowledge, and competence an influencer has in specific regions (Farrington-Darby and Wilson, 2006; Xiang et al., 2016). It is considered a dynamic state, which can be continuously enhanced through learning and experience (Herling, 2000). Characteristics such as broad knowledge, effective communication skills, and problem-solving capabilities contribute to the perception of an influencer's expertise. Influencers are deemed experts when they possess significant expertise, granting them expert power (Kim et al., 2020). This perception leads audiences to listen, trust, seek advice from influencers, and possibly influence their attitudes (Lou and Yuan, 2019; Yuan and Lou, 2020). Additionally, if consumers think the influencer is an expert, it can increase consumers' PSR with the influencer, thereby increasing consumers' brand attitude and purchase intention (Xie and Feng, 2023).

This thesis posits that perceived influencer expertise strengthens the relationship between PSI and perceived value. Initially, it is hypothesised that consumers' perceived value of influencer's post escalates when they have PSI (Goldberg and Allen, 2008). For instance, Hsieh and Lee (2021) demonstrate that consumers' PSI with smart speakers enhances both perceived usefulness and ease of use of these devices. Similarly, consumers PSI with mobile shopping applications is shown to elevate their perceived value. Consequently, this thesis argues that this positive correlation between PSI and perceived value persists within the context of influencer marketing.

This thesis contends that the impact of PSI on perceived value is enhanced by higher levels of perceived influencer expertise as opposed to lower levels. Specifically, when consumers regard an influencer's expertise as superior, the influence of PSI on perceived value is notably strengthened. This argument is anchored in the premise that an influencer's expertise significantly shapes consumer reception of information. As discussed above, influencers deemed experts are likely to offer more credible, effective, and beneficial advice, thereby elevating consumers' valuation of the information presented (Vieira et al., 2022).

Research hypotheses summary
H1: PSI will positively influence perceived value of influencer's social media content on social media
H2: Perceived value of influencer's social media content will positively influence PSR on social media
H3: Perceived value of influencer's social media content will positively influence trust in the influencer
H4: Trust in the influencer will positively influence PSR on social media
H6: PSR will positively influence brand attitude on social media
H7: Brand attitude will positively influence purchase intention on social media
H8: PSR will positively influence intention to support on social media
H9: Intention to support will positively influence purchase intention on social media
H10: Perceived expertise will moderate (strengthen) the relationship between PSI and perceived value
H11: Perceived authenticity will moderate (strengthen) the relationship between perceived value and trust
H12: Interaction frequency will moderate (strengthen) the relationship between perceived value and PSR

Table 6.1 Research hypotheses summary

From a utilitarian viewpoint, influencers with greater expertise are capable of delivering high-quality, substantial, and valuable content, enhancing consumer perceptions of value (Vieira et al., 2022; Dootson et al., 2016). From a hedonic perspective, influencers of high expertise excel in creating more engaging and entertaining content, thereby increasing consumer enjoyment and satisfaction (Lin et al., 2018; Lee and Theokary, 2021). When influencers are perceived as having lower expertise, their ability to produce meaningful content that aids decision-making or entertains is diminished. This can lead consumers to perceive little or no value in the influencer's posts. Thus, perceived expertise serves as a moderating factor that positively strengthens the relationship between PSI and perceived value.

H10: Perceived expertise moderates (strengthens) the relationship between PSI and perceived value on social media

### 6.4.2 Perceived authenticity

Influencer authenticity is critical, serving as the foundation upon which consumers measure the genuineness and relatability of an influencer (Campbell and Farrell, 2020). In this thesis, perceived authenticity can be defined as the degree to which an individual perceives an influencer as real and down-to-earth person (Shoenberger and Kim, 2022; Jun and Yi, 2020). The influencer's authenticity plays an essential role in consumers' perception of the influencer and the effectiveness of influencer marketing (Audrezet et al., 2020). Consumers discern the intent behind influencer posts, valuing content that seems driven by the influencer's genuine interests and desires rather than by brand agendas (Kim and Kim, 2021). This perception of authenticity fosters a more positive attitude towards the brands and products featured, subsequently enhancing consumers' willingness to pay (Kapitan et al., 2021; Moulard et al., 2015; Audrezet et al., 2020). The authenticity of an influencer is crucial for fostering long-term relationships with consumers, as it promises content that feels more authentic and trustworthy (Zhang, 2022; Shoenberger and Kim, 2022). Influencers who embody authenticity are viewed as more reliable, as they are perceived to act on intrinsic motivations and personal satisfaction, leading consumers to trust them for sharing genuine experiences over those motivated by external incentives (Shoenberger and Kim, 2022).

This thesis posits that the perceived authenticity of influencers strengthens the connection between perceived value and trust. First, perceived value positively influences trust (Portal et al., 2018). The authenticity of influencers, manifested through intrinsic motivations and self-gratification, convinces social media users of the influencers' genuine brand experiences over mere interest-driven endorsements, thereby enhancing trust levels in the influencers (Jun and Yi, 2020). Furthermore, Kim and Kim (2021) observe that authenticity attributes, such as reliability and honesty, in influencers heighten viewer' trust.

The thesis further contends that this effect is enhanced with higher versus lower levels of perceived influencer authenticity. In essence, when consumers perceive greater authenticity in an influencer, the influence of perceived value on trust is significantly enhanced. The rationale is that perceived value's impact on trust intensifies when influencers are regarded as authentic, where the shared content is motivated by genuine experiences and self-satisfaction rather than solely by commercial interests (Jun and Yi, 2020). Such influencers, exemplifying dependability and honesty, foster a heightened level of consumer trust (Kim and Kim, 2021). Conversely, this effect will be weaker when influencers are not viewed as authentic as such trust in those influencers are difficult to develop. Thus, this paper argues

that perceived authenticity acts as a moderating element, strengthening the relationship between perceived value and trust.

H11: Perceived authenticity moderates (strengthens) the relationship between perceived value and trust on social media

### 6.4.3 Interaction frequency

In the media environment, the frequency of interaction is positively linked to media engagement. In this thesis, **interaction frequency** refers to the frequency individuals consume influencer posts on social media. Research demonstrates that increased media exposure enhances the likelihood of forming relationships with its creators (Gabriel et al., 2018; Pressrove and Pardun, 2016). In traditional media contexts, frequent consuming social media content leads to a range of positive outcomes, including heightened engagement, PSR, emotional empathy, character identification, perceived realism, and content enjoyment (Kühne and Opree, 2020). Similarly, in the social media environment, a higher frequency of use correlates with greater dependency and even addiction among users (Allahverdi, 2022).

This thesis argues that the frequency of interactions strengthens the relationship between perceived value and PSR on social media platforms. First, it is posited that perceived value has a positive impact on PSR (Zhong et al., 2021). For example, Dhiman et al. (2023) discover that consumers' perception of the value in AI tools increases their PSR with these tools. Similarly, both informative and entertainment values have been found to positively influence consumers' PSR with brand employees on social media (Venciute et al., 2023). Although empirical research validating this relationship within influencer marketing is lacking, the above analysis leads to the prediction that such a relationship persists in the influencer marketing context.

The thesis further argues that this effect is further enhanced with higher interaction frequencies compared to lower ones. When consumers interact more frequently with influencers, the impact of perceived value on PSR is significantly enhanced. This assertion is underpinned by the Uses and Gratifications Theory (U&G), which suggests that one of the motivations for media consumption is to fulfil relationship-building needs, fostering long-term connections between consumers and influencers (Flecha-Ortiz et al., 2023; Wang, 2023). High-frequency interactions imply that consumers are exposed to more content from

influencers, indicating a time investment in the relationship, which acts as an indicator of relationship maintenance (Donath, 2007; Ellison et al., 2014). Previous studies have shown that increased interaction with a hotel brand's social media account leads to the development of closer, stronger relationships (Su et al., 2015). Further research supports the idea that frequent interactions with favoured celebrities on social media enhance self-congruency and relationship strength (Kim and Kim, 2020). Conversely, lower interaction frequencies suggest a lack of interest in and commitment to the relationship, hindering its development. Consequently, this paper suggests that interaction frequency acts as a moderating factor, strengthening the relationship between perceived value and PSR.

H12: Interaction frequency moderates (strengthens) the relationship between perceived value and PSR on social media

#### 6.5 Chapter summary

This chapter presented the research model focused on the PSR development process and outcomes, informed by the AIDA model and Tukachinsky and Stever's (2018) PSR framework, and supported by existing literature and qualitative insights. The framework is structured around four stages: exposure initiation, experimentation, intensification, and integration. Based on this, twelve research hypotheses were formulated and presented in Figure 6.1. These hypotheses were empirically tested using a quantitative approach. The methodology and results of this quantitative investigation are detailed in Chapters 7 and 9, respectively. The forthcoming chapter will focus on the quantitative research design.

# Chapter 7: Quantitative methodologies for PSR model validation

#### 7.1 Chapter introduction

The previous chapter discusses the conceptual framework and research hypotheses, and this chapter is devoted to designing the study to test the conceptual framework and the twelve research hypotheses posited in Chapter 6. Detailed in Chapter 3 and illustrated in Table 3.1, this chapter unveils research phase 4: quantitative study 1 (PSR measure development), phase 5: quantitative study 2 (PSR measure development), and a comprehensive discussion on phase 6: an online survey for quantitative study 3 (testing the conceptual framework).

This chapter is structured as follows: initially, it will explore the conceptual framework's components set for empirical testing in this study, addressing the model's complexity and the peculiarities of PSR development stages. It will then briefly present the operationalisation of key constructs, followed by an introduction to the questionnaire design and study execution. Subsequently, the sampling strategy, data analysis methods, and pilot testing will be presented. The chapter will conclude by summarising potential biases and the measures implemented to mitigate them.

### 7.2 The tested model

Chapter 6 presented a comprehensive four-stage conceptual framework elucidating the development of PSR with influencers by consumers, including exposure initiation, experimentation, intensification, and integration. Testing this model empirically necessitates a pragmatic approach. The initial stage—exposure initiation—concentrates on the factors that enhance consumer social media involvement, capturing the preliminary interaction with influencers without presupposing a consumer's interest in deepening this involvement. The subsequent stages transition from experimentation to integration, evaluating the dynamics of consumer-influencer relationships, with each phase signifying an intensification of the connection.

Assessing the full model within a single study presents challenges, particularly due to the distinct nature of the initial stage compared to the latter three. The first stage can be examined through an experimental research design to explore how varying conditions affect consumer

social media involvement (Mitchell and James, 2001), whereas a survey design is appropriate for understanding the PSR development process from the second to the fourth stage. This study aims to address **Research Objective 3: to explore the PSR development process and test its outcomes**, thus focusing on the latter three stages.

Incorporating the first stage with the other stages for simultaneous testing is feasible, albeit with limitations. One significant concern is the participants' potential reluctance to recall their initial interactions with favoured influencers accurately. As Schwarz (1999) highlights researchers can employ techniques to help participants search their memory, but participants do not follow this process, which could provide inaccurate data and decrease research validity. Considering these factors, this study tests the conceptual framework from the second to the fourth stage. The tested model is presented in Figure 7.1, and as expounded in Chapter 6.3 and demonstrated in Table 6.1, twelve research hypotheses are posited for testing in this study.

### 7.3 Conceptualisation and operationalisation

As already demonstrated above, ten constructs are involved in the tested framework, and how to measure them needs to be determined at this stage. These constructs comprise PSI, Perceived Value, Trust, Brand Attitude, Intention to Support, Purchase Intention, Perceived Expertise, Perceived Authenticity, PSR, and Interaction Frequency, as illustrated in Figure 7.1. The process to identify suitable measurements adheres to the scale selection protocol suggested by Haws et al. (2022).

This protocol includes three essential steps. The initial step involves defining each construct and delineating its main domain characteristics, with comprehensive discussions on construct definitions presented in Chapter 6.3 and presented in Table 7.1. Following this, the operationalisation process identifies the most fitting scales through a literature review, ensuring alignment with the study's conceptualisations and emphasising scale validity and reliability (Peter, 1981; Haws et al., 2022). The final step entails evaluating the scale items to ascertain if adjustments or modifications are necessary. Figure 7.1 The empirically tested model



This thesis argues that there is a need to develop a new PSR measurement instrument since there is no appropriate scale to measure this construct. In this study, PSR is conceptualised as *"individual dynamic psychological connection"*, but the current PSR scale captures *"people involvement"*, which was developed by Rubin et al. (1985). *"People involvement"* encompasses various psychological processes, including transportation, PSI, identification, and worship, making it a broad concept beyond the scope of PSR and not in line with the main domain characteristics of PSR in this study (Brown, 2015).

Construct	Definition
Parasocial interaction	Individuals' perception of direct, reciprocal and one-to-many communication with influencers through social media (Hartmann and Goldhoorn, 2011).
Perceived value of post	As the overall evaluation of relevant factors when engaging with influencers on social media (Zeithaml, 1998; Lou and Yuan, 2019).
Perceived influencer expertise	Consumers' perception of the influencer's knowledge in a particular field (Kim et al., 2020).
Trust	An individual's belief that an influencer will provide consistent and competent quality (Jun and Yi, 2020).
Perceived influencer authenticity	The degree to which an individual perceives an influencer as as a real and down-to-earth person (Shoenberger and Kim, 2022; Jun and Yi, 2020).
Brand attitude	The summary evaluation of people's perception of a brand (Mitchell and Olson, 1981).
Purchase intention	Consumers' intention to purchase a product or service under influencers' influence based on their evaluation (Hwang and Zhang, 2018).
Intention to support	The extent to which the individual is willing to help the influencers maintain and increase popularity.
Interaction frequency	The frequency with which individuals consume influencer posts on social media.

#### Table 7.1 Constructs and their definitions

Furthermore, the extant PSR scale includes items ill-suited for the online context, owing to their origins in traditional media (Wahab et al., 2022). An example being the item "*I like hearing the voice of my favourite newscaster in my home*" from Rubin et al. (1985), which does not transition well to the online setting, particularly within the influencer marketing context. The online environment introduces novel features absent in traditional settings, like enhanced convenience and media accessibility, necessitating a new PSR scale that accurately captures PSR in digital contexts (Tsiotsou, 2015; Kim, 2022).

The PSR measurement development is guided by Churchill (1979) and Gerbing and Anderson (1988) since the guidelines are widely used in developing a new measurement scale and get published in the marketing field, such as emotional attachment scale development (Jiménez and Voss, 2014), online engagement scale development (Paruthi and Kaur, 2017) and online brand community engagement (Baldus et al., 2015). The PSR scale development includes two quantitative studies: **research phase 4, quantitative study 1** to check scale reliability and validity, and **research phase 5: quantitative study 2** to refine the scale. Detailed information about the PSR measurement development process and the operationalisation of other constructs are presented in Chapter 8.

### 7.4 Questionnaire design

The design of the questionnaire is grounded in a set of principles aimed at crafting an effective and reliable instrument for gathering data (Peterson, 2000). Initially, it is crucial to determine the optimal number of questions, which involves using the fewest questions possible to frame effective inquiries. This strategy helps to shorten the time required for participants to complete the questionnaire, thereby enhancing their focus and, consequently, improving the quality and reliability of the data collected (Peterson, 2000). Furthermore, the phrasing of the questions must be clear and precise. That is, the questions devised by the researcher should align with the participants' understanding. Given that this study includes participants from diverse cultures and regions, it steers clear of using reversed or negated items to eliminate potential confusion (Herche and Engelland, 1996).

Moreover, to bolster research objectivity, this study opts for a self-administered questionnaire (an online survey) over questionnaires administered by others (e.g., via telephone) to reduce researcher involvement (Peterson, 2000). Additionally, it is the researcher's responsibility to increase and verify that participants remain attentive when responding to the questions (Kostyk et al., 2019; Paas and Morren, 2018). Given the variability in participants' ages, genders, backgrounds, and knowledge levels, which could influence the responses, employing straightforward and understandable language is essential to facilitate smooth participation and completion of the questionnaire by all participants (Lietz, 2010).

#### 7.4.1 Questionnaire structure

Building on the aforementioned principles, the questionnaire is segmented into four parts and detailed in Appendix 13. The initial section introduces the participant information sheet, which serves to acquaint participants with the study's purpose and ethical considerations. To circumvent ethical issues, the participant information sheet details participants' rights and the ethical guidelines set forth by the College Research Ethics Committee. It emphasises the voluntary basis of participation and explains how the data will be gathered and utilised to safeguard participant privacy. Participants may commence with the questionnaire only after consenting to participate in the study.

The questionnaire's second part includes filtering questions alongside questions about consumers' social media behaviours. To ascertain participant eligibility for this study, filter questions are employed to identify qualified participants (Bryman and Bell, 2011). Additionally, questions regarding participants' social media behaviours aim to glean insights into their usage patterns and preferences. To guarantee that all participants possess an equivalent understanding necessary for answering the questions, the study includes several specifically designed questions (Peterson, 2000). This research targets participants who have been following influencers on social media, requiring them to name a favourite influencer and indicate the typical content posted by these influencers. Furthermore, the study is interested in the outcomes of PSR, such as brand attitudes, necessitating participants to write a brand name that their chosen influencer endorses.

The third portion of the questionnaire includes scale items to measure the ten constructs in this study. Nine constructs are measured on a 7-point Likert scale to measure participants' attitudes (1 as strongly disagree, 2 as disagree, 3 as somewhat disagree, 4 as neither agree nor disagree, 5 as somewhat agree, 6 as agree, 7 as strongly agree), while one construct (perceived value) employs a 7-point semantic differential scale for the same purpose. The rationale behind the scale items is presented in Chapter 7.3 and Chapter 8.2. The Likert and semantic differential scales, widely applied for assessing consumer attitudes, are featured here. The Likert scale presents statements, enabling participants to express their level of agreement or disagreement through predetermined response options (Mueller, 1986). Recognised for its simplicity and effectiveness, the Likert scale aptly captures participants' attitudes (Mueller, 1986).

There are various scoring scale formats, such as 5-point, 7-point and 10-point (Dawes, 2008). Some researchers also assert that 5-9 points are acceptable (Peterson, 2000). This research does not consider a finer scale (e.g., 9- or 10-point Likert scale) since the finer scale could cause participants to spend more time thinking and may cause participants to struggle with the options (Dawes, 2008). When the questionnaire includes too many questions, finer scales may induce impatience among participants, thereby adversely affecting data quality. Between the 5-point and 7-point Likert scales, the 5-point scale tends to yield higher mean scores and less precise data in comparison with the 7-point scale, especially in web-based surveys (Dawes, 2008; Finstad, 2010). Consequently, the 7-point Likert scale is identified as more suitable for this study.

The semantic differential scale is designed to capture the extent to which participants associate an objective with pairs of contrasting adjectives (Mueller, 1986). Given its high correlation with the Likert scale and brevity, the semantic differential scale is both efficient to administer and reliable (Mueller, 1986). A potential concern is that certain pairs of adjectives may not be applicable to some attitudinal objects. To address this, the study ensures the semantic differential scale's application in measuring perceived value is relevant and meaningful. In line with the Likert scale, the semantic differential scale also utilises a 7-point format.

The questionnaire's concluding section collects demographic information from participants. This research gathers data on age, gender, and other pertinent background variables to facilitate a thorough analysis. For example, to test how different demographic groups (e.g., gender) may vary in their attitudes (e.g., PSR). As this study recruits participants via Prolific—a professional online panel—there is no need to collect demographic information through the questionnaire since Prolific provides comprehensive participant details. Instead, participants are asked to manually enter their Prolific ID at the end of questionnaire, ensuring efficient data analysis.

Both open-ended and close-ended questions are utilised in the questionnaire. Open-ended questions, employed in the second part, aim to garner extensive information. Despite their relatively high cost, they afford participants a liberating environment for responding, thereby furnishing researchers with rich data (Mueller, 1986; Brennan, 1997). An example is, *"Please write below ONE name of your favourite social media influencer who promotes brands on social media platforms"*. The majority of questions in this study are close-ended,

noted for their simplicity in creation and for requiring less time from participants to answer (Mueller, 1986).

Moreover, assessing respondents' attentiveness and response accuracy is vital; thus, the study incorporates one Instructional Manipulation Check (IMC) question (Kostyk et al., 2021; Paas and Morren, 2018). An IMC, a type of seriousness check question, is proficient in identifying inattentive responses, including speeding and straight-lining (Meade and Craig, 2012; Paas and Morren, 2018). Given the survey's brevity, a single IMC is deemed sufficient. The IMC's format aligns with Buchanan and Scofield (2018), which states "*This is a quality check. Please choose 'strongly disagree' if you are reading this*". This question is positioned in the survey's midst. Furthermore, at the survey's onset, participants are courteously informed, "*Please note that there are checks in this survey to assess whether you are reading the survey questions properly*" (Paas and Morren, 2018).

### 7.4.2 Pilot test

After the survey's creation and development, a pilot test was conducted, which is critical for identifying potential issues and enhancing the main research's feasibility and workability (Teijlingen and Hundley, 2001). The pilot test also aids in determining the time required to complete the survey. Three marketing PhD students and six marketing master's students were selected for the pilot test. They were invited to participate and asked to review the survey critically, assessing the clarity of questions, the survey layout, the presence of typos, and offering suggestions for improvement (Peterson, 2000). Feedback indicated that while the questions were clear, the layout, particularly on mobile devices, required refinement, which the researcher addressed. The data from these participants were excluded from the main study's data analysis.

#### 7.5 Sampling approach

The study's sampling frame targets individuals who have established long-term relationships with influencers on social media. Given the research's focus on empirically testing how individuals develop PSR with social media influencers, identifying participants who have followed influencers for an extended period is crucial, excluding those who do not meet the criteria. Following Kostyk et al. (2021), filtering out undesirable respondents is vital for enhancing data quality. Participants must meet specific criteria: 1) be 18 years or older, 2)

be active social media users, and 3) have followed social media influencers for a significant duration (three months or more). The three-month threshold is based on qualitative data suggesting that this period is necessary to form a stable relationship with influencers. To precisely target the desired respondents, the study initially conducted a pre-screen study to identify individuals with established relationships with influencers on social media, subsequently targeting these qualified respondents for the main study.

This approach offers three main benefits. Firstly, it effectively identifies qualified participants. Since only a subset of the population has established relationships with influencers on social media, a pre-screen study can precisely filter participants who have such relationships, thereby enhancing the sample's alignment with the sampling frame. Secondly, the method can screen out professional respondents, thereby bolstering the study's validity. Participants on online panels might not genuinely meet the questionnaire requirements but may bypass screening questions for economic gain (Dennis, 2001; Brüggen et al., 2011). Pre-screening can substantially lower the likelihood of including professional respondents. Thirdly, employing split questionnaires can decrease the total number of questions, thereby enhancing the questionnaire's quality. A reduced number of questions leads to a more focused questionnaire and improved completion rates (Vriens et al., 2001).

The study utilises an online platform to recruit qualified respondents. Platforms like Prolific facilitate swift access to target participants for rapid data collection (Dennis, 2001). Nonetheless, the use of online panels for data collection may raise concerns, particularly regarding professional respondents who might reduce data quality through repeated survey participation or incentive-driven motivations. However, it is argued that a larger number of participants are intrinsically motivated by an interest in the research topic and a desire to contribute (Dennis, 2011; Brüggen et al., 2011), which, along with increased sample sizes, may mitigate this issue (Brüggen et al., 2011).

Prolific is employed in this study as a professional crowdworking platform for respondent recruitment for three reasons. Compared to other intermediary service online panels like Qualtrics, Prolific offers researchers complete control over sampling and study administration, positioning it as a self-service platform (Peer et al., 2021). It is known for providing high-quality data; research indicates that Prolific yields superior data quality in terms of attentiveness and honesty compared to other well-known online panels, such as Amazon Mechanical Turk, Qualtrics, and CloudResearch (Peer et al., 2021; Douglas et al., 2023).

Furthermore, Prolific offers a pre-screening feature that enhances the transparency of the participant population, significantly reducing the chances of including unqualified respondents who may dishonestly complete the questionnaire for financial gain or those who attempt to participate repeatedly (Palan and Schitter, 2018). Researchers have the option to initiate a pre-screening project, creating a brief survey with no more than five questions to identify individuals qualified for the study. Upon completion of this short survey, researchers can pinpoint eligible participants and invite them to the main study through a "Custom Allowlist" function. Additionally, Prolific presents a fair and transparent mechanism for questionnaire completed the questionnaire and safeguarding participants' rights, including legal payment.

In this study, the sample size is determined from a precision standpoint, employing the A Priori Procedure to calculate a credible sample size (see Figure 7.2). The precision perspective focuses on precision, which is the degree of accuracy between sample statistics and the true population (Trafimow, 2018). It concerns the confidence in the probability that researchers will reach the precision criteria (Trafimow, 2018). Following this logic, the procedure involves setting precision criteria and confidence levels, with an equation for sample size calculation provided. For this study, the precision level is designated as 0.1 (f=0.1), with a 95% confidence level (Z=1.96), necessitating a minimum of 385 respondents to achieve the desired precision and confidence.

### **Figure 7.2 The A Priori Procedure Equation**

$$n = \left(\frac{Z_C}{f}\right)^2,$$

where

•  $Z_C$  is the z-score that corresponds to the desired probability of being close, and

• *n* is the minimum sample size necessary to meet specifications for closeness and confidence. *Source: Trafimow (2019)* 

This methodological approach boasts significant benefits, notably ensuring a high degree of sampling precision. Sampling imprecision is a factor in low research reproducibility, a concern in marketing and other disciplines striving for scientific rigour through reproducible

research (Trafimow et al., 2020; Evanschitzky and Armstrong, 2013). Unfortunately, existing research in consumer behaviour demonstrates poor precision, underlining the importance of prioritising high precision in initial research designs to support replication efforts (Trafimow et al., 2020).

Additionally, this approach facilitates obtaining a large sample size, addressing the risk of Type II errors—failing to reject a false null hypothesis (McQuitty, 2004). A larger sample size is preferable, as it increases the likelihood of achieving a normal distribution of data, thereby ensuring a more reliable sample size. For instance, with a constant confidence level, a larger sample size correlates with higher precision (Trafimow et al., 2021), underscoring the methodological advantages of this study's approach.

### 7.6 Execution of the quantitative study 3

To enhance the response rate and response rate, this study implements incentives for participants (Cobanoglu and Cobanoglu, 2003). Various forms of incentives exist, including lotteries, physical products, and donations; however, this research employs monetary rewards (Cobanoglu and Cobanoglu, 2003; Jobber et al., 2004; Gendall and Healey, 2010). Monetary incentives are chosen for their proven effectiveness in boosting response rates (Jobber et al., 2004) and the acknowledgement that participants invest time in completing the questionnaire, warranting compensation to ensure their rights. The incentives recommended by Prolific are determined based on the minimum wage rates in the UK. For the pre-screening study, participants were compensated £0.1 each for answering five questions in 2 minutes. For the main study, the payment was £1.2 per person for 7-minute questionnaire completion, with immediate payment following questionnaire completion.

The study opts for an online survey among various survey modes, such as telephone, mail, face-to-face, and online survey (Couper, 2011). Online survey, being self-administered, reduce researcher involvement and enhance objectivity (Peterson, 2000). They offer a cost-effective, straightforward method of data collection without geographical constraints, suiting this study's requirements (Couper, 2011). While some may argue online surveys are prone to inattentive responses, potentially compromising data quality, this study employs various techniques to monitor participant attention, such as IMC and data quality checks (Paas and Morren, 2018).

For the main study, the online survey platform chosen is Qualtrics. Qualtrics is a professional platform renowned for its comprehensive survey design capabilities (Qualtrics, 2024), extensively used in existing studies (e.g., Sahaym et al., 2021). This questionnaire leverages Qualtrics functions, like "*piped text*," which dynamically replaces "*your favourite influencer*" with the name of an influencer provided by respondents in subsequent questions. This customisation minimises confusion and enhances the effectiveness of communication. Additionally, the questionnaire utilises the rich content editor function, allowing the researchers to highlight key parts of questions (e.g., through bolding) to draw respondents' attention, further ensuring data quality and engagement.

As discussed in the above section, to identify qualified participants, this study initially conducted a pre-screen study to identify individuals meeting the set criteria. Concurrently, a pilot test of the main study was carried out to ensure the survey's readiness for release. Subsequently, the main study was conducted to gather data through Prolific. The following section provides a detailed introduction to these processes.

### 7.6.1 Pre-screen study

As already demonstrated above, the pre-screen study's objective is to identify participants who have established long-term relationships with influencers on social media. The prescreen survey was developed and conducted on Prolific, utilising the platform for participant recruitment. The pre-screen questionnaire comprises two segments. The initial segment contains a detailed participant information sheet, which informs participants about the study's purpose and data confidentiality measures. Only those who consent to participate can proceed to complete the questionnaire. The questionnaire's second segment features four screening questions designed to verify respondents' eligibility and enhance precision. These questions are structured according to the principle that each question should address a single topic, employing closed-ended multiple-choice formats (Teijlingen and Hundley, 2001). The questions are binary, requiring participants to select either "Yes" or "No." The specifics of the pre-screen questionnaire are detailed in Appendix 14.

The pre-study successfully recruited 400 participants through Prolific, receiving 400 responses. Each participant was compensated £0.1 for answering five questions, a process designed to take no more than two minutes. Qualified respondents were those who 1) are active social media users, 2) follow influencers on social media, 3) have followed some

influencers for more than three months, and 4) continue to engage with these influencers' content. Following these criteria, 343 respondents were deemed qualified and subsequently added to Prolific's "Customer Allowlist". The qualification rate among respondents was 83.5%.

### 7.6.2 Main study

The main study utilised Prolific for the recruitment of participants, specifically targeting those previously identified as qualified candidates. Given the necessity for a minimum valid sample size of 387, from a standpoint of accuracy and precision, the study enlisted 454 respondents from the qualified pool. Each participant received £1.2 for completing the questionnaire in seven minutes, yielding a total of 454 responses. The response rate is 100%.

### 7.7 Data analysis techniques

The approach to data analysis was established prior to the collection of data, aiming to reduce the researchers' degree of freedom (Yoo et al., 2015; Wicherts et al., 2016). This is critical, as excessive freedom for researchers can be unethical and contravene principles of scientific research design (Wicherts et al., 2016). Consequently, this study formulated a detailed a priori data analysis plan and adhered strictly to the data analysis procedure. The strategy for data analysis was crafted with the research objectives in mind to ensure consistency (Yoo et al., 2015).

### 7.7.1 Data cleaning

Data cleaning is an essential preliminary step to data analysis, tasked with identifying outliers, detecting invalid responses made carelessly, and enhancing the quality of the data (Saunders et al., 2023). For the identification of careless responses, two methods are recognised: prior checks and post hoc checks (Curran, 2016). Given that prior checks occur before data collection, this stage employs post hoc checks. The research utilised multiple inattention checks, acknowledging that relying on a single method would be insufficient for identifying all inattentive respondents. The use of several checks can significantly improve data quality compared to a single technique (Meade and Craig, 2012; Curran, 2016). Accordingly, the main study implemented six data-cleaning techniques to pinpoint inattentive respondents and enhance data quality. These techniques, widely adopted in

marketing research, have received endorsement for their effectiveness from the academic community (Meade and Craig, 2012; Curran, 2016).

The first technique involves verifying the uniqueness of the participant ID (Aust et al., 2013). This technique effectively identifies participants who completed the survey multiple times (Aust et al., 2012). By manually inspecting respondents' IP addresses, researchers can ensure that each participant only completed the survey once. Subsequently, the dataset is examined for missing values (Vriens et al., 2002). The IDs of all 454 participants were found to be unique, indicating that no one completed the questionnaire multiple times.

The second technique examines response times. Checks on completion time can serve as a supplementary tool for assessing data quality. This means that while researchers can use these checks to identify data of low validity manually, they should not rely solely on them. The benefit of checking completion times is their ability to reveal whether respondents' times are unusually fast or slow (Meade and Craig, 2012). Nevertheless, it is important to note that not all quick responses are of low quality; some studies have shown that faster response times can indicate greater stability and accessibility for some respondents (Zhang and Conrad, 2014). In this main study, the guideline is that response times should fall within the range of Mean-3\*S.D. to Mean+3\*S.D. (Leys et al., 2013; Berger and Kiefer, 2021). Applying this criterion, the study identified 11 responses that took an unusually long time and found no instances of respondents completing the survey too quickly.

The third technique involves a specific quality check question: "*This is a quality check. Please choose strongly disagree if you are reading this*". This approach has been utilised in previous studies, such as by Chen and Yao (2021). According to this rule, 14 respondents failed to select "strongly disagree," indicating a lack of attention to the survey questions. The failure rate for this quality check question was 3.08%, which is lower than the failure rates reported in similar research (e.g., Buchanan and Scofield, 2018 reported a 5% failure rate). This improvement may be attributed to this study's strategy of informing participants at the outset that a quality check would be conducted, as recommended by Paas and Morren (2018).

Fourthly, screening questions are deployed to sift out participants who do not meet the study's criteria. Given that the target population consists of individuals who have sustained long-term relationships with influencers, it is crucial to exclude those who have followed influencers for no more than three months.

Furthermore, to pinpoint outlier responses, an extreme response check is implemented (Hyman and Sierra, 2012). Extreme responding, which eschews content in favour of selecting only the most extreme options, whether positive or negative, is known as a response style rather than content-based responding (Baumgartner and Steenkamp, 2001). Participants who exhibit an extreme response style disregard the questions' content, affecting the overall data quality, introducing response bias, and undermining the validity of the data. The analysis of extreme responses proves effective in identifying such outliers (Baumgartner and Steenkamp, 2001; Weijters et al., 2010). Consequently, it is imperative for researchers to detect these outliers to mitigate response bias and enhance the quality of the data. The criterion employed by this study involves removing the lowest and highest 0.5% of mean responses (Hyman and Sierra, 2012), leading to the identification of four extreme responses.

Following the implementation of six rules, 49 responses were eliminated, resulting in 405 valid responses for subsequent data analysis. This number of valid responses surpasses the minimum sample size of 387, calculated from the standpoint of accuracy/precision in determining sample size.

### 7.7.2 Descriptive analysis

After data cleaning, this study should conduct descriptive analysis. Descriptive analysis can provide rich information about the sample, such as demographic information (Pallant, 2020).

### 7.7.3 Scale reliability and validity testing

Ensuring the reliability and validity of scales is crucial, so this study employs various techniques to verify the scales' reliability and validity. The prerequisite for testing the research model involves confirming the reliability and validity of the measurement model, as this is a logical necessity (Fornell and Larcker, 1981). To evaluate the measures' reliability and validity, the study initially examines the internal consistency of the scale items by using reliability analysis through SPSS (28.0) (Knapp, 2002). A Cronbach's alpha or Correlation Coefficient value exceeding 0.7 indicates high internal consistency among the scale items for each construct (Haws et al., 2022).

Subsequently, Confirmatory Factor Analysis (CFA) is conducted using SPSS AMOS (28.0) as the initial phase of a two-step Structural Equation Modelling (SEM) process to assess

model fit and the reliability and validity of the measures (Anderson and Gerbing, 1988). The model's goodness of fit is evaluated based on criteria from Byrne (2016) and Collier (2020), including Chi-square, degrees of freedom (df), p-value, Chi-square/df ratio, Root Mean Square Error of Approximation (RMSEA), Comparative Fit Index (CFI), Tucker-Lewis Index (TLI), and Incremental Fit Index (NFI). Specifically, a Chi-square/df ratio between 1 and 2 suggests a good fit, and a ratio between 2 and 5 indicates an acceptable fit. An RMSEA value close to 0 signifies the best fit, below 0.5 denotes a good fit, with 0.9 serving as the acceptable threshold. A TLI greater than 0.9 is considered an acceptable fit, and an a NFI value exceeding 0.9 also denotes an acceptable fit. The summary of the model fit indices is presented in Table 7.2.

Assessed model fit indices								
Measure	sure Poor Acceptable		Excellent					
Chi-square/df	Above 5	Between 2 and 5 Between 1 and						
RMSEA	Higher than 0.08	Between 0.05 and 0.08	Lower than 0.05					
CFI	Lower than 0.9	Higher than 0.9						
TLI	Lower than 0.9	Higher than 0.9						
NFI	Lower than 0.9	Higher than 0.9						

Table 7.2 The summary of the model fit indices

Composite Reliability (CR), also known as Omega, serves as an effective indicator of the reliability of scale items (Dunn et al., 2013). A CR value exceeding 0.7 indicates that the scales demonstrate high consistency (Hair et al., 2014). Furthermore, assessing construct validity is crucial. As stated by Peter (1981), construct validity is foundational for the testing and development of theories. To evaluate convergent validity, CFA was performed using SPSS AMOS (28.0). Good convergent validity is suggested if 1) all standardised loadings exceed 0.5, 2) all values in the standardised residual covariance matrix are below 4, and 3) all Average Variance Extracted (AVE) values are above 0.5 (Hair et al., 2014; Anderson and Gerbing, 1988). To assess discriminant validity, the Heterotrait-Monotrait ratio (HTMT) was applied using SPSS AMOS (28.0), adopting the HTMT.85 criterion due to its high sensitivity rate (Campbell and Fiske, 1959; Henseler et al., 2014). The criteria for reliability and validity tests are summarised in Table 7.3.

The summary of the reliability and validity test criteria							
Reliability and validity	Criteria/test	Recommend values	Source				
Internal	Cronbach's alpha/Correlation Coefficient Higher than 0.		Haws et al. (2022)				
consistency	Composite Reliability (CR) / Omega	Higher than 0.7	Hair et al. (2014)				
Convergent validity	Standardised loadings	Higher than 0.5	Hair et al. (2014)				
	The values of standardised residual covariance matrix	Lower than 4	and Anderson and Gerbing				
	Average Variance Extracted (AVE)	Higher than 0.5	(1988)				
Discriminant validity	Heterotrait-monotrait ratio (HTMT)	Lower than 0.85	Campbell and Fiske (1959) and Henseler et al. (2014)				

Table 7.3 The Summary of the reliability and validity test criteria

# 7.7.4 Research hypotheses testing

An essential preliminary step involves conducting a correlation analysis to examine the relationships among constructs using SPSS (28.0). This step is crucial before proceeding with path analysis, as it helps determine the direction and strength of the relationships (Pallant, 2020). The Pearson Correlation Coefficient (r) is used to indicate the direction and strength of these relationships. A positive r value suggests a positive correlation, whereas a negative r value indicates a negative correlation (Pallant, 2020). The absolute value of r denotes the strength of the relationship, with r=0 indicating no correlation,  $0.10 \le |r| \le 0.29$  representing a weak correlation,  $0.30 \le |r| \le 0.49$  signifying a moderate correlation, and  $0.50 \le |r| \le 1$  indicating a strong correlation (Cohen, 1988).

Following this, path analysis is conducted through SEM, representing the subsequent phase in SEM to examine the model (Anderson and Gerbing, 1988). Additionally, full model mediation analysis is performed using SPSS AMOS (28.0), guided by Bollen (1989) and Zhao et al. (2010). Then, full model moderation analysis is carried out through SPSS AMOS (28.0), following the guidance of Spiller et al. (2013) and Collier (2020).

### 7.8 Possible bias and the efforts to minimise bias

Reflecting on and minimising bias is critical for maintaining objectivity, so this section will highlight potential biases within the study and efforts undertaken to mitigate them. Previous sections have detailed these efforts and summarising them provides valuable insight into ensuring research integrity. In quantitative studies, measurement bias can undermine construct validity, for instance, through the use of inappropriate scales for measuring constructs (Haws et al., 2022; Rossiter, 2002). To counter this, the researcher clearly defined each construct and selected scales that are both reliable and valid according to existing literature. Among the ten constructs in the conceptual framework, nine utilise scales from established research. Due to the lack of an appropriate scale for measuring PSR, a new PSR scale was developed and detailed in Chapter 8.

Sampling and participant recruitment decisions can also introduce bias. It is crucial to enlist participants who possess sufficient knowledge to respond accurately (Lietz, 2010). Recruiting unqualified participants may compromise data validity. To address this, the study established specific criteria for participant recruitment, including a pre-screening process to identify individuals who met these qualifications. Additionally, a small sample size can result in a Type II error, which occurs when a false null hypothesis is not rejected (McQuitty, 2004). Adopting a scientific approach to sample size determination based on precision, this study set the precision level at 0.1 and the confidence level at 95%, thereby minimising this risk (Trafimow, 2018).

Questionnaire design is another factor that can generate bias. To enhance research objectivity, a self-administered online questionnaire was utilised, reducing researcher involvement compared to other methods, such as face-to-face surveys (Peterson, 2000). Ambiguities in question phrasing and structure can adversely affect participants' ability to process information; thus, the survey was crafted following specific guidelines proposed by Peterson (2000) to ensure reliability. The study employed Qualtrics, a professional online survey platform, to create a clear and structured survey. Moreover, a pilot test involving nine marketing PhD and master's students was conducted to verify the survey's functionality and identify any issues, further efforts to ensure the study's rigour and reduce potential biases.

As highlighted by many scholars, response bias can significantly impact the validity of data (Kostyk et al., 2019). Factors such as participant inattention and the presence of professional respondents can undermine the quality of responses (Meade and Craig, 2012; Dennis, 2001).

To mitigate these issues, this study employs IMC check and other methods to detect inattentive responses (Meade and Craig, 2012; Paas and Morren, 2018). A pre-screening study is also used to effectively lower the likelihood of encountering professional respondents, and an increase in sample size is anticipated to further diminish biases (Brüggen et al., 2011). Prolific is selected as the online panel for its superior data quality compared to other panels (Peer et al., 2021; Douglas et al., 2023). This study also provides incentives to enhance the response rate and speed, ensuring respect for participants' rights (Cobanoglu and Cobanoglu, 2003; Kostky et al., 2021). Additionally, informed consent is sought at the survey's outset to confirm voluntary participation.

Moreover, the potential for bias arising from the degree of freedom in data processing, such as P-hacking, is recognised (Kerr, 1998; Wicherts et al., 2016). To counter this, the study establishes a prior data analysis protocol, outlining the data analysis approach and rules before the commencement of data collection and analysis, thereby reducing flexibility. The study also maintains transparency throughout the research process, including preserving original data outputs and conducting periodic meetings with supervisors.

# 7.9 Chapter conclusion

This chapter outlines the methodology of Quantitative Study 3, designed to test the conceptual framework. By adhering to a pre-planned approach, the study aims to effectively achieve **Research objective 3: to explore the formation process of PSR development and test its outcome**. Detailed discussions cover the portions of the conceptual framework subjected to empirical testing, the design and execution of the survey, the scientific approach to sampling, and data analysis techniques. Furthermore, the chapter recapitulates potential biases within the study and outlines efforts made to minimise these biases and uphold objectivity. The results of this study will be detailed in Chapter 9. The next Chapter presents the measurements for PSR formation model and PSR measurement instrument development.

# Chapter 8: Measurements for PSR formation model

### 8.1 Chapter introduction

The previous chapter discussed the methodology of the quantitative study, while this chapter concentrates on the operationalisation and the measurements of the ten constructs. The extant literature offers suitable measures for the nine key constructs in the model (PSI, perceived value, trust, brand attitude, intention to support, purchase intention, perceived expertise, perceived authenticity, and interaction frequency), leading to adaptations of these measures from the literature for the nine key constructs.

However, the literature lacks an accurate PSR measurement instrument to capture the main domain characteristics of PSR in this thesis. As MacKenzie (2003) points out, employing inappropriate construct measures can undermine construct validity and the internal validity of research. Consequently, this chapter is devoted to developing a PSR measurement instrument that is both highly reliable and valid, capable of precisely measuring PSR online through a triangulation approach. The scale development process, guided by Churchill (1979) and Gerbing and Anderson (1988), aims to establish a reliable and valid measure for PSR. This step is critical before data collection and contributes to fulfilling **Research Objective 3: to explore the formation process of PSR development and test its outcome**. The newly developed PSR measurement will also aid future PSR studies in the online context in selecting an appropriate scale. Moreover, the working definition of PSR, outlined in Chapter 5.4.2, will be amended based on expert feedback. Through recommendations and suggestions from experts, the working definition of PSR will be updated in this chapter to meet **Research Objective 2: to propose new definitions of PSI and PSR based on their unique prerequisites**.

The structure of this chapter is organised as follows. Firstly, the operationalisation of the nine constructs will be discussed. Subsequently, the chapter will address the rationale behind developing a new PSR measurement. Following this, the chapter will detail the PSR measurement development process, based on Churchill (1979) and Gerbing and Anderson (1988), comprising four stages: 1) specifying the main domain characteristics of PSR, 2) generating an initial pool of scale items, 3) conducting initial item purification (Research Phase 3: expert survey & Research Phase 4: quantitative study 1), and 4) refining and assessing the scale (Research Phase 5: quantitative study 2).

#### 8.2 Constructs' operationalisation

As detailed in Chapter 7, the conceptual framework tested encompasses ten constructs, with the appropriate measurements for nine of these constructs being identified according to guidelines proposed by Haws et al. (2022). These scales, recognised within the existing literature, have been adapted from pre-established scales. The comprehensive list of construct measurements, including the names of the constructs, the scale items, and their sources, is presented in Table 8.1. The rationale and the sources for these measurements are discussed below. However, the existing literature does not provide suitable scales for measurement for the focal construct, PSR, which this thesis aims to address. The justification for creating a new PSR measurement and the process for its development are explored in the subsequent section.

### 8.2.1 PSI

As previously explored in Chapters 2.4.1 and 2.6.1, scholars have proposed two primary perspectives on the conceptualisation of PSI, leading to differing approaches to the operationalisation of PSI. The first perspective on PSI definition emphasises the "relationship" aspect between one party and another, with scale items such as "*I feel the streamer is like an old friend*" (e.g., Xu et al., 2020) being employed to measure PSI. However, this conceptualisation of PSI is considered not rigorous as it conflates PSI's main domain characteristics with those of PSR. This study adopts the alternative conceptualisation of PSI, as discussed in Chapters 2 and 6.3.2.1, viewing it as an individual's perception of direct, reciprocal, and one-to-many communication with influencers through social media (Hartmann and Goldhoorn, 2011).

The scales created by Hartmann and Goldhoorn (2011), consisting of six items and assessed on a seven-point Likert scale (ranging from 1 strongly disagree to 7 strongly agree), are deemed most suitable since they can capture the "interaction" view of PSI. Three reasons justify the selection of this scale: firstly, the main domain characteristics of PSI as proposed by Hartmann and Goldhoorn (2011) match the definition of PSI in this thesis; secondly, the scale demonstrates high reliability (Cronbach's alpha = 0.87); and thirdly, this scale has been effectively applied in influencer marketing research (e.g., Chen et al., 2022). Thus, the PSI scale utilised in this thesis comprises six items, which have been adjusted to better fit the context of the current research.

Table 8.1 C	<b>Operationalisation</b>	of study constructs
-------------	---------------------------	---------------------

Construct	Dimension	Item	Operationalisation of study constructs Scale	Source		
Construct	Dimension	PSI1	When I am watching this influencer's social media content, I have the feeling that this influencer is aware of me.	Source		
		PSI2	When I am watching this influencer's social media content, I have the feeling that this influencer knows I am there.			
Parasocial		PSI3	When I am watching this influencer's social media content, I have the feeling that this influencer knows I am aware of them.	Hartmann and		
interaction		PSI4	When I am watching this influencer's social media content, I have the feeling that this influencer knows I pay attention to them.	Goldhoorn (2011)		
	PSI		When I am watching this influencer's social media content, I have the feeling that this influencer knows that I react to them.			
		PSI6	When I am watching this influencer's social media content, I have the feeling that this influencer reacts to what I say or do.			
	Perceived utilitarian value	UV1	Ineffective/Effective			
		UV2	Unhelpful/helpful	T 137		
			UV3	Not functional/ functional	Luo and Yuan (2019)	
			value	UV4	Unnecessary/necessary	(2015)
Perceived value		UV5	Impractical/practical			
of posts		HV1	Not fun/fun			
	Perceived hedonic value	HV2	Dull/exciting	Luo and Yuan		
		HV3	Not delightful/delightful	(2019)		
		HV4	Not thrilling/thrilling	()		
		HV5	Enjoyable/unenjoyable			
		CE1	This influencer is an expert in topics related to their content.			
Perceived influencer		CE2	This influencer is experienced in topics related to their content.	Ladhari et al.		
expertise		CE3	This influencer is knowledgeable in topics related to their content.	(2020)		
		CE4	This influencer is qualified in topics related to their content.			

Table 8.1	(cont'd	) <b>O</b>	perationalisation	of	study	constructs
-----------	---------	------------	-------------------	----	-------	------------

Operationalisation of study constructs				
Construct	Dimension	Item	Scale	Source
		T1	This influencer can be relied upon on their content.	
Trust		T2	I believe what this influencer says and that they would not try to take advantage of the followers.	Alboqami
Trust		Т3	This influencer is straightforward and honest even though their self-interests are involved.	(2023)
		T4	This influencer would not tell a lie even if they could gain by it.	
		PIA1	This influencer is real.	Shoenberger
Perceived influencer authenticity		PIA2	This influencer is authentic.	and Kim
authentienty		PIA3	This influencer is truthful.	(2022)
		BA1	This brand is good.	
Brand attitude		BA2	This brand is pleasant.	Torres et al. (2019)
		BA3	This brand is favourable.	(2017)
Purchase intention		PI1	How likely would you be to purchase this brand endorsed by this influencer?	Gong and Li (2017)
		ITS1	I will support this influencer to see them succeed.	
		ITS2	I will support this influencer because I want to.	
		ITS3	This influencer's success will make me happy.	Kim and
Intention to support		ITS4	I will support the influencer if they need help from fans like me.	Drumwright
		ITS5	I feel good when I support this influencer.	(2016)
		ITS6	I pay attention to this influencer's needs.	
		ITS7	I care about this influencer's success.	
Interaction frequency		IF	How often do you view the messages (e.g., text, picture, or video) posted by your favourite influencer on social media?	Su et al. (2015)

#### 8.2.2 Perceived value of posts

In this thesis, the concept of the perceived value of posts can be reflected in two dimensions: the perceived utilitarian value and the perceived hedonic value of the posts, as detailed in Chapter 6.3.2.2. The perceived utilitarian value of the posts refers to the perceived usefulness of the posts when reading or watching a post (Meske et al., 2019). Upon reviewing relevant literature, the measurement scale developed by Lou and Yuan (2019) emerges as particularly suitable for this construct. This scale comprises five items rated on a 7-point semantic differential scale and has demonstrated high reliability and adaptability in the context of influencer marketing (Cronbach's alpha = 0.884; CR = 0.915; AVE = 0.684). Overall, the perceived utilitarian value scale used in this thesis consists of five scale items.

The perceived hedonic value of posts focuses on viewers' perception of the fun and playfulness that a media user experiences from reading or watching a post (Hughes et al., 2019). In line with this definition, the measurement scale adopted in this research from Lou and Yuan (2019) has five items anchored by a 7-point semantic differential scale. This scale exhibits high reliability and validity (Cronbach's alpha = 0.89; CR = 0.918; AVE = 0.693) and is well-suited for the influencer marketing context. The scale for the perceived hedonic value used in this thesis also comprises five items.

### 8.2.3 Perceived influencer expertise

Within the scope of this thesis, perceived influencer expertise is defined as consumers' perception of the influencer's knowledge in a particular field (Kim et al., 2020). The scale developed by Ladhari et al. (2020) is identified as the most appropriate for measuring this construct, consisting of four items and employing a five-point Likert scale ranging from 1 (strongly disagree) to 5 (strongly agree). Three key reasons underpin the selection of this scale: firstly, it aligns well with the thesis's definition of perceived content expertise; secondly, it demonstrates high reliability and validity (Cronbach's alpha = 0.961; AVE = 0.860); and thirdly, its applicability and effectiveness have been proven in influencer marketing research. Consequently, the scale used in this thesis for perceived content expertise includes four items, modified to fit the specific context of this research.

In this thesis, trust is conceptualised as an individual's belief that an influencer will provide consistent and competent quality (Jun and Yi, 2020). Consistent with this definition, the scale developed by Alboqami (2023) is identified as the most suitable due to its alignment with the trust definition and successful application within the influencer marketing context. This scale comprises four items and employs a 5-point Likert scale ranging from 1 (strongly disagree) to 5 (strongly agree), demonstrating high reliability and validity (Cronbach's alpha = 0.931; CR = 0.950; AVE = 0.592). The trust scale utilised in this thesis thus includes four scale items.

#### 8.2.5 Perceived influencer authenticity

Perceived influencer authenticity refers to the degree to which an individual perceives an influencer as as a real and down-to-earth person (Shoenberger and Kim, 2022; Jun and Yi, 2020). Upon evaluating the relevant literature on influencer authenticity, the scales used by Shoenberger and Kim (2022) are deemed most fitting. This scale, consisting of three items rated on a seven-point Likert scale from 1 (strongly disagree) to 7 (strongly agree), aligns with the thesis's definition of perceived influencer authenticity and has shown high reliability (Cronbach's alpha = 0.84) and applicability in the influencer marketing context. The scale for perceived influencer authenticity in this thesis, therefore, includes three items.

### 8.2.6 Brand attitude

Brand attitude is the summary evaluation of people's perception of a brand (Mitchell and Olson, 1981). The scale employed by Torres et al. (2019) captures brand attitude and has been effectively adapted to the influencer marketing setting. This scale includes three items and is measured on a 7-point Likert-type scale ranging from 1 (strongly disagree) to 7 (strongly agree), showing high reliability and validity (Cronbach's alpha= 0.982; AVE=0.951; CR=0.983). Accordingly, the brand attitude scale utilised in this thesis is comprised of three items, adjusted to align with the current research context.

#### 8.2.7 Purchase intention

Purchase intention refers to consumers' intention to purchase a product or service under influencers' influence based on their evaluation (Hwang and Zhang, 2018). The scale developed by Gong and Li (2017), consisting of a single item measured on a 7-point scale, is identified as the most suitable for this construct. This scale is primarily employed in the context of celebrity endorsements on social media. The single-item measure is sufficient to measure purchase intention since purchase intention is 'singularly concrete' in nature and is narrowly defined (Rossiter, 2002; Malhotra et al., 2012). Multiple items may not provide more information for purchase intention but could lead to verbose information. Therefore, this scale is used in this thesis and modified to suit the current research context.

#### 8.2.8 Intention to support

Intention to support can be defined as the extent to which the individual is willing to help the influencers maintain and increase popularity in the long run. Upon examining the relevant literature, the scale used by Kim and Drumwright (2016) was selected for its alignment with the definition of intention to support and its successful application within the social media context. This scale comprises seven items and is rated on a 7-point Likert-type scale from 1 (strongly disagree) to 7 (strongly agree), demonstrating high reliability (Cronbach's alpha = 0.93). Consequently, the intention to support scale in this thesis includes seven items, modified to the current study context.

#### 8.2.9 Interaction frequency

Interaction frequency is operationalised as the frequency at which media users visit their favourite influencers' social media accounts. For this thesis, a single-item scale, modified from Su et al. (2015), is employed to measure this construct.

The operationalisation of the nine constructs has been outlined above and detailed in Table 7.1. Despite the comprehensive approach to these constructs, the literature lacks a precise PSR scale that captures the dynamic and complex nature of PSR in the online environment. Consequently, the development of a new PSR scale that accurately measures PSR in the online environment is essential. The justification for this new scale and the process of its development are to be elaborated upon in the subsequent sections.

#### 8.3 Rationale for developing a new PSR measurement

Identifying the rationale for developing a new scale for an existing construct is crucial, as it enables researchers to decide whether an adequate existing scale is available to measure the construct (Churchill, 1979). This study reviews previous literature and existing PSR measurement scales, proposing two reasons why a new PSR scale is necessary.

First, upon reviewing current PSR research and developed scales, it becomes apparent that the existing PSR scales do not capture the main domain characteristic of PSR as proposed in this thesis, discussed in Chapter 5.4.2. The most commonly used PSR scale, developed by Rubin et al. (1985), appears 11 times according to the systematic literature review. Rubin et al. (1985) introduced a widely used scale for measuring the "*psychological involvement*" of media users in the traditional media environment, termed "PSI." This term's application, in lieu of PSR, represents a misuse. This "PSI" scale comprises 20 items assessing individuals' interpersonal involvement with their media consumption. This psychological involvement includes a range of psychological processes such as transportation, PSI, identification, and worship, thereby extending well beyond the PSR's intended scope (Brown, 2015).

Another widely used scale for measuring PSR, devised by Rubin and Perse (1987), is mentioned 10 times in the systematic literature review. It originates from the scale by Rubin et al. (1985), comprising ten items. However, Rubin and Perse (1987) fail to justify theoretically the decision to exclude certain items while retaining others, nor do they detail the scale reduction process. Such oversight may undermine the construct validity and potentially affect the scale's reliability. Furthermore, Rubin and Perse (1987) also incorrectly use "PSI" in their study, defining it as "interpersonal involvement," conceptualised as a sense of friendship. In contrast, this thesis conceptualises PSR as "dynamic attachment," which does not coincide with the main domain characteristics of "interpersonal involvement" as defined by Rubin and Perse (1987). Consequently, from the perspective of conceptualisation, an appropriate PSR scale has yet to be identified.

Another justification for the necessity of a new PSR scale lies in the fact that existing scales were developed within the context of traditional media environments, such as television. Certain items from these scales are not suitable for the online environment, necessitating their removal (Wahab et al., 2022). For instance, an item from Rubin et al. (1985), "*I like hearing the voice of my favourite newscaster in my home*", which is hard to adapt to the online environment. Similarly, some items do not align with the contemporary patterns of

online media consumption (e.g., "*I look forward to watching my favourite newscaster on tonight's news*") (Kim, 2022). The practice of omitting these unsuitable items from the original PSR scales, such as Yang and Ha (2021) adapting Rubin and Perse (1987) to include nine instead of ten items, lacks rigour, potentially undermining construct validity and internal consistency (Kim, 2022). New features have emerged in the online environment compared with the traditional environment, such as convenience and media accessibility. Thus, developing a new PSR measurement instrument that can accurately measure PSR is essential, especially in the online environment (Tsiotsou, 2015; Kim, 2022).

Therefore, developing a new PSR measurement can contribute to PSR studies. The current absence of a precise measure for online PSR necessitated the development of a PSR scale that is both reliable and valid, aiming specifically to assess PSR in the online context through a triangulation approach. This initiative is crucial for meeting research objective 3, which aims to investigate the formation process of PSR development and its outcomes. The development process, informed by Churchill (1979) and Gerbing and Anderson (1988), strives for high validity and reliability, addressing existing literature calls for an online-specific PSR scale (e.g., Kim, 2022). This development process incorporated multiple validation sources, including expert surveys and two quantitative studies, to ensure that the new online PSR scale is reliable and valid, thus significantly contributing to future PSR studies in the online environment.

#### 8.4 PSR measurement development process

This thesis adheres to the measurement scale development guidelines proposed by Churchill (1979) and Gerbing and Anderson (1988). The adoption of these guidelines is justified by their widespread use in the marketing field for developing new measurement scales, as evidenced by the creation of scales such as the emotional attachment scale (Jiménez and Voss, 2014), the online engagement scale (Paruthi and Kaur, 2017), and the online brand community engagement scale (Baldus et al., 2015).

The development of the PSR scale involves four steps. The initial step involves defining the main domain characteristic of PSR and establishing the boundaries of what is to be included in and excluded from the PSR definition (Churchill, 1979). This definition is derived from a systematic review of the literature presented in Chapter 2 and qualitative insights discussed in Chapter 5. Subsequently, the working definition is refined based on expert feedback, as

detailed in Section 8.5. The second phase generates a preliminary set of scale items through literature review and analysis of qualitative data. In the third phase, the item pool is refined through expert and consumer surveys (during research phases 3 and 4, respectively), aiming to streamline the item set and evaluate the scale's reliability and validity. The final phase involves a consumer survey (research phase 5) to refine and confirm the reliability and validity of the developed PSR scale. An in-depth discussion of the PSR scale development process is provided below and summarised in Table 8.2.

# 8.4.1 Step 1: Specify the main domain characteristics of PSR

As recommended by Churchill (1979), the initial step in developing the PSR measurement scale involves delineating the main domain characteristics of PSR. This process draws on a comprehensive analysis of past discussions (Chapter 2), prerequisites (Chapter 2), and criteria for a good definition criterion (Chapter 5). Consequently, the working definition is proposed as "an individual's dynamic perception of a valuable psychological connection, unilaterally developed with the other party based on repeated exposure through media or technology".

### 8.4.2 Step 2: Generate initial scale item pool

This stage culminates in the identification of 40 potential scale items, detailed in Appendix 15. To generate these initial items, two approaches were employed: a review of the literature and qualitative data discussion in Chapter 5.2. The literature review aimed to pinpoint possible themes and scale items reflecting PSR from existing studies, which yielded 16 items (Churchill, 1979). The analysis of qualitative data provided an additional 24 items (Churchill, 1979).

The literature suggests that PSR scales should be multi-item rather than single-item scales (e.g., Ledbetter and Meisner, 2021; Sokolova and Perez, 2021), due to their greater reliability in measuring constructs (Churchill, 1979). While some researchers argue that a single item can be as valid as multi-item scales (Bergkvist and Rossiter, 2007), multi-item scales are preferred for their ability to encapsulate more information and characteristics of the construct (Baumgartner and Homburg, 1996). Given the complexity of PSR, which is impossible to capture psychometric properties of PSR through single-item measures, this thesis adopts a multi-item approach for assessing PSR.
PSR Scale Development Process								
Steps	Methods	Analysis approach	Results					
Step 1: Specify main domain characteristic	Systematic literature review (Chapter 2) Qualitative data analysis (Chapter 5)	Content analysis; Thematic analysis	The PSR working definition needs to be evaluated; Good PSR definition criteria are proposed.					
Step 2: Generate initial	Literature review (n=16)		Generate 40 scale items					
scale item pool	Research phase 2: Semi-structured interview (n= 24)	Content analysis	(n=40)					
Stop 2: Initial itom	Research phase 3: Experts survey (delete 20 and retain 20)		Delete 32 scale items and retain 8					
Step 3: Initial item purification	Research phase 4: Quantitative study 1: Consumer survey (delete 12 and retain 8)	Exploratory Factor Analysis (EFA); Confirmatory Factor Analysis (CFA); Reliability test	scale items (n=8)					
Step 4: Assessment of the scaleResearch phase 5: Quantitative study 2: Consumer survey		Exploratory Factor Analysis (EFA); Confirmatory Factor Analysis (CFA); Reliability test; Structural Equation Modeling (SEM)	Retain 8 scale items with three dimensions (n=8)					

Scholars hold different views regarding the dimensions of PSR. Broadly, these perspectives fall into two categories. The majority of scholars conceive of PSR as a unidimensional construct, as argued by Yuan and Lou (2020) and Kim (2022). Conversely, a number of studies propose that PSR is multi-dimensional (e.g., Tsiotsou, 2016; Presgrove and Pardun, 2016). For instance, Tsiotsou (2016) delineates two dimensions of PSR among online community members: parasocial identification and parasocial problem-solving. Similarly, Presgrove and Pardun (2016) identify two dimensions of PSR between consumers and organisations, namely feelings of companionship and affective connection to the organisation. The varying conceptualisations of PSR, its main domain characteristics, and the diverse research contexts within the online environment likely contribute to this plurality of views on its dimensions.

The second approach to generating the initial item pool for the PSR scale involves analysing qualitative data. While some scholars contend that undertaking qualitative research is not an essential step in scale development, qualitative investigations can significantly contribute to the generalisation of initial items (Hinkin, 1995; Carpenter, 2017). It is noteworthy that the majority of scale development studies utilise qualitative data to generate initial scale items (e.g., Shimul et al., 2019). Following Churchill's (1979) recommendation, this study employs qualitative data to generate more relevant PSR scale items. Through content and thematic analyses detailed in Chapter 5, the researcher generates items that encapsulate the keywords and themes identified. These items undergo a rigorous evaluation process to ensure their clarity and representativeness, which is a back-and-forth process. A total of 24 relevant scale items are generated through this method. In conclusion, 40 PSR scale items are identified by combining a literature review and qualitative data analysis.

# 8.4.3 Step 3: Initial scale items purification

Scale item purification is a critical step in scale development, serving to effectively reduce the number of items (Gerbing and Anderson, 1988). Excessively long or cumbersome scale items can lead to protracted surveys, potentially diminishing respondents' patience and leading to low validity (Lietz, 2010). Given that participants exhibit diverse response styles, shortening the scale items can more effectively capture their attention. The current PSR scale typically comprises no more than ten items (e.g., Yuan et al., 2021; Sokolova and Perez, 2021), underscoring the necessity of reducing the 40 PSR scale items identified at this phase. Consequently, this study employs three distinct studies to refine the item pool: an expert survey to evaluate scale items and revise the PSR definition, an online survey to refine the initial item pool, a subsequent survey to further refine and valid the new PSR scale.

## Research phase 3: expert survey

This study employs an expert survey to refine scale items and revise the definition of PSR. Utilising expert survey is a widely accepted and critical method for developing scales in marketing, enhancing face validity (Hardesty and Bearden, 2004; Rossiter, 2002). Although face validity and content validity are often used interchangeably, face validity can be defined as "*the extent to which the "objects" (people) that were measured with the instrument agreed that the instrument was valid*" (Knapp, 2002, p.102). Its importance lies in its fundamental role in ensuring construct validity (Churchill, 1979; Hardesty and Bearden, 2004).

In scale development, typically two types of experts are consulted: academic and industry professionals. Academic experts were selected for this study due to their role in enhancing face validity. These experts were chosen based on their previous experience with PSR research and publication history. A total of 35 experts in the relevant field were identified. Seventeen experts initially responded to the questionnaire, though five incomplete submissions were excluded. Further review revealed that one expert submitted responses twice. The second response amended the first response and provided more comments was retained, resulting in eleven valid responses for data analysis, some of which included additional feedback.

The expert survey questionnaire comprises three sections. The initial section presents a plain language statement detailing the survey's purpose. The second section focuses on the PSR working definition, offering the newly working definition of PSR for evaluation. Experts are asked to rate how clear, comprehensive, and capturing the essence of the construct the definition of PSR is, which is measured by a 7-point Likert Scale. This scale format is a widely recognised analytical tool in marketing (Dawes, 2008). An open-ended question invites experts to suggest improvements to the PSR definition. The final section addresses the scale items, presenting 40 initial PSR scale items related to PSR for experts to evaluate in terms of clarity and relevance to the PSR definition, again employing a 7-point Likert Scale. This section also includes two open-ended questions for additional comments on the PSR scale items.

The initial stage of executing an expert survey involves sending an email invitation to the selected experts. Given potential geographical constraints, an online survey is deemed the most suitable method. The invitation, detailed in Appendix 16, includes a link to the survey created on Qualtrics, a widely used digital survey platform among academic researchers (Qualtrics, 2024). Experts are encouraged to participate by clicking on the provided link, with the questionnaire designed to take around fifteen minutes to complete.

The prior item retention decision rule of this study is that only items rated as "reflects slightly", "reflects moderately", and "reflects extremely well" by at the latest 80% of experts will be retained (Lichtenstein et al., 1990). This 80% threshold aligns with Zaichkowsky (1985) and surpasses the 75% consensus minimum for item retention (Hardesty and Bearden, 2004). This approach ensures all expert opinions are considered in evaluating the face validity of the items (Hardesty and Bearden, 2004).

Following this prior item retention decision rule, 20 scale items were removed for not adequately reflecting PSR, and 20 scale items were retained. These retained items underwent further examination to assess their clarity. Items rated by more than 20% of experts as "extremely unclear," "moderately unclear," or "slightly unclear" were subject to potential clarification. However, upon review of expert feedback regarding the 20 retained items, it was concluded that no clarifications were necessary. Consequently, these 20 scale items proceeded to the next stage of testing for reliability and validity.

#### Amending the PSR definition

The preliminary definition of PSR, discussed in Chapter 5.4.2, was formulated drawing on existing literature (MacKenzie, 2003; Churchill, 1979; Gilliam and Voss, 2013), qualitative findings, and criteria for good definitions (Murrow and Hyman, 1994). It conceptualises *an individual's dynamic perception of a valuable psychological connection, unilaterally developed with the other party based on repeated exposure through media or technology*.

As highlighted in Chapter 5.4.2, this working definition offers several advantages over existing definitions. This study solicited feedback from experts in the PSR field to refine the working definition, ensuring its clarity, comprehensiveness, and credibility. Experts were invited to assess the face validity of the newly proposed PSR scale items and to provide feedback on the working definition (Hardesty and Bearden, 2004; Rossiter, 2002). The

detailed study design of the expert survey is detailed in Chapter 8.4.3.1. Experts were asked to evaluate the definition's clarity, comprehensiveness, and its encapsulation of the PSR construct's essence. An open-ended question was included to gather detailed comments.

Feedback from five experts on the PSR definition was both valuable and enlightening, contributing significantly to the refinement of the working definition. Their insights and suggestions are compiled in Appendix 17. Taking these into account, the working definition of PSR has been revised to an individual's dynamic psychological attachment to another party, developed unilaterally through media and perceived as personally meaningful.

#### Research phase 4: online consumer survey - quantitative study 1

The current retention of 20 items within the PSR scale exceeds the item count of other existing scales, necessitating a refinement to streamline the newly developed scale. Consequently, the **primary aim of this study is to reduce PSR scale items**. Secondly, the paucity of literature concerning the dimensions of PSR underscores the necessity to delineate the appropriate dimensions of this construct (Gerbing and Anderson, 1988; Haws et al., 2022). Examination of the current 20 PSR scale items reveals thematic congruencies among some items, suggesting inherent dimensions within the PSR scale. Thus, **the second aim is to elucidate the dimensions of PSR**. Thirdly, prevailing research advocates for the high internal consistency, reliability, and validity of scale items (Knapp, 2002). It is imperative, therefore, to ensure the PSR scale items exhibit a strong internal correlation (Haws et al., 2022). The third objective, therefore, is to evaluate and enhance the reliability and validity of the PSR scale.

### Questionnaire design

This questionnaire is structured into three distinct sections. The initial section presents the participant information sheet, designed to brief participants on the study's objectives, ethical considerations, and the measures in place to safeguard their privacy. In line with the guidelines stipulated by the College Research Ethics Committee, this sheet emphasises the voluntary nature of participation and details the procedures for data collection and usage, aiming to mitigate any ethical concerns and protect participant privacy.

The subsequent section comprises filter questions alongside queries regarding consumers' social media behaviours. The implementation of filter questions is critical to promptly ascertain the eligibility of participants for this study (Bryman and Bell, 2011). To ensure a uniform understanding necessary for providing informed responses, the questionnaire introduces several carefully crafted questions (Peterson, 2000). Specifically, the study targets participants with a history of following influencers over extended periods. Accordingly, respondents are requested to name a favoured influencer, indicate the typical content posted by these influencers, and how long they have been following them. The final section of the questionnaire is dedicated to the 20 PSR scale items, employing a 7-point Likert scale for responses. with 1 indicating "strongly disagree", 2 indicating "disagree", 3 indicating "somewhat disagree", 4 indicating "neither agree nor disagree".

Notably, this questionnaire omits demographic questions, as Prolific provides detailed demographic information of participants. This approach reduces the questionnaire's length, enhancing participant attention. Participants are, however, required to enter their Prolific IDs manually for identification purposes.

The study utilises an online survey method, which is advantageous for its cost-effectiveness, ease of administration, and ability to overcome geographical constraints (Couper, 2011). Qualtrics serves as the chosen platform, recognised for its advanced functionalities and widespread adoption in research (e.g., Sahaym et al., 2021; Qualtrics, 2024). Features such as piped text are employed to personalise the survey experience. This feature automatically updates subsequent questions to include the name of an influencer identified by the respondent, thus reducing confusion and enhancing the survey's communicative efficiency.

## Pilot test

Prior to the main study, a pilot test was conducted to identify and rectify potential issues, thereby ensuring the survey's overall feasibility (Teijlingen and Hundley, 2001). This test involved three marketing PhD students and two marketing postgraduate students, who were asked to evaluate the survey for clarity, layout, typographical errors, and provide general feedback for improvement (Peterson, 2000). Their assessments confirmed the survey's clarity and ease of comprehension, indicating no need for further modifications. Data from these pilot participants were excluded from the analysis of the main study.

#### Sampling approach

The sampling frame for this study encompasses individuals who have fostered long-term relationships with social media influencers. Given the study's emphasis on empirically examining how people develop PSR with influencers on social media, it is critical to enlist participants who have consistently followed influencers over time whilst excluding those who do not meet the criteria. As highlighted by Kostyk et al. (2021), filtering out unsuitable respondents is vital for enhancing the quality of data.

Specifically, participants must meet the following criteria: 1) be aged 18 years or older, 2) be active social media users, and 3) have followed social media influencers for an extended period (at least three months). The decision to set a three-month threshold is informed by qualitative data, which suggests that it typically takes this duration for individuals to establish a stable relationship with influencers. To accurately target the desired respondents, this research initially conducted a pre-screening study to identify individuals who have developed relationships with influencers on social media. The main study subsequently focused on these pre-qualified respondents. The justification for undertaking the pre-screening study is discussed in detail in Chapter 7.6.

This pre-study garnered 400 responses, aiming to select a minimum of 200 participants meeting the criteria. Each respondent was compensated £0.10 for completing a brief survey of five questions, which required no more than 2 minutes to complete. Only those participants who 1) use social media, 2) follow social media influencers, 3) have been following at least one influencer for more than three months, and 4) continue to engage with these influencers' updates were deemed eligible.

The pre-study achieved a response rate of 100%, receiving 400 responses. According to the predefined criteria, 91 respondents were deemed ineligible, resulting in a total of 309 qualified participants. These participants were subsequently added to the "Customer Allowlist," earmarking them for the invitation to the next phase of the research, specifically quantitative study 1 (research phase 4).

The determination of the main study's sample size is guided by the number of scale items, a subject of ongoing debate, particularly when conducting Exploratory Factor Analysis (EFA) to assess scale validity. Scholars like Beavers et al. (2013) argue that it is challenging to establish an adequate sample size since the outcomes are significantly influenced by it,

rendering the stability of EFA results difficult due to this influence. Nonetheless, identifying an appropriate sample size is crucial for reducing the degrees of freedom (Yoo et al., 2015; Wicherts et al., 2016).

Other researchers, such as Hutcheson & Sofroniou (2006), suggest that a sample size ranging between 150 to 300 responses is reasonable. However, this broad range does not offer precise guidance. This study employs the item ratio rule, referencing various studies that tailor their approach based on the actual number of scale items (Osborne and Costello, 2005). The item-to-sample size ratio applied herein is 1:10, meaning that for each scale item, the sample size should be multiplied by 10. With 20 PSR scale items present, the calculation dictates a sample size of 20\*10=200, surpassing the minimum of 150 participants suggested for initial structure exploration (Beavers et al., 2013). Consequently, this study aims to recruit 220 participants and successfully receive 220 responses. Participants were compensated £1.05 each for their time, with the survey taking no longer than 6 minutes to complete.

#### Data analysis

The data analysis for this study unfolds in three distinct phases: data cleaning, EFA and CFA. Employing various data-cleaning techniques is pivotal for identifying outliers, detecting carelessly invalid responses, and enhancing data quality (Leys et al., 2013; Berger and Kiefer, 2021).

#### Data cleaning

The data cleaning techniques are thoroughly examined in Chapter 7.7.1. This study utilises four techniques for this purpose: participant ID verification (Aust et al., 2013), completion time checks (Meade and Craig, 2012), screening question checks, and evaluation for extreme response patterns (Hyman and Sierra, 2012). Following the specified data cleaning protocols, 14 responses were removed, resulting in 205 valid responses available for further analysis.

## EFA

The study initially performs Harman's one-factor test to evaluate the presence of common method bias, adhering to the guidance of Fuller et al. (2016). No common method bias is confirmed through two indicators: 1) the Eigenvalues of three factors surpassing 1, and 2) the highest variance explained is 35.39%, which falls below the threshold of 40%. EFA is

conducted prior to CFA to both refine the scale items and investigate the dimensionality of the PSR scale. This approach is advocated by Gerbing and Anderson (1988), who suggest EFA's necessity in the absence of concrete evidence regarding the interrelations among constructs and scale items. Prior to executing EFA, several critical decisions regarding the methodology must be made, including the choice of extraction method, factor extraction approach, determination of the number of factors to retain, and the rotation method to be applied.

The initial decision involves selecting an appropriate extraction method. Components analysis and factor analysis are the two techniques commonly employed in EFA. Despite their functional similarities and potential to yield analogous outcomes, they are underpinned by distinct mathematical and theoretical foundations (Beavers et al., 2013). While components analysis is frequently utilised for item reduction, it does not facilitate a deeper exploration of variance differentiation (Osborne and Costello, 2005). In contrast, factor analysis is adept at revealing data patterns and aiding in item reduction (Osborne and Costello, 2005). Therefore, this study opts for factor analysis as the extraction method, aiming to elucidate the underlying structure of the PSR scale more effectively.

The choice of factor extraction method constitutes the second decision point in this study, following the guidelines proposed by Osborne and Costello (2005). Specifically, maximum likelihood is employed if the data are normally distributed; alternatively, principal axis factoring is utilised if the data distribution deviates from normality. Subsequent to this, the method of rotation—pertinent to the data's structure—was addressed, with this study opting for oblique rotation (Field, 2018). This choice is justified on the grounds that the PSR scale items are interrelated, rendering oblique rotation suitable (Beavers et al., 2013). Moreover, oblique rotation is acknowledged for yielding results that are both more precise and replicable (Osborne and Costello, 2005). Consequently, the criteria for factor retention are summarised in Table 8.3.

As a result, a principal axis factor analysis (FA) incorporating oblique rotation (direct oblimin) was initially applied to the 20 items. The Kaiser-Meyer-Olkin (KMO) measure confirmed the sampling adequacy for the subsequent analysis, KMO = 0.890,  $\chi 2(df) = 1842.531(190)$ , p < 0.001. The analysis revealed five factors, distinguished by eigenvalues exceeding 1, together accounting for 53.71% of the variance. However, due to the fact that the factor loading values of seven items were below 0.5, these items were excluded. Subsequent EFA with the remaining thirteen scale items were conducted. The Kaiser-Meyer-

Olkin measure once again verified sampling adequacy, KMO = 0.839,  $\chi 2(df) = 1100.802(78)$ , p < 0.001, yet three items with factor loadings below 0.5 were removed. A further EFA on the remaining ten PSR scale items indicated a good fit, KMO = 0.860,  $\chi 2(df) = 1164.412(91)$ , p < 0.001, with a total variance explained of 50.636%.

	Criteria for factor retention in EFA						
Rule	Criteria	Rules	Source				
А	Kaiser-Meyer-Olkin (KMO)	Higher than 0.5	Beavers et al. (2013); Kaiser (1974)				
В	Scree test	The number of datapoints above the "break"	Osborne and Costello (2005); Cattell (1966)				
С	Kaiser criterion	Eigenvalues should equal or higher than 1	Beavers et al. (2013); Kaiser (1960)				
D	Variance extracted	75 – 90% of the variance should be accounted for; however, 50% of the variance explained is acceptable	Beavers et al. (2013)				

Table 8.3 Criteria for factor retention in EFA

The determinations based on the established decision rules indicated that PSR comprises three factors: intimacy, relationship maintenance, and real-life relationship intention. First, all factors loading values above 0.5 (Beavers et al., 2013). Second, the scree plot identified a "break" point at four. Third, the eigenvalues of the three factors exceeded 1. Fourth, collectively, these factors explained 54.76% of the variance, surpassing the acceptable threshold of 50% (Beavers et al., 2013). Initially, a reliability analysis using SPSS confirmed the reliability and validity of these dimensions. Table 8.4 illustrates that Cronbach's alpha values for the three dimensions of PSR exceed 0.7, indicating high reliability (Haws et al., 2022).

# Table 8.4 Quantitative study 1 – CFA results

	Quantitative Study 1: CFA Results									
Construct	Dimension	Scale item		Median	Mode	S.D.	CFA Factor loading	Cronbach's alpha	Average variance extracted (AVE)	Composite reliability (CR)
		The other party makes me feel comfortable, as if I am with family.	3.72	4.00	4.00	1.52	0.80		0.53	0.77
	Intimacy	I think I understand the other party quite well.	4.22	4.00	5.00	1.37	0.60	0.76		
		I treat the other party as a friend.	3.01	3.00	2.00	1.56	0.76			
Parasocial	Relationship maintenance	I want to support the other party.	5.21	5.00	6.00	1.19	0.84		0.55	
relationship		I would like to continue engaging with the other party.	5.74	6.00	6.00	0.85	0.68	0.77		0.78
		If the other party appeared on a different media platform, I would watch it.	5.50	6.00	6.00	1.24	0.70			
	Real-life relationship intention	I would like to be friends with the other party in real life.	4.93	5.00	5.00	1.39	0.96	0.77	0.67	0.80
		I would like to meet the other party in person.	5.08	5.00	5.00	1.34	0.65	0.77	0.67	0.80

The subsequent phase involves conducting CFA, the outcomes of which are delineated in Table 8.4. This procedure aligns with the methods outlined in Chapter 7. Following the exclusion of two scale items, the PSR model demonstrates a good model fit, as indicated by  $\chi^2$  (df = 16) = 24.516, p = 0.031, Chi-Square/df = 1.442, RMSEA = 0.047, CFI = 0.986, NFI = 0.957, TLI = 0.977 (Byrne, 2016). All AVE values are above 0.5, and all CR values are above 0.7. For the first-order reflective scale reliability, Cronbach's alpha/coefficient alpha values are between 0.76 and 0.77, higher than 0.7 (Haws et al., 2022). CR values are higher than 0.7 (Hair et al., 2014). For convergent validity, the standardised residual covariance matrix values are all lower than 4, all standardised loadings are higher than 0.5, and all AVE values are higher than 0.5, indicating good convergent validity (Anderson and Gerbing, 1988).

Discriminant validity is assessed through the HTMT ratio via SPSS AMOS. The HTMT.85 criterion, known for its high sensitivity rate, serves as the benchmark (Campbell and Fiske, 1959; Henseler et al., 2014). The HTMT outcomes, presented in Table 8.5, reveal interconstruct correlation values ranging from 0.375 to 0.489, all of which are below the 0.85 threshold. This indicates the absence of a discriminant validity issue within the model (Henseler et al., 2014), thus establishing discriminant validity.

Quantitative Study 1 – HTMT Matrix							
Intimacy Maintenance Real-life							
Intimacy							
Maintenance	0.402						
Real-life	0.489	0.375					

Table 8.5 Quantitative study 1 – HTMT Matrix

Consequently, eight PSR scale items are maintained at this stage, encapsulating three dimensions. The forthcoming step is to refine and valid the PSR scale.

#### 8.4.4 Scale validation and reliability (Research phase 5: quantitative study 2)

At this stage, it is crucial to refine the PSR scale, further test the reliability and validity of the PSR scale and confirm the stability of its three dimensions. Specifically, this study focuses on internal consistency, test-retest reliability, convergent validity, discriminant validity, and the nomological network of the newly developed scale (Haws et al., 2022). The

assessment of internal consistency, convergent, and discriminant validity is in line with quantitative study 1.

Test-retest reliability refers to the consistency and reproducibility of the same measurement of the same individuals in the same condition (Knapp, 2002; Aldridge et al., 2017). In essence, test-retest aims to verify the stability of the respondents' item responses over time (Haws et al., 2022). Evaluating test-retest reliability is essential in new scale development, as researchers must ensure that the scale remains stable over time (Haws et al., 2022). Various methods exist to test test-retest reliability, and this study utilises 95% confidence intervals along with their Intraclass Correlation Coefficients (ICC) to demonstrate test-retest reliability, as this approach yields more precise results (Aldridge et al., 2017).

Furthermore, to confirm that the scale has high construct validity, a nomological network with constructs with linkages needs to be developed and tested (Peter, 1981). The nomological network is developed and illustrated in Figure 8.1, drawing on existing literature.





Quantitative study 2 also adopts the online survey approach. The questionnaire's design is consistent with the structure discussed in Chapter 7. Influencer credibility is assessed using seven scale items by Sokolova and Perez (2021). The scales for measuring brand attitude and purchase intention are explored in Section 8.2 above and depicted in Table 8.2. The sampling frame and criteria remain unchanged from the previous study. This study also implements a two-step process for participant recruitment. Following Hinkin (1995), the study adheres to the recommended ratio of items to responses of 1:10. Given the total of 19 scale items, a minimum of 190 participants is required.

As previously discussed in Chapter 7, the pre-screen study was conducted via Prolific to identify eligible participants. This pre-screen study comprised five questions, which took no

longer than two minutes to complete. A total of 400 participants were recruited, each receiving a £0.1 incentive, from which 309 qualified participants were identified. Given the requirement to conduct a test-retest analysis, the study recruited participants from both the previous quantitative study 1 and those newly identified. Quantitative study 2 managed to recruit 220 participants and obtained 220 responses, with the questionnaire completion time not exceeding six minutes. The researcher compensated each participant £1.05.

The data analysis procedure encompassed data cleaning, CFA for the PSR scale, CFA for the nomological model, path analysis for testing the nomological network, and test-retest testing. Following data cleaning rules, 14 responses were removed, resulting in 206 valid responses. Among these, 154 participants had participated in quantitative study 1, and 52 were new respondents.

The subsequent step involved performing CFA to assess the newly developed PSR scale using AMOS (28.0). The CFA results, presented in Table 8.6, indicate a good fit for the model:  $\chi^2$  (df = 16) = 31.810, p = 0.011, Chi-Square/df = 1.988, RMSEA = 0.069, CFI = 0.979, NFI = 0.959, TLI = 0.963, (Byrne, 2016). All AVE values exceeded 0.5, all CR values surpassed 0.7, and all Cronbach's alpha values were above 0.7, which indicate high reliability and validity.

Regarding first-order reflective scale reliability, for intimacy, Cronbach's alpha was 0.75; for relationship maintenance, Cronbach's alpha was 0.77; for real-life relationship intention, the coefficient alpha was 0.89. All values exceeded the 0.7 threshold, demonstrating high reliability (Haws et al., 2022). Similarly, all CR values exceeded 0.7, indicating good reliability (Hair et al., 2014). For convergent validity, all standardised residual covariance matrix values were below 4, and all standardised loadings exceeded 0.5, showcasing strong convergent validity (Anderson and Gerbing, 1988). For discriminant validity, the HTMT ratio results, displayed in Table 8.7, revealed inter-construct correlation values ranging from 0.536 to 0.58, all below the 0.85 threshold, thus confirming the absence of a discriminant validity issue (Henseler et al., 2014). Consequently, discriminant validity was established.

#### Quantitative study 2: Results of reliability, confirmatory factor analysis - PSR three dimensions Average CFA Cronbach's Compositereliability variance Mode S.D. Factor Scale item Median Construct Dimension Mean alpha extracted (CR) loading (AVE) I treat the other party as my friend. 5 1.59 3.53 4.00 0.814 I think I understand the other party quite 5 0.647 4.87 5.00 1.15 Intimacy well. 0.75 0.51 0.76 The other party makes me feel 4.72 5 1.40 0.671 5.00 comfortable, as if I am with family. I want to support the other party. 5.22 5 0.637 5.00 1.10 Parasocial I would like to continue engaging with the 5.59 6 0.96 0.835 6.00 Relationship relationship other party. 0.77 0.78 0.55 maintenance If the other party appeared on a different 5.39 6.00 1.20 0.732 6 media platform, I would watch it. I would like to be friends with the other 4.86 5.00 0.86 6 1.43 party in real life. Real-life 0.89 0.89 0.80 relationship I would like to meet the other party in 4.92 1.51 0.932 5.00 6 person.

## Table 8.6 Quantitative study 2: PSR CFA results

Subsequently, the study CFA for the nomological model using AMOS. The final CFA results are displayed in Table 8.8, indicating an acceptable fit: fit  $\chi 2$  (df = 125) = 261.204, p = 0.000, Chi-Square/df = 2.09, RMSEA = 0.073, CFI = 0.948, NFI = 0.907, TLI = 0.937 (Byrne, 2016). All AVE values exceeded 0.5, all CR values surpassed 0.7, and all Cronbach's alpha values were above 0.7. Regarding reliability, all Cronbach's alpha/coefficient alpha values were higher than 0.7 (Haws et al., 2022), and all CR values exceeded 0.7, indicating good reliability (Hair et al., 2014).

Quantitative study 2 - PSR HTMT Matrix							
	Real-life	Maintenance	Intimacy				
Real-life							
Maintenance	0.561						
Intimacy	0.580	0.536					

Table 8.7 Quantitative study 2 – PSR HTMT Matrix

For convergent validity, all standardised residual covariance matrix values were below 4, all standardised loadings were above 0.5, and all AVE values exceeded 0.5, illustrating strong convergent validity (Anderson and Gerbing, 1988). Regarding discriminant validity, the HTMT ratio results, presented in Table 8.9, showed inter-construct correlation values ranging from 0.222 to 0.58, all below the 0.85 threshold, thus affirming the absence of a discriminant validity issue (Henseler et al., 2014), and establishing discriminant validity.

The study further executed SEM path analysis through AMOS to examine nomological validity. The results, illustrated in Table 8.10, revealed an acceptable model fit: with  $\chi^2$  (df = 142) = 294.299, p = 0.000, Chi-Square/df = 2.073, RMSEA = 0.072, CFI = 0.945, NFI = 0.901, TLI = 0.934, (Byrne, 2016). The path analysis indicated that influencer credibility significantly and positively influences PSR ( $\beta$ =0.64, p<0.001), PSR significantly and positively influences PSR ( $\beta$ =0.001), PSR exerts a significant positive influence on purchase intention ( $\beta$ =0.13, p=0.029), and brand attitude significantly and positively impacts purchase intention ( $\beta$ =0.68, p<0.001), thereby supporting the nomological validity.

	Quantitative study 2: Results of reliability, confirmatory factor analysis - Nomological model									
Construct	Dimension	Scale item	Mean	Median	Mode	S.D.	CFA Factor loading	Cronbach's alpha/ Correlation coefficient	Average variance extracted (AVE)	Composite reliability (CR)
		I treat the other party as my friend.	3.53	4.00	5	1.60	0.657			
	Intimacy	I think I understand the other party quite well.	4.87	5.00	5	1.15	0.655	0.751	0.51	0.75
		The other party makes me feel comfortable, as if I am with family.	4.72	5.00	5	1.40	0.812			
Parasocial	Relationship maintenance	I want to support the other party.	5.22	5.00	5	1.10	0.723	0.768	0.55	0.78
relationship		I would like to continue engaging with the other party.	5.59	6.00	6	0.96	0.852			
		If the other party appeared on a different media platform, I would watch it.	5.39	6.00	6	1.20	0.632			
	Real-life relationship	I would like to be friends with the other party in real life.	4.86	5.00	6	1.43	0.932	- 0.890	0.80	0.80
		I would like to meet the other party in person.	4.92	5.00	6	1.51	0.861			0.89

# Table 8.8 Quantitative study 2: Nomological model CFA results

Construct	Dimension	Scale item	Mean	Median	Mode	S.D.	CFA Factor loading	Cronbach's alpha/ Correlation coefficient	Average variance extracted (AVE)	Composite reliability (CR)
		This influencer is expert in topics related to their content.	5.57	6.00	6	1.17	0.841			
		This influencer is skilled in topics related to their content.	5.78	6.00	6	1.07	0.879			
		This influencer is knowledgeable in topics related to their content.	5.84	6.00	6	0.97	0.875			
Credibility		This influencer is qualified in topics related to their content.	5.47	6.00	6	1.21	0.763	0.924	0.65	0.93
		I trust in information provided by this influencer.	5.61	6.00	6	1.04	0.738			
		This influencer's social media content is reliable.	5.82	6.00	6	0.91	0.773			
		Overall, I recommend this influencer's social media content.	6.02	6.00	6	0.84	0.731			
		This brand is good.	5.49	6.00	6	1.10	0.909			
Brand attitude		This brand is pleasant.	5.45	6.00	6	1.04	0.916	0.947	0.86	0.95
		This brand is favourable.	5.45	6.00	6	1.08	0.954			

Quantitative Study 2 - Nomological model HTMT Matrix						
	Credibility	Maintenance	Intimacy	Real-life	Attitude	
Credibility						
Maintenance	0.391					
Intimacy	0.434	0.536				
Real-life	0.548	0.561	0.58			
Attitude	0.399	0.224	0.306	0.222		

Table 8.9 Quantitative study 2: Nomological model HTMT Matrix

The final step involved testing the test-retest reliability using SPSS, employing 95% confidence intervals alongside ICC to report test-retest reliability, a method known for generating more accurate results (Aldridge et al., 2017). The confidence interval type applied was Bias-corrected and accelerated (BCa). Although the study originally included 154 participants from Quantitative Study 1, only 79 of these provided consistent responses regarding their favourite influencer across both studies. Consequently, the test-retest analysis was conducted using these 79 responses. The retest results are detailed in Table 8.11.

Significant differences in mean scores were observed for only two out of the eight scale items: "I think I understand the other party quite well" and "The other party makes me feel comfortable as if I am with family". For the item "I think I understand the other party quite well", the mean score was higher in Quantitative Study 2 (M=4.99, SE=0.12) compared to Quantitative Study 1 (M=4.30, SE=0.15). This difference, 0.69, BCa 95% CI [0.41, 0.95], was significant, t(78)=4.58, p<0.001, and represented an effect of d=0.52. For "The other party makes me feel comfortable, as if I am with family", participants show higher with their favourite influencers in Quant Study 2 (M=4.75, SE=0.15) than in Quant Study 1 (M=3.89, SE=0.15). This difference, 0.86, BCa 95% CI [0.58, 1.14], was significant, t(78)=5.39, p<0.001, and represented an effect of d=0.61. In addition, all eight scale items' test-retest correlations were significant, between r=0.371 and r=0.778, p<0.001.

	Quantitative study 2: Nomological network - Path analysis									
No.	Hypothesis			Estimate	S.E.	C.R.	P-value	Test Results		
1	Influencer credibility	-	PSR	0.64	0.15	7.04	***	Supported		
2	PSR	•	Brand attitude	0.35	0.08	4.47	***	Supported		
3	PSR		Purchase intention	0.13	0.09	2.19	0.029	Supported		
4	Brand attitude		Purchase intention	0.68	0.09	11.76	***	Supported		

 Table 8.10 Quantitative study 2: Nomological network – Path analysis

\*\*\*Significant at 0.01 level

Table 8.11 Quantitative stud	y 2: Test-retest 79 responses
------------------------------	-------------------------------

	Quantitative study 2: Test-retest 79 responses												
Dimension/	Quant study 1		Quant study 2		Difference	T-value	P-value	Cohen's d	BCa 95%CI	Correlation			
Construct	Mean	SE	Mean	SE	Difference	1-value	P-value	Conen's d	BCa 95%CI	Correlation			
Intimacy	3.81	0.13	4.43	0.11	0.62	5.31	<0.01	0.6	[0.39, 0.87]	0.556***			
Maintenance	5.5	0.1	5.47	0.09	-0.04	-0.05	0.655	-0.05	[-1.08, 0.11]	0.610***			
Real-life	5.06	0.13	4.97	0.15	-0.09	-0.79	0.643	-0.09	[-0.28, 0.11]	0.695***			
PSR	4.78	0.00	4.96	0.09	0.18	2.44	<0.01	0.65	[0.04. 0.32]	0.699***			

\*\*\*Significant at 0.01 level

For the PSR intimacy dimension, on average, participants show higher intimacy with their favourite influencers in Quant Study 2 (M=4.43, SE=0.11) than in Quant Study 1 (M=3.81, SE=0.13). This difference, 0.62, BCa 95% CI [0.39, 0.87], was significant, t(78)=5.31, p<0.001, and represented an effect of d=0.60. However, the test-retest correlation was significant, r=0.556, p<0.001. Therefore, the findings show that their mean scores differed for the intimacy dimension, but the two tests were correlated.

For the PSR relationship maintenance dimension, on average, participants show lower relationship maintenance with their favourite influencers in Quant Study 2 (M=5.47, SE=0.09) than in Quant Study 1 (M=5.50, SE=0.10). This difference, -0.04, BCa 95% CI [-0.18, 0.11], was not significant, t(78)=-0.55, p=0.655, and represented an effect of d=-0.05. Meanwhile, the test-retest correlation was significant, r=0.61, p<0.001. Therefore, the findings show that their mean scores had no difference for the relationship maintenance dimension, and the two tests were correlated.

For the PSR real-life relationship dimension, on average, participants show lower real-life relationship with their favourite influencers in Quant Study 2 (M=4.97, SE=0.15) than in Quant Study 1 (M=5.06, SE=0.13). This difference, -0.09, BCa 95% CI [-0.28, 0.11], was not significant, t(78)= -0.79, p=0.643, and represented an effect of d=-0.09. Meanwhile, the test-retest correlation was significant, r=0.695, p<0.001. Therefore, the findings show that their mean scores had no difference for the relationship maintenance dimension, and the two tests were correlated.

For PSR, on average, participants show higher PSR with their favourite influencers in Quant Study 2 (M=4.96, SE=0.09) than in Quant Study 1 (M=4.78, SE=0.00). This difference, 0.18, BCa 95% CI [0.04, 0.32], was significant, t(78)=2.44, p<0.001, and represented an effect of d=0.65. However, the test-retest correlation was significant, r=0.699, p<0.001. Therefore, the findings show that the mean scores had no difference for PSR, and the two tests were highly correlated.

The final stage of this study has successfully verified the high reliability and validity of the newly developed PSR scale, consisting of eight scale items, and has also confirmed the three dimensions of PSR. Consequently, a summary table detailing the PSR scale items is provided in Appendix 18.

#### 8.5 Chapter summary

This chapter presents the process involved in the development of the PSR scale and the evaluation of the PSR definition. The development of the PSR scale comprised four critical steps: identifying the main domain characteristics of PSR, generating the initial pool of PSR scale items, refining the initial scale items, and validating and verifying the scale's reliability and validity.

A comprehensive literature review and semi-structured interviews facilitated the identification of 40 initial PSR scale items. Following the analysis of feedback from eleven experts, 20 items were discarded, leaving 20 for further examination. Subsequent to quantitative study 1, which garnered 205 valid consumer responses, 12 items were eliminated, leaving eight retained scale items and identifying the three dimensions of PSR. Quantitative study 2, with 206 valid responses, not only confirmed the three dimensions of PSR but also established the validity and reliability of the new PSR scale. Additionally, the PSR definition was evaluated and revised accordingly. The next chapter will present the findings from testing the conceptual framework, constituting the sixth phase of study, quantitative study 3.

# Chapter 9: Empirical findings of the PSR formation model

## 9.1 Chapter introduction

The previous chapter outlined the operationalisation of constructs and introduced a new PSR measurement in the online environment for use in this study. This chapter details the findings from testing the conceptual framework. Chapter 6 introduced a PSR development process model comprising four stages (refer to Figure 6.1). However, as extensively discussed in Section 7.2, due to the thesis's scope, the model's complexity, and practical considerations, this study empirically examined three stages of the PSR development process: experimentation, intensification, and integration (referred to as Model 1). This conceptual model is developed based on existing literature and qualitative insights.

The study utilised an online survey, garnering 405 valid responses for the assessment of the proposed model (Model 1). Data analysis adhered to the methods outlined in Section 7.7, encompassing data cleaning, descriptive analysis, scale validity and reliability assessment, testing of research hypotheses, mediation analysis, and moderation analysis. Following recommendations from modification indices, Model 1 was revised to Model 2, incorporating an additional relationship to improve model fit. The twelve research hypotheses were examined, and extra relationships are presented.

The structure of this chapter is as follows. Firstly, the rationale of this study will be clarified. Subsequently, the respondents' profiles will be depicted. The third section will examine the testing of Model 1 and the alterations leading to Model 2. The fourth section will elucidate the mediation and moderation analyses conducted. The fifth section will explore the posthoc analysis. Lastly, a summary of this chapter's key points will be provided.

### 9.2 Study rationale

Chapter 6 theorises consumers' PSR development process and empirically tests the conceptual model to validate and generalise the model is needed. The tested model (Model 1) is shown in Figure 9.1. Using a quantitative approach to validate the model allows a more rigorous and objective approach to testing the PSR development process.

# Figure 9.1 Model 1



This study significantly augments the existing body of literature on PSR by offering a comprehensive and empirical framework for understanding the dynamics of consumer connections with social media influencers. Such an approach enables the generalisation of quantitative findings and the application of relevant theories. In particular, this study tackles **research objective 3: to explore the formation process of PSR development and test its outcome.** The study design is discussed in Chapter 7.

## 9.3 Respondents' information

The Quantitative Study 3 showcases a broad age range among participants, spanning from 18 to 73 years old, with an average age of 34. This age distribution is in line with the global demographic of social media users in 2023 (Newberry, 2023). The gender distribution reveals a predominance of female participants (61.2%) over male participants (38.8%), diverging from the global trend of more male than female social media users reported by Kemp (2023). Nonetheless, this finding is consistent with previous studies by Lim et al. (2020) and Yuan and Lou (2020), which observed that females are more inclined to form long-term connections with influencers. The participants predominantly reside in Europe, representing a wide array of nationalities.

Regarding social media platform usage, Instagram emerges as the most frequently used platform (44%), followed by YouTube (20.2%), TikTok (11.9%), Facebook (10.9%), Twitter (6.9%), Reddit (5.9%), and Twitch (0.2%). The preferred categories of influencers among respondents are fashion and lifestyle (each at 16.05%), beauty (12.1%), gaming (10.12%), and entertainment (7.16%). The data indicates that respondents have maintained long-term engagements with their favourite influencers, with 35.6% following theirs for two to five years, 22.7% for one to two years, 19.3% for more than five years, 15.6% for six months to one year, and 6.9% for three to six months. In terms of visit frequency to their favourite influencers' social media accounts, around 40% of respondents do so several times per week, whereas a mere 2.2% visit less than once a month.

# 9.4 Conceptual model testing

Prior to assessing the conceptual framework, this study undertook an evaluation of the measurement model's reliability and validity. The rationale behind the chosen data analysis

techniques was detailed in Chapter 7. Establishing the reliability and validity of the measurement model is a critical prerequisite for testing the research model due to its logical necessity (Fornell and Larcker, 1981). Peter (1981) further underscores that construct validity is essential for theory testing and development. Thus, this study commenced with an assessment of the scale items' internal consistency using SPSS reliability analysis (Knapp, 2002). Following this, CFA was performed using SPSS AMOS, marking the initial phase in a two-step SEM process to evaluate model fit and verify the reliability and validity of the measures (Anderson and Gerbing, 1988).

### 9.4.1 Scale reliability and validity

The CFA is conducted to test the reliability and validity of the measurement model. The model shows acceptable fit with  $\chi 2$  (df=907) =1940.173, p=0.00, Chi-Square/df = 2.139, RMSEA = 0.053, CFI = 0.929, TLI = 0.922 (Byrne, 2016).

## Reliability

The results from the CFA are detailed in Table 9.1. First, reliability analysis was conducted using SPSS. Table 9.1 reveals that the Cronbach's alpha/Correlation Coefficient values for the constructs and their dimensions range from 0.76 to 0.95, all surpassing the 0.7 threshold, thus indicating high internal consistency for each construct's scale items (Haws et al., 2022). Upon calculating the CR value, it was established that all constructs and their dimensions are reliable. Table 9.1 indicates that the CR values for each construct and its dimensions vary from 0.77 to 0.95, exceeding the minimum standard of 0.7. This suggests that the constructs and their dimensions exhibit high consistency (Hair et al., 2014).

## Convergent validity

Convergent validity of the constructs is demonstrated through several indicators as detailed in Table 9.1. All constructs' standardised loadings exceed 0.5, and the values in the standardised residual covariance matrix are below 4. Furthermore, all AVE values range from 0.52 to 0.87, surpassing the benchmark of 0.5. This indicates high convergent validity (Hair et al., 2014; Anderson and Gerbing, 1988).

		Quant	titative st	udy 3: CF.	A results						
Construct	Dimension	Scale item	Mean	Median	Mode	S.D.	CFA Factor loading	Cronbach's alpha/ Correlation Coefficient	Average variance extracted (AVE)	Composite reliability (CR)	
		I have the feeling that this influencer is aware of me.	2.88	2.00	1	1.70	0.849				
		I have the feeling that this influencer knows I am there.	2.98	2.00	1	1.75	0.878				
		I have the feeling that this influencer knows I am aware of them.	3.52	4.00	5	1.91	0.845				
PSI		I have the feeling that this influencer knows I pay attention to them.	3.53	4.00	5	1.92	0.872	0.95	0.74	0.95	
		I have the feeling that this influencer knows that I react to them.	3.39	3.00	1	1.88	0.898				
		I have the feeling that this influencer reacts to what I say or do.	2.79	2.00	1	1.73	0.822				
		This influencer is an expert in topics related to their content.	5.57	6.00	6	1.19	0.834				
Perceived		This influencer is experienced in topics related to their content.	5.92	6.00	6	0.99	0.903	0.01	0.54	0.02	
expertise		This influencer is knowledgeable in topics related to their content.	5.89	6.00	6	1.03	0.902	0.91	0.74	0.92	
		This influencer is qualified in topics related to their content.	5.44	6.00	6	1.37	0.8				

 Table 9.1 Quantitative study 3 CFA

Construct	Dimension	Scale item	Mean	Median	Mode	S.D.	CFA Factor loading	Cronbach's alpha/ Correlation Coefficient	Average variance extracted (AVE)	Composite reliability (CR)
	utilitarian	Effective	5.92	6.00	6	0.88	0.647			
		Helpful	5.78	6.00	6	1.08	0.759			0.83
		Functional	5.54	6.00	6	1.21	0.733	0.82	0.50	
		Necessary	4.70	5.00	5	1.41	0.685			
Perceived		Practical	5.43	5.00	5	1.15	0.695			
value		Fun	6.21	6.00	7	0.92	0.604			
		Exciting	5.67	6.00	6	1.15	0.755			
	hedonic	Delightful	5.70	6.00	6	1.15	0.807	0.84	0.52	0.84
	-	Thrilling	4.94	5.00	5	1.31	0.674			
		Enjoyable	6.36	7.00	7	0.84	0.754			

Table 9.1 (cont'd) Quantitative study 3 CFA

		Quant	itative st	tudy 3: CF	A results					
Construct	Dimension	Scale item	Mean	Median	Mode	S.D.	CFA Factor loading	Cronbach's alpha/ Correlation Coefficient	Average variance extracted (AVE)	Composit reliabilit (CR)
Perceived Authenticity		This influencer is real.	6.37	7.00	7	0.80	0.693			0.85
		This influencer is authentic.	6.08	6.00	6	1.00	0.873	0.85	0.66	
		This influencer is truthful.	5.93	6.00	6	1.07	0.865			
		This influencer can be relied upon on their content.		6.00	6	0.96	0.773			
		I believe what this influencer says and that they would not try to take advantage of the followers.	5.45	6.00	6	1.38	0.888		0.68	0.89
Trust		This influencer is straightforward and honest even though their self- interests are involved.	5.69	6.00	6	1.18	0.877	0.88		
		This influencer would not tell a lie even if they could gain by it.	4.86	5.00	6	1.46	0.742			
		I treat the other party as my friend.	4.67	5.00	5	1.48	0.823			
PSR	Intimacy	I think I understand the other party quite well.		5.00	5	1.26	0.749	0.80	0.58	0.81
		The other party makes me feel comfortable, as if I am with family.	3.40	3.00	3	1.62	0.717			

 Table 9.1 (cont'd) Quantitative study 3 CFA

								Cronbach's	Auorogo	
Construct	Dimension	Scale item	Mean	Median	Mode	S.D.	CFA Factor loading	alpha/ Correlation Coefficient	Average variance extracted (AVE)	Composite reliability (CR)
PSR		I want to support the other party.	4.94	5.00	5	1.35	0.809			
	Relationship maintenance	I would like to continue engaging with the other party.	5.55	6.00	6	1.06	0.747	0.76	0.53	0.77
		If the other party appeared on a different media platform, I would watch it.	5.35	6.00	6	1.32	0.616			
	Real-life	I would like to be friends with the other party in real life.	4.81	5.00	5	1.52	0.87		0.78	0.88
	relationship intention	I would like to meet the other party in person.	4.93	5.00	5	1.55	0.899	0.88		
		This brand is good.	5.54	6.00	6	1.19	0.908			
Brand Attitude		This brand is pleasant.	5.52	6.00	6	1.12	0.938	0.95	0.87	0.95
		This brand is favourable.	5.48	6.00	6	1.13	0.952			

# Table 9.1 (cont'd) Quantitative study 3 CFA

# Table 9.1 (cont'd) Quantitative study 3 CFA

	Quantitative study 3: CFA results												
Construct	Dimension	Scale item	Mean	Median	Mode	S.D.	CFA Factor loading	Cronbach's alpha/ Correlation Coefficient	Average variance extracted (AVE)	Composite reliability (CR)			
		I will support this influencer to see they succeed.	5.05	5.00	5	1.26	0.846						
		I will support this influencer because I want to.	5.42	6.00	6	1.26	0.768						
		This influencer's success will make me happy.	5.00	5.00	5	1.32	0.814						
Support		I will support this influencer if they need help from fans like me.	4.75	5.00	5	1.44	0.896	0.94	0.69	0.94			
		I feel good when I support this influencer.	4.76	5.00	4	1.32	0.895						
		I pay attention to this influencer's needs.	4.12	4.00	4	1.46	0.753						
		I care about this influencer's success.	4.71	5.00	5	1.37	0.825						

#### **Discriminant validity**

Discriminant validity is assessed using the HTMT through SPSS AMOS. The study adopts the HTMT.85 criterion, recognised for its high sensitivity rate (Campbell and Fiske, 1959; Henseler et al., 2014). The HTMT results, as illustrated in Table 9.2, show inter-construct correlation values ranging between 0.214 and 0.756. These values are all below the threshold of 0.85, suggesting the absence of discriminant validity issues within the model (Henseler et al., 2014). Hence, discriminant validity is established.

	Quantitative study 3 - HTMT Matrix												
	PSI	Perceived expertise	Trust	Perceived authenticity	Intention to support	Brand attitude	PSR	Perceived value					
PSI													
Perceived expertise	0.274												
Trust	0.279	0.51											
Perceived authenticity	0.214	0.415	0.752										
Intention to support	0.441	0.322	0.614	0.556									
Brand attitude	0.233	0.382	0.435	0.384	0.462								
PSR	0.519	0.383	0.602	0.556	0.756	0.416							
Perceived value	0.25	0.575	0.624	0.563	0.514	0.413	0.483						

Table 9.2 Quantitative study 3 - HTMT Matrix

# 9.4.2 Correlation analysis

Prior to conducting path analysis, a correlation analysis was performed to examine the relationships among constructs using SPSS. Table 9.3 presents the outcomes of this analysis, indicating that there are significant correlations between constructs for each hypothesis, with all p-values falling below 0.001. Specifically, H1 demonstrates a weak correlation between PSI and perceived values. Hypotheses 2, 5, 6, and 9 reveal moderate correlations between the constructs. In addition, Hypotheses 3, 4, 7, and 8 exhibit strong correlations among the constructs involved.

	Qua	ntitative S	tudy 3 - Correlation	Analysis		
No.		Hypothesis	5	r	P-value	Strength
1	PSI		Perceived value	.249**	<.001	Week
2	Perceived value		PSR	.464**	<.001	Moderate
3	Perceived value		Trust	.614**	<.001	Strong
4	Trust		PSR	.574**	<.001	Strong
5	PSR	<b>→</b>	Purchase intention	.372**	<.001	Moderate
6	PSR	-	Brand attitude	.399**	<.001	Moderate
7	Brand attitude		Purchase intention	.650**	<.001	Strong
8	PSR		Support	.737**	<.001	Strong
9	Support		Purchase intention	.468**	<.001	Moderate

Table 9.3 Quantitative study 3 - Correlation Analysis

\*\*Significant at 0.05 level

### 9.4.3 Path analysis

#### Model 1

Path analysis is conducted using SEM through SPSS AMOS to test the research hypotheses. It is also the second step that Anderson and Gerbing (1988) suggest when testing the research model. The model was developed in AMOS and is shown in Figure 9.2. The model shows acceptable model fit with  $\chi^2$  (df=683) = 1495.398, p = 0.000, Chi-Square/df = 2.189, RMSEA = 0.054, CFI = 0.933, NFI = 0.884, TLI = 0.927 (Byrne, 2016).

Table 9.4 shows the path analysis results for model 1. For H1, PSI significantly positively influences perceived value ( $\beta$ =0.465, p<0.001), so H1 was supported. For H2, perceived value significantly positively influences PSR ( $\beta$ =1.09, p=0.002), so H2 was supported. For H3, Perceived value positively influences Trust ( $\beta$ =0.869, p<0.001), so H3 was supported. For H4, Trust does not positively influence Trust ( $\beta$ =-0.222, p=0.467), so H4 was rejected. For H5, PSR does not positively influence Purchase intention ( $\beta$ =-0.062, p=0.609), so H5 was rejected. For H6, PSR significantly positively influences Brand attitude ( $\beta$ =0.529, p<0.001), so H6 was supported. For H7, brand attitude significantly positively influences purchase intention ( $\beta$ =0.571, p<0.001), so H7 was supported. For H8, PSR significantly positively influences Intention to support ( $\beta$ =0.894, p<0.001), so H8 was supported. For H9, Intention to support significantly positively influences Purchase intention ( $\beta$ =0.267, p=0.019), so H9 was supported.

Figure 9.2 Model 1 in AMOS



		Quan	titative Study 3	- Model 1	- Path an	alysis		
No.	]	Hypothesis		Estimate	S.E.	C.R.	P- value	Test Results
1	PSI		Perceived value	0.465	0.019	6.400	***	Supported
2	Perceived value		PSR	1.09	0.952	3.049	0.002	Supported
3	Perceived value	-	Trust	0.869	0.22	7.685	***	Supported
4	Trust		PSR	-0.222	0.418	-0.728	0.467	Rejected
5	PSR	•	Purchase intention	-0.062	0.187	-0.511	0.609	Rejected
6	PSR		Brand attitude	0.529	0.061	9.461	***	Supported
7	Brand attitude		Purchase intention	0.571	0.067	12.079	***	Supported
8	PSR		Support	0.894	0.074	13.702	***	Supported
9	Support		Purchase intention	0.267	0.155	2.341	0.019	Supported

Table 9.4 Quantitative study 3 – Model 1 – Path analysis results

\*\*\*Significant at 0.01 level

#### Alternative model - Model 2

Based on the modification indices from Model 1, it was suggested that incorporating another relationship between PSI and PSR could enhance the model, indicating a potential direct link between PSI and PSR. This direct relationship is supported by prior literature. For instance, Hu (2016) find that in traditional media contexts, such as movies and talk shows, a heightened level of viewers' PSI significantly bolsters their PSR. Consequently, the pathway linking PSR to purchase intention was removed, and a new pathway from PSI to PSR was introduced. The revised model, developed in AMOS, is depicted in Figure 9.3. This alternative model exhibits an improved fit compared to Model 1, as evidenced by the following metrics:  $\chi 2$  (df = 683) = 1439.955, p = 0.000, Chi-Square/df = 2.108, RMSEA = 0.052, CFI = 0.938, NFI = 0.888, TLI = 0.932 (Byrne, 2016).

Table 9.5 shows the path analysis results of the alternative model. For H1, PSI significantly positively influences perceived value ( $\beta$ =0.337, p<0.001), so H1 was supported. For H2, perceived value significantly positively influences PSR ( $\beta$ =0.406, p=0.012), so H2 was supported. For H3, Perceived value positively influences Trust ( $\beta$ =0.846, p<0.001), so H3 was supported. For H4, Trust does not positively influence Trust ( $\beta$ =-0.284, p=0.039), so H4 was rejected.

Figure 9.3 Model 2 in AMOS


For H6, PSR significantly positively influences Brand attitude ( $\beta$ =0.524, p<0.001), so H6 was supported. For H7, brand attitude significantly positively influences purchase intention ( $\beta$ =0.565, p<0.001), so H7 was supported. For H8, PSR significantly positively influences Intention to support ( $\beta$ =0.89, p<0.001), so H8 was supported. For H9, Intention to support significantly positively influences Purchase intention ( $\beta$ =0.212, p<0.001), so H9 was supported. For the new path between PSI and PSR, PSI significantly positively influences PSR ( $\beta$ =0.336, p<0.001).

	Quantitative Study 3 - Model 1 - Path analysis									
No.	Hypothesis			Estimate	S.E.	C.R.	P- value	Test Results		
1	PSI		Perceived value	0.337	0.019	5.154	***	Supported		
2	Perceived value	+	PSR	0.406	0.397	2.517	0.012	Supported		
3	Perceived value	•	Trust	0.846	0.188	8.034	***	Supported		
4	Trust	-	PSR	0.284	0.19	2.059	0.039	Supported		
5	PSR	-	Purchase intention	0.524	0.06	9.44	***	Supported		
6	PSR	•	Brand attitude	0.565	0.063	12.677	***	Supported		
7	Brand attitude	•	Purchase intention	0.89	0.071	13.887	***	Supported		
8	PSR		Support	0.212	0.059	4.856	***	Supported		
9	Support		Purchase intention	0.337	0.019	5.154	***	Supported		
#	PSI		PSR	0.336	0.032	7.33	***	Supported		

Table 9.5 Quantitative study 3 – Model 2 – Path analysis results

\*\*\*Significant at 0.01 level

#### 9.4.4 Mediation analysis

The full model mediation analysis is conducted through SPSS AMOS. The mediation analysis is guided by Bollen (1989) and Zhao et al. (2010).

#### PSI - Perceived value - PSR

First, the full mediation effect model without the direct path from PSI to PSR was developed in AMOS. The model shows acceptable model fit with  $\chi 2$  (df = 684) = 1495.661, p = 0.000,

Chi-Square/df = 2.187, RMSEA = 0.054, CFI = 0.933, NFI = 0.884, TLI = 0.927 (Byrne, 2016). Then, the full alternative model with the direct effect (the direct path from PSI to PSR) was created in AMOS. The model shows acceptable model fit with  $\chi^2$  (df = 683) = 1439.955, p = 0.000, Chi-Square/df = 2.108, RMSEA = 0.052, CFI = 0.938, NFI = 0.888, TLI = 0.932 (Byrne, 2016). The model fit with direct effect is improved than the model without direct effect. Then, the two models were compared with the Chi-square test. The Chi-square test results suggested a significant difference between the two models ( $\Delta\chi^2$ = 55.706,  $\Delta$ df=1, p=0.00), indicating both direct and indirect effects (Bollen, 1989). Meanwhile, PSI positively influences perceived value ( $\beta$ =0.337, p<0.001), perceived value positively influences PSR ( $\beta$ =0.406, p=0.012), PSI positively influences PSR ( $\beta$ =0.336, p<0.001), which suggesting perceived value complementary mediate the effect of PSI on PSR and the indirect effect is accounted for 28.9% of the total effect (Zhao et al., 2010).

#### Perceived value - Trust - PSR

First, the full mediation effect model without the direct path from perceived value to PSR was developed in AMOS. The model shows acceptable model fit with  $\chi^2$  (df = 684) = 1449.446, p = 0.000, Chi-Square/df = 2.119, RMSEA = 0.053, CFI = 0.937, NFI = 0.887, TLI = 0.932, GFI = 0.835 and AGFI = 0.812 (Byrne, 2016). Then, the full alternative model with the direct effect (the direct path from perceived value to PSR) was created in AMOS. The model shows acceptable model fit with  $\chi^2$  (df = 683) = 1439.955, p = 0.000, Chi-Square/df = 2.108, RMSEA = 0.052, CFI = 0.938, NFI = 0.888, TLI = 0.932 (Byrne, 2016). The model fit with direct effect is improved than the model without direct effect. Then, the two models were compared with the Chi-square test. The Chi-square test results suggested a significant difference between the two models ( $\Delta\chi^2$ =9.491,  $\Delta$ df=1, p=0.002), indicating both direct and indirect effects (Bollen, 1989). Meanwhile, perceived value positively influences trust ( $\beta$ =0.846, p<0.001), trust positively influences PSR ( $\beta$ =0.284, p=0.039), perceived value positively influences PSR ( $\beta$ =0.012), which suggesting trust complementary mediate the effect of perceived value on PSR and the indirect effect is accounted for 37.2% of the total effect (Zhao et al., 2010).

#### PSR - Brand Attitude - Purchase Intention

First, the full mediation effect model without the direct path from PSR to purchase intention was developed in AMOS. The model shows acceptable model fit with  $\chi^2$  (df = 683) = 1439.955, p = 0.000, Chi-Square/df = 2.108, RMSEA = 0.052, CFI = 0.938, NFI = 0.888,

TLI = 0.932 (Byrne, 2016). Then, the full alternative model with the direct effect (the direct path from PSR to purchase intention) was created in AMOS. The model shows acceptable model fit with  $\chi 2$  (df = 682) = 1439.568, p = 0.000, Chi-Square/df = 2.111, RMSEA = 0.052, CFI = 0.937, NFI = 0.888, TLI = 0.932 (Byrne, 2016). The model fit with direct effect decreases than the model without direct effect. Then, the two models were compared with the Chi-square test. The Chi-square test results suggested no difference between the two models ( $\Delta\chi 2$ =0.387,  $\Delta$ df=1, p=0.534), indicating an indirect effect and no direct effect (Bollen, 1989). Meanwhile, PSR positively brand attitude ( $\beta$ =0.524, p<0.001), brand attitude positively influences purchase intention ( $\beta$ =0.573, p<0.001), PSR does not influence purchase intention ( $\beta$ =0.73, p=0.537), which suggesting brand attitude indirect-only mediate the effect of PSR on purchase intention (Zhao et al., 2010).

#### PSR - Intention to Support - Purchase Intention

First, the full mediation effect model without the direct path from PSR to purchase intention was developed in AMOS. The model shows acceptable model fit with  $\chi^2$  (df = 683) = 1439.955, p = 0.000, Chi-Square/df = 2.108, RMSEA = 0.052, CFI = 0.938, NFI = 0.888, TLI = 0.932 (Byrne, 2016). Then, the full alternative model with the direct effect (the direct path from PSR to purchase intention) was created in AMOS. The model shows acceptable model fit with  $\chi^2$  (df = 682) = 1439.568, p = 0.000, Chi-Square/df = 2.111, RMSEA = 0.052, CFI = 0.937, NFI = 0.888, TLI = 0.932 (Byrne, 2016). The model fit with direct effect decreases than the model without direct effect. Then, the two models were compared with the Chi-square test. The Chi-square test results suggested no significant difference between the two models ( $\Delta\chi^2$ =0.387,  $\Delta$ df=1, p=0.534), indicating no indirect and direct effects (Bollen, 1989). Meanwhile, PSR positively influences intention ( $\beta$ =0.275, p=0.013), PSR does not influence purchase intention ( $\beta$ =-0.73, p=0.537), which suggesting intention to support indirect-only mediate the effect of PSR on purchase intention (Zhao et al., 2010).

#### 9.4.5 Moderation analysis

To test the moderation effect of perceived expertise on PSI and perceived value and the moderation effect of perceived authenticity on perceived value and trust, the mixed model method is used in the full structural model (Collier, 2020). For testing H10, the composited moderator, perceived expertise and the composited interaction term "CenterPSI X Expertise" are included in the full structure model in AMOS. The results of

the moderation analysis are shown in Table 9.6. As is shown in Table 9.6, perceived expertise positively influences perceived value ( $\beta$ =0.328, p<0.001), but the moderation effect is not established ( $\beta$ =0.016, t-value=1.208, p=0.227). The next step is to run floodlight analysis and visualise the effects through SPSS PROCESS, and the plot is shown in Figure 9.4 (Spiller et al., 2013). The moderation relationship was assessed using PROCESS bootstrapping procedure that computed 5,000 samples at -1SD., Mean, +1SD and 95 confidence interval (model 1), and the results suggest the moderation effect is not significant (R2 change = 0.0015, p=0.34,  $\beta$  = 0.02, SE = 0.02, 95%CI [-0.02, 0.05]). Therefore, H10 is rejected.

Table 9.6 Structural and Moderation Test Results (Mixed Model Method) -H10&H11

Structural and Moderation Test Results (Mixed Model Method)									
No.	Hypothesis	Unstandardized Estimates	T-Value	P-Value	Test Restuls				
	Perceived Expertise -> Perceived Value	0.328	10.154	< 0.001					
H10	PSI X Perceived Expertise -> Perceived Value	0.016	1.208	0.227	Rejected				
	Perceived Authenticity ->Trust	0.277	2.959	0.003					
H11	Perceived Value X Perceived Authenticity ->Trust	-0.059	-1.536	0.125	Rejected				

Figure 9.4 The plot of the moderating effect of perceived expertise on the relationship between PSI and perceived value



For testing H11, the composited moderator, perceived authenticity, and the composited interaction term "CenterValue\_X\_Authenticity" are included in the full structure model in AMOS, and the testing model is presented in Table 9.6. As presented in Table 9.6, perceived authenticity positively influences trust ( $\beta$ =0.277, p=0.003), but the moderation effect is not established ( $\beta$ =-0.059, t-value=-1.536, p=0.125). The next step is to run floodlight analysis

and visualise the effects through SPSS PROCESS and the plot is shown in Figure 9.5 (Spiller et al., 2013). The moderation relationship was assessed using PROCESS bootstrapping procedure that computed 5,000 samples at -1SD., Mean, +1SD and 95 confidence interval (model 1), and the results suggest the moderation effect is not significant (R2 change = 0.0004, p=0.49,  $\beta = -0.03$ , SE = 0.05, 95%CI [-0.13, 0.06]). Therefore, H11 is rejected.

# Figure 9.5 The plot of the moderating effect of perceived authenticity on the relationship between perceived value and trust



For testing H12, the full model moderation test through AMOS results is shown in Table 9.7. The results suggest no statistical relationship between frequency and PSR ( $\beta$ =0.027, p=0.349), but there is a statistical relationship between interaction variable and PSR ( $\beta$ =0.087, p=0.016). Therefore, it can be concluded that the moderation effect of interaction frequency on the relationship between perceived value and PSR is not statistically significant.

Structural and Moderation Test Results (Mixed Model Method) - H12									
No.	Hypothesis	Unstandardized Estimates	T-Value	P-Value	Test Restuls				
	Frequency -> PSR	0.027	0.936	0.349					
H12	Perceived value X Frequency -> PSR	0.087	2.403	0.016	Rejected				

Table 9.7 Structural and Moderation Test Results (Mixed Model Method) – H12

The next step is to run floodlight analysis and visualise the effects through SPSS PROCESS, and the plot is shown in Figure 9.6 (Spiller et al., 2013). The moderation relationship is assessed using PROCESS bootstrapping procedure that computed 5,000 samples at -1SD., Mean, +1SD and 95 confidence interval (model 1), and the results suggest the moderation effect is not significant (R2 change = 0.0005, p=0.61,  $\beta$  = 0.22, SE = 0.42, 95%CI [-0.61, 0.10]). Although there is no statistically significant effect, the moderating plot shows

interaction frequency strengthens the relationship between perceived value and PSR. Therefore, H12 is rejected.





#### 9.5 Post-hoc analysis

Post-hoc analysis is advocated by some scholars as a more ethical and effective method to explore findings that were not anticipated prior to data analysis, offering a foundation for future research directions (Hollenbeck and Wright, 2016). This approach is distinct from practices such as P-hacking or HARKing (Hypothesizing After the Results are Known) in that it employs a transparent process to disclose unforeseen findings, thus maintaining the integrity of the research (Hollenbeck and Wright, 2016). In this thesis, the post-hoc analysis concentrates on examining how individual demographic and behavioural characteristics influence the strength of PSR. Specifically, the study seeks to ascertain whether variations in age and gender among individuals correspond to differing intensities of PSR, thereby contributing to a nuanced understanding of PSR development.

To test the relationship between age and PSR, this study adds another link between age and PSR in Model 2 and then runs the analysis through AMOS. The model shows acceptable model fit with  $\chi 2$  (df = 721) = 1516.319, p = 0.000, Chi-Square/df = 2.103, RMSEA = 0.052, CFI = 0.934, NFI = 0.883, TLI = 0.925 (Byrne, 2016). The result shows no statistical relationship between age and PSR ( $\beta$  = 0.004, p=0.904). Therefore, it can be concluded that PSR does not vary from age.

To test the relationship between gender and PSR, this thesis adds another link between gender and PSR in Model 2 and then runs the analysis through AMOS. The model shows acceptable model fit with  $\chi 2$  (df = 721) = 1508.133, p = 0.000, Chi-Square/df = 2.092, RMSEA = 0.052, CFI = 0.935, NFI = 0.883, TLI = 0.930 (Byrne, 2016). The result shows that PSR varies by gender ( $\beta$  = 0.07, p=0.044). The female' PSR (M=4.86, S.E.=0.07) is higher than the Male's PSR (M=4.82, S.E.=0.08), and the difference is 0.03. Therefore, it can be concluded that females are more likely to have PSR than males.

## 9.6 Chapter summary

This chapter has detailed the process and findings from testing the conceptual framework. Initially, it introduced Model 1, which is subject to examination in this study and comprises 12 research hypotheses. Following the data analysis protocols established in Chapter 7, 405 valid responses were analysed. Upon confirming the reliability and validity of the scales, Model 1 was tested. The modification indices revealed opportunities for enhancing Model 1, leading to the development and refinement of Model 2 based on these insights.

Subsequently, a mediation analysis was carried out, affirming the mediation mechanism within the model. A moderation analysis was then performed; however, the results indicated that the three proposed moderation effects did not achieve statistical significance. The chapter concluded with a post-hoc analysis exploring the relationship between respondents' geographic and behavioural information and their PSR, identifying some positive associations.

The next chapter will have a detailed discussion, bridge the quantitative findings with the existing body of literature, offering a comprehensive discussion on how the current study's results align with, diverge from, or extend previous research insights.

## Chapter 10: Discussion

#### 10.1 Chapter introduction

This chapter discusses the hypothesis testing explored in the preceding chapter (Chapter 9) and brings to conclusion the three research objectives introduced in Chapter 1. These objectives encompass 1) to identify differences between PSI and PSR in the offline and online environments, 2) to propose a new definition of PSR (and PSI) based on their unique prerequisites, and 3) to explore the formation process of PSR development and test its outcome. In particular, this chapter delves into how the empirical findings from the research align with or diverge from existing literature on PSR development.

The structure of this chapter is organised as follows. Firstly, it tackles the first research objective by summarising the arguments in current literature alongside seven prerequisites to distinguish between PSI and PSR. Subsequently, the second research objective is addressed through an evaluation of the newly proposed definitions for PSR and PSI. Following this, the chapter explores the development of PSR measurement. Then, it presents the findings from hypothesis testing and discusses their relation to the existing literature. Finally, it will cover additional observations that have emerged.

# 10.2 Research objective 1: The difference between PSI and PSR in the offline and online environments

The first research objective focuses on the conceptual issues surrounding PSI and PSR. Within the domains of media, psychology, and marketing, PSI and PSR emerge as critical constructs for interpreting individual behaviours (Boehmer, 2016; Iannone et al., 2018; Sokolova and Perez, 2021). Notably, in marketing, especially in influencer marketing, these concepts play a significant role in explaining consumer behaviour both offline and online (Bond, 2021; Noor et al., 2022). Research in this area suggests a correlation between consumers' engagement, intimacy with influencers, and their purchasing behaviours regarding influencer-endorsed products, with consumers' PSI and PSR with influencers (Farivar et al., 2022; Manchanda et al., 2022). However, conceptual ambiguities persist.

The systematic literature review, presented in Chapter 2, highlights the ongoing debate over the main domain characteristics of PSI and PSR in both offline and online contexts. In offline settings, five perspectives on PSI's main domain characteristics and four on PSR's have been identified. A significant portion of scholars argue that "*relationship*" is a main characteristic of PSI, deeming it indistinguishable from PSR. Some researchers even propose that PSI and PSR represent the same concept (e.g., So and Shen, 2016; Penttinen et al., 2022), leading to their interchangeable use in some studies (e.g., Chang and Kim, 2022). Despite calls for clearer distinctions and precise definitions from some academics (e.g., Hartmann and Goldhoorn, 2011; Dibble et al., 2016), these conceptual issues remain unresolved, a situation mirrored in the online environment.

In the online context, the review of PSI and PSR studies reveals four distinct interpretations for the main domain characteristics of each concept. Over half of these studies regard PSI as a "users' relationship", emphasising the "connection" or "relationship" users form with others (e.g., Liao et al., 2022). Approximately one-third of the literature continues to use PSI and PSR interchangeably, underlining the need for clarity and distinction as highlighted by researchers (e.g., Utz et al., 2021). Despite these calls to action, the literature still lacks consensus on definitive criteria to distinguish PSI from PSR effectively.

This thesis argues that the above-mentioned practices lack rigour and can lead to low construct validity and internal validity (MacKenzie, 2003). As emphasises by current literature, a clear and precise definition of a construct lays the groundwork for theory development, making it crucial to differentiate similar concepts, define them accurately, and select the appropriate concept for subsequent theoretical exploration (MacKenzie, 2003; Gilliam and Voss, 2013). Drawing on existing literature and theories related to social interaction and relationships, this thesis proposes seven prerequisites to systematically distinguish between PSI and PSR, thereby enhancing understanding and differentiation of the two concepts. These prerequisites, applicable in both offline and online environments, are summarised in Table 10.1.

The first prerequisite is *one of the parties participating in PSI/PSR*. In both online and offline environments, only people who are exposed to the media environment will generate PSI (Oliver et al., 2019), while there is no similar prerequisite for PSR participants. The second prerequisite is *the other parties participating in PSI/PSR*. The other party of PSI and PSR is similar. In the online environment, the other party can be a real person, a media character, or an anthropomorphic object (Lee, 2018; Kim et al., 2020; Han and Yang, 2018). The range of possible parties with whom a user can develop PSI/PSR is wider online than in traditional media.

Prerequisites for Distinguishing PSI and PSR in Offline and Online Environments									
	1. One of the parties participating in PSI/PSR	2. The other parties participating in PSI/PSR	3. The nature of PSI/PSR	4. The main domain characteristics of PSI/PSR	5. The process of PSI/PSR (from viewer perspective)	6. The time when PSI/PSR can occur	7. The platforms where PSI/PSR can occur		
Offline PSI	media users (Oliver et al., 2019)	media characters, includes real and fictional characters (Cummins and Cui, 2014)	one-way unequal interaction	media viewers' perception of direct and reciprocal communication (Cummins and Cui, 2014; Oliver et al., 2019)	triggered from mutual awareness, attention, and adjustment (Hartmann and Goldhoorn, 2011)	occurs in the media exposure period (Cummins and Cui, 2014; Oliver et al., 2019)	traditional media, including television, movie, radio music, video games and book (Lim and Kim, 2011; So and Nabi 2013; Bentley, 2014)		
Offline PSR	individual (Bernhold and Metzger, 2020)	media characters, includes real and fictional characters (Stephens et al. 1996; Gabriel et al., 2018)	individual's connection (Dias et al., 2017; Bernhold and Metzger, 2020)	the connection (Dias et al., 2017; Bernhold and Metzger, 2020)	triggered from experience at least one-time parasocial interaction or a cross-situational, stable, and schematic cognitive pattern of images and interaction scripts that includes affective aspects (Bernhold and Metzger, 2020)	occurs from the relationship establish to the relationship breakup			

# Table 10.1 The summary table of prerequisites to distinct PSI and PSR in the offline and online environments

Prerequisites for Distinguishing PSI and PSR in Offline and Online Environments									
	1. One of the parties participating in PSI/PSR	2. The other parties participating in PSI/PSR	3. The nature of PSI/PSR	4. The main domain characteristics of PSI/PSR	5. The process of PSI/PSR (from viewer perspective)	6. The time when PSI/PSR can occur	7. The platforms where PSI/PSR can occur		
Online PSI	people who are exposure in the online environment, e.g., new media or technology products users (Kelton and Pennington, 2020)	real person, media characters, and anthropomorphic objects (Lee, 2018; Kim et al., 2020)	one-way unequal interaction	perception of direct and reciprocal communication (Choi et al., 2019; Kelton and Pennington, 2020)	triggered from mutual awareness, attention, and adjustment (Hartmann and Goldhoorn, 2011)	occurs in the exposure period (Yuksel and Labrecque, 2016; Kelton and Pennington, 2020)	through mass and new media, including website, applications, social media, extended media and technology products (Chiu and Huang, 2015; Zhou and Jia, 2018; Tseng et al., 2022)		
Online PSR	individual (Burnasheva and Suh, 2022)	real person, media characters, and anthropomorphic objects (Han and Yang, 2018; Hwang et al., 2020)	individual's connection (Feder, 2020)	the connection (Feder, 2020)	triggered from experience at least one-time parasocial interaction or a cross- situational, stable, and schematic cognitive pattern of images and interaction scripts that includes affective aspects (Bernhold and Metzger, 2020)	occurs from the relationship establish to the relationship breakup			

# Table 10.1 (cont'd) The summary table of prerequisites to distinct PSI and PSR in the offline and online environments

Prerequisite three is the nature of PSI/PSR. This thesis argues that the nature of PSI and PSR is different. While PSI is a one-way unequal interaction, PSR is an individual's connection (Hartmann and Goldhoorn, 2011; Feder, 2020). Prerequisite four is the main domain characteristics of PSI/PSR. PSI is a user's perception of direct behaviour, whereas PSR is the connection (Cummins and Cui, 2014; Feder, 2020). Prerequisite five is the process of PSI/PSR. PSI and PSR are both individuals' automated psychological processes, but there are differences in how they occur. PSI is triggered by mutual awareness, attention, and adjustment (Hartmann and Goldhoorn, 2011; Cummins and Cui, 2014). Establishing PSR requires at least one PSI experience and, in exceptional cases, cognitive experience (Bernhold and Metzger, 2020). Prerequisite six is the time when PSI/PSR can occur, which illustrates the difference between the necessary conditions for PSI and PSR. The occurrence of PSI requires media users to be exposed to the media environment (Kelton and Pennington, 2020), but the occurrence of PSR requires people to voluntarily establish a relationship with the other party until the relationship breaks up. Finally, prerequisite seven is *the platforms* where PSI/PSR can occur. While media platforms are needed to facilitate PSI (e.g., social media), they are not necessary for PSR (Tseng et al., 2022).

In summary, the first research objective has been addressed. This thesis argues that PSI and PSR should be viewed as distinct concepts due to their differing main domain characteristics and nature. PSI is described as the viewer's perception of direct and reciprocal communication with another entity (Hartmann and Goldhoorn, 2011), whereas PSR emphasises the connection between individuals and another entity (Feder, 2020). The identification of seven prerequisites addresses calls from some scholars (e.g., Hartmann and Goldhoorn, 2011) for measures to resolve longstanding debates over the main domain characteristics of PSI and PSR. This can systematically aid scholars in differentiating between these two closely related concepts. Furthermore, the exploration of these prerequisites in both offline and online contexts contribute to a deeper understanding of PSI and PSR, as well as to the conceptualisation of the two terms.

# 10.3 Research objective 2: The new definitions of PSI and PSR based on their unique prerequisites

Developing a well-defined construct is crucial for the robust development and testing of theories (Summers, 2001). While clear prerequisites aid in distinguishing between PSI and PSR, the conceptualisation of these terms requires further refinement. A comprehensive

definition encompasses numerous criteria (Murrow and Hyman, 1994). Merely differentiating between constructs is one attribute of a sound definition. Other important aspects include outlining the construct's main domain characteristics, employing unambiguous terminology, and aligning with existing research (MacKenzie, 2003). As highlighted in Chapter 2, current definitions of PSI and PSR are limited, prompting the second objective of this thesis to develop precise and high-quality definitions for these concepts.

To refine the conceptualisation of PSR, this thesis adopts a "bottom-up" logic and employs several techniques (Murrow and Hyman, 1994; Gilliam and Voss, 2013. Chapter 2 begins with a review of existing definitions of PSR, identifying ten critical components and eight prevalent issues within the current literature. Chapter 5 delves into qualitative data analysis to understand consumer's perceptions of PSR, leading to the proposal of twelve criteria for defining PSR, which are detailed in Table 5.6. The working definition of PSR is proposed as "an individual's dynamic perception of a valuable psychological connection, unilaterally developed with the other party based on repeated exposure through media or technology". Next, based on experts' comments, the PSR working definition is amended as an individual's dynamic psychological attachment to another party, which is developed unilaterally through media and perceived as personally meaningful.

This final definition of PSR, as thoroughly assessed in Chapters 5 and 8, offers significant improvements over existing definitions. It clarifies the roles of the involved parties, limiting the 'other' entities to those perceived as personally meaningful. It encapsulates PSR's main domain characteristics, such as inequality, dynamism, and influence, derived from literature and qualitative insights. Furthermore, this definition explicates the development process of PSR, highlighting the temporal aspect of its formation. Defining PSR as an "individual's dynamic psychological attachment" rather than a general emotional state reduces ambiguity. The details of the evaluation of this definition are discussed below.

PSR is a widespread phenomenon where individuals can develop connections with various entities through media engagement. The reason to define PSR as "individuals" and "another party" is fundamental, representing the two key components of PSR and serving as a criterion for a robust PSR definition. The scope of "the other party" in PSR is broad in the online environment, encompassing real persons like social media influencers (e.g., Cheung et al., 2022), media characters such as fictional entities (Stephens et al., 1996), even anthropomorphic objects like animals (Jacobson et al., 2022) and AR applications (Yuan et

al., 2021). This thesis recommends that the specification of these parties should be contingent upon the research context.

Defining PSR as a "dynamic psychological attachment" introduces new value, diverging from previous definitions that often depicted PSR as an intimate or close, yet static, connection (e.g., Dias et al., 2017; Liao et al., 2022). This new definition reflects the variety of psychological states, strengths, emotions, and feelings individuals might hold towards the other entity. Qualitative data reveals that individuals perceive their relationships with media figures as dynamic, for example, "distance", "close", and "private", characterised by varying degrees of closeness and privacy, thus underscoring PSR as an evolving process that unfolds over time (Bernhold, 2019).

Moreover, this thesis argues the depiction of PSR as merely a "connection" (e.g., Young et al., 2013) or "relationship" (e.g., Sundermann and Munnukka, 2022) as commonly presented in existing literature. Instead, it argues for understanding PSR as an "individual's psychological attachment," emphasising the conscious and motivational aspects underpinning these relationships. This attachment is not incidental but reflects individuals' deliberate efforts to satisfy diverse needs and intrinsic motivations, such as entertainment, information-seeking, and relationship-building (Kim, 2020; Zafar, Qiu, and Shahzad, 2020; Chiu and Huang, 2015). The term "attachment" suggests that PSR often represents a long-term bond rather than a fleeting emotional response, highlighting its significance in the online media environments.

The term "developed unilaterally" highlights the inherently unequal nature of PSR (Yuan et al., 2016). Despite involving two parties, the individual and the entity, the formation of the attachment is predominantly driven by the individual's wishfulness, without the other entity's conscious consent to establish such a relationship. This aspect is relatively underexplored in the literature, underscoring the one-sidedness of PSR, where individuals may engage in repeated interactions to foster a connection with an entity that remains unaware of their existence (Reinikainen et al., 2020; Balaban et al., 2022). This relationship dynamic is characterised by a unique form of inequality: the individual is acutely aware of the other entity, but this awareness is not reciprocated (Lim et al., 2020; Feder, 2020).

The inclusion of "through media" in the definition of PSR is crucial, signifying the medium through which these relationships are cultivated. Research findings and qualitative data support the notion that the development of PSR necessitates the involvement of media

channels such as television, radio, books, movies, websites, social media, e-commerce platforms, and technological devices (Goldberg and Allen, 2008; Xiang et al., 2016; Nanda and Banerjee, 2020; Lee and Cho, 2020). This element of the definition underscores the pivotal role of media in facilitating the growth of PSR.

Furthermore, being "perceived as personally meaningful" encapsulates the significance and impact of PSR on individuals. In PSR, the entity is valued by the individual, providing both utilitarian and hedonic benefits (Guo et al., 2021; Lou and Yuan, 2019). This perceived value influences the individual's tolerance towards different viewpoints and can lead to changes in attitudes or behaviours based on the recommendations or endorsements of the entity. This influence, frequently cited by interviewees, highlights the profound effect that PSR can have on an individual's perspectives and decisions.

In sum, this new PSR definition adds value to the existing PSR definitions. It is evaluated as a clear, high-quality, parsimonious, unambiguous definition that does not include antecedents and consequences, which meet the good definition criteria (Gilliam and Voss, 2013; Murrow and Hyman, 1994; MacKenzie, 2003).

Regarding the conceptualisation of PSI, despite debates over its main domain characteristics, the seven prerequisites for identifying these characteristics should correspond with Hartmann and Goldhoorn's (2011) interpretation. Hartmann and Goldhoorn (2011) define PSI as "this idea of parasocial interaction as the experience of TV viewers to be engaged in an immediate, personal, and reciprocal encounter with a TV performer". This definition enhances the understanding of PSI by focusing on the 'interaction' aspect, which accurately captures the nature of viewers' experiences within a media context. Hartmann and Goldhoorn's (2011) work has been instrumental in delineating PSI and in exploring the psychological processes viewers undergo during such interactions. Nevertheless, qualitative data suggests that the definition of PSI can be further refined to "an individual's perception of a direct, reciprocal and one-to-many communication with another party through mass and new media as one-to-one". This refinement aims to provide a more precise understanding of PSI by emphasising the perception of personal communication within mass and new media environments.

#### 10.4 PSR measurement

This thesis introduces a new PSR measurement instrument designed for online environment, contributing significantly to the search for an appropriate scale in future studies on PSR in the online environment. As discussed in Chapters 7 and 8, the PSR scale most frequently cited in existing literature is grounded in "psychological involvement" as its main domain characteristic, a definition developed by Rubin et al. (1985) through a scale comprising 20 items. Another commonly used scale, devised by Rubin and Perse (1987), modifies the former with ten scale items.

However, these prevalent scales fail to fully capture the main domain characteristics of PSR proposed in this thesis as "psychological attachment." This term implies a narrower focus than "psychological involvement," which encompasses a range of psychological processes such as transportation, PSI, identification, and worship (Brown, 2015). Additionally, these scales are developed in the traditional media environment, which cannot precisely be modified in the online environment. These gaps in the literature underline the need for a new PSR scale that is both highly reliable and valid, capable of precisely measuring PSR in online contexts through a triangulation approach.

Diverging from the commonly used unidimensional approach to PSR measurement, the PSR scale proposed in this thesis adopts a multidimensional perspective. It comprises three dimensions: intimacy, relationship maintenance, and real-life relationship intention. These dimensions collectively offer a more nuanced understanding and accurate representation of the complex dynamics between consumers and influencers in PSR.

The intimacy dimension of PSR aims to encapsulate the emotional connection consumers form with influencers, where, in some instances, consumers view their preferred influencers as friends (Whang and Im, 2021). The relationship maintenance dimension reflects the continuous engagement and efforts consumers invest to sustain their relationship with influencers, driven by the personal meaning of the PSR and a desire not to lose the established bond (Zhong et al., 2021). The dimension of real-life relationship intention focuses on the consumers' aspirations to transition their parasocial relationships into actual social relationships, as highlighted by interviewees expressing a desire to meet their favourite influencers in person and develop real-life social connections.

The development of the PSR scale for online contexts addresses the current scholars' call for an updated measurement tool, recognising that existing scales are predominantly based on traditional media settings, such as television (Kim, 2022). This new PSR measurement instrument is specifically suitable for the influencer marketing context and utilises a triangulation method for validation, incorporating expert surveys and two quantitative studies. This comprehensive approach ensures the reliability and validity of the newly developed PSR scale in the online environment, making a significant contribution to future research in PSR. By employing this new scale, forthcoming studies can more accurately capture the dynamic nature of PSR and potentially extend its application to other online contexts, such as PSR between consumers and AI assistants.

#### 10.5 Research objective 3: PSR development process and its outcome

The third research objective of this thesis is to explore the PSR development process and its outcomes. Chapter 2 discusses how previous studies have examined the antecedents and consequences of PSR, commonly assuming uniformity in the quality, strength, and stages of PSR. However, this approach overlooks the inherently dynamic and complex nature of PSR, a point that has been underscored by some scholars, such as Tukachinsky and Stever (2018), who argue that PSR should be understood as a developmental process rather than a static concept.

For instance, Pressrove and Pardun (2016) find a positive correlation between the amount of time spent on social media and the strength of PSR. Similarly, Farivar et al. (2022) observe that the duration of following an influencer on Instagram has a positive impact on PSR. Despite these acknowledgements of PSR as a process, the literature review reveals that Tukachinsky and Stever (2018) are among the few who have attempted to outline the PSR development stages theoretically in the traditional media environment. While their proposed model offers valuable insights and serves as an important theoretical foundation, its lack of empirical testing leaves questions about its reliability and validity unanswered. Therefore, this thesis aims to fill this gap by examining the PSR development process in the online environment, thereby providing a deeper understanding of its complex dynamics.

The investigation employed 25 semi-structured interviews with social media users regarding their development of PSR with their favourite influencers. The qualitative data garnered from these interviews confirmed the four stages of the PSR development process as initially

proposed by Tukachinsky and Stever (2018): initial exposure, experimentation, intensification, and integration. Furthermore, the analysis of the qualitative data led to the identification of key themes representing critical factors within each stage of development. Drawing on both the qualitative study's findings and existing theoretical frameworks, a conceptual framework for understanding the PSR development process was developed.

The conceptual framework and research hypotheses are empirically tested and confirmed through quantitative study, although it did not establish the anticipated moderating relationships. This thesis marks a pioneering effort in empirically examining the development of consumers' PSR with influencers on social media platforms and their impact on marketing-related outcomes, such as brand attitude and purchase intention. The outcomes of the quantitative hypotheses testing are detailed in Table 10.2.

Given these findings, future research is encouraged to build upon this thesis by exploring additional marketing-related outcomes that may be influenced by PSR. Moreover, there is an opportunity to test the applicability of the developed model in varied contexts, such as the relationship between consumers and voice assistants or brands.

Research hypothesis	Result
H1: PSI will positively influence perceived value of influencer's social media content on social media	Supported
H2: Perceived value of influencer's social media content will positively influence PSR on social media	Supported
H3: Perceived value of influencer's social media content will positively influence trust in the influencer	Supported
H4: Trust in the influencer will positively influence PSR on social media	Supported
H5: PSR will positively influence purchase intention on social media	Supported
H6: PSR will positively influence brand attitude on social media	Supported
H7: Brand attitude will positively influence purchase intention on social media	Supported
H8: PSR will positively influence intention to support on social media	Supported
H9: Intention to support will positively influence purchase intention on social media	Supported
H10: Perceived expertise will moderate (strengthen) the relationship between PSI and perceived value	Rejected
H11: Perceived authenticity will moderate (strengthen) the relationship between perceived value and trust	Rejected
H12: Interaction frequency will moderate (strengthen) the relationship between perceived value and PSR	Rejected

Table 10.2 The summary table of research hypothesis testing results

#### 10.5.1 PSI and perceived value (H1)

PSI represents a typical perception among individuals when consuming media, characterised by consumers' direct, reciprocal, and one-to-many communication with another party through media (Hartmann and Goldhoorn, 2011). Essentially, during media consumption, individuals engage in a conversational exchange with another entity (Tukachinsky et al., 2020). PSI's intensity is significantly influenced by the communicative strategies of the media figures, including both their verbal and non-verbal cues (Cummins and Cui, 2014). For instance, direct address approaches can foster stronger PSI compared to non-direct addresses (Oliver et al., 2019).

In social media environments, influencers frequently employ strategies that evoke psychological closeness, such as portraying themselves as "friends" to their viewers, thereby enhancing the sense of interactivity (Hartmann and Goldhoorn, 2011). Qualitative evidence supports the notion that influencers' varied techniques consistently spark consumers' PSR. This research corroborates the significance of PSI in the progression towards PSR.

The quantitative findings confirm a positive relationship between PSI and perceived value within an influencer marketing context, thereby extending their interrelation. From a hedonic value standpoint, existing literature has established that PSI fosters consumers' positive emotions, such as satisfaction (Chiu and Huang, 2015; Choi et al., 2019). This study broadens our understanding by demonstrating that PSI with influencers also engenders hedonic feelings in individuals. Current research predominantly investigates the positive association between people's PSI with non-human entities (e.g., mobile shopping applications) and hedonic value. This study progresses our comprehension by illustrating that people's PSI with human entities, specifically influencers, positively affects their perceived hedonic value.

From a utilitarian value perspective, previous studies have highlighted the positive impact of consumers' PSI with non-human entities (e.g., websites and smart speakers) on their perception of utilitarian value (e.g., ease of use and usefulness) (Goldberg and Allen, 2008; Hsieh and Lee, 2021). This study corroborates these findings and expands their applicability to the human-centric influencer marketing context.

The qualitative findings illustrate that consumers experience PSI when engaging with influencers' posts, influencing their perception of both hedonic (e.g., enjoyment) and

utilitarian values (e.g., usefulness). The quantitative data bolster the qualitative study's outcomes. The results of hypothesis testing in the quantitative phase confirm the positive relationship between PSI and perceived value in the influencer marketing context.

This investigation primarily concentrates on two dimensions of perceived value, hedonic and utilitarian values, acknowledging the existence of additional facets of value delineated by various models, such as symbolic value. Consequently, this study's insights pave the way for future research to examine the nexus between PSI and other perceived value dimensions (e.g., symbolic value) across different contexts (e.g., virtual influencers).

#### 10.5.2 Perceived value and PSR (H2)

This study underscores the pivotal role of perceived value in the PSR development process, identifying it as a fundamental driver. It demonstrates that the value and benefits individuals derive from an influencer's content positively influence their PSR development with influencers (Zhong et al., 2021). The qualitative evidence suggests that if individuals believe an influencer will consistently deliver value, they are more inclined to engage and sustain the relationship to fulfil their diverse needs. Notably, consumers' perceived value from influencers' posts directly enhances their PSR with these influencers.

This study finds a positive correlation between consumers' perceived value from influencers' posts and their PSR with these influencers, thereby enriching our comprehension of the dynamic between these constructs within the influencer marketing context. From a hedonic perspective, previous studies have established a positive linkage between perceived value and consumers' PSR with non-human entities, such as virtual assistants (Noor et al., 2022) and brands (Zhong et al., 2021). This study contributes to the discourse by providing evidence of a similar positive relationship between perceived hedonic value and consumers' PSR with human influencers, signifying that these favourable effects are also present in influencer marketing. From a utilitarian standpoint, the findings indicate that perceived utilitarian value from an influencer's content strengthens PSR, corroborating the qualitative insights of Quelhas-Brito et al. (2020). Thus, this study broadens the scope of the positive relationship between perceived values and pSR in the context of influencer marketing.

Other dimensions of perceived value should also be considered and tested for future studies. Although this study focuses on hedonic and utilitarian values, additional dimensions, such as symbolic value identified in qualitative research, merit examination for their impact on PSR. Future research could also assess the relationship between perceived value and consumers' PSR with other entities (e.g., mobile applications and animal influencers), thereby expanding the breadth of understanding of PSR.

#### 10.5.3 Perceived value and trust (H3)

Although existing literature presents conflicting findings regarding the relationship between perceived value and trust, this study demonstrates that the perceived value of influencers' posts positively influences consumers' trust in them. Some studies have identified a positive relationship between these constructs (Dastan and Gecti, 2014; Sharma and Klein, 2020), while others have not found such a relationship to exist (e.g., Albayrak et al., 2019).

One potential reason for these differing outcomes may lie in the varied research contexts. Scholars have explored the relationship between these constructs across diverse settings, including traditional retail brands (Achmad et al., 2020), websites (Albayrak et al., 2019), and social media platforms (Sharma and Klein, 2020). This study, concentrating on social media and influencer marketing, corroborates previous research focused on social media online groups, suggesting that consumers' perceived value can enhance trust in the online buying group (Sharma and Klein, 2020). The qualitative findings indicate that consumers are likely to trust influencers if they recognise both hedonic and utilitarian values in the content. Thus, this study empirically substantiates a positive relationship between perceived value and trust within the context of influencer marketing.

It should be noted that, in this study, perceived value is treated as a single construct, measured through two dimensions: hedonic and utilitarian value, rather than considering these dimensions as separate constructs. Previous research investigating the relationship between hedonic value and trust, and utilitarian value and trust separately, has yielded conflicting results. For instance, Lou and Yuan (2019) discover that the informative value of influencer posts positively affects users' trust in the advertised content, whereas the entertainment value does not have a significant impact on users' trust.

This study provides evidence that the overall perceived value of influencers' posts positively influences consumers' trust in influencers. It opens avenues for future research to explore various aspects of perceived value on trust further. Specifically, in influencer marketing,

different types of influencers deliver distinct values. For instance, qualitative findings suggest that entertainment influencers primarily offer hedonic value, whereas technology influencers provide utilitarian value. This study did not concentrate on a particular influencer type, thereby enabling future research to examine the relationships across diverse influencer domains.

#### 10.5.4 Trust and PSR (H4)

Trust emerges as a critical construct in influencer marketing, serving as a metric to assess and forecast the quality of the relationship between individuals and influencers (Kim and Kim, 2021). This study centres on trust as the belief in an influencer's consistent and competent quality delivery (Jun and Yi, 2020; Chaudhuri and Holbrook, 2001). The findings reveal that trust in influencers positively affects their PSR with influencers. Qualitative evidence supports that during the PSR development process, increased trust in a favoured influencer subsequently strengthens the PSR.

Aligning with existing research on consumer and brand relationships, this study extends the positive relationship between trust and PSR to the influencer marketing context. Current consumer and brand relationship studies have found that consumer trust in brands contributes to consumer-brand relationship development, such as brand loyalty (Li et al., 2020). This study corroborates that trust similarly strengthens PSR in influencer marketing.

Moreover, this finding highlights the critical role of trust in the PSR development process. While previous research primarily examined influencers' characteristics, such as trustworthiness (Yuan et al., 2016; Yuan et al., 2020) and credibility (Sokolova and Perez, 2021), in influencing PSR, this study provides additional evidence that trust in influencers also exerts a positive impact on PSR.

The findings further elucidate the significant mediating role of trust in influencers within the relationship between the perceived value of influencers' posts and consumers' PSR with those influencers. Notably, the indirect effect constitutes 37.2% of the total effect. This indicates that during the PSR development process, as consumers perceive value in the influencers' posts, their trust in the influencers increases, subsequently enhancing the consumer-influencer relationship. Moreover, this suggests that PSR is cultivated over time through the consistent generation of content by influencers, with consumers gravitating

towards influencers who consistently deliver reliable content. Drawing on this insight, future research could delve into other potential mediators to further elucidate the positive relationship between perceived value and PSR.

#### 10.5.5 PSR and purchase intention (H5)

A significant finding from this study is that consumers' PSR with influencers do not directly lead to an increase in purchase intention. This challenges the previously established direct link between PSR and purchase intention, as shown in existing literature, which has largely demonstrated a positive correlation between these constructs (Hwang and Zhang, 2018; Agnihotri and Bhattacharya, 2020). For instance, Hwang and Zhang (2018) discover that viewers' PSR with digital celebrities positively influences their purchase intentions. Similarly, Farivar et al. (2021) report that followers' PSR with influencers significantly boosts their intention to purchase. Contrary to these findings, this study finds no direct positive relationship between PSR and purchase intention, suggesting that PSR with influencers does not necessarily trigger a positive purchase intention for the products or brands endorsed or recommended by the influencer.

However, this study finds that brand attitude and intention to support indirectly only mediate the relationship between PSR and purchase intention. Qualitative evidence suggests that consumers exhibit purchase intentions towards products or brands endorsed by influencers with whom they develop PSR, driven by a desire to support their favoured influencers or due to positive brand attitudes. In essence, a higher purchase intention emerges only when consumers hold a positive attitude towards the brand endorsed by their preferred influencer or when they intend to support these influencers. The detailed discussion will be presented below.

Since this study tests the relationship between PSR and purchase intention in an influencer marketing context, future research can replicate the study and further test their relationship in other research contexts (e.g., AI marketing and live-streaming commerce). In influencer marketing, influencers mainly share their experience with their followers, but for live-streaming to commence, live-streamers sell products. The positive relationship between consumers' PSR with live-streamers and purchase intention may be established in live-streaming commerce. In addition, future research can explore other mediators to explain the relationship between PSR and purchase intention.

#### 10.5.6 PSR and brand attitude (H6)

Brand attitude plays a significant role in the outcome of PSR (Reinikainen et al., 2021). In influencer marketing, influencers advocate for products and brands to their followers. This content appeals to consumers because it aids in their purchasing decisions (Yang and Ha, 2021). Brand attitude overall evaluation of the brand influencers endorsed. According to qualitative research, consumers who regularly view their favourite influencers' content tend to develop a favourable view of the endorsed brands. Quantitative studies further validate the significant impact influencers have on their followers, which can enhance their attitude toward the brand that influencers endorse.

Quantitative results indicate a positive correlation between PSR and brand attitude, aligning with prior research that highlights the substantial influence of consumers' PSR with influencers on their perception of brands (Balaban et al., 2022). This discovery carries vital implications for brands and marketers, underscoring influencer marketing as an effective digital marketing strategy. Therefore, marketers are encouraged to collaborate with influencers who have a dedicated fanbase, allowing them to engage with potential customers to boost positive perceptions of the brand. Furthermore, researchers can investigate additional PSR-related outcomes, such as brand trust and brand loyalty.

#### 10.5.7 Brand attitude and purchase intention (H7)

As discussed above, this study finds that brand attitude serves as a mediator between PSR and consumers' intention to purchase, suggesting that the efficacy of influencer marketing in enhancing purchase intentions is not merely due to the long-standing connections between consumers and influencers (Balaban et al., 2022). Instead, a positive perception of the endorsed brand leads to a higher intention to purchase. This insight implies that the psychological attachment between consumers and influencers does not directly result in greater purchase intentions. Instead, influencers' detailed and authentic sharing of their experience helps consumers evaluate the brand, increasing consumers' purchase intention (Balaban et al., 2022).

The qualitative data support the positive relationship between brand attitude and purchase intention. The qualitative finding reveals that consumers evaluate the brands influencers

endorse or share in their posts. Consumers will increase their purchase intention if they generate a positive attitude toward the brand.

This research aligns with existing studies demonstrating the positive connection between brand attitude and purchase intention (Tseng and Wang, 2023). Echoing Gong and Li (2017) and Reinikainen et al. (2021), it also discovers that in the context of influencer marketing, a positive brand attitude significantly enhances consumer's purchase intention. This carries important implications for marketers: when collaborating with influencers, the focus should be on shaping consumers' perceptions, attitudes, and evaluations rather than on direct sales. A positive consumer attitude towards a brand can subsequently enhance their intention to make a purchase. Future research could examine the relationship between brand attitude and purchase intention within different contexts and explore additional factors that mediate the link between PSR and the intention to purchase.

#### 10.5.8 PSR and intention to support (H8)

Intention to support is a pivotal outcome of influencer marketing, encapsulating the extent to which the individual is willing to help the influencers maintain and increase popularity in the long run. The theory of relationship investment posits that individuals are inclined to make both tangible and intangible contributions towards another party in pursuit of a continuous relationship (Luo et al., 2009; Donabedian, 2021). This concept is corroborated by qualitative data, revealing interviewees' intentions to support their favourite influencers, thereby fostering their encouragement and elevating their fame. For instance, consumers exhibit significant behavioural engagement with influencers, aimed at augmenting the views, likes, and comments on their posts.

This study's findings reveal that consumers' PSR with influencers have a positive impact on their willingness to offer support. Qualitative evidence underscores consumers' enthusiasm for supporting their favoured influencers, suggesting that the drive to back influencers mirrors the inclination to support real-life friends.

To the best of the researcher's knowledge, this study provides the first empirical evidence of the link between PSR and intention to support. Prior research has shown that followers' PSR can foster a readiness to disseminate information on social media (Labrecque, 2014; Kim and Kim, 2020; Chen et al., 2021) and generate positive E-WOM on these platforms (Hwang

and Zhang, 2018). This study extends the understanding of PSR's influence by highlighting intention to support as a significant outcome. Building upon this discovery, future research could examine this relationship in varying contexts, such as consumers' PSR with brands. Furthermore, recognising support intention as a crucial effect of influencer marketing opens avenues for exploring additional outcomes linked to PSR.

### 10.5.9 Intention to support and purchase intention (H9)

As discussed above, intention to support is one of the key influencer-related outcomes, which reflects consumer relationship investment to maintain long-term relationships (Donabedian, 2021). Qualitative data highlight consumers' awareness that influencers' financial well-being often depends on collaborations with brands. Consequently, many express a willingness to support their favourite influencers' success by purchasing products and brands endorsed by them. Quantitative analysis confirms this, establishing that consumers' intention to support positively influences their purchase intention.

To the best of the researcher's knowledge, this study provides the first empirical evidence of a link between intention to support and purchase intention. While previous investigations have identified consumers' support intentions, such as offering virtual gifts or buying related products (Donabedian, 2021; Lu et al., 2018), and noted the influence of favourite influencers on heightened purchase intention (Farivar et al., 2021; Hwang and Zhang, 2018), the direct positive association between intention to support and purchase intention has remained unexplored. Furthermore, although Donabedian (2021) posits a theoretical link between fans' PSR and their relationship investment towards favoured idols, this hypothesis had not been empirically validated prior to this study.

As discussed above, this study does not establish a positive correlation between PSR and purchase intention. However, it identifies the intention to support as a mediator in the relationship between PSR and purchase intention exclusively. This quantitative discovery offers fresh insights into consumer behaviour within digital environment, indicating that although consumers may form a strong psychological attachment with influencers, this attachment alone does not directly lead to an intention to purchase. Instead, the intention to support emerges as a critical factor in influencing their purchasing decisions.

This revelation holds significance for influencers, brands, and marketers, suggesting that efforts should not be solely focused on fostering strong connections with their follower base but also on encouraging their support. Such strategies could enhance consumers' intentions to make purchases, thereby increasing the effectiveness of influencer marketing. Future studies can further explore additional mediators that could explain the relationship between PSR and purchase intention. Moreover, forthcoming research could validate these findings in varied contexts, such as live-streaming commerce.

### 10.5.10 The moderating role of perceived expertise (H10)

This study thoroughly examines the role of perceived expertise in the psychological mechanisms at play during the second stage of PSR development. The findings reveal that consumers' perceptions of direct and reciprocal one-on-one communication directly and positively enhance perceived value (Tseng et al., 2022). This observation unveils a key theoretical insight: consumers' perception of content value is affected by PSI, and this influence is unaffected by perceived influencer expertise.

The absence of a significant moderating effect of perceived influencer expertise on the relationship between PSI and perceived value suggests that consumers can discern both utilitarian and hedonic values in the influencer's content through PSI, regardless of whether they view the influencer as an expert. Thus, it can be inferred that in the latter half of the second stage of PSR development, consumers have already formed certain expectations and recognition of the influencer's expertise based on previous interactions. This acknowledgement of influencer expertise leads consumers to believe that the influencer can provide high-quality content or address consumer needs effectively (Vieira et al., 2022; Lin et al., 2018).

Future research could further explore the role of perceived influencer expertise during the second stage of PSR development. As discussed in Chapter 5, the qualitative study has shown that both the first and second stages are crucial for consumers' perception of influencer characteristics, with a particular emphasis on influencer expertise during the second stage. This indicates a shift in consumer focus from general characteristics in the first stage to a deeper engagement with the influencer's perceived expertise in the second stage. Further investigations could examine the potential moderating effects of perceived

influencer expertise and other influencer characteristics, such as authenticity and similarity, in the early part of the second stage.

#### 10.5.11 The moderating role of perceived authenticity (H11)

This study provides an in-depth examination of the psychological mechanisms that facilitate consumers' transition from the second to the third stage of PSR development. It was found that consumers develop trust towards influencers in the intensification stage upon perceiving the value offered by influencer posts during the experimentation stage. This thesis hypothesised that the perceived influencer authenticity would strengthen this relationship. However, empirical studies did not observe this significant moderating effect. Nevertheless, the insignificance of this moderating effect leaves room for further interpretation.

The lack of a significant moderating effect of perceived influencer authenticity on the relationship between perceived value and trust suggests that when consumers appreciate and derive substantive value on both utilitarian and hedonic levels, they enter into the intensification stage, starting to establish significant relationships with influencers. Therefore, perceived value is proven to be the most critical link between the experimentation and intensification stages. Although the perceived influencer authenticity is particularly important for consumers to establish trust with influencers (Jun and Yi, 2020), it can be further inferred that during the transition from the second to the third phase, consumers have already addressed and resolved their concerns about the authenticity of influencers, consumers have already resolved their concerns about authenticity in the experimentation stage, establishing confidence in the influencers (Lee and Johnson, 2021; Nah, 2022).

Based on the analysis above, perceived influencer authenticity may play a significant role in the first and second stages of PSR development. Undoubtedly, perceived influencer authenticity has been proven to be one of the critical factors for consumers' establishing long-term connections with influencers (Agnihotri et al., 2023). As discussed in Chapter 5 the early stages of PSR, such as the first and second stages, are crucial for consumers' evaluation of influencers. Future research could further investigate the potential moderating effects of influencer authenticity and other characteristics in the first and second stages of relationship building with influencers, thereby enhancing our understanding of how influencer characteristics influence the PSR development process.

#### 10.5.12 The moderating role of interaction frequency (H11)

This study explores the impact of interaction frequency on the transition from the second to the third stage in the development of PSR with influencers. It was discovered that when consumers perceive both utilitarian and hedonic values in influencer content during the experimentation stage, they enter into the intensification stage, establishing long-term and relatively stable PSR with the influencer. As previously analysed, perceived value is a critical factor for consumers to progress from the experimentation to the intensification stage in their relationship with influencers. This research hypothesised that a high frequency of interaction with the influencer during this process would further strengthen this relationship. However, empirical findings suggest that while frequent interactions do enhance this relationship, the moderating effect is not significantly impactful. This leads to further discussion on the matter.

The lack of a significant moderating effect of interaction frequency on the relationship between perceived value and PSR suggests that when consumers appreciate the value provided by influencer content, they develop psychological attachments to the influencer (Dhiman et al., 2023; Venciute et al., 2023). Despite the significance of considering time and interaction frequency in establishing and maintaining long-term relationships, it is inferred that as consumers move from the experimentation to the intensification stage, significant time investment to strengthen relationships with influencers becomes less critical. Instead, consumers have already invested time in familiarising themselves with the influencers and are prepared to maintain these relationships. A study indicated that although relationships improve with the frequency of interaction in an online environment, this effect plateaus after three months (Pennington and Hall, 2020). Given qualitative research where consumers reported developing relatively long-term and stable PSR with influencers around the three-month mark, it is conjectured that the influence of interaction frequency on the PSR establishment process is more pronounced earlier, possibly during the first or second stage.

The discussions above suggest new avenues for future research. The positive impact of interaction frequency on the PSR establishment process is most likely to manifest in the early stages of PSR development, such as the first and second stages. Future studies could delve deeper into the influence of interaction frequency during the initial stages of the PSR development process, such as during exposure initiation and experimentation stages, to

enrich our understanding of the role of interaction frequency in the PSR establishment process.

#### 10.6 Additional observation

Following the re-specification of the original conceptual model (Model 1), a significant link between PSI and PSR was established, culminating in the development of the revised Model 2 (as depicted in Figure 9.3 and outlined in Table 9.5). Current literature posits various assumptions regarding the relationship between PSI and PSR. (Bernhold and Metzger, 2020), with a prevalent view among scholars suggesting that viewers are likely to form PSR after repeated PSI experiences (e.g., Bernhold and Metzger, 2020). Prior research demonstrates the direct correlation between PSI and PSR within traditional media settings, including movies and talk shows, where Hu (2016) observes that increased PSI levels notably enhance viewers' PSR.

The findings of this empirical investigation reveal a direct influence of PSI on PSR, thereby identifying PSI as an antecedent of PSR. This insight is of significant importance as it proves that PSI and PSR are two different concepts that have been previously interchangeably used or considered identical by some researchers (e.g., So and Shen, 2016). This clarification is anticipated to assist future scholars in comprehensively understanding the relationship between PSI and PSR.

Furthermore, these results offer valuable guidance for influencers and brands aiming to develop long-term connections with their audience. It underscores the importance of crafting content that creates a sense of direct communication with consumers, thereby increasing their PSR. Based on this finding, scholars could further explore this relationship within other research contexts.

This finding is also valuable for influencers and brands to develop long-term relationships with consumers. It is necessary to create content that makes people feel the influencers or brands directly communicate with them, thereby increasing consumers' PSR.

#### 10.7 Chapter summary

This chapter discusses the findings to address the three research objectives presented in Chapter 1. For research objective 1: to identify differences between PSI and PSR in the offline and online environments, this thesis adopts a systematic literature review approach, employing a historical method. By reviewing 172 PSI and PSR studies, this thesis proposes seven prerequisites to distinguish PSI and PSR in both environments. This contributes to the literature by precisely identifying PSI and PSR, facilitating their correct application in future studies.

For research objective 2, to propose a new definition of PSR (and PSI) based on their unique prerequisites, this thesis utilises a systematic literature review and expert surveys. It proposes a new definition of PSR as an individual's dynamic psychological attachment to another party, developed unilaterally through media and perceived as personally meaningful. This new PSR definition offers added value over existing definitions by reflecting the complex and dynamic nature of PSR. Moreover, a newly developed three-dimensional PSR measurement comprising eight scale items effectively captures consumer PSR in the online environment, contributing to future studies in finding an appropriate measurement for online PSR.

For research objective 3, to explore the formation process of PSR development and test its outcome, this thesis employs semi-structured interviews and online consumer surveys. It proposes four stages in the PSR development process: exposure initiation, experimentation, intensification, and integration. The key concepts of each stage and their interrelationships are also established. This advances the existing literature on how consumers develop PSR with influencers and the outcomes of such relationships.

Finally, this chapter discusses additional observations and illustrates the direct positive relationship between PSI and PSR, based on modifications to the original conceptual model. This finding assists researchers in understanding that PSI and PSR are distinct concepts, with PSI serving as the antecedent of PSR. The following chapter will conclude the thesis.

# Chapter 11: Contributions and future research

#### 11.1 Chapter introduction

This thesis enhances our comprehension of two important consumer phenomena in media consumption: PSI and PSR. It delineates seven prerequisites for differentiating PSI and PSR within both online and offline contexts. The thesis proposes a refined definition of PSR, which adeptly captures its complex and dynamic nature. This clarity significantly aids in understanding the concept of PSR. Furthermore, the study introduces a new, reliable and valid measurement instrument for measuring PSR in the online environment, thus advancing the current PSR scales. The outcomes of this research enrich the body of literature concerning the nuanced development process of PSR by evaluating its formation and several critical outcomes. Additionally, the findings underscore the importance of developing long-term customer relationships, offering practical insights for brands, influencers, and marketers.

The layout of this chapter is organised as follows. Firstly, this chapter explores the theoretical contributions of this thesis. Subsequently, it delineates the practical implications and recommendations derived from the study. Concluding, the chapter outlines the research limitations and proposes directions for future research.

# 11.2 Theoretical contributions

From a theoretical perspective, this thesis first contributes to advancing the conceptualisation of PSR's complex and dynamic nature. Current literature conceptualises PSR in a static manner, which overlooks the complex nature of PSR (e.g., Lee and Park, 2022; Cheung et al., 2022). Some scholars use "connection" (e.g., Young et al., 2013) or "relationship" (e.g., Sundermann and Munnukka, 2022) to describe PSR, which does not reflect the conscious and motivational origin of the relationship. Also, most of the current PSR definitions do not qualify as good definition standards. For example, some definitions are ambiguous since the definitions include similar concepts (e.g., interpersonal relationships). The ambiguous definition of PSR is insufficient to capture the complex and dynamic nature of PSR.

Through several techniques guided by a "bottom-up" logic, this thesis theoretically elaborates PSR (Murrow and Hyman, 1994; Gilliam and Voss, 2013). This new PSR definition advances the conceptualisation of PSR and defines it as an individual's dynamic psychological attachment to another party, which is developed unilaterally through media and perceived as personally meaningful. This new PSR definition implies the inequality, dynamic, and influence characteristics of PSR, which added new value to the existing PSR definition, which represents people who can have various psychological statuses, strengths, emotions, and feelings toward the other part. This new PSR definition provides new insights into understanding PSR's complex and dynamic nature.

#### Second, this thesis contributes by distinguishing between two similar concepts: PSI and

**PSR**. PSI and PSR are essential yet distinct concepts in marketing, psychology, and media, aiding in the understanding of consumer behaviour. However, current research disputes the main domain characteristics of PSI and PSR, along with the misuse of the two terms in both online and offline environments (Oliver et al., 2019; Breves et al., 2019). For instance, some scholars argue that the main domain characteristics of PSI should centre on "relationship", thereby entangling it with the concept of PSR (e.g., Lee and Lee, 2022). Additionally, some scholars assert that PSI and PSR represent the same concept (e.g., So and Shen, 2016; Penttinen et al., 2022), but this view lacks sufficient evidence. Due to the unclear definitions of PSI and PSR, many authors employ these terms interchangeably, leading to conceptual ambiguity and difficulties in differentiating between the two concepts conceptually (MacKenzie, 2003; Wacker, 2004). Although attempts have been made to distinguish between PSI and PSR, the debate has not yet received significant attention (e.g., Hartmann and Goldhoorn, 2011; Oliver et al., 2019).

Based on the systematic literature review of 172 studies addressing PSI and PSR in both offline and online contexts, this thesis identifies seven prerequisites to differentiate PSI from PSR across these environments, summarised in Table 10.1. This thesis responds to calls from existing literature for a thorough differentiation between these closely related concepts (Hartmann and Goldhoorn, 2011; Oliver et al., 2019). The prerequisites substantially advance the field, addressing the issue of current studies that use PSI and PSR interchangeably without clear distinction. Given the differing conditions for PSI and PSR in offline versus online contexts, these seven prerequisites offer clear criteria to assist future researchers in effectively understanding the differences between the terms in various media environments.

The third theoretical contribution of this thesis is to provide a comprehensive overview of the shifts in PSI and PSR within online environments compared to traditional media, thereby broadening their conceptual scope. Current literature mainly explores the antecedents and consequences of PSI and PSR (Deng et al., 2022; Farivar et al., 2022) but lacks a deep understanding of how these constructs transition from traditional to online media environments. Employing a typological approach, this study enhances our knowledge regarding the impact of media environment changes on PSI and PSR conditions and scopes, offering a systematic comprehension of the unique attributes of PSI and PSR across different media settings.

#### The fourth theoretical contribution focuses on the operationalisation of online PSR.

This thesis provides a reliable and valid online PSR measurement, advancing the existing literature in precisely measuring PSR online. In the current PSR literature, there is no conceptual adequate to capture the dynamic nature of PSR in the online environment. Rubin et al. (1985) and Rubin and Perse (1987) are the two most often used PSR scales that do not capture the main domain characteristics of PSR proposed in this thesis as "*psychological attachment*". Meanwhile, the two most used scales are developed based on the traditional media environment, and some of the scale items are not suitable and cannot adapt to the online environment. Therefore, developing a new online PSR is necessary.

This thesis responds to the calls of existing literature to develop a **new online PSR measurement** (e.g., Kim, 2022). To the best of the researcher's knowledge, this thesis is the first study to advance reliable and valid online PSR. Based on the scientific scale development process proposed by Churchill (1979) and Gerbing and Anderson (1988), this thesis develops a new PSR measurement scale with 12 scale items and three dimensions (intimacy, relationship maintenance, and real-life relationship intention). This new PSR scale significantly advances the PSR scale by capturing the multi-dimensional complex nature of PSR compared to the unidimensional PSR scales in the existing literature.

The fifth theoretical contribution of this thesis concerns its theoretical proposal for the dynamic PSR development process in influencer marketing. Existing studies have tested the antecedents and consequences of PSR, and the default of these studies is that the quality, strength and stages of PSR are the same (Noor et al., 2022; Leite and Baptista, 2022). However, these papers fail to capture PSR's dynamic and complex nature. Tukachinsky and Stever (2018) theoretically propose four stages of the PSR development process, which can be seen as a starting point since this model is mainly based on the traditional media

environment and without empirical testing. Hence, it is necessary to explore how consumers develop PSR with influencers.

Through qualitative study and reviewing existing literature, this study advances the existing literature regarding the theoretically proposed PSR development model in the online environment. This thesis confirms the four stages of the PSR development process proposed by Tukachinsky and Stever (2018), including initial exposure, experimentation, intensification, and integration. Critical concepts in each stage are also proposed to contribute to comprehensively and systematically understanding consumers' psychological process from the first interaction with an influencer to the outcomes.

The sixth theoretical contribution of this thesis is validation and generalising the PSR development process. To the best of the researcher's knowledge, this is the first study that offers empirical evidence of the PSR development process in influencer marketing. For example, Tukachinsky and Stever (2018) theoretically propose the model without empirically testing it. This study experientially tests the PSR conceptual model in influencer marketing, contributing to understanding consumers' relationship development process with influencers.

This thesis advances the role of PSI in the PSR development process. The current literature has assumptions about the relationship between PSI and PSR (Bernhold and Metzger, 2020). This study empirically finds that consumers who experience PSI are the antecedent of PSR, which plays a crucial role in the second experimentation stage. This finding also advances the connection and difference between PSI and PSR.

The results of the quantitative study also confirm that perceived value is the core concept in determining consumers from experimentation to the intensification stage. This study contributes to the existing literature by providing new insights into the importance of perceiving hedonic and utilitarian value from influencers as a PSR driver. This illustrates that more value perception from influencers contributes to developing long-term relationships.

In line with the existing literature on consumer-brand relationship development (Li et al., 2020), this study finds that trust is one of the critical antecedents of PSR in consumer relationships with influencers. This study advances the existing literature in understanding

not only influencers' characteristics (e.g., influencer trustworthiness) but also how trust in influencers positively influences the strength of PSR.

The outcome of PSR extends the existing literature on understanding the mechanism of influencer marketing effectiveness. Current literature provides evidence of a positive relationship between PSR and purchase intention (Hwang and Zhang, 2018; Agnihotri and Bhattacharya, 2020). Current studies suggest that PSR directly influence consumers' purchase intention (Farivar et al., 2021). This study advances the influencer marketing mechanism that consumers have purchase intention because they have a high attitude towards the brand or they have an intention to support the influencers and not only because they have established PSR with influencers. To the best of the researcher's knowledge, this study first proposes intention to support as one of the influencer-related outcomes of PSR and empirically test the relationship. This finding advances consumers' psychological and behavioural outcomes of PSR related to influencers.

#### **11.3 Practical implications**

This thesis offers valuable practical implications for influencers, brands and markets, particularly in influencer marketing strategies. The findings of this study provide an in-depth understanding of modern consumer behaviour in the digital age and how to develop long-term relationships with them.

**First, the PSR model can guide influencers and companies to develop long-term consumer relationships**. This study confirms that consumers who develop PSR need to go through different stages, so the findings emphasise the importance of following the PSR development model to establish the relationships. In the first exposure initiation stage, it is essential to increase consumer involvement. On this ground, several factors can effectively increase consumer involvement. For example, the designed social media content needs to satisfy consumers' various needs. Influencers and brands can design more entertaining, rich information and promotional content with a daily life tone to increase consumer involvement and satisfy their needs. Also, influencers can increase their attractiveness, including physical, social, and voice attractiveness, to gain more involvement. The design of the posts can adopt a visual appeals strategy, such as using aesthetically appealing photos, attractive titles and professional video equipment (Kostyk and Huhmann, 2021; Al-Emadi and Ben Yahia, 2020).
Meanwhile, the content of posts needs to use various linguistic styles (e.g., psychological closeness) and non-verbal cues to trigger viewers' PSI. For example, influencers can adopt the ask-questions strategy to make viewers believe they have two-way reciprocity communications. Furthermore, this study finds that perceived value is critical in the experimentation stage, determining whether consumers will enter the intensification stage. Therefore, this study suggests that social media posts need to continuously provide consumers with both hedonic value and utilitarian value to make them feel enjoyable and useful. By doing so, consumers' trust will increase and contribute to PSR development.

#### Second, this study contributes to brands choosing appropriate influencers to cooperate.

This study finds that more PSR strength can lead to high effectiveness of influencer marketing. Therefore, brands can employ the newly developed PSR scales to measure the strength of the relationship between consumers and potential influencers as a complement to the existing techniques. Brands can conduct a quantitative study to test the potential influencers' followers' PSR strength and choose the highest influencers to cooperate with.

Third, the findings of this thesis can guide influencers and brands to generate a more effective influencer marketing strategy. This study confirms that one of the main reasons that consumers have the intention to purchase the brands and products that influencers endorse is because they have the intention to support the influencers. Specifically, the findings show that when consumers have the intention to support the influencers, their purchase intention increases. Therefore, when designing the posts, influencers and brands can employ some strategies to increase consumers' intention to support influencers. In addition, the marketing campaign can focus on increasing consumers' attitudes to brands, which is also effective in increasing consumers' purchase intention.

**Fourth, this thesis contributes to industry-led research**. This thesis provides a clear and in-depth explanation of the difference between PSI and PSR, which benefits marketers in understanding the two different consumer phenomena. As better conceptualisation leads to better operationalisation, this thesis provides accurate, valid and reliable PSR scales, which can be used in industry-led research to explore consumer behaviour further and achieve effective online marketing campaigns.

**Fifth, this thesis makes contributions to society.** This thesis refines the PSR development process model in influencer marketing, enhancing our comprehension of these dynamics and underscoring the potent role influencers assume as value drivers for consumers. By

elucidating the mechanisms underlying PSR in influencer marketing, this research helps consumers to critically assess and evaluate the influences exerted upon them. Consequently, this thesis supports society, consumers, and influencers in cultivating more authentic interactions and disseminating content that is both socially valuable and positive. Ultimately, this contributes to improving individual well-being and satisfaction in the long run.

Sixth, this thesis contributes to policymakers by demonstrating how the PSR development model can develop long-term relationships between policymakers and the public. By leveraging these relationships, policymakers can more effectively promote policies that align closely with public well-being and acceptance. Additionally, this thesis suggests that policymakers collaborate with influencers to address urgent societal issues, such as public health during the COVID-19 pandemic. For example, influencers can play a crucial role in spreading health-related knowledge and practices among their followers to protect public health. This approach can enhance policy communication and ensure that beneficial measures and advice reach a broader audience.

#### 11.4 Limitations and future research directions

Given that this thesis mainly focuses on investigating PSI and PSR in influencer marketing, there are several limitations that have been identified, which can be addressed in future research.

**Firstly**, the PSR model proposed in this thesis includes four stages: exposure initiation, experimentation, intensification, and integration. Since this study aims to explore consumers' PSR development process with influencers, it empirically tests the second to the fourth stages. However, the first stage, exposure initiation, is only theoretically proposed and not empirically tested. Future research can further empirically test the relationship between the three identified critical conditions, consumers' motivations, influencer attractiveness, and post-appeal and consumer involvement, to empirically validate the first stage of the model.

**Secondly,** the PSR development model proposed in this thesis is developed and tested in the influencer marketing context. However, the generalisation of this model in other research contexts needs further examination. For instance, the PSR between consumers and brands and the PSR between consumers and AI assistants could be explored to assess the model's generalisation beyond influencer marketing. Additionally, it should be acknowledged that

this thesis did not control for different categories of influencers, such as virtual influencers. Future research could focus on incorporating various influencer types to enhance the generalisation of this model and address this limitation.

**Thirdly,** the newly developed PSR measurement scale has established its reliability and validity in the influencer marketing context. This new PSR measurement scale contains three dimensions: intimacy, relationship maintenance, and real-life relationship intention. However, in online environments, the other parties might be non-human objectives, such as brands (Chang and Kim, 2022) and AI assistance (Whang and Im, 2021), and it remains uncertain whether consumers would have a real-life relationship intention with these other parties. Therefore, future research can investigate how to adapt or modify this newly developed PSR measurement scale to those different contexts.

**Fourthly**, the PSR development model incorporates key concepts that focus on the process by which consumers develop PSR with influencers. However, this model could be further tested and enhanced. On the one hand, future research might explore the potential roles of other key concepts in the PSR development process (e.g., customer engagement) and PSR outcomes (e.g., continued intention to follow), to advance the understanding of PSR development. On the other hand, a significant additional observation in this thesis is the identification of a direct link between PSI and PSR. The study finds that PSI positively and directly influences PSR during the second stage of the PSR development process. This finding provides a starting point for future research to further investigate the relationship between PSI and PSR.

**Fifthly**, the model measurement took place at a single point in time. Given the dynamic nature of the phenomenon under study, which encompasses the evolving psychological attachments and fluctuating intensities inherent to PSR, this cross-sectional approach limits the understanding of changes in PSR over time. To capture the dynamic and complex nature of PSR, future studies could incorporate time into research design, using experimental methods or longitudinal research approaches. Such research approaches would allow for the examination of the temporal evolution of PSR, offering insights into the developmental stages of PSR and testing the causal relationships between these critical constructs (Mitchell and James, 2001).

Sixthly, the quantitative research in this thesis primarily involves samples from Western countries. Considering the significant impact of culture on consumer decision-making,

future research could test the model in other cultural contexts to generate cross-cultural comparisons.

Lastly, the latest literature reveals that consumers can form PSR with other novel entities, such as virtual influencers, AI assistants (Noor et al., 2022), and AR devices (Yuan et al., 2021). These novel parties have distinct natures and characteristics compared to human influencers. Therefore, future research can test existing PSR-related theories in these new research contexts to understand how PSR manifest differently. Furthermore, researchers can broaden the scope of PSR to identify more types of other parties in the future. Additionally, future studies could also test the PSR development model with these novel parties to expand the generalisation of the model. Considering the differing effects and influences of various ranges of influencer categories based on the number of followers, future studies could further clarify whether the PSR development process with different categories of influencers may have any differences or similarities to enrich the understanding of consumers' PSR development with various influencer categories.

## Appendices

#### Appendix 1: PRISMA 2020 guided systematic literature review

			Location where
Section and topic	Item #	Checklist item	item is reporte
Fitle	1	Identificatio noncet de la avatematica version.	
Title Abstract	1	Identify the report as a systematic review.	
Abstract	2	See the PRISMA 2020 for Abstracts checklist (table 2).	
ntroduction	2		
Rationale	3	Describe the rationale for the review in the context of existing knowledge.	
Objectives	4	Provide an explicit statement of the objective(s) or question(s) the review addresses.	
Methods			
Eligibility criteria	5	Specify the inclusion and exclusion criteria for the review and how studies were grouped for the syntheses.	
nformation sources	6	Specify all databases, registers, websites, organisations, reference lists and other sources searched or consulted to identify studies. Specify the date when each source was last searched or consulted.	
Search strategy	7	Present the full search strategies for all databases, registers and websites, including any filters and limits used.	
Selection process	8	Specify the methods used to decide whether a study met the inclusion criteria of the review, including how many reviewers screened each record and each report retrieved, whether they worked independently, and if applicable, details of automation tools used in the process.	
Data collection process	9	Specify the methods used to collect data from reports, including how many reviewers collected data from each report, whether they worked independently, any processes for obtaining or confirming data from study investigators, and if applicable, details of automation tools used in the process.	
Data items	10a	List and define all outcomes for which data were sought. Specify whether all results that were compatible with each outcome domain in each study were sought (e.g. for all measures, time points, analyses), and if not, the methods used to decide which results to collect.	
	10b	List and define all other variables for which data were sought (e.g. participant and intervention characteristics, funding sources). Describe any assumptions made about any missing or unclear information.	
Study risk of bias assessment	11	Specify the methods used to assess risk of bias in the included studies, including details of the tool(s) used, how many reviewers assessed each study and whether they worked independently, and if applicable, details of automation tools used in the process.	
ffect measures	12	Specify for each outcome the effect measure(s) (e.g. risk ratio, mean difference) used in the synthesis or presentation of results.	
ynthesis methods	13a	Describe the processes used to decide which studies were eligible for each synthesis (e.g. tabulating the study intervention	
	13b	characteristics and comparing against the planned groups for each synthesis (item #5)). Describe any methods required to prepare the data for presentation or synthesis, such as handling of missing summary statistics,	
	120	or data conversions.	
	13c 13d	Describe any methods used to tabulate or visually display results of individual studies and syntheses. Describe any methods used to synthesize results and provide a rationale for the choice(s). If meta-analysis was performed,	
	13e	describe the model(s), method(s) to identify the presence and extent of statistical heterogeneity, and software package(s) used. Describe any methods used to explore possible causes of heterogeneity among study results (e.g. subgroup analysis, meta- regression).	
	13f	Describe any sensitivity analyses conducted to assess robustness of the synthesised results.	
Reporting bias Issessment	14	Describe any methods used to assess risk of bias due to missing results in a synthesis (arising from reporting biases).	
Certainty assessment Results	15	Describe any methods used to assess certainty (or confidence) in the body of evidence for an outcome.	
Study selection	16a	Describe the results of the search and selection process, from the number of records identified in the search to the number of studies included in the review, ideally using a flow diagram (see fig 1).	
	16b	Cite studies that might appear to meet the inclusion criteria, but which were excluded, and explain why they were excluded.	
Study characteristics	17	Cite each included study and present its characteristics.	
lisk of bias in studies	18	Present assessments of risk of bias for each included study.	
esults of individual tudies	19	For all outcomes, present, for each study: (a) summary statistics for each group (where appropriate) and (b) an effect estimate and its precision (e.g. confidence/credible interval), ideally using structured tables or plots.	
Results of syntheses	20a	For each synthesis, briefly summarise the characteristics and risk of bias among contributing studies.	
	20b	Present results of all statistical syntheses conducted. If meta-analysis was done, present for each the summary estimate and its precision (e.g. confidence/credible interval) and measures of statistical heterogeneity. If comparing groups, describe the direction of the effect.	
	20c	Present results of all investigations of possible causes of heterogeneity among study results.	
	20d	Present results of all sensitivity analyses conducted to assess the robustness of the synthesised results.	
Reporting biases	21	Present assessments of risk of bias due to missing results (arising from reporting biases) for each synthesis assessed.	
Certainty of evidence	22	Present assessments of certainty (or confidence) in the body of evidence for each outcome assessed.	
Discussion	23a	Provide a general interpretation of the results in the context of other evidence.	
	23b	Discuss any limitations of the evidence included in the review.	
	23c	Discuss any limitations of the review processes used.	
	23d	Discuss implications of the results for practice, policy, and future research.	
Other information	24a	Provide registration information for the review, including register name and registration number, or state that the review was not	
protocol	24b	registered. Indicate where the review protocol can be accessed, or state that a protocol was not prepared.	
	240 24c	Describe and explain any amendments to information provided at registration or in the protocol.	
upport	240	Describe and explain any amendments to information provided at registration of in the protocol. Describe sources of financial or non-financial support for the review, and the role of the funders or sponsors in the review.	
Competing interests	26	Declare any competing interests of review authors.	
vailability of data, code, and other	27	Report which of the following are publicly available and where they can be found: template data collection forms; data extracted from included studies; data used for all analyses; analytic code; any other materials used in the review.	

Source: Page et al. (2021)

	The Re	eviewed PSI De	finitions in the	Offline Env	ironment			
Authors	Definitions	One of the parties participating in PSI	The other parties participating in PSI	The nature of PSI	The main domain characteristics of PSI	The process of PSI (from viewer's perspective)	The time when PSI can occur	The platform where PSI can occur
Rubin et al. (1985)	Parasocial interaction was conceptualized as interpersonal involvement of the media user with what he or she consumes.	media user	he or she consumes		interpersonal involvement			media
Rubin and Perse (1987)	Parasocial interaction was conceptualized as a sense of affective interpersonal involvement with media personalities.		media personalities		affective interpersonal involvement			media
Perse and Rubin (1989)	Parasocial interaction is a perceived interpersonal relationship on the part of a television viewer with a mass media persona.	television viewer	mass media persona		perceived interpersonal relationship			TV
Grant et al. (1991)	PSI is a relationship between viewers and television personalities.	viewers	television personalities		relationship			TV
Conway and Rubin (1991)	PSI: a one-way relationship or bond of intimacy that develops over time between audience members and media personae, has contained interpersonal attraction.	audience members	media personae		one-way relationship or bond of intimacy			
Kim and Rubin (1997)	PSI is an imaginary face-to-face relationship of audience members with media personalities	audience members	media personalities		imaginary face-to-face relationship			media
Skumanich and Kintsfather (1998)	PSI is the one-sided interpersonal involvement of a media user with a program character.	a media user	program character		one-sided interpersonal involvement			media

	The Re	eviewed PSI De	efinitions in the	Offline Env	ironment			
Authors	Definitions	One of the parties participating in PSI	The other parties participating in PSI	The nature of PSI	The main domain characteristics of PSI	The process of PSI (from viewer's perspective)	The time when PSI can occur	The platform where PSI can occur
Papa et al. (2000)	An audience member forms a relationship with a performer that is perceived as analogous to the interpersonal relationships of people in a primary, face-to-face group.	audience member	performer		relationship			
Greenwood (2008)	Parasocial interaction (or PSI) refers to the development of an imagined friendship with a media persona such that the viewer comes to feel that they ''know'' a media character or personality and forms an interpersonal attachment that is somewhat analogous to one they would form with an actual friend.	viewer	media persona		an imagined friendship			media
Greenwood and Long (2009)	"para-social interaction"— or the imagined intimacy that develops over time as audience members get to know a particular media persona.	audience members	media persona		imagined intimacy	over time		media
Moyer-Guse and Nabi (2010)	PSI is defined as a "seeming face-to- face relationship between spectator and performer"	spectator	performer		seeming face- to-face relationship			
Sun (2010)	Parasocial relationships are one-sided interpersonal relationships that "television viewers establish with media characters", allowing viewers to perceive a special connection with media characters.	television viewers	media characters		one-sided interpersonal relationships			TV

	The Re	eviewed PSI De	efinitions in the	Offline Env	ironment			
Authors	Definitions	One of the parties participating in PSI	The other parties participating in PSI	The nature of PSI	The main domain characteristics of PSI	The process of PSI (from viewer's perspective)	The time when PSI can occur	The platform where PSI can occur
Hartmann and Goldhoorn (2011)	This idea of parasocial interaction as the experience of TV viewers to be engaged in an immediate, personal, and reciprocal encounter with a TV performer.	TV viewers	a TV performer		an immediate, personal, and reciprocal encounter			TV
Lim and Kim (2011)	parasocial interaction is a pseudo- intimate relationship between the audience and media personalities such as talk radio hosts and TV shopping program hosts.	audience	media personalities		a pseudo- intimate relationship			radio or TV
So and Nabi (2013)	This imaginary friendship that develops between an audience and media personality is referred to as parasocial relationship, and the general communicative phenomenon that takes place in this relationship is called parasocial interaction.	audience	media personality		communication			media
Bentley (2014)	"Interpersonal involvement of the media user with what he or she consumes".	media user	he or she consumes		interpersonal involvement			media
Cummins and Cui (2014)	users' feeling of being involved in an interaction with the performer during media exposure	users	performer		being involved in an interaction		during media exposure	media
Tian and Yoo (2015)	the relationship in which "the viewer is engaged in a role relationship with a television persona"	viewer	television persona		relationship			TV

	The Re	eviewed PSI De	finitions in the	Offline Env	ironment			
Authors	Definitions	One of the parties participating in PSI	The other parties participating in PSI	The nature of PSI	The main domain characteristics of PSI	The process of PSI (from viewer's perspective)	The time when PSI can occur	The platform where PSI can occur
Knoll et al. (2015)	viewer responses to media personae as being composed of different cognitive, emotional, and/or behavioural processes. These processes follow on initial impression formation (or persona recognition), can emerge into different interaction patterns, can change dynamically within the course of media exposure, and are strongly influenced both by persona and viewer variables.	viewer	media personae		different cognitive, emotional, and/or behavioural processes			
So and Shen (2016)	parasocial interaction, an imaginary friendship between an audience and a character.	an audience	a character		imaginary friendship			
Oliver et al. (2019)	parasocial interactions are conceptualized as viewer perceptions while viewing– more akin to feelings of social engagement or "conversational give-and-take" with the character during the viewing experience itself.	viewer	character		perceptions		while viewing/ during the viewing experience itself	
Bi et al. (2021)	PSI is a pseudo-intimate relationship (Rubin & Step, 2000), and is based on the media effect of audiences developing illusionary unreciprocated connections to media personas (Levy & Windahl, 1984).	audiences	media personas		pseudo- intimate relationship, illusionary unreciprocated connections		media effect	

	The review	wed PSR defin	nitions in the of	fline envi	ronment			
Authors	Definitions	One of the parties participatin g in PSR	The other parties participating in PSR	The nature of PSR	The main domain characteristics of PSR	The process of PSR (from viewer's perspective)	The time when PSR can occur	The platform where PSR can occur
Stephens et al. (1996)	Feelings of friendship and intimacy on the viewer's part-with remote "personae" such as soap opera regulars, news anchors, and talk-show hosts.	viewer	personae such as soap opera regulars, news anchors, and talk- show hosts		feelings of friendship and intimacy			
Sun and Wu (2012)	Emotional attachment to sports celebrities (including those from a losing team) can be characterized as a parasocial relationship.		sports celebrities		emotional attachment			
Young et al. (2013)	Psychological connections with media figures, referred to as parasocial relationships.		media figures		psychological connections			
Hoffner and Cohen (2014)	A parasocial relationship (PSR) with a media figure reflects the sense of a real social bond in that the relationship is deeply felt and has many of the characteristics of "real" relationships.		media figure		sense of a real social bond			

#### Appendix 3: The reviewed PSR definitions in the offline environment

	The review	wed PSR defi	nitions in the of	filine envi	ironment			
Authors	Definitions	One of the parties participatin g in PSR	The other parties participating in PSR	The nature of PSR	The main domain characteristics of PSR	The process of PSR (from viewer's perspective)	The time when PSR can occur	The platform where PSR can occur
Behm- Morawitz et al. (2017)	Parasocial relationship refers to the formation of a one-way "pseudorelationship" in which the viewer imagines a face-to-face relationship with the performer that is unreciprocated.	viewer	performer		one- way/unrecipro cated/ imagines a face-to-face relationship			
Dias et al. (2017)	Viewer has the illusion of a face-to- face relationship with a fictional character or a celebrity (real person) and therefore feels part of a close, intimate relationship.	viewers	a fictional character or a celebrity (real person)		illusion face- to-face relationship			
Gabriel et al. (2018)	Parasocial relationships are one-sided psychological bonds with specific media figures such as favorite celebrities or fictional characters.		specific media figures such as favorite celebrities or fictional characters		one-sided psychological bonds			
Bernhold (2019)	Parasocial relationships are the durable and continuing bonds viewers form with television characters.	viewers	television characters		durable and continuing bonds			

		The reviewed P	SR definitions in	the offline	environment			
Authors	Definitions	One of the parties participating in PSR	The other parties participating in PSR	The nature of PSR	The main domain characteristics of PSR	The process of PSR (from viewer's perspective)	The time when PSR can occur	The platform where PSR can occur
Bernhold and Metzger (2020)	A parasocial relationship is a long-term mental schema about one's relationship with a character that is fueled by repeated parasocial interactions, or a "cross-situational, stable, and schematic cognitive pattern of images and interaction scripts that includes affective aspects".	one	character		a long-term mental shema/ relationship	repeated parasocial interactions, or a "cross- situational, stable, and schematic cognitive pattern of images and interaction scripts that includes affective aspects		
Bond (2021)	PSRs are affective bonds audiences foster with media characters and celebrities that last beyond episodic exposure (Dibble et al., 2016; Horton & Wohl, 1956).	audiences	media characters		affective bonds		last beyond episodic exposure	

	The revie	ewed PSI defini	tions in the onlin	e environ	iment			
Authors	Definitions	One of the parties participating in PSI	The other parties participating in PSI	The nature of PSI	The main domain characteristics of PSI	The process of PSI (from viewer's perspective)	The time when PSI can occur	The platform where PSI can occur
Thorson and Rodgers (2006)	Interpersonal involvement of the media user with what he or she consumes.	media user	what he or she consumes		interpersonal involvement			
Goldberg and Allen (2008)	Interpersonal involvement of the user with the Web site, and the extent to which users perceive themselves as interacting with a stable personality.	user	website		interpersonal involvement			website
Lee and Jang (2011)	"intimacy at a distance" (Horton & Wohl, 1956, p. 215), media audience develops the illusion of intimacy with media personalities, such as news anchor and TV show host, in the complete absence of direct interpersonal encounter.	media audience	media personalities		illusion of intimacy			
Keng et al. (2011)	PSI paradigm to describe one-sided, parasocial relationships between the mass media and audiences.	audiences	mass media		one-sided, parasocial relationships			

	The revie	ewed PSI defini	tions in the onlin	e environ	ment			
Authors	Definitions	One of the parties participating in PSI	The other parties participating in PSI	The nature of PSI	The main domain characteristics of PSI	The process of PSI (from viewer's perspective)	The time when PSI can occur	The platform where PSI can occur
Lueck (2012)	Audiences interact, relate to and develop relationships with a celebrity.	audiences	celebrity		relationship			
Colliander and Erlandsson (2013)	The illusion of a face-to-face relationship with a media performer.		media performer		illusion of a face-to-face relationship			
Lee (2013)	Feelings as imagined intimacy, personal interest, and attributional confidence.				Feelings as imagined intimacy, personal interest, and attributional confidence.			
Men and Tsai (2013) a	Audiences perceive having an intimate and personal relationship with media personalities, such as fictional TV characters and news hosts (Horton & Wohl, 1956; Russell & Stern, 2006).	audiences	media personalities		an intimate and personal relationship			

.

	The reviewed PSI definitions in the online environment										
Authors	Definitions	One of the parties participating in PSI	The other parties participating in PSI	The nature of PSI	The main domain characteristics of PSI	The process of PSI (from viewer's perspective)	The time when PSI can occur	The platform where PSI can occur			
Men and Tsai (2013) b	A user's interpersonal involvement with a media personality (including companies' SNS representatives) through mediated communication.	users	media personality		interpersonal involvement			mediated communi cation			
Tsai and Men (2013)	A user's interpersonal involvement with a media personality (including brands' SNS representatives) through mediated communication.	users	media personality		interpersonal involvement			mediated communi cation			
Labrecque (2014)	An illusionary experience, such that consumers interact with personas (i.e., mediated representations of presenters, celebrities, or characters) as if they are present and engaged in a reciprocal relationship.	consumers	personas (i.e., mediated representatio ns of presenters, celebrities, or characters)		illusionary experience						
Men and Tsai (2015)	Imagined intimacy and personal relationship with mediated characters.		mediated characters		imagined intimacy and personal relationship						

	The revie	ewed PSI defini	tions in the onlin	e enviror	ment			
Authors	Definitions	One of the parties participating in PSI	The other parties participating in PSI	The nature of PSI	The main domain characteristics of PSI	The process of PSI (from viewer's perspect ive)	The time when PSI can occur	The platform where PSI can occur
Chiu and Huang (2015)	Individuals' behavior in establishing and maintaining a relationship of friendship or intimacy with others using the SNSs.	individual	others		a relationship of friendship or intimacy		using SNS	SNS
Kim and Song (2016)	An imaginary relationship to a person whom one knows well but who barely knows the other.	one	person		imaginary relationship			
Yuksel and Labrecque (2016)	PSI occur during media exposure when media users respond to media personae in a quite similar way with real social interactions, whereas the relationships that emerge from such one-sided interactions are defined as PSR.	media users	media personae				during media exposure	
Xiang et al. (2016)	A one-sided relationship a user forms with other users on a SCP, especially with celebrities or experts, which stems from imagined intimacy or illusion.	user	other user (celebrities or experts)		one-sided relationship			social commerc e platform

	The revie	ewed PSI defini	tions in the onlin	e environ	iment			
Authors	Definitions	One of the parties participating in PSI	The other parties participating in PSI	The nature of PSI	The main domain characteristics of PSI	The process of PSI (from viewer's perspect ive)	The time when PSI can occur	The platform where PSI can occur
Kim et al. (2016)	Publics feel personally attached to and develop perceived personal relationships with public figures such as television characters (Rubin & McHugh, 1987).	publics	public figures		personally attached to and develop perceived personal relationships			
Lee and Watkins (2016)	Labrecque (2014) defines PSI as, "an illusionary experience, such that consumers interact with personas (i.e., mediated representations of presenters, celebrities, or characters) as if they are present and engaged in a reciprocal relationship".	consumers	personas		an illusionary experience engaged in a reciprocal relationship			
Sanz-Blas et al. (2017)	Audience intervention in TV shows/communication media and, thus, the relationship with the viewer and other members of the audience (Skumanich and Kintsfather, 1998).	audience	other members of the audience		relationship			TV shows/co mmunicat ion media

	The revie	ewed PSI defini	itions in the onlin	e enviror	iment			
Authors	Definitions	One of the parties participating in PSI	The other parties participating in PSI	The nature of PSI	The main domain characteristics of PSI	The process of PSI (from viewer's perspect ive)	The time when PSI can occur	The platform where PSI can occur
Tsai and Men (2017)	PSI refers to the media audience's perception of having an intimate and personal relationship with media personalities, such as TV characters, show hosts, and news anchors (Russell and Stern 2006).	audience	media personalities		perception of having an intimate and personal relationship			
Kim and Kim (2017)	On a website, people may experience a one-sided interaction, that is, parasocial interaction (PSI).	people			one-sided interaction			website
Gong and Li (2017)	Fans mostly engage in one-way interpersonal interaction with the celebrity, which is called "parasocial interaction".	fans	celebrity		one-way interpersonal interaction			
Hu et al. (2017)	An illusive sense of mutual awareness and intimacy with media personas (e.g. celebrities, news hosts, characters).		media personas (e.g. celebrities, news hosts, characters)		illusive sense of mutual awareness and intimacy			

	The reviewed PSI definitions in the online environment										
Authors	Definitions	One of the parties participating in PSI	The other parties participating in PSI	The nature of PSI	The main domain characteristics of PSI	The process of PSI (from viewer's perspect ive)	The time when PSI can occur	The platform where PSI can occur			
Lee and Lee (2017)	An intimate relationship between customers and the hotel.	customers	hotel		intimate relationship						
De Jans et al. (2018)	The relationship media users have with media personalities.		media personalities		relationship						
Lee (2018)	PSI refers to media-enabled connections between users and media personalities.	users	media personalities		media- enabled connections						
Daniel et al. (2018)	A viewer's one-way experience with a media personality is known as PSI when it feels like an actual interaction to the viewer.	viewer	media personality		one-way experience						
Phua et al. (2018)	A one-sided personal relationship that an individual develops with a media character.	individual	media character		one-sided personal relationship						

ŤΠ

	The revie	ewed PSI defini	tions in the onlin	e environ	ment			
Authors	Definitions	One of the parties participating in PSI	The other parties participating in PSI	The nature of PSI	The main domain characteristics of PSI	The process of PSI (from viewer's perspect ive)	The time when PSI can occur	The platform where PSI can occur
Jin (2018)	A media user develops the illusion of intimacy with media figures.	media user	media figures		illusion of intimacy			
Zhou and Jia (2018)	Virtual conversation that is stimulated by the website's brand attraction, media "persona," and communications strategy to improve the interaction quality between the consumer and the website information and make the consumer feel like he/ she is having a face-to- face dialogue (i.e., social presence).	consumer	website		a face-to-face dialogue			website
Breves et al. (2019)	The nonreciprocal cognitive, affective, and conative interactions of media users with the media personality.	media user	media personality		interaction			
Jin and Muqaddam (2019)	Parasocial interaction (PSI) refers to the feeling of companionship or illusion of friendship with media figures.		media figures		feeling of companionsh ip or illusion of friendship			

The reviewed PSI definitions in the online environment										
Authors	Definitions	One of the parties participating in PSI	The other parties participating in PSI	The nature of PSI	The main domain characteristics of PSI	The process of PSI (from viewer's perspect ive)	The time when PSI can occur	The platform where PSI can occur		
Choi et al. (2019)	Interactions among community members regarding shared interests are likely to evoke affective responses such as intimacy, friendship, and empathy.	community members	community members		interactions					
Liu et al. (2019)	PSI refers to the extent to which a media user perceives a media persona as an intimate social partner.	media user	media persona		relationship					
Shan et al. (2019)	The seeming face-to-face and one- sided quasi-interactions between a media viewer and a media character.	media viewer	media character		seeming face- to-face and one-sided quasi- interactions					
Jin and Ryu (2020)	PSI refers to a media user's "illusion of face-to-face relationship with a media personality".	media user	media personality		illusion of face-to-face relationship					
Zheng et al. (2020)	Interpersonal involvement of the media user with what he or she consumes.	media user	what he or she consumes		interpersonal involvement					

	The revie	ewed PSI defini	tions in the onlin	e enviror	iment			
Authors	Definitions	One of the parties participating in PSI	The other parties participating in PSI	The nature of PSI	The main domain characteristics of PSI	The process of PSI (from viewer's perspect ive)	The time when PSI can occur	The platform where PSI can occur
Aw and Labrecque (2020)	Parasocial interactions are one-sided relationships that individuals form with mediated persona, including celebrities.	individuals	mediated persona, including celebrities		one-sided relationships			
Sokolova and Kefi (2020)	The theory of para-social interaction (PSI) defines the relationship between a spectator and a performer with an illusion of intimacy as for the 'real' interpersonal relationships.	spectator	performer		relationship			
Boerman (2020)	A PSI is the illusion of having an intimate, personal relationship with a media personality		media personality		intimate and personal relationship			
Sakib et al. (2020)	PSI is the mechanism whereby a media consumer believes that the media character is directly communicating to him or her, resulting in a one-sided feeling of closeness.	media consumer	media character		feeling of closeness			

Ъъ

	The revie	ewed PSI defini	tions in the onlin	e environ	ment			
Authors	Definitions	One of the parties participating in PSI	The other parties participating in PSI	The nature of PSI	The main domain characteristics of PSI	The process of PSI (from viewer's perspect ive)	The time when PSI can occur	The platform where PSI can occur
Kim (2020)	PSI is defined as the illusionary relationship viewers develop with media performers.	viewers	media performers		illusionary relationship			
Kelton and Pennington (2020)	Parasocial interaction is a media user's perception of personal contact and interpersonal interaction with media figures in the absence of any direct interactions.	media user	media figures		perception		in the absence of any direct interactio ns	
Kim et al. (2020)	The interpersonal involvement of the media user with what he or she consumes.	media user	what he or she consumes		interpersonal involvement			
Handarkho (2020)	Emotional relationship where the form tends to be a one-way bond and usually occurs between individuals and their idols, such as a celebrity.	individuals	idols, such as a celebrity		emotional relationship			

	The revie	ewed PSI defini	tions in the onlin	e environ	ment			
Authors	Definitions	One of the parties participating in PSI	The other parties participating in PSI	The nature of PSI	The main domain characteristics of PSI	The process of PSI (from viewer's perspect ive)	The time when PSI can occur	The platform where PSI can occur
Nanda and Banerjee (2020)	A specific case of social interaction that happens in a one-sided way.				a specific case of social interaction in a one-sided way			
Ledbetter and Meisner (2021)	Viewer's sense of having a relationship with a media figure.	viewers	media figure		relationship			
Xu et al. (2021)	The intimacy between media users and media characters can be regarded as a one-sided interaction.	media users	media characters		Intimacy			
Kang et al. (2021)	Parasocial interaction refers to a self- defined, one-way relationship with a mediated personality (Brown, 2015).		mediated personality		self-defined, one-way relationship			
Yılmazdoğa n et al. (2021)	PSI is described as interaction of media users with people covered in the media (Grant et al., 1991; Rubin et al., 1985). This is an emotional relationship (Zhang et al., 2020).	media users	people		interaction/e motional relationship			media

	The revie	ewed PSI defini	tions in the onlin	e environ	ment			
Authors	Definitions	One of the parties participating in PSI	The other parties participating in PSI	The nature of PSI	The main domain characteristics of PSI	The process of PSI (from viewer's perspect ive)	The time when PSI can occur	The platform where PSI can occur
Youn and Jin (2021)	Parasocial interaction (PSI) refers to an illusionary mediated experience such that media users interact with media personas as if they are present and engaged in reciprocal relationships (Horton & Wohl, 1956).							
Handarkho (2021)	Para-social interaction is a relationship founded on emotions between an individual and an idolized or admired people such as influencers, respected person or celebrities (Sokolova and Kefi, 2020; Ward, 2016).	individual	an idolized or admired people such as influencers, respected person or celebrities		a relationship founded on emotions			
Kim and Kim (2021)	People experience a one-sided relationship when interacting with media personae in the mass media (Auter & Palmgreen, Citation2000; Horton & Wohl, 1956).	people	media personae		experience a one-sided relationship when interacting		in mediated encounte rs with performe rs	in the mass media

	The revie	ewed PSI defini	tions in the onlin	e enviror	iment			
Authors	Definitions	One of the parties participating in PSI	The other parties participating in PSI	The nature of PSI	The main domain characteristics of PSI	The process of PSI (from viewer's perspect ive)	The time when PSI can occur	The platform where PSI can occur
Chen et al. (2021)	Parasocial relationships, as a kind of psychological relationship experienced by an audience in mediated encounters with performers, definitely play an important role in interactive marketing (Labrecque, 2014).	audience	performers		psychological relationship		in mediated encounte rs	in mediated encounter s
Tsai et al. (2021)	PSI as a user's perceived interpersonal involvement with a media character, including brands' bot representatives, through mediated communication.	users	media character, including brands' bot representativ es		perceived interpersonal involvement			through mediated communi cation
Hsieh and Lee (2021)	Horton and Wohl (1956) posited that parasocial interaction is a "simulacrum of conversational give and take" (p. 215) that occurs between users and mass media performers.	users	mass media performers		simulacrum of conversationa l give and take			

'n

	The revie	ewed PSI defini	tions in the onlin	e environ	ment			
Authors	Definitions	One of the parties participating in PSI	The other parties participating in PSI	The nature of PSI	The main domain characteristics of PSI	The process of PSI (from viewer's perspect ive)	The time when PSI can occur	The platform where PSI can occur
Hoabin Ye et al. (2021)	Parasocial interaction refers to the imagined intimacy between media characters and an audience (Perse & Rubin, 1989).	audience	media character		imagined intimacy			
Jin et al. (2021)	PSI refers to the relationship viewers and audiences form with fictional or real characters presented in media outlets such as film, TV, radio, and so on.	viewers and audiences	fictional or real characters		relationship			media outlets such as film, TV, radio and so on
Lee et al. (2021)	PSI refers to a sense of mutual awareness that is limited to the duration of media consumption (Dibble et al., 2016).				mutual awareness		limited to the duration of media consump tion	

	The reviewed PSI definitions in the online environment											
Authors	Definitions	One of the parties participating in PSI	The other parties participating in PSI	The nature of PSI	The main domain characteristics of PSI	The process of PSI (from viewer's perspect ive)	The time when PSI can occur	The platform where PSI can occur				
Penttinen et al. (2022)	Parasocial interactions (Horton and Wohl 1956) or parasocial relationships (Rubin and McHugh 1987) are one-sided, imaginary connections that consumers experience with media personalities such as artists, online influencers, or even fictional characters (Hartmann and Goldhoorn 2011; Rubin and McHugh 1987).	consumers	media personalities such as artists, online influencer, or even fictional characters		one-sided, imaginary connections that consumers experience							
Yousaf (2022)	PSI is a relationship between media users and a media persona.	media users	media persona		relationship							
Hudders and De Jans (2022)	Parasocial interaction refers to the 'one-sided mediated form of social interaction between audience and media characters' (p. 5) (Liebers and Schramm, 2019).	audience	media characters		one-sided mediated form of social interaction							

	The reviewed PSI definitions in the online environment											
Authors	Definitions	One of the parties participating in PSI	The other parties participating in PSI	The nature of PSI	The main domain characteristics of PSI	The process of PSI (from viewer's perspect ive)	The time when PSI can occur	The platform where PSI can occur				
Lo et al. (2022)	PSI is defined as the digital interactions between viewers and social actors, conceptualized by the feeling of reciprocity that is comprised of mutual awareness, attention, and adjustment.	viewers	social actors		digital interaction, the feeling of reciprocity that is comprised of mutual awareness, attention, and adjustment.							
AW et al. (2022)	Consumers can generate a sense of social relationship with their digital voice assistants through frequent interactions.	consumers	digital voice assistants		a sense of social relationship		through frequent interactio ns					
Chen et al. (2022)	Audiences usually maintain a unidirectional attachment relationship with media characters, which is similar to but different from social relationships (Rasmussen, 2018).	audiences	media characters		unidirectional attachment relationship							

	The revie	ewed PSI defini	tions in the onlin	e enviror	nment			
Authors	Definitions	One of the parties participating in PSI	The other parties participating in PSI	The nature of PSI	The main domain characteristics of PSI	The process of PSI (from viewer's perspect ive)	The time when PSI can occur	The platform where PSI can occur
Wang and Hu (2022)	The process by which users attach themselves to and build relationships with media celebrities with whom they are not acquainted in real life (Ding and Qiu, 2017).	users	media celebrities		attach themselves to and build relationships			
Chang and Kim (2022)	Media such as video, television, radio, or movies give the illusion of having an actual relationship with the characters shown in media.		characters		give the illusion of having an actual relationship			media such as video, television , radio, or movies
Tseng et al. (2022)	Parasocial interaction as the illusive experience that is perceived by the audience that a "seemingly face-to- face friendship" Horton and Richard Wohl (1956).							
Kim (2022)	The concept of parasocial interaction (PSI) is defined as the illusionary relationships viewers develop with media characters (Horton and Wohl Citation1956).	viewer	media characters		illusionary relationships			

	The reviewed PSI definitions in the online environment											
Authors	Definitions	One of the parties participating in PSI	The other parties participating in PSI	The nature of PSI	The main domain characteristics of PSI	The process of PSI (from viewer's perspect ive)	The time when PSI can occur	The platform where PSI can occur				
Masuda et al. (2022)	Horton and Wohl (1956) defined PSR as the enduring relationships that users form with a mediated performer.	users	mediated performer		enduring relationships							
Zhang et al. (2022)	The illusion of face-to-face relationships between individuals and media figures (e.g., presenters, actors, or celebrities) Horton and Wohl (1956).											
Kim (2022)	PSI is referred to as the illusionary relationships viewers develop with media personae (Horton and Wohl, 1956).	viewers	media personae		the illusionary relationships							
Lee and Lee (2022)	Parasocial interactions can be defined as a viewer's perceived relationship with a media personality (Papa et al., 2000).	viewers	media personality		perceived relationship							
Deng et al. (2022)	A perceived short- or long-term social relationship (positive or negative) with a media figure (Giles, 2002).		media figure		short-term or long-term social relationship							

.

Appendix 4 (cont'd): The re	viewed PSI definitions	in the on	line environment
-----------------------------	------------------------	-----------	------------------

	The reviewed PSI definitions in the online environment											
Authors	Definitions	One of the parties participating in PSI	The other parties participating in PSI	The nature of PSI	The main domain characteristics of PSI	The process of PSI (from viewer's perspect ive)	The time when PSI can occur	The platform where PSI can occur				
Wei et al. (2022)	PSI can be defined as unilateral psychology associations that media audiences develop with media figures (Rubin and Step, 2000).	media audiences	media figures		unilateral psychology							
Manchanda et al. (2022)	As per the definition of Giles (2002, p. 279), "interaction between users of mass media and representations of humans appearing in the media ("media figures," such as presenters, actors, and celebrities) and can produce a form of relationship, to which the user responds as though in a typical social relationship".	users of mass media	media figures, such as presenters, actors and celebrities		a form of relationship, to which the users respond as though in a typical social relationship							
Liao et al. (2022)	Perceived parasocial interaction refers to the extent to which an audience perceives a media performer as an intimate conversational partner (Dibble, Hartmann, and Rosaen 2016; Horton and Wohl 1956).	audience	media performer		an intimate conversationa l partner							

	The review	wed PSR defini	tions in the onl	ine enviro	onment			
Authors	Definitions	One of the parties participating in PSR	The other parties participating in PSR	The nature of PSR	The main domain characteristics of PSR	The process of PSR (from viewer's perspective)	The time when PSR can occur	The platform where PSR can occur
Richards and Calvert (2015)	The deliverers of those rewards are at times popular and familiar media characters, with whom children have sometimes formed close, one-sided, emotionally tinged relationships called parasocial relationships.	children	popular and familiar media characters		close, one- sided, emotionally tinged relationships			
Tsiotsou (2015)	The one-sided interpersonal relationships developed between members of SNSs groups which may include cognitive, affective and behavioral responses.	members of SNSs groups	members of SNSs groups		one-sided interpersonal relationships			
Tsiotsou (2016)	The one-sided interpersonal relationships consumers develop with the members of a consumption community.	consumers	members of a consumption community		one-sided interpersonal relationships			
Yuan et al. (2016)	Parasocial relationships are psychological connections that users form unilaterally with media personalities/celebrities through virtual media.	users	media personalities/ celebrities		psychological connections			
Escalas and Bettman (2017)	Parasocial relationships are one-sided relationships that a media user establishes with a media figure.	media user	media figure		one-sided relationships			
Chung and Cho (2017)	Parasocial relationships refers to intimate relationships between audiences and celebrities.	audiences	celebrities		intimate relationships			

	The review	wed PSR defini	tions in the onl	ine enviro	onment			
Authors	Definitions	One of the parties participating in PSR	The other parties participating in PSR	The nature of PSR	The main domain characteristics of PSR	The process of PSR (from viewer's perspective)	The time when PSR can occur	The platform where PSR can occur
Hwang and Zhang (2018)	Parasocial relationships can be defined as unilateral relationships that a media audience develops with a media character.	media audience	media character		unilateral relationships			
Lim and Kim (2018)	A passive and unilateral relationship in which one party is known but the other party is not.	one party	the other party		unilateral relationship			
Iannone et al. (2018)	One-sided relationships with media figures.		media figures		one-sided relationships			
Han and Yang (2018)	The perception of imaginary in interpersonal relationships between people (viewers) and media characters.	people (viewers)	media characters		imaginary in interpersonal relationships			
de Bérail et al. (2019)	A parasocial relationship is a relationship an individual forms with someone he does not actually know by consuming media about that person.	individual	someone		relationship	by consuming media		
Munnukka et al. (2019)	Illusionary relationship and involvement with the vlogger.		vlogger		Illusionary relationship and involvement			

	The review	wed PSR defini	tions in the onl	ine envire	onment			
Authors	Definitions	One of the parties participating in PSR	The other parties participating in PSR	The nature of PSR	The main domain characteristics of PSR	The process of PSR (from viewer's perspective)	The time when PSR can occur	The platform where PSR can occur
Lee and Cho (2020)	A parasocial relationship is defined as the degree to which a media consumer perceives they have developed a social relationship with a media character.	media consumer	media character		social relationship			
Hu et al. (2020)	The emotional bonds that audiences form with media personas.	audiences	media personas		emotional bonds			
Zafar et al. (2020)	The media-enabled associations among individuals and celebrities.	individuals	celebrities		media-enabled associations			
Lim et al. (2020)	PSR, which can be defined as a one- sided intimate relationship that an audience feels toward a media performer, based on repeated encounters with the performer through mediated reality.	audience	media performer		one-sided intimate relationship	repeated encounters with the performer		mediated reality
Reinikainen et al. (2020)	PSRs are imaginary relationships with media performers that begin with spending time with the performer through media consumption and that are characterised by perceived relational development with the performer and knowing the performer well.		media performers		imaginary relationships	begin with spending time with the performer through media consumption		

	The review	wed PSR defini	tions in the onl	ine enviro	onment			
Authors	Definitions	One of the parties participating in PSR	The other parties participating in PSR	The nature of PSR	The main domain characteristics of PSR	The process of PSR (from viewer's perspective)	The time when PSR can occur	The platform where PSR can occur
Agnihotri and Bhattacharya (2020)	Mutual relationship characterized by intimacy and liking.				mutual relationship			
Yuan and Lou (2020)	A socioemotional bond between media personae and audiences.	media personae	audiences		socioemotiona 1 bond			
Sokolova and Perez (2021)	Para-social relationships can be defined as "illusory" relationships created between a spectator and a performer. Such relationships can be observed between celebrities and their fans that feel the proximity with their idols as in a real relationship.	spectator	performer		"illusory" relationships			
Yuan et al. (2021)	Emotional and long-term bonding between consumers and media characters can be defined as a parasocial relationship.	consumers	media characters		emotional and long-term bonding			
Feder (2020)	Strong one-sided connections "arise when individuals are repeatedly exposed to a media persona, and the individuals develop a sense of intimacy, perceived friendship, and identification with the [athlete]".	individuals	media persona (athlete)		one-sided connections	individuals are repeatedly exposed to a media persona		
	The reviewed PSR definitions in the online environment							
------------------------------	--	--	--	-------------------------	---	---	--	--
Authors	Definitions	One of the parties participating in PSR	The other parties participating in PSR	The nature of PSR	The main domain characteristics of PSR	The process of PSR (from viewer's perspective)	The time when PSR can occur	The platform where PSR can occur
Hwang et al. (2020)	A motive to create relationships that fans form with celebrities and in turn form a virtual intimacy in the sport media context.	fans	celebrities		relationships			
Farivar et al. (2021)	Parasocial relationship refers to a one- sided relationship an audience perceives with a media person.	audience	media person		one-sided relationship			
Utz et al. (2021)	Parasocial relationships are asymmetrical relationships that develop between media users and media figures.	media users	media figures		asymmetrical relationships			
Chen et al. (2021)	The imagined, one-sided quasi- relationship between a media user and a media character (Labrecque, 2014; Rubin & Step, 2010), with social media influencers.	media user	media character/soc ial media influencers		imagined, one- sided, quasi- relationship	encountered through media		media
Reinikainen et al. (2021)	A parasocial relationship refers to the illusion of a face-to-face relationship with a person who is encountered through media (Horton and Wohl, 1956).		person		illusion of a face-to-face relationship			
Aw and Chuah (2021)	Parasocial relationships denote one- sided relationships that a person develops with mediated personas,	person	mediated personas, including		one-sided relationships			

	The reviewed PSR definitions in the online environment							
Authors	Definitions	One of the parties participating in PSR	The other parties participating in PSR	The nature of PSR	The main domain characteristics of PSR	The process of PSR (from viewer's perspective)	The time when PSR can occur	The platform where PSR can occur
Yuan et al. (2021)	Parasocial relationship refers to intimate relationship between an actor and an audience.	audience	actor		Intimate relationship			
Whang and Im (2021)	A person engaged in a parasocial interaction feels reciprocity and rapport with the counterpart (Hartmann, 2008; Horton & Wohl, 1956), and gradually develops an illusory interpersonal relationship with the counterpart such as friendship and intimacy over time. This relationship is referred to as a parasocial relationship (Stern et al., 2015).	person	counterpart		illusory interpersonal relationship, such as friendship and intimacy	over time		
Breves et al. (2021)	PSR extends the concept by describing the cross-situational relationships between media characters and media users (Schramm Citation2008).	media users	media characters		cross-situational relationships			
Yuan et al. (2021)	Parasocial relationship with an AR application refers to an emotional bonding experience associated with the users' perception of the interactivity and vividness features of AR technology.	users	AR application		emotional bonding experienceuse rs' perception of the interactivity and vividness features of AR technology			

	The reviewed PSR definitions in the online environment							
Authors	Definitions	One of the parties participating in PSR	The other parties participating in PSR	The nature of PSR	The main domain characteristics of PSR	The process of PSR (from viewer's perspective)	The time when PSR can occur	The platform where PSR can occur
Yang and Ha (2021)	A parasocial relationship is defined as a pseudo friendship (Lee and Watkins, 2016) and psychological associations between an audience and a performer (Horton and Wohl, 1956).	audience	performer		pseudo friendship and psychological associations			
Tsfati et al. (2022)	Citizens' feelings of closeness and pseudo-friendships with politicians as similar to real-life relationships (Gabriel et al., 2018).	citizen	politicians		feelings of closeness and pseudo- friendship as similar to real- life relationships			

	The reviewed PSR definitions in the online environment							
Authors	Definitions	One of the parties participating in PSR	The other parties participating in PSR	The nature of PSR	The main domain characteristics of PSR	The process of PSR (from viewer's perspective)	The time when PSR can occur	The platform where PSR can occur
Balaban et al. (2022)	Usersrepresents a long-lasting relationship that could occur outside an exposure situation as a result of previous media exposure (Eyal & Dailey, 2012; Klimmt et al., 2006).	users			long-lasting relationship		as a result of previous media exposur e, could occur outside an exposur e situation	
Wahab et al. (2022)	PSR refers to followers' psychological association with SMCs, which drives them to establish a sense of familiarity and linkages (Zafar et al., 2020).	followers	SMCs		psychological association, a sense of familiarity and linkages			
Farivar et al. (2022)	Parasocial relationship is a bond-based attachment, in which followers consider that they are in a relationship with an influencer.	followers	influencer		bond-based attachment			
Lee and Park (2022)	An individual's perceived emotional intimacy with media personnel.	individual	media personnel		perceived emotional			

	The reviewed PSR definitions in the online environment							
Authors	Definitions	One of the parties participating in PSR	The other parties participating in PSR	The nature of PSR	The main domain characteristics of PSR	The process of PSR (from viewer's perspective)	The time when PSR can occur	The platform where PSR can occur
Sundermann and Munnukka (2022)	A parasocial relationship (PSR) is defined by an illusionary face-to-face relationship between a media performer and a media consumer (Horton and Wohl 1956).	media consumer	media performer		an illusionary face-to-face relationship			
Liao et al. (2022)	Perceived parasocial interaction refers to the extent to which an audience perceives a media performer as an intimate conversational partner (Dibble, Hartmann, and Rosaen 2016; Horton and Wohl 1956).	audience	media performer		intimate conversational partner			
Cheung et al. (2022)	Parasocial relationships are conceptualized as one-sided relationships that consumers develop with SMIs (Horton and Wohl, 1956).	consumers	SMIs		one-sided relationships			
Jacobson et al. (2022)	A parasocial relationship is a one-sided relationship that occurs between media users and people or characters who are influential on the media (Ballantine and Martin, 2005).	media users	people or characters who are influentical on the media		a one-dided relationship			media
Noor et al. (2022)	the illusionary bond that viewers form with characters played by performers in media such as television and theater.	viewers	characters		illusionary bond			media such as televisio n and theater

	The reviewed PSR definitions in the online environment							
Authors	Definitions	One of the parties participating in PSR	The other parties participating in PSR	The nature of PSR	The main domain characteristics of PSR	The process of PSR (from viewer's perspective)	The time when PSR can occur	The platform where PSR can occur
Leite and Baptista (2022)	PSR is a long-term relationship that can last beyond media exposure (Giles, 2002; Schramm & Hartmann, 2008; Tukachinsky & Sangalang, 2016).				long-term relationship		beyond media exposur e	
Burnasheva and Suh (2022)	Parasocial relationships are one-sided, psychological links that individuals establish with celebrities through virtual media (Horton and Wohl, 1956; Rubin and Step, 2000).	individuals	celebrities		one-sided, psychological links			through virtual media

### Appendix 6: Antecedents of PSR

	Antecedents of PSR				
	Antecedents	References			
	Amount of time spend on social media	Pressrove and Pardun (2016)			
	Motivations	Yuan et al. (2016); Lee and Cho, 2020; Zafar et al. (2020)			
	Need to belong	Escalas and Bettman (2017); Iannone et al. (2018)			
	Empathy	Hwang and Zhang (2018)			
	Self-esteem	Sun and Wu, 2012; Hwang and Zhang (2018)			
	Security	Han and Yang (2018)			
	Social anxiety	de Bérail et al. (2019)			
	Audience participation	Munnukka et al. (2019)			
	Wishful identification	Lim et al. (2020)			
Viewer-related	Emotional engagement	Lim et al. (2020)			
factors	Chronic ostracism	Iannone et al. (2018)			
	Exposure	Utz et al. (2021); Sokolova and Perez (2021); Gabriel et al. (2018)			
	Materialism	Sun and Wu (2012)			
	Interest	Sun and Wu (2012)			
	People contact	Hoffner and Cohen (2014)			
	Life-quality	Bernhold and Metzger (2020)			
	Friendship	Bernhold and Metzger (2020)			
	Attachment anxiety	Bernhold and Metzger (2020)			
	Self-disclosure	Chung and Cho (2017)			
	Self-presence	Farivar et al. (2022)			

# Appendix 6 (cont'd): Antecedents of PSR

Antecedents of PSR				
	Antecedents	References		
	Attractiveness	Han and Yang (2018); Sokolova and Perez (2021); Yuan and Lou (2020); Reinikainen et al. (2021); Masuda et al. (2022)		
	Similarity	Hu et al. (2020); Yuan and Lou (2020)		
	Expertise	Yuan and Lou (2020)		
	Identify distinctiveness	Hu et al. (2020)		
	Identity prestige	Hu et al. (2020)		
	Opinion leader	Quelhas-Brito et al. (2020)		
	Features endorser	Yuan et al. (2021)		
	Perceived human likeness	Whang and Im (2021)		
	Trustworthiness	Yuan and Lou (2020)		
The other party-	Influencer credibility	Yuan and Lou (2020)		
related factors	Endorser type	Agnihotri and Bhattacharya (2020)		
	Influencer coolness	Reinikainen et al. (2021)		
	Leverage	Yuan et al. (2021)		
	Fashionable	Yuan et al. (2021)		
	Creativity	Cheung et al. (2022)		
	Affinity	Yuan et al. (2021)		
	Social media celebrities	Wahab et al. (2022)		
	Non-followers vs followers	Breves et al. (2021)		
	Followers vs non-followers	Balaban et al. (2022)		
	Attitude homophily	Noor et al. (2022)		
	Influencers' intimate self-disclosure	Leite and Baptista (2022)		

# Appendix 6 (cont'd): Antecedents of PSR

	Antecedents of PSR				
	Antecedents	References			
	Features of purchasing platform	Yuan et al. (2021)			
	Perceived usability of blog	Quelhas-Brito et al. (2020)			
	Source credibility	Yuan et al. (2016)			
Other factors	Attitude toward product placement	Dias et al. (2017)			
	Design quality	Cheung et al. (2022)			
	Following length	Farivar et al. (2022)			
	Technology quality	Cheung et al. (2022)			

# Appendix 7: Consequences of PSR

Consequences of PSR				
	Consequences	References		
	Satisfaction	Han and Yang (2018)		
	Offline intention	Pressrove and Pardun (2016)		
	Stereotypes of mental illness	Hoffner and Cohen (2014)		
	Social distance	Hoffner and Cohen (2014)		
	Loneliness	Bernhold (2019)		
Viewer-related factors	Discouraging teen pregancy	Behm-Morawitz et al. (2019)		
	Depressive symtoms	Bernhold and Metzger (2020)		
	Belonging	Derrick et al. (2009)		
	Self-serving motive	Aw and Chuah (2021)		
	Flow experience	Yuan et al. (2021)		
	Subjective well-being	Noor et al. (2022)		

# Appendix 7 (cont'd): Consequences of PSR

	Consequences of PSR				
	Consequences	References			
	Forgiveness to the celebrity transgression	Finsterwalder et al. (2017)			
	Perceived credibility of the vlogger	Munnukka et al. (2019)			
	Wishful identification	Hu et al. (2020)			
	Repeated viewing	Lim et al. (2020)			
	Intention to watch influencers' video	Sokolova and Perez (2021)			
	Stickiness	Hu et al. (2020)			
	Perceived credibility of the vlogger	Munnukka et al. (2019)			
	Loyalty	Tsfati et al. (2022)			
	Influencer credibility	Reinikainen et al. (2020); Balaban et al. (2022)			
	Attitude to character	Gabriel et al. (2018)			
The other party-	Believing character's promises	Gabriel et al. (2018)			
related factors	Believing character's controversal statements	Gabriel et al. (2018)			
	Influencer coolness	Reinikainen et al. (2021)			
	Ronceptual persuasion knowledge	Breves et al. (2021)			
	Identification	Wahab et al. (2022)			
	Voting for the candidate	Tsfati et al. (2022)			
	Social relationships	Tsiotsou (2016); Tsiotsou (2015)			
	Voting behaviour	Gabriel et al. (2018)			
	Communication accuracy	Lee and Park (2022)			
	Communication credibility	Lee and Park (2022)			
	Communication competence	Lee and Park (2022)			
	Persuasive resistance	Breves et al. (2021)			

# Appendix 7 (cont'd): Consequences of PSR

	Consequences of PSR				
	Consequences	References			
	Attitude to brand	Agnihotri and Bhattacharya (2020); Lee and Cho (2020); Yuan et al. (2016); Reinikainen et al. (2021); Balaban et al. (2022)			
	Ad attitude	Lee and Cho (2020)			
	Attitude toward product placement	Breves et al. (2021)			
	Attitude toward branded content	Chen et al. (2021)			
	Customer equity drivers	Yuan et al. (2016)			
	Product attutude	Dias et al. (2017)			
	Product interest	Yuan and Lou (2020)			
	Evaluation	Whang and Im (2021); Utz et al. (2021)			
	Self-brand connection	Escalas and Bettman (2017)			
	Value equity	Yuan et al. (2021)			
Brand-related	Brand equity	Yuan et al. (2021)			
factors	Relationship equity	Yuan et al. (2021)			
	Engagement with branded content	Chen et al. (2021)			
	Problem engagement	Farivar et al. (2022)			
	Brand engagement	Zhong et al. (2021)			
	Online engagement	Pressrove and Pardun (2016)			
	Consume	Cheung et al. (2022)			
	Contribute	Cheung et al. (2022)			
	Create	Cheung et al. (2022)			
	Trust in branded content	Chen et al. (2021)			
	Brand trust	Reinikainen et al. (2020)			
	Source credibility	Sundermann and Munnukka (2022)			

# Appendix 7 (cont'd): Consequences of PSR

	Consec	quences of PSR					
	Consequences	References					
	Source trustwothiness	Chung and Cho (2017)					
	Continuance intention	Han and Yang (2018); Hwang et al. (2020)					
	Youtube addiction	de Bérail et al. (2019)					
	Electronic world-of-mouth	Hwang and Zhang (2018); Balaban et al. (2022)					
Brand-related factors	Purchase intention	Hwang and Zhang (2018); Farivar et al. (2021); Agnihotri and Bhattacharya (2020); Yang and Ha (2021); Wahab et al. (2022); Masuda et al. (2022); Leite and Baptista (2022); Breves et al. (2021); Balaban et al. (2022)					
	Urge to buy	Zafar et al. (2020)					
	Repeat purchase intention	Yuan et al. (2021)					
	Impulse buying	Zafar et al. (2020)					

### Appendix 8: Interview guide

#### Semi-Structured Interview Guide

#### Developing a measurement of parasocial relationship in the online environment

#### Introduction

- Thank the interviewee for taking part.
- Briefly introduce myself (the researcher).
- Remind the interviewee briefly of the topic and the study's context.
- Ensure the interviewee that there are no right or wrong answers, and that as the researcher I am interested in the interviewee's opinion.
- Inform the interviewee that the interview may take up to 60 minutes (approximately).
- Ask the interviewee if they have any questions before start.
- Ask the interviewee for permission to record the conversation. The interviews will be recorded for transcription, after which the recordings will be deleted.
- Turn on the recording.
- Reassure the interviewee that the data is kept only for research purposes, and is confidential, and will be anonymous. Remind the interviewee that they have the right to withdraw anytime without providing any reasons to me as the researcher, and in that case any data provided by the interviewee will be deleted.

#### **Questions:**

#### Part 1: Warm-up questions

Q1: Can you tell me what video-based social media you use? *Probing*: What do you like about this media site?

#### Part 2: Explore online users' parasocial relationship with influencers

Can you recall any influencers you follow on this media site and think of one of them?

Q2: Can you tell me about this influencer? (Influencer name/field) *Probing*: Can you give a more detailed description of this influencer?

Q3: Do you visit this influencer's social media account? If you do, for what reason? How often?

Q4: What do these videos/contents mean to you? Probing: How do you feel about the content of the video?

Q5: Some fans say that when they watch their favourite influencers' videos, they feel that the influencer is talking directly to them as if they are having a conversation. Have you experienced this yourself?

Probing: Do you have further examples of this?

Q6: Have you ever engaged or reacted to this influencer? *Probing*: Can you give some examples of how you interact with this influencer? (like/comment/share with others).

Q7: Have you had real/direct interaction with this influencer outside of the online environment? *Probing*: (If yes) Can you give a more detailed description of what happened?

### Appendix 8 (cont'd): Interview guide continued

#### Part 3: I would now like to know how you developed this relationship with this influencer.

Q8: Do you remember the occasion when you first noticed this influencer? *Probing*: Could you say something more about that?

Q9: What would you say was the reason that you decided to become one of the fans/followers of the influencer?

Probing: Can you give a more detailed description of what happened?

Q10: Has anything changed in your relationship with the influencer since you have become a (loyal) fan/follower? *Probing*: Could you say something more about that?

Q11: Why do you keep following this influencer? *Probing*: what does this mean to you? *Follow-up*: What would make you stop following this influencer?

Q12: Can you tell me the instances when you wanted to do or did something to help this influencer? *Probing*: Why did you want to help them?

What did you actually do when you helped this influencer?

Q13: In your opinion, how are fans influenced by their favourite influencers when it comes to buying influencer-endorsed products?

Probing: Have you ever purchased any products because this influencer endorsed them?

#### Part 4: The concept of parasocial relationship

Q14: In your own words, how would you describe your relationship with this influencer?

Q15: Some followers consider influencers they follow as their friends. Have you ever had the same feeling?

*Follow-up*: Do you think the relationship between you and this influencer is different from the relationship between you and your real-life friends/family members? *Probing*: Could you say something more about that?

#### Finally, we conclude this interview by asking a few questions about you.

#### Closure

• Seek the demographic information upon permission, including name, age, gender, years of internet use, years of social media use, and frequency of using social media (daily).

- Thank the interviewee once again.
- Turn the recording off.



College of Social Sciences

#### **Participant Information Sheet**

Study title: An exploration of the influence of online parasocial interaction and parasocial relationship on consumer behaviour

Researcher: Jie Sheng Student ID: School & Subject: Adam Smith Business School & Marketing Cluster PGR Programme Title: Management Supervisors: Dr. Alena Kostyk Email: Alena.Kostyk@glasgow.ac.uk Dr. Kalliopi Chatzipanagiotou Email: Kalliopi.Chatzipanagiotou@glasgow.ac.uk

You are being invited to take part in a research study. Before you decide to take part, it is important for you to understand why the research is being done and what it will involve. Please read the following information carefully and discuss it with others if you wish. Ask the researchers if there is anything that is not clear or if you would like more information. Take some time to decide whether or not you wish to take part.

Thank you for reading this.

#### FAQs

#### 1. What is the purpose of the study?

The purpose of this research is to investigate how consumers perceived the direct behaviour from the influencers and their long-term relationships with influencer influence online consumer behaviours. This research could benefit for participants in terms of helping participants reflect on their online behaviour and re-evaluate the impact of social networking on participants' lives. At the same time, participants will evaluate their relationship with online celebrities and influencers, and benefit for participants assessing the impact of online celebrities and influencers on their well-being and behaviours.

#### 2. Why have I been chosen?

You are invited to participate in this research project because the researcher is interested in knowing social media users' long-term relationships with influencers. You, as an online and social media user,

### Appendix 9 (cont'd): Participant information sheet

are the best person to participate in this research. Approximately 20 other participants will be recruited for the study.

#### 3. Do I have to take part?

Your participation is entirely voluntary: if you decide to take part and later change your mind, you are free to withdraw from the project at any time without prejudice, for example, to your job, studies or well-being and without providing a reason. In addition, your relationship with the researcher will not be affected by your decision to participate or not. There are no right or wrong answers. In the case of respondent withdrawal, the data provided by the respondent will be deleted.

#### 4. What will happen if I take part?

The interview will take approximately 45 minutes. Interviews will take place over Zoom. Interviews will only be audio-recorded with participants' consent. Before the interview, the researchers will ask the interviewee for permission to record the conversation (audio-recorded) and then turn on the recording. During the interview, you will be asked a series of open-ended questions about your experience with social media influencers. The discussion will be organized in a friendly, semistructured fashion and with some probe questions. The interview will NOT be video-recorded.

The interview will be audio-recorded. The audio file will be stored securely for the purposes of transcribing. Once the transcript of your interview is completed, the researchers will delete the audio file. The transcript will not be stored together with any of your personal data, there will be no personal identifiers (e.g., name) in the transcript file.

You will not be at physical or psychological risk and should experience no discomfort resulting from the research study methods.

#### 5. Will my taking part in this study be kept confidential?

Confidentiality will be maintained. All data will be collected anonymously, confidentially, and stored securely. Only the researcher will have access to the research materials, which will be kept on a password protected computer.

Only basic personal data will be collected – specifically, your first name and an email address – for the purposes of scheduling the Zoom interview. This information will then be discarded after the interview is completed. The transcript of your interview will not be stored together with any of your personal data, there will be no personal identifiers (e.g., name) in the transcript file.

Please note that assurances on confidentiality will be strictly adhered to unless evidence of wrongdoing or potential harm is uncovered. In such cases the University may be obliged to contact relevant statutory bodies/agencies.

### Appendix 9 (cont'd): Participant information sheet

#### 6. What will happen to the results of the research study?

The results of this research will be used to in PhD thesis in the University of Glasgow, future conference papers, or future journal publications, both print and online. Full confidentiality of your information will be kept by the researchers, that is, by all people helping with the study. Any information provided by you is only for academic use and will be kept securely: research data will be openly available. Interview transcripts CONTAINING NO PERSONAL INFORMATION of the participants will be stored electronically on a password-protected computer in encrypted files for 10 years after the completion of the research project. Participants should aware that data may be shared/archived or re-used, including PhD thesis, future conference papers, future journal publications, both print and online. Copy of this summary of results will be presented to you directly if you wish to request it (contact email: @student.gla.ac.uk).

#### 7. Who has reviewed the study?

This project has been considered and approved by the College Research Ethics Committee.

#### 8. Contact for Further information

If you have any complaints about any aspect of the project, the way it is being conducted or any questions about your rights as a research participant, then you may contact the researcher Jie Sheng at @student.gla.ac.uk or the supervisor Dr Alena Kostyk at Alena.Kostyk@glasgow.ac.uk and Dr Kalliopi Chatzipanagiotou at Kalliopi.Chatzipanagiotou@glasgow.ac.uk.

In addition, if you have any complaints regarding the conduct of this research project, you can also contact the College of Social Sciences Acting Lead for Ethical Review, Dr Benjamin Franks at socsciethics-lead@glasgow.ac.uk.

End of Participant Information Sheet

### Appendix 10: Consent form



College of Social Sciences

### **Consent Form**

Title of Project: An exploration of the influence of online parasocial interaction and parasocial relationship on consumer behaviour Name of Researcher: Jie Sheng

#### Please tick as appropriate

Yes		No		I confirm that I have read and understood the Participant Information Sheet for the above study and have had the opportunity to ask questions.
Yes		No		I confirm that I am 18 years old or over.
Yes		No		I understand that my participation is voluntary and that I am free to withdraw at any time, without giving any reason.
Yes		No		I consent to interviews being audio-recorded
Yes		No		I acknowledge that participants will be referred to by pseudonym.
l ag	ree	that		
Yes		No		All names and other material likely to identify individuals will be anonymised.
Yes		No		The material will be treated as confidential and kept in secure storage at all times.
Yes		No		The material may be used in PhD thesis, future conference papers, future journal publications, both print and online.
Yes		No		Other authenticated researchers will have access to this data only if they agree to preserve the confidentiality of the information as requested in this form.
Yes		No		Other authenticated researchers may use my words in publications, reports, web pages, and other research outputs, only if they agree to preserve the confidentiality of the information as requested in this form
Yes		No		I acknowledge the provision of a Privacy Notice in relation to this research project.
_				this research study  part in this research study
Nam	ne of	f Part	ticipant .	Signature
Date	e			
Nam	ne of	Res	earcher	Jie ShengSignature
Date	e			

# Appendix 11: Interviewee profile

						Interviewe	e list =25				
Name	Gender	Age	Occupation	Year s of use Inter net	Years of use social media	Time spends on social media (Daily)	Used video- based social media	Favourite Influencer(s)	Influencer field	Way of contact	Interview Duration
Interviewee 1	Female	30	Working full-time	20	13	4 hours	Bilibili	Pamela Reif	Fitness	Zoom	40 min
Interviewee 2	Male	26	Working part-time	15	8	3 hours	Bilibili TikTok	Silence Wang	Music	Zoom	38min
Interviewee 3	Female	28	Student	16	16	3 hours	Bilibili Weibo	Xiaochao Yuanzhang Oh_Emma	Entertainm ent; Lifestyle	Zoom	43min
Interviewee 4	Female	24	Student	10	10	6 hours	Bilibili Weibo	Xianxunheboluo you Kim Seokjin	Thought leaders	Zoom	45min
Interviewee 5	Male	38	Student	15	15	4 hours	TikTok YouTube Instagram	Jeremy Fragrance	Perfume	Zoom	45min
Interviewee 6	Female	31	Working full-time	23	12	1 hour	LinkedIn	Simon Sinek Taj Lavani	Entreprene ur	Zoom	32min
Interviewee 7	Female	26	Working full-time	10	10	5 hours	TikTok, Bilibili Weibo	Jiaqi Li	Beauty	Zoom	40min
Interviewee 8	Male	27	Student	20	10	2 hours	YouTube	Sam Chui The Detail Geek	Entertainm ent; Leisure	Zoom	39min

# Appendix 11 (cont'd): Interviewee profile

						Interviewe	e list =25				
Name	Gender	Age	Occupation	Year s of use Inter net	Years of use social media	Time spends on social media (Daily)	Used video- based social media	Favourite Influencer(s)	Influencer field	Way of contact	Interview Duration
Interviewee 9	Female	24	Student	10	10	2 hours	YouTube Bilibili TikTok	Oh_Emma Daoyueshe Jianyuerumian	Beauty; Lifestyle; Entertain ment	Zoom	45min
interviewee 10	Female	27	Student	20	19	3-4 hours	YouTube; Bilibili; Weibo	Wulie; Chizhihuoyan; E_volution	Art; Video games	Zoom	32 min
Interviewee 11	Female	30	Student	18	10	1 hour	WeChat moments	Yigongzi Ouyangchunxiao Jiaqi LI	History; Fitness; Beauty	Zoom	31min
Interviewee 12	Female	29	Working full-time	20	12	2-3 hours	WeChat moments; Bilibili; TikTok	Liurun; Cathy Chen	Managem ent; Marketing	Zoom	29 min
Interviewee 13	Female	35	Working full-time	24	17	45 min	TikTok, Instagram	Jillian Harris; Abbey Sharp	Lifestyle; Food and nutrition	Zoom	34 min
Interviewee 14	Female	29	Student	19	11	4 hours	Instagram; Facebook	Tessbegg; Nimai Delogado; Sherylsandberg	Vegan; Fitness; Female leader	Zoom	30min
Interviewee 15	Female	37	Working part-time	20	15	1-2 hours	YouTube	ouTube Lisa Eldrige; Claire Dim; Charlotte Tilburry		Zoom	28min
Interviewee 16	Male	25	Student	15	13	5 hours	Instagram; TikTok; YouTube	Farahrdara; Raditya Dika	Model; Writer	Zoom	23min

# Appendix 11 (cont'd): Interviewee profile

						Interviewe	e list =25				
Name	Gender	Age	Occupation	Year s of use Inter net	Years of use social media	Time spends on social media (Daily)	Used video- based social media	Favourite Influencer(s)	Influencer field	Way of contact	Interview Duration
Interviewee 17	Female	32	Student	24	15	A lot	Instagram	Vimoh; AndheriWestShit Posting; fraz_invsoutnutrit ion	Opinion leader; Food	Zoom	28min
Interviewee 18	Female	30	Working full-time	17	14	1-2 hours	Instagram; TikTok	Xenia Adonts; Carin Olsson (@parisinfourmo nths); Amanda du Pont	Lifestyle; Sustainabl e fashion	Zoom	30min
Interviewee 19	Male	45	Working full-time	27	21	1 hour	Instagram; TikTok	JK Rowling	Cook; Writer	Zoom	26min
Interviewee 20	Male	42	Working full-time	20	Over 10	2-3 hours	Instagram TikTok	Courtney Act Rupaul	Drag queens	Zoom	15min
Interviewee 21	Male	26	Working full-time	16	14	2-3 hours	TikTok; Bilibili	Hearthstone Valila; LOL DYS Tianyi Luo	Game; AI singer	Zoom	24min
Interviewee 22	Female	26	Working full-time	14	14	1-2 hours	Instagram			Zoom	25min
interviewee 23	Female	23	Student	13	Over 10	A lot	Instagram; YouTube			Zoom	26min

	Interviewee list =25														
Name	Gender	Age	Occupation	Year s of use Inter net	Years of use social media	Time spends on social media (Daily)	Used video- based social media	Favourite Influencer( s)	Influencer field	Way of contact	Interview Duration				
Interviewee 24	Male	26	Student	20	17	5 hours	YouTube; Bilibili	Andaxiong Greg Doucette	Psychotherapist; Fitness	Zoom	29min				
Interviewee 25	Male	23	Student	Over 10	Over 10	1-2 hours	YouTube; Instagram	Toress Pit Messi Marques	Lifestyle; Football; Technology	Zoom	21min				

No	Theme	Sub-theme	Initial Code			
		Accumulation of time	Accumulation of time			
		Habit	Become habit			
1	Time		Few visit			
		Interaction frequency	Visit every day			
			Visit every week			
			Influencer preference			
			Positive attitude to influencer			
			Respect to influencers			
			Get support from influencer			
		Emotion	Worship			
			Emotional attachment			
			Like influencer			
			Love influencer			
			Want to have fun			
			Want to find interesting things			
			Want to relax			
			Get discount			
			Want to get diverse information			
2	Psychological		Want to get latest information			
	closeness	Motivation	Want to get useful information			
			Want to learn new knowledge			
			Want to search information			
			Want to connect with friends			
			Want to get connection with home country			
			Want to get connection with the world			
			Want to see influencer undated posts			
			Same character			
			Same interest			
			Same opinion			
		Belongingness	Common people			
			Similarity			
			Same background			
			Same life journey			

### Appendix 12: Thematic analysis of the interview

		Thematic analysis	of the interview
No	Theme	Sub-theme	Initial Code
			Enjoyable
			Escape
			Fun
		Hedonic value	Funny
			Happiness
			Interesting
			Joy
			Kill time
			Relax
			Convenience to purchase products
			Get discount information
3	Value		Get influencer personal information
			Get influencer update information
		Utilitarian Value	Get influencer work-related information
		Otintarian value	Get product feedback
			Get useful information
			Insights
			Inspiration
			Learn new knowledge
			Reflect myself
		Symbolic value	Resonance
			Ideal self

Appendix 12 (cont'd): Thematic analysis of the interview

		Thematic analysis	of the interview
No	Theme	Sub-theme	Initial Code
			Care needs
		Influencer efforts to establish relationship	Update frequency
		Ĩ	Influencer self-exposure
		Follow influencer	Follow influencer
			Attend offline events
			Go to airport
			Create content
			Intention to support
			Let more people know this influencer
		Support influencer	Let other people follow this influencer
			Participate online campaign
			Become more popular and famous
			Purchase album
4	Inequality		Reward
			Rich
			Support job
			Support life
			Voice support
			Cognitive engagement
		Engagement with influencers online	Behavioural engagement
			Two-way engagement online
		Interest to influencer	Want to know more about the influencer

# Appendix 12 (cont'd): Thematic analysis of the interview

		Thematic analysis	of the interview
No	Theme	Sub-theme	Initial Code
			Tolerant of different opinion
		Influencer-related	Real interaction with influencers
		influential	Offline training
			Become real-life friend
			Brand preference
			Interest to influencer-created products
5	Influential		Positive attitude to product
5	mnuentiai		Purchase intention
		Brand-related influential	Purchase products
			Trust brand
			Try the brand
		Connection	Connection
6	Connection	Bonds	Bonds
	Connection	Relate	Relate
		Relationship	Relationship

# Appendix 12 (cont'd): Thematic analysis of the interview



# Study title: An exploration of the influence of online parasocial interaction and parasocial relationship on consumer behaviour

You are being invited to take part in a research study. Before you decide whether to take part, it is important that you to understand why the research is being done and what it will involve. Please take time to read the following information. Ask us if there is anything that is not clear or if you would like more information. Take time to decide whether or not you wish to take part. Thank you for reading this.

#### 1. What is the purpose of the study?

The purpose of this research is to investigate how parasocial interaction and parasocial relationship influence online consumer behaviours.

### 2. Do I have to take part?

Participation in this study is completely voluntary and you are free to stop participating at any time. In the case of respondent withdrawal, the data provided by the respondent will be deleted.

### 3. What does taking part involve?

You will be provided with a link to an online questionnaire. You are then asked to complete the questions to your best knowledge. There are no right or wrong answers. The questionnaire will take approx. 7 minutes. After completion, you are asked to return/submit the questionnaire. Answering "Yes" to the consent question, returning of the questionnaire, and clicking on the "Proceed to the Next Page" icon on the first page imply consent.

#### 4. Will my taking part in this study be kept confidential?

Please note that assurances on confidentiality will be strictly adhered to unless evidence of wrongdoing or potential harm is uncovered. In such cases, the University of Glasgow may be obliged to contact relevant statutory bodies/agencies.

#### 5. What will happen to the results of the research study?

The results of this research will be used to in PhD thesis in the University of Glasgow, future conference papers, or future journal publications.

### 6. Who has reviewed the study?

This project has been considered and approved by the College Research Ethics Committee. If you have any complaints regarding the conduct of this research project, you can also contact the College of Social Sciences Lead for Ethical Review, Dr Susan Batchelor at socsci-ethics-lead@glasgow.ac.uk.

#### **Researcher and Supervisor's Contact Details for Further Information:**

Researcher: Jie Sheng E-mail address: @student.gla.ac.uk Supervisor: Dr. Alena Kostyk, Senior Lecturer in Marketing E-mail: Alena.Kostyk@glasgow.ac.uk Supervisor: Dr. Kalliopi Chatzipanagiotou, Senior Lecturer in Marketing

E-mail: Kalliopi.Chatzipanagiotou@glasgow.ac.uk

Q1. Please indicate whether you consent to take part in this study:

[] I agree to take part in this research study

[] I do not agree to take part in this research study (this will terminate the study)

Please note that there are checks in this survey to assess whether you are reading the survey questions attentively.

Q2. What social media platform do you use the most in general?

- [ ] Instagram
- [] YouTube
- [] Facebook
- [ ] TikTok
- [] Twitter
- [] Reddit
- [ ] Other, please indicate

Q3. Please write below **ONE** name of your favourite **social media influencer** who promotes brands on social media platforms.

In this research, social media influencer refers to people who actively gain people's attention on multiple social media platforms and could influence their fans' behaviour in a particular field by generating texts, images, videos and live streaming.

Q4. Please choose the type of content this influencer typically posts:

- [] Fashion
- [ ] Health & Wellness
- [ ] Beauty
- [] Lifestyle
- [ ] Food
- [ ] Pet & Animal
- [ ] Sports & Fitness
- [] Travel
- [ ] Parenting
- [] Gaming
- [] Technology

- [] Entertainment
  [] Entrepreneurship
  [] Education
  [] News
  [] Book
  [] Science
  [] Chess
  [] Politics
  [] Cleaning
  [] Finance
  [] LGBTQ
  [] Watch
  [] Art & Music
  [] Other, please indicate
- No more than three months
   Three months to six months
   Six months to one year
   One to two years
   Two to five years
   More than five years

Q6. Please write below a name of ONE brand that this influencer promotes: []

#### Part two

Q7. How often do you view the messages (e.g., text, picture, or video) posted by your favourite influencer on social media?

(Su et al.,2015)

- [ ] Less than once per month
- [] Once per month
- [ ] Several time per month
- [ ] Once per week
- [ ] Several time per week
- [ ] Once per day
- [ ] Several times per day

Q8. Please answer the following questions thinking of this influencer.

Indicate to what extent you agree or disagree with the following statements.

There are no wrong or right answers; therefore, please choose the response you personally find the most appropriate.

When I am watching this influencer's social media content,

PSI (Hartmann and Goldhoorn, 2011)	1	2	3	4	5	6	7
I have the feeling that this influencer is							
aware of me.							
I have the feeling that this influencer							
knows I am there.							
I have the feeling that this influencer							
knows I am aware of them.							
I have the feeling that this influencer							
knows I pay attention to them.							
I have the feeling that this influencer							
knows that I react to them.							
I have the feeling that this influencer							
reacts to what I say or do.							

Q9. Please answer the following questions thinking of this influencer.

Indicate to what extent you agree or disagree with the following statements.

There are no wrong or right answers; therefore, please choose the response you personally find the most appropriate.

When I am watching this influencer's social media content,

Expertise (Ladhari et al., 2020)	1	2	3	4	5	6	7
This influencer is an expert in topics							
related to their content.							
This influencer is experienced in topics							
related to their content.							
This influencer is knowledgeable in							
topics related to their content.							
This influencer is qualified in topics							
related to their content.							

Q10. Please answer the following questions thinking of this influencer.

There are no wrong or right answers; therefore, please choose the response you personally find the most appropriate.

Please describe your overall evaluation of this influencer's social media content:

Perceived utilitarian	1	2	3	4	5	6	7	
value (Lou and Yuan,								
2019)								
Ineffective								Effective
Unhelpful								Helpful
Not functional								Functional
Unnecessary								Necessary
Impractical								Practical

Q11. Please answer the following questions thinking of this influencer.

There are no wrong or right answers; therefore, please choose the response you personally find the most appropriate.

Please describe your overall evaluation of this influencer's social media content:

Perceived utilitarian	1	2	3	4	5	6	7	
value (Lou and Yuan,								
2019)								
Not fun								Fun
Dull								Exciting
Not delightful								Delightful
Not thrilling								Thrilling
Unenjoyable								Enjoyable

Q12. Please answer the following questions thinking of this influencer.

Indicate to what extent you agree or disagree with the following statements.

There are no wrong or right answers; therefore, please choose the response you personally find the most appropriate.

1	2	3	4	5	6	7
	1	1 2	1     2     3	1     2     3     4	1     2     3     4     5	1     2     3     4     5     6

Q13. Please answer the following questions thinking of this influencer.

Indicate to what extent you agree or disagree with the following statements.

There are no wrong or right answers; therefore, please choose the response you personally find the most appropriate.

Trust (Alboqami, 2023)	1	2	3	4	5	6	7
This influencer can be relied upon on							
their content.							
I believe what this influencer says and							
that they would not try to take advantage							
of the followers.							
This influencer is straightforward and							
honest even though their self-interests							
are involved.							
This influencer would not tell a lie even							
if they could gain by it.							
This is a quality check. Please do not							
answer this question if you are reading							
this.							

Q14. Please answer the following questions thinking of this influencer.

Indicate to what extent you agree or disagree with the following statements.

There are no wrong or right answers; therefore, please choose the response you personally find the most appropriate.

PSR	1	2	3	4	5	6	7
This influencer makes me feel							
comfortable, as if I am with family.							
I think I understand this influencer quite							
well.							
I treat this influencer as a friend.							
I want to support this influencer.							
I would like to continue engaging with							
this influencer.							

If this influencer appeared on a different media platform, I would watch it.				
I would like to be friends with this influencer in real life.				
I would like to meet this influencer in				
person.				

Part three – PSR outcomes

Q15. Please answer the following questions thinking of this influencer.

Indicate to what extent you agree or disagree with the following statements.

There are no wrong or right answers; therefore, please choose the response you personally find the most appropriate.

Brand attitude (Torres et al., 2019)	1	2	3	4	5	6	7
This brand is good.							
This brand is pleasant.							
This brand is favourable.							

Q16. Please answer the following question thinking of this influencer and the brand.

There are no wrong or right answers; therefore, please choose the response you personally find the most appropriate.

<b>Purchase intention</b> (Gong and Li, 2017)	1	2	3	4	5	6	7
How likely would you be to purchase this brand endorsed by this influencer?							

Q17. Please answer the following questions thinking of this influencer.

Indicate to what extent you agree or disagree with the following statements.

There are no wrong or right answers; therefore, please choose the response you personally find the most appropriate.

Intention	to	support	(Kim	and	1	2	3	4	5	6	7
Drumwright	t, 20	16)									

I will support this influencer to see they				
succeed.				
I will support this influencer because I				
want to.				
This influencer's success will make me				
happy.				
I will support this influencer if they need				
help from fans like me.				
I feel good when I support this				
influencer.				
I pay attention to this influencer's needs.				
I care about this influencer's success.				

### Part four – ID

Please enter your Prolific ID:

# Appendix 14: Pre-screen study questions

	Pre-screen study questions	
No.	Question	Answer
1	Do you consent to participant this study?	Yes/No
2	Are you a social media user?	Yes/No
3	Do you follow influencers on social media?	Yes/No
4	Have you been following some influencers on social media for more than three months?	Yes/No
5	Are you still watching these influencers' updates?	Yes/No
# Appendix 15: PSR scale items initial pool

	PSR scale items initial pool			
No.	Scale items	Source		
1	I look forward to seeing the other party's updates.	Sokolova and Perez (2021)		
2	I want to learn more about the other party.	Iannone et al. (2018)		
3	If the other party appeared on a different media platform, I would watch it.	Escalas and Bettman (2017)		
4	I miss seeing the other party when they do not post an update.	Agnihotri and Bhattacharya (2020)		
5	I have the feeling that I know the other party very well.	Utz et al. (2021)		
6	I can feel the other party's emotions in certain situations.	Chung and Cho (2017)		
7	I think I understand the other party quite well.	Chung and Cho (2017)		
8	I treat the other party as a friend.	Richards and Calvert (2015)		
9	The other party makes me feel comfortable, as if I am with family.	Han and Yang (2018)		
10	The other party reminds me of myself.	Chung and Cho (2017)		
11	I would like to meet the other party in person.	Hu et al. (2020)		
12	I would like to follow what the other party is saying and doing.	Escalas and Bettman (2017)		
13	I feel fascinated by the other party.	Zafar et al. (2020)		
14	I feel sorry for the other party when they make a mistake.	Escalas and Bettman (2017)		
15	I can rely on information I get from the other party.	Hwang and Zhang (2018)		
16	When the other party shows me how they feel about something, it helps me make up my own mind about the issue.	Munnukka et al. (2019)		
17	I would like to be friends with the other party in real life.	Interview		
18	I want to support the other party.	Interview		
19	I would like to continue engaging with the other party.	Interview		
20	I feel connected with the other party.	Interview		
21	I feel close to the other party.	Interview		
22	I am emotionally attached to the other party.	Interview		
23	I worship the other party.	Interview		
24	The other party is similar to me.	Interview		
25	The other party reflects my ideal-self.	Interview		

	PSR scale items initial pool			
No.	Scale items	Source		
26	I know the other party cares about me like family.	Interview		
27	If given a chance, I would give virtual money to the other party.	Interview		
28	I can tolerate the other party's different opinion.	Interview		
29	The other party influenced my decisions.	Interview		
30	I think I have a distance relationship with the other party.	Interview		
31	I regard my connection with the other party as private.	Interview		
32	I can learn new information from the other party.	Interview		
33	I can get support from the other party.	Interview		
34	I can get encouragement from the other party.	Interview		
35	It is fun for me to watch or listen to the other party.	Interview		
36	I prefer to see this other party as opposed to others.	Interview		
37	I like to create content about the other party.	Interview		
38	I want the other party become more popular.	Interview		
39	I would like to purchase things created by the other party.	Interview		
40	The other party resonates with me.	Interview		

### Appendix 16: Experts survey

Dear Professor xxx,

I hope that this email finds you well.

My name is Jie Sheng, and I am a doctoral researcher at the University of Glasgow, Scotland, working under the supervision of Dr. Alena Kostyk and Dr. Kalliopi Chatzipanagiotou.

My PhD focuses on how online parasocial interactions and parasocial relationships influence consumer behaviour. As no appropriate parasocial relationship scale could be identified in the existing literature, I am developing one.

I am contacting you to ask for your help. I hope that as an expert in the field, you can help me develop an appropriate scale for the core construct of my thesis, parasocial relationship. Your response will be of great value in developing an appropriate scale to formally measure parasocial relationship.

If you choose to participate in the study, you will be invited to answer a few questions about the completeness and clarity of the definition of parasocial relationship, and the scale items that intend to measure it. The survey takes approximately 15 minutes to complete and has been approved by the Ethics board at the University of Glasgow.

Link to the survey: <u>https://uofg.qualtrics.com/jfe/form/SV\_3lwyovToTgvp3ue</u>

I thank you in advance for your kind help!

Kind regards, Jie

# Appendix 17: Experts comments about PSR working definition

Experts' comments about PSR working definition			
No.	Comments		
1	I did not find it particularly clear. It does not indicate explicitly that the exposure is to the other party, and it doesn't indicate explicitly that the value is solely to the focal individual. The definition also does not address potentially different motivations/perceptions of the connection. For example, there is no distinction between benefits accruing to the individual from other signaling (i.e., describing the other as in a relationship with the individual to the individual's friends) versus self-signaling (building up one's self-perception/self-worth by convincing oneself of the relationship).		
2	Psychological connection might be, I am not sure, a little bit narrow, can it cover all the cases?		
3	Very convoluted and incomplete.		
4	Does the repeated exposure need to be only through media or technology? As long as the other party is completely unaware of the other's existence it still is a parasocial relationship.		
5	What is "valuable"? Does it have to be "unilaterally developed"? There is debate in the field over this idea. What's the difference between "media" and "technology"? Overall, I think this definition lacks the necessary specificity.		

# Appendix 18: The summary of PSR scale items

	The summary of PSR Scale Items			
No.	Dimensions	Scale items	Source	Reason for deletion
1		I have the feeling that I know the other party very well.	Utz et al. (2021)	
2	Intimacy	I think I understand the other party quite well.	Chung and Cho (2017)	
3		The other party makes me feel comfortable, as if I am with family.	Han and Yang (2018)	
4		I want to support the other party.	Interview	
5	Relationship maintenance	I would like to continue engaging with the other party.	Interview	
6		If the other party appeared on a different media platform, I would watch it.	Escalas and Bettman (2017)	
7	Real-life	I would like to be friends with the other party in real life.	Interview	
8	relationship intention	I would like to meet the other party in person.	Hu et al. (2020)	
9		I look forward to seeing the other party's updates.	Sokolova and Perez (2021)	Deleted - First EFA
10		I am emotionally attached to the other party.	Interview	Deleted - First EFA
11		I want to learn more about the other party.	Iannone et al. (2018)	Deleted - First EFA
12		I feel close to the other party.	Interview	Deleted - First EFA
13		I worship the other party.	Interview	Deleted - First EFA
14		I treat the other party as a friend.	Richards and Calvert (2015)	Deleted - First EFA
15		I miss seeing the other party when they do not post an update.	Agnihotri and Bhattacharya (2020)	Deleted - First EFA

# Appendix 18 (cont'd): The summary of PSR scale items

	The summary of PSR Scale Items			
No.	Dimensions	Scale items	Source	Reason for deletion
16		I would like to follow what the other party is saying and doing.	Escalas and Bettman (2017)	Deleted - First EFA
17		It is fun for me to watch or listen to the other party.	Interview	Deleted - First EFA
18		I can feel the other party's emotions in certain situations.	Chung and Cho (2017)	Deleted - Second EFA
19		I feel connected with the other party.	Interview	Deleted - Second EFA
20		The other party resonates with me.	Interview	Deleted - Second EFA
21		I prefer to see this other party as opposed to others.	Interview	Deleted - Experts
22		I like to create content about the other party.	Interview	Deleted - Experts
23		I want the other party become more popular.	Interview	Deleted - Experts
24		I would like to purchase things created by the other party.	Interview	Deleted - Experts
25		I feel fascinated by the other party.	Zafar et al. (2020)	Deleted - Experts
26		I feel sorry for the other party when they make a mistake.	Escalas and Bettman (2017)	Deleted - Experts
27		I can rely on information I get from the other party.	Hwang and Zhang (2018)	Deleted - Experts
28		I can learn new information from the other party.	Interview	Deleted - Experts
29		I can get support from the other party.	Interview	Deleted - Experts

	The summary of PSR Scale Items			
No.	Dimensions	Scale items	Source	Reason for deletion
30		I can get encouragement from the other party.	Interview	Deleted - Experts
31		The other party reminds me of myself.	Chung and Cho (2017)	Deleted - Experts
32		The other party reflects my ideal-self.	Interview	Deleted - Experts
33		I know the other party cares about me like family.	Interview	Deleted -
34		If given a chance, I would give virtual money to the other party.	Interview	Experts Deleted - Experts
35	-	I can tolerate the other party's different opinion.	Interview	Deleted - Experts
36		The other party influenced my decisions.	Interview	Deleted - Experts
37		I think I have a distance relationship with the other party.	Interview	Deleted - Experts
38		I regard my connection with the other party as private.	Interview	Deleted - Experts
39		When the other party shows me how they feel about something, it helps me make up my own mind about the issue.	Munnukka et al. (2019)	Deleted - Experts
40		The other party is similar to me.	Interview	Deleted - Experts

### References

- Abratt, R. and Bendixen, M. (2018) 'The role of Brand Authenticity in developing Brand Trust', *Journal of Strategic Marketing*, 27(8), pp. 714–729. doi:10.1080/0965254x.2018.1466828.
- Achmad, G.N., Hudayah, S. and Ilmi, Z. (2020) 'The influence of hedonic value and utilitarian value on Brand Trust and Loyalty', *International Journal of Economics, Business and Accounting Research*, 4(03). doi:10.29040/ijebar.v4i03.1287.
- Agnihotri, A. and Bhattacharya, S. (2020) 'Endorsement effectiveness of celebrities versus social media influencers in the materialistic cultural environment of India', *Journal* of International Consumer Marketing, 33(3), pp. 280–302. doi:10.1080/08961530.2020.1786875.
- Agnihotri, D. *et al.* (2023) 'Investigating the impact of authenticity of social media influencers on followers' purchase behavior: Mediating Analysis of parasocial interaction on Instagram', *Asia Pacific Journal of Marketing and Logistics*, 35(10), pp. 2377–2394. doi:10.1108/apjml-07-2022-0598.
- Al-Ababneh, M.M. (2020) 'Linking Ontology, Epistemology and Research Methodology', Science & Philosophy, 8(1), pp. 75–91. doi:10.23756/sp.v8i1.500.
- Albaum, G. (1997) 'The Likert Scale revisited', *International Journal of Market Research*, 39(2), pp. 1–21. doi:10.1177/147078539703900202.
- Albayrak, T. *et al.* (2019) 'Customer loyalty towards travel agency websites: The role of trust and hedonic value', *Journal of Quality Assurance in Hospitality & Tourism*, 21(1), pp. 50–77. doi:10.1080/1528008x.2019.1619497.
- Alboqami, H. (2023) 'Trust me, I'm an influencer! causal recipes for customer trust in Artificial Intelligence Influencers in the retail industry', *Journal of Retailing and Consumer Services*, 72, p. 103242. doi:10.1016/j.jretconser.2022.103242.
- Aldridge, V.K., Dovey, T.M. and Wade, A. (2017) 'Assessing test-retest reliability of psychological measures', *European Psychologist*, 22(4), pp. 207–218. doi:10.1027/1016-9040/a000298.
- Al-Emadi, F.A. and Ben Yahia, I. (2020) 'Ordinary celebrities related criteria to harvest fame and influence on social media', *Journal of Research in Interactive Marketing*, 14(2), pp. 195–213. doi:10.1108/jrim-02-2018-0031.
- Allahverdi, F. (2022) 'Relationship between perceived social media addiction and social media applications frequency usage among university students'. *Psychology in the Schools*, 59(6), pp.1075-1087.
- Amanah, D. and Harahap, D.A. (2020) 'Visual appeal model for consumer online impulsive purchases in Indonesia'. *International Journal of Scientific & Technology Research*, 9(6), pp.388-397.

- Anderson, J.C. and Gerbing, D.W. (1988) 'Structural equation modeling in practice: A review and recommended two-step approach.', *Psychological Bulletin*, 103(3), pp. 411–423. doi:10.1037//0033-2909.103.3.411.
- Ao, L. *et al.* (2023) 'Impact of social media influencers on customer engagement and purchase intention: A meta-analysis', *Sustainability*, 15(3), p. 2744. doi:10.3390/su15032744.
- Arantola, H. (2002) 'Consumer bonding- a conceptual exploration', *Journal of Relationship Marketing*, 1(2), pp. 93–107. doi:10.1300/j366v01n02\_06.
- Archibald, M.M. (2015) 'Investigator triangulation', *Journal of Mixed Methods Research*, 10(3), pp. 228–250. doi:10.1177/1558689815570092.
- Archibald, M.M. *et al.* (2019) 'Using zoom videoconferencing for qualitative data collection: Perceptions and experiences of researchers and participants', *International Journal of Qualitative Methods*, 18, p. 160940691987459. doi:10.1177/1609406919874596.

Argyle, M. (2017) Social Interaction: Process and products. Routledge.

- Arsel, Z. (2017) 'Asking questions with reflexive focus: A tutorial on designing and conducting interviews', *Journal of Consumer Research*, 44(4), pp. 939–948. doi:10.1093/jcr/ucx096.
- Asghar, H.M. (2015) 'Measuring information seeking through Facebook: Scale Development and initial evidence of information seeking in Facebook Scale (ISFS)', *Computers in Human Behavior*, 52, pp. 259–270. doi:10.1016/j.chb.2015.06.005.
- Audrezet, A., de Kerviler, G. and Guidry Moulard, J. (2020) 'Authenticity under threat: When Social Media influencers need to go beyond self-presentation', *Journal of Business Research*, 117, pp. 557–569. doi:10.1016/j.jbusres.2018.07.008.
- Aust, F. et al. (2013) 'Seriousness checks are useful to improve data validity in online research', Behavior Research Methods, 45(2), pp. 527–535. doi:10.3758/s13428-012-0265-2.
- Aw, E.C.-X. and Chuah, S.H.-W. (2021) "stop the unattainable ideal for an ordinary me!" fostering parasocial relationships with social media influencers: The role of selfdiscrepancy', *Journal of Business Research*, 132, pp. 146–157. doi:10.1016/j.jbusres.2021.04.025.
- Aw, E.C.-X. and Labrecque, L.I. (2020) 'Celebrity endorsement in social media contexts: Understanding the role of parasocial interactions and the need to belong', *Journal of Consumer Marketing*, 37(7), pp. 895–908. doi:10.1108/jcm-10-2019-3474.
- Aw, E.C.-X. *et al.* (2022) 'Alexa, what's on my shopping list? transforming customer experience with Digital Voice assistants', *Technological Forecasting and Social Change*, 180, p. 121711. doi:10.1016/j.techfore.2022.121711.
- Babin, B.J., Darden, W.R. and Griffin, M. (1994) 'Work and/or fun: Measuring hedonic and utilitarian shopping value', *Journal of Consumer Research*, 20(4), p. 644. doi:10.1086/209376.

- Bagozzi, R.P., Gopinath, M. and Nyer, P.U. (1999) 'The role of emotions in marketing', *Journal of the Academy of Marketing Science*, 27(2), pp. 184–206. doi:10.1177/0092070399272005.
- Baker, J. (1995) The Need for Theory in Marketing. In: Baker M.J. et al. (eds). *Marketing Theory and Practice*. Palgrave, London.
- Balaban, D.C., Szambolics, J. and Chirică, M. (2022) 'Parasocial relations and Social Media Influencers' persuasive power. exploring the moderating role of product involvement', *Acta Psychologica*, 230, p. 103731. doi:10.1016/j.actpsy.2022.103731.
- Baldus, B.J., Voorhees, C. and Calantone, R. (2015) 'Online brand community engagement: Scale development and validation', *Journal of Business Research*, 68(5), pp. 978–985. doi:10.1016/j.jbusres.2014.09.035.
- Bales, R.F. (1950). The analysis of small group interaction. American Sociological Review, 15, pp.257-264
- Baumgartner, H. and Homburg, C. (1996) 'Applications of structural equation modeling in marketing and Consumer Research: A Review', *International Journal of Research in Marketing*, 13(2), pp. 139–161. doi:10.1016/0167-8116(95)00038-0.
- Baumgartner, H. and Steenkamp, J.-B.E.M. (2001) 'Response styles in marketing research: A cross-national investigation', *Journal of Marketing Research*, 38(2), pp. 143–156. doi:10.1509/jmkr.38.2.143.18840.
- Beavers, A. S., Lounsbury, J. W., Richards, J. K., Huck, S. W., Skolits, G. J. & Esquivel, S. L., (2013) 'Practical Considerations for Using Exploratory Factor Analysis in Educational Research', *Practical Assessment, Research, and Evaluation*, 18(1)
- Behera, R.K. et al. (2020) 'Personalized Digital Marketing Recommender engine', Journal of Retailing and Consumer Services, 53, p. 101799. doi:10.1016/j.jretconser.2019.03.026.
- Behm-Morawitz, E. *et al.* (2017) 'Examining the effects of MTV's *16 and pregnant* on adolescent girls' sexual health: The implications of character affinity, pregnancy risk factors, and health literacy on Message Effectiveness', *Health Communication*, 34(2), pp. 180–190. doi:10.1080/10410236.2017.1399506.
- Belanche, D. et al. (2021) 'Understanding influencer marketing: The role of congruence between influencers, products and consumers', *Journal of Business Research*, 132, pp. 186–195. doi:10.1016/j.jbusres.2021.03.067.
- Belch, G. and Belch, M., 2021. Advertising and promotion: an integrated marketing communications perspective. 12th ed. New York, NY: McGraw-Hill Education.
- Bentley, J.M. (2014) 'Best practices in noncommercial radio fundraising: A practitioner perspective', *International Journal of Nonprofit and Voluntary Sector Marketing*, 19(4), pp. 250–261. doi:10.1002/nvsm.1505.
- Berger, A. and Kiefer, M. (2021) 'Comparison of different response time Outlier Exclusion Methods: A Simulation Study', *Frontiers in Psychology*, 12. doi:10.3389/fpsyg.2021.675558.

- Bergkvist, L. and Rossiter, J.R. (2007) 'The predictive validity of multiple-item versus single-item measures of the same constructs', *Journal of Marketing Research*, 44(2), pp. 175–184. doi:10.1509/jmkr.44.2.175.
- Berkelaar, B.L. (2017) 'Different ways new information technologies influence conventional organizational practices and employment relationships: The case of cybervetting for personnel selection', *Human Relations*, 70(9), pp. 1115–1140. doi:10.1177/0018726716686400.
- Bernhold, Q.S. (2019) 'Parasocial relationships with disliked television characters, depressive symptoms, and loneliness among older adults', *Journal of Applied Communication Research*, 47(5), pp. 548–570. doi:10.1080/00909882.2019.1679384.
- Bernhold, Q.S. and Metzger, M. (2020) 'Older adults' parasocial relationships with favorite television characters and depressive symptoms', *Health Communication*, 35(2), pp. 168–179. doi:10.1080/10410236.2018.1548336.
- Bi, Y., Yin, J. and Kim, I. (2021) 'Fostering a young audience's media-induced travel intentions: The role of Parasocial Interactions', *Journal of Hospitality and Tourism Management*, 47, pp. 398–407. doi:10.1016/j.jhtm.2021.04.011.
- Boddy, C.R. (2016) 'Sample size for qualitative research', *Qualitative Market Research: An International Journal*, 19(4), pp. 426–432. doi:10.1108/qmr-06-2016-0053.
- Boehmer, J. (2016) 'Does the game really change? how students consume mediated sports in the age of Social Media', *Communication & Sport*, 4(4), pp. 460–483. doi:10.1177/2167479515595500.
- Boerman, S.C. (2020) 'The effects of the standardized Instagram disclosure for Micro- and Meso-influencers', *Computers in Human Behavior*, 103, pp. 199–207. doi:10.1016/j.chb.2019.09.015.
- Bollen, K.A. (1989) Structural equations with latent variables. New York u.a.: Wiley.
- Bond, B.J. (2021) 'The development and influence of parasocial relationships with television characters: A longitudinal experimental test of prejudice reduction through parasocial contact', *Communication Research*, 48(4), pp. 573–593. doi:10.1177/0093650219900632.
- Bonus, J.A., Matthews, N.L. and Wulf, T. (2021) 'The impact of moral expectancy violations on audiences' parasocial relationships with movie heroes and villains', *Communication Research*, 48(4), pp. 550–572. doi:10.1177/0093650219886516.
- Bos, W. and Tarnai, C. (1999) 'Content analysis in empirical social research', *International Journal of Educational Research*, 31(8), pp. 659–671. doi:10.1016/s0883-0355(99)00032-4.
- Braun, V. and Clarke, V. (2006) 'Using thematic analysis in psychology', *Qualitative Research in Psychology*, 3(2), pp. 77–101. doi:10.1191/1478088706qp0630a.

- Braun, V. and Clarke, V. (2020) 'One size fits all? what counts as quality practice in (reflexive) thematic analysis?', *Qualitative Research in Psychology*, 18(3), pp. 328– 352. doi:10.1080/14780887.2020.1769238.
- Braun, V. and Clarke, V. (2022) *Thematic analysis a practical guide*. Los Angeles, Calif: Sage.
- Braxton, D. and Lau-Gesk, L. (2020) 'The impact of collective brand personification on Happiness and Brand Loyalty', *European Journal of Marketing*, 54(10), pp. 2365– 2386. doi:10.1108/ejm-12-2019-0940.
- Brennan, Mike (1997), "The Effect of Question Tone on Responses to Open-Ended Questions," *Marketing Bulletin*, 8, pp.66-72.
- Breves, P. et al. (2021) 'Blind Trust? the importance and interplay of parasocial relationships and advertising disclosures in explaining influencers' persuasive effects on their followers', *International Journal of Advertising*, 40(7), pp. 1209–1229. doi:10.1080/02650487.2021.1881237.
- Breves, P. *et al.* (2021) 'Reducing resistance: The impact of nonfollowers' and followers' parasocial relationships with social media influencers on persuasive resistance and advertising effectiveness', *Human Communication Research*, 47(4), pp. 418–443. doi:10.1093/hcr/hqab006.
- Breves, P.L. *et al.* (2019) 'The perceived fit between Instagram influencers and the endorsed Brand', *Journal of Advertising Research*, 59(4), pp. 440–454. doi:10.2501/jar-2019-030.
- Bridges, E. and Florsheim, R. (2008) 'Hedonic and utilitarian shopping goals: The online experience', *Journal of Business Research*, 61(4), pp. 309–314. doi:10.1016/j.jbusres.2007.06.017.
- Brinkmann, S. (2022) *Qualitative interviewing: Conversational knowledge through research interviews*. New York, NY: Oxford University Press.
- Brosnan, K., Babakhani, N. and Dolnicar, S. (2019) "I know what you're going to ask me" why respondents don't read survey questions', *International Journal of Market Research*, 61(4), pp. 366–379. doi:10.1177/1470785318821025.
- Brown, W.J. (2015) 'Examining four processes of audience involvement with media personae: Transportation, Parasocial Interaction, identification, and worship', *Communication Theory*, 25(3), pp. 259–283. doi:10.1111/comt.12053.
- Bruggen, E. and Willems, P. (2009) 'A critical comparison of offline focus groups, online focus groups and E-delphi', *International Journal of Market Research*, 51(3), pp. 363–381. doi:10.2501/s1470785309200608.
- Bruhn, M., Schoenmueller, V. and Schäfer, D.B. (2012) 'Are social media replacing traditional media in terms of brand equity creation?', *Management Research Review*, 35(9), pp. 770–790. doi:10.1108/01409171211255948.
- Bryman, A. (2006) 'Integrating quantitative and qualitative research: How is it done?', *Qualitative Research*, 6(1), pp. 97–113. doi:10.1177/1468794106058877.

- Bryman, A. and Bell, E. (2011) 'Chapter 1: Business Research Strategies', in *Business* research methods. Oxford: Oxford Univ. Press.
- Buchanan, E.M. and Scofield, J.E. (2018) 'Methods to detect low quality data and its implication for psychological research', *Behavior Research Methods*, 50(6), pp. 2586–2596. doi:10.3758/s13428-018-1035-6.
- Buder, J. and Schwind, C. (2012) 'Learning with personalized recommender systems: A psychological view', *Computers in Human Behavior*, 28(1), pp. 207–216. doi:10.1016/j.chb.2011.09.002.
- Bun, P. et al. (2017) 'Low cost devices used in virtual reality exposure therapy', Procedia Computer Science, 104, pp. 445–451. doi:10.1016/j.procs.2017.01.158.
- Burnasheva, R. and Suh, Y.G. (2022) 'The moderating role of parasocial relationships in the associations between celebrity endorser's credibility and emotion-based responses', *Journal of Marketing Communications*, 28(4), pp. 343–359. doi:10.1080/13527266.2020.1862894.
- Byrne, B.M. (2016) Structural equation modeling with Amos: Basic Concepts, applications, and programming. 3rd edn. Routledge.
- Cairns-Lee, H., Lawley, J. and Tosey, P. (2021) 'Enhancing researcher reflexivity about the influence of leading questions in interviews', *The Journal of Applied Behavioral Science*, 58(1), pp. 164–188. doi:10.1177/00218863211037446.
- Campbell, C. and Farrell, J.R. (2020) 'More than meets the eye: The functional components underlying influencer marketing', *Business Horizons*, 63(4), pp. 469– 479. doi:10.1016/j.bushor.2020.03.003.
- Campbell, D.T. and Fiske, D.W. (1959) 'Convergent and discriminant validation by the multitrait-multimethod matrix.', *Psychological Bulletin*, 56(2), pp. 81–105. doi:10.1037/h0046016.
- Cao, J. et al. (2022) 'Influencing factors of users' shift to buying expensive virtual gifts in live streaming: Empirical evidence from China', Frontiers in Psychology, 13. doi:10.3389/fpsyg.2022.997651.
- Carpenter, S. (2017) 'Ten steps in scale development and reporting: A guide for researchers', *Communication Methods and Measures*, 12(1), pp. 25–44. doi:10.1080/19312458.2017.1396583.
- Casado-Aranda, L.-A., Martínez-Fiestas, M. and Sánchez-Fernández, J. (2018) 'Neural effects of environmental advertising: An fmri analysis of Voice Age and temporal framing', *Journal of Environmental Management*, 206, pp. 664–675. doi:10.1016/j.jenvman.2017.10.006.
- Casaló, L.V., Flavián, C. and Ibáñez-Sánchez, S. (2020) 'Influencers on Instagram: Antecedents and consequences of opinion leadership', *Journal of Business Research*, 117, pp. 510–519. doi:10.1016/j.jbusres.2018.07.005.
- Chae, H. *et al.* (2020) 'Impact of product characteristics of limited edition shoes on perceived value, Brand Trust, and purchase intention; focused on the scarcity

message frequency', *Journal of Business Research*, 120, pp. 398–406. doi:10.1016/j.jbusres.2019.11.040.

- Chang, D.R. and Kim, Q. (2022) 'A study on the effects of background film music valence on para-social interaction and consumer attitudes toward Social Enterprises', *Journal* of Business Research, 142, pp. 165–175. doi:10.1016/j.jbusres.2021.12.050.
- Chang, E.-C. and Tseng, Y.-F. (2013) 'Research note: E-store image, perceived value and perceived risk', *Journal of Business Research*, 66(7), pp. 864–870. doi:10.1016/j.jbusres.2011.06.012.
- Chang, H.H., Rizal, H. and Amin, H. (2013) 'The determinants of consumer behavior towards email advertisement', *Internet Research*, 23(3), pp. 316–337. doi:10.1108/10662241311331754.
- Chang, Y. and Thorson, E. (2023) 'Media multitasking, counterarguing, and brand attitude: Testing the mediation effects of advertising attention and cognitive load', *Computers in Human Behavior*, 139, p. 107544. doi:10.1016/j.chb.2022.107544.
- Chatterjee, S. et al. (2021) 'Cross-disciplinary issues in International Marketing: A Systematic Literature Review on International Marketing and Ethical Issues', *International Marketing Review*, 38(5), pp. 985–1005. doi:10.1108/imr-12-2020-0280.
- Chatzopoulou, G., Sheng, C. and Faloutsos, M. (2010) 'A first step towards understanding popularity in YouTube', 2010 INFOCOM IEEE Conference on Computer Communications Workshops [Preprint]. doi:10.1109/infcomw.2010.5466701.
- Chaudhuri, A. and Holbrook, M.B. (2001) 'The chain of effects from Brand Trust and brand affect to brand performance: The Role of Brand Loyalty', *Journal of Marketing*, 65(2), pp. 81–93. doi:10.1509/jmkg.65.2.81.18255.
- Chen, C. and Yao, M.Z. (2021) 'Strategic use of Immersive Media and narrative message in virtual marketing: Understanding the roles of Telepresence and Transportation', *Psychology & Marketing*, 39(3), pp. 524–542. doi:10.1002/mar.21630.
- Chen, K., Lin, J. (Elaine) and Shan, Y. (2021) 'Influencer marketing in China: The roles of Parasocial Identification, consumer engagement, and inferences of manipulative intent', *Journal of Consumer Behaviour*, 20(6), pp. 1436–1448. doi:10.1002/cb.1945.
- Chen, T.Y., Yeh, T.L. and Lee, F.Y. (2021) 'The impact of internet celebrity characteristics on followers' impulse purchase behavior: The mediation of attachment and Parasocial Interaction', *Journal of Research in Interactive Marketing*, 15(3), pp. 483–501. doi:10.1108/jrim-09-2020-0183.
- Chen, X., Hyun, S.S. and Lee, T.J. (2022) 'The effects of parasocial interaction, authenticity, and self-congruity on the formation of consumer trust in online travel agencies', *International Journal of Tourism Research*, 24(4), pp. 563–576. doi:10.1002/jtr.2522.
- Cheung, M.L. *et al.* (2022) "'I follow what you post!": The Role of Social Media Influencers' content characteristics in consumers' online brand-related activities

(Cobras)', *Journal of Retailing and Consumer Services*, 66, p. 102940. doi:10.1016/j.jretconser.2022.102940.

- Chi, T. (2018) 'Mobile Commerce website success: Antecedents of consumer satisfaction and purchase intention', *Journal of Internet Commerce*, 17(3), pp. 189–215. doi:10.1080/15332861.2018.1451970.
- Chiu, C.-M. and Huang, H.-Y. (2015) 'Examining the antecedents of user gratification and its effects on individuals' social network services usage: The moderating role of Habit', *European Journal of Information Systems*, 24(4), pp. 411–430. doi:10.1057/ejis.2014.9.
- Choi, S. *et al.* (2019) 'Travelers' parasocial interactions in online travel communities', *Journal of Travel & Tourism Marketing*, 36(8), pp. 888–904. doi:10.1080/10548408.2019.1657053.
- Chu, S. and Chen, H. (2019) 'Impact of consumers' corporate social responsibility-related activities in social media on brand attitude, electronic word-of-mouth intention, and purchase intention: A Study of Chinese Consumer Behavior', *Journal of Consumer Behaviour*, 18(6), pp. 453–462. doi:10.1002/cb.1784.
- Chung, S. and Cho, H. (2017) 'Fostering parasocial relationships with celebrities on social media: Implications for celebrity endorsement', *Psychology & Marketing*, 34(4), pp. 481–495. doi:10.1002/mar.21001.
- Churchill, G.A. (1979) 'A paradigm for developing better measures of marketing constructs', *Journal of Marketing Research*, 16(1), p. 64. doi:10.2307/3150876.
- Claffey, E. and Brady, M. (2014) 'A model of consumer engagement in a virtual customer environment', *Journal of Customer Behaviour*, 13(4), pp. 325–346. doi:10.1362/147539214x14185702584405.
- Claffey, E. and Brady, M. (2019) 'An empirical study of the impact of consumer emotional engagement and affective commitment in firm-hosted Virtual Communities', *Journal* of Marketing Management, 35(11–12), pp. 1047–1079. doi:10.1080/0267257x.2019.1601125.
- Clark, M.S. and Waddell, B. (1985) 'Perceptions of exploitation in communal and exchange relationships', *Journal of Social and Personal Relationships*, 2(4), pp. 403–418. doi:10.1177/0265407585024002.
- Claycomb, C., Porter, S.S. and Martin, C.L. (2000) 'Riding the wave: Response Rates and the Effects of Time Intervals between Successive Mail Survey Follow-up Efforts', *Journal of Business Research*, 48(2), pp. 157–162. doi:10.1016/s0148-2963(98)00098-8.
- Cloutier, J. *et al.* (2008) 'Are attractive people rewarding? sex differences in the neural substrates of facial attractiveness', *Journal of Cognitive Neuroscience*, 20(6), pp. 941–951. doi:10.1162/jocn.2008.20062.
- Coates, A.E. *et al.* (2020) "it's just addictive people that make addictive videos": Children's understanding of and attitudes towards influencer marketing of food and

beverages by YouTube video bloggers', *International Journal of Environmental Research and Public Health*, 17(2), p. 449. doi:10.3390/ijerph17020449.

- Cobanoglu, C. and Cobanoglu, N. (2003) 'The effect of incentives in web surveys: Application and ethical considerations', *International Journal of Market Research*, 45(4), pp. 1–13. doi:10.1177/147078530304500406.
- Cohen, J. (1988) *Statistical Power Analysis for the behavioral sciences*. Hillsdale, NJ: L. Erlbaum Associates.
- Coker, K.K., Smith, D.S. and Altobello, S.A. (2015) 'Buzzing with disclosure of social shopping rewards', *Journal of Research in Interactive Marketing*, 9(3), pp. 170–189. doi:10.1108/jrim-06-2014-0030.
- Colliander, J. and Erlandsson, S. (2013) 'The blog and the bountiful: Exploring the effects of disguised product placement on blogs that are revealed by a third party', *Journal of Marketing Communications*, 21(2), pp. 110–124. doi:10.1080/13527266.2012.730543.
- Collier, J.E. (2020) *Applied structural equation modeling using amos: Basic to advanced techniques.* New York: Routledge.
- Conde, R. and Casais, B. (2023) 'Micro, Macro and mega-influencers on Instagram: The power of persuasion via the parasocial relationship', *Journal of Business Research*, 158, p. 113708. doi:10.1016/j.jbusres.2023.113708.
- Conway, J.C. and Rubin, A.M. (1991) 'Psychological predictors of television viewing motivation', *Communication Research*, 18(4), pp. 443–463. doi:10.1177/009365091018004001.
- Côté, J. (1999) 'The influence of the family in the development of talent in sport', *The Sport Psychologist*, 13(4), pp. 395–417. doi:10.1123/tsp.13.4.395.
- Couper, M.P. (2011) 'The future of modes of data collection', *Public Opinion Quarterly*, 75(5), pp. 889–908. doi:10.1093/poq/nfr046.
- Crick, J.M. (2020) 'Qualitative research in marketing: What can academics do better?', *Journal of Strategic Marketing*, 29(5), pp. 390–429. doi:10.1080/0965254x.2020.1743738.
- Croes, E. and Bartels, J. (2021) 'Young adults' motivations for following social influencers and their relationship to identification and buying behavior', *Computers in Human Behavior*, 124, p. 106910. doi:10.1016/j.chb.2021.106910.
- Cummins, R.G. and Cui, B. (2014) 'Reconceptualizing address in television programming: The effect of address and affective empathy on viewer experience of parasocial interaction', *Journal of Communication*, 64(4), pp. 723–742. doi:10.1111/jcom.12076.
- Curran, P.G. (2016) 'Methods for the detection of carelessly invalid responses in Survey Data', *Journal of Experimental Social Psychology*, 66, pp. 4–19. doi:10.1016/j.jesp.2015.07.006.

- Dabbous, A. and Barakat, K.A. (2020) 'Bridging the online offline GAP: Assessing the impact of brands' Social Network Content Quality on brand awareness and purchase intention', *Journal of Retailing and Consumer Services*, 53, p. 101966. doi:10.1016/j.jretconser.2019.101966.
- Daniel, E.S., Crawford Jackson, E.C. and Westerman, D.K. (2018) 'The influence of social media influencers: Understanding online vaping communities and parasocial interaction through the lens of Taylor's six-segment Strategy Wheel', *Journal of Interactive Advertising*, 18(2), pp. 96–109. doi:10.1080/15252019.2018.1488637.
- Dastan, I. and Gecti, F. (2014) 'Relationships among utilitarian and hedonic values, brand affect and brand trust in the smartphone industry', *Journal of Management Research*, 6(2), p. 124. doi:10.5296/jmr.v6i2.5261.
- Daugherty, T. and Hoffman, E. (2013) 'EWOM and the importance of capturing consumer attention within social media', *Journal of Marketing Communications*, 20(1–2), pp. 82–102. doi:10.1080/13527266.2013.797764.
- Dawadi, S., Shrestha, S. and Giri, R.A. (2021) 'Mixed-methods research: A discussion on its types, challenges, and criticisms', *Journal of Practical Studies in Education*, 2(2), pp. 25–36. doi:10.46809/jpse.v2i2.20.
- Dawes, J. (2008) 'Do data characteristics change according to the number of scale points used? an experiment using 5-point, 7-point and 10-point scales', *International Journal of Market Research*, 50(1), pp. 61–104. doi:10.1177/147078530805000106.
- de Bérail, P., Guillon, M. and Bungener, C. (2019) 'The relations between YouTube addiction, social anxiety and parasocial relationships with youtubers: A moderatedmediation model based on a cognitive-behavioral framework', *Computers in Human Behavior*, 99, pp. 190–204. doi:10.1016/j.chb.2019.05.007.
- de Groot, J.I. (2022) 'The personalization paradox in Facebook advertising: The mediating effect of relevance on the personalization–brand attitude relationship and the moderating effect of intrusiveness', *Journal of Interactive Advertising*, 22(1), pp. 57–74. doi:10.1080/15252019.2022.2032492.
- De Jans, S., Cauberghe, V. and Hudders, L. (2018) 'How an advertising disclosure alerts young adolescents to sponsored vlogs: The moderating role of a peer-based advertising literacy intervention through an informational vlog', *Journal of Advertising*, 47(4), pp. 309–325. doi:10.1080/00913367.2018.1539363.
- Derrick, J.L., Gabriel, S. and Hugenberg, K. (2009) 'Social surrogacy: How favored television programs provide the experience of belonging', *Journal of Experimental Social Psychology*, 45(2), pp. 352–362. doi:10.1016/j.jesp.2008.12.003.
- De Toni, D. *et al.* (2021) 'The symbolic value of wine, moderating and mediating factors and their relationship to consumer purchase intention', *International Journal of Wine Business Research*, 34(2), pp. 190–211. doi:10.1108/ijwbr-01-2021-0006.
- De Veirman, M., Cauberghe, V. and Hudders, L. (2017) 'Marketing through Instagram influencers: The impact of number of followers and product divergence on brand attitude', *International Journal of Advertising*, 36(5), pp. 798–828. doi:10.1080/02650487.2017.1348035.

- de Vries, L., Gensler, S. and Leeflang, P.S.H. (2017) 'Effects of traditional advertising and social messages on brand-building metrics and customer acquisition', *Journal of Marketing*, 81(5), pp. 1–15. doi:10.1509/jm.15.0178.
- Delgado-Ballester, E. and Luis Munuera-Alemán, J. (2005) 'Does brand trust matter to brand equity?', *Journal of Product & Brand Management*, 14(3), pp. 187–196. doi:10.1108/10610420510601058.
- DelVecchio, D., Henard, D.H. and Freling, T.H. (2006) 'The effect of sales promotion on post-promotion Brand Preference: A meta-analysis', *Journal of Retailing*, 82(3), pp. 203–213. doi:10.1016/j.jretai.2005.10.001.
- Dencheva, V. (2024) *Global Influencer Market Size 2024, Statista*. Available at: https://www.statista.com/statistics/1092819/global-influencer-market-size/ (Accessed: 22 March 2024).
- Deng, Z., Benckendorff, P. and Wang, J. (2022) 'From interaction to relationship: Rethinking parasocial phenomena in travel live streaming', *Tourism Management*, 93, p. 104583. doi:10.1016/j.tourman.2022.104583.
- Dennis, J.M. 2001, "Are Internet Panels Creating Professional Respondents?", Marketing research, 13(2), pp. 34.
- Dias, J.A., Dias, J.G. and Lages, C. (2017) 'Can negative characters in soap operas be positive for product placement?', *Journal of Business Research*, 71, pp. 125–132. doi:10.1016/j.jbusres.2016.10.010.
- Dibble, J.L., Hartmann, T. and Rosaen, S.F. (2015) 'Parasocial interaction and parasocial relationship: Conceptual clarification and a critical assessment of measures', *Human Communication Research*, 42(1), pp. 21–44. doi:10.1111/hcre.12063.
- Donabedian, T. (2021) 'They're worth my investment': Cultivating intimacy through fanlead financial and support initiatives among BTS fans', *Asia Marketing Journal*, 22(4). doi:10.53728/2765-6500.1370.
- Donath, J. (2007) 'Signals in Social Supernets', *Journal of Computer-Mediated Communication*, 13(1), pp. 231–251. doi:10.1111/j.1083-6101.2007.00394.x.
- Douglas, B.D., Ewell, P.J. and Brauer, M. (2023) 'Data quality in online human-subjects research: Comparisons between Mturk, prolific, CloudResearch, Qualtrics, and Sona', *PLOS ONE*, 18(3). doi:10.1371/journal.pone.0279720.
- Duff, B.R.-L. and Sar, S. (2014) 'Seeing the big picture: Multitasking and Perceptual Processing Influences on ad recognition', *Journal of Advertising*, 44(3), pp. 173–184. doi:10.1080/00913367.2014.967426.
- Dunn, T.J., Baguley, T. and Brunsden, V. (2013) 'From alpha to omega: A practical solution to the pervasive problem of internal consistency estimation', *British Journal* of Psychology, 105(3), pp. 399–412. doi:10.1111/bjop.12046.
- Easterby-Smith, M. et al. (2018) 'Chapter 3: The philosophy of management and business research', in *Management & Business Research*. Los Angeles: SAGE, pp. 46–65.

Easterby-Smith, M. et al. (2021) Management and business research. London: Sage.

- EBSCO (2023) Ultimate databases: EBSCO, EBSCO Information Services, Inc. | www.ebsco.com. Available at: https://www.ebsco.com/products/researchdatabases/ultimate-databases (Accessed: 01 October 2023).
- Ellis, R.J. and Holmes, J.G. (1982) 'Focus of attention and self-evaluation in social interaction.', *Journal of Personality and Social Psychology*, 43(1), pp. 67–77. doi:10.1037//0022-3514.43.1.67.
- Ellison, N.B. *et al.* (2014) 'Cultivating social resources on social network sites: Facebook Relationship Maintenance Behaviors and their role in social capital processes', *Journal of Computer-Mediated Communication*, 19(4), pp. 855–870. doi:10.1111/jcc4.12078.
- Escalas, J.E. and Bettman, J.R. (2017) 'Connecting with celebrities: How consumers appropriate celebrity meanings for a sense of belonging', *Journal of Advertising*, 46(2), pp. 297–308. doi:10.1080/00913367.2016.1274925.
- Esteban-Santos, L. et al. (2018) 'Fashion bloggers: Communication tools for the Fashion Industry', Journal of Fashion Marketing and Management: An International Journal, 22(3), pp. 420–437. doi:10.1108/jfmm-10-2017-0101.
- Etemad-Sajadi, R. and Ghachem, L. (2015) 'The impact of hedonic and utilitarian value of online avatars on E-service quality', *Computers in Human Behavior*, 52, pp. 81–86. doi:10.1016/j.chb.2015.05.048.
- Evanschitzky, H. and Armstrong, J.S. (2013) 'Research with in-built replications: Comment and further suggestions for replication research', *Journal of Business Research*, 66(9), pp. 1406–1408. doi:10.1016/j.jbusres.2012.05.006.
- Fang, J., Wen, C., George, B. and Prybutok, V.R. (2016). 'Consumer heterogeneity, perceived value, and repurchase decision-making in online shopping: The role of gender, age, and shopping motives'. *Journal of Electronic Commerce Research*, 17(2), pp.116-131.
- Fang, S., Zhang, C. and Li, Y. (2020) 'Physical attractiveness of service employees and customer engagement in tourism industry', *Annals of Tourism Research*, 80, p. 102756. doi:10.1016/j.annals.2019.102756.
- Farivar, S., Wang, F. and Turel, O. (2022) 'Followers' problematic engagement with influencers on social media: An attachment theory perspective', *Computers in Human Behavior*, 133, p. 107288. doi:10.1016/j.chb.2022.107288.
- Farivar, S., Wang, F. and Yuan, Y. (2021) 'Opinion leadership vs. para-social relationship: Key factors in Influencer marketing', *Journal of Retailing and Consumer Services*, 59, p. 102371. doi:10.1016/j.jretconser.2020.102371.
- Farrington-Darby, T. and Wilson, J.R. (2006) 'The nature of expertise: A Review', *Applied Ergonomics*, 37(1), pp. 17–32. doi:10.1016/j.apergo.2005.09.001.
- Fazli-Salehi, R. *et al.* (2022) 'Social media reviewing channels: The role of channel interactivity and vloggers' self-disclosure in consumers' parasocial interaction',

*Journal of Consumer Marketing*, 39(2), pp. 242–253. doi:10.1108/jcm-06-2020-3866.

- Feder, L. (2020) 'From ESPN to Instagram Live: The Evolution of Fan–athlete interaction amid the coronavirus', *International Journal of Sport Communication*, 13(3), pp. 458–464. doi:10.1123/ijsc.2020-0233.
- Field, A.P. (2018) *Discovering statistics using IBM SPSS statistics*. London: SAGE Publications.
- Finstad, K. (2010) 'Response Interpolation and Scale Sensitivity: Evidence Against 5-Point Scales', *Journal of usability studies*, 5(3), pp. 104–110.
- Finsterwalder, J., Yee, T. and Tombs, A. (2017) 'Would you forgive Kristen Stewart or Tiger Woods or maybe Lance Armstrong? exploring consumers' forgiveness of celebrities' transgressions', *Journal of Marketing Management*, 33(13–14), pp. 1204–1229. doi:10.1080/0267257x.2017.1382553.
- Fisch, C. and Block, J. (2018) 'Six tips for your (systematic) literature review in Business and Management Research', *Management Review Quarterly*, 68(2), pp. 103–106. doi:10.1007/s11301-018-0142-x.
- Flecha-Ortiz, J.A. *et al.* (2023) 'Hedonic and utilitarian gratifications to the use of TikTok by Generation Z and the parasocial relationships with influencers as a mediating force to purchase intention', *Journal of Interactive Advertising*, 23(2), pp. 114–127. doi:10.1080/15252019.2023.2195403.
- Fornell, C. and Larcker, D.F. (1981) 'Evaluating structural equation models with unobservable variables and measurement error', *Journal of Marketing Research*, 18(1), p. 39. doi:10.2307/3151312.
- Frampton, J.R. and Fox, J. (2021) 'Monitoring, creeping, or surveillance? A synthesis of online social information seeking concepts', *Review of Communication Research*, 9, pp. 1–42. doi:10.12840/issn.2255-4165.025.
- Francis, J.J. et al. (2010) 'What is an adequate sample size? Operationalising data saturation for theory-based interview studies', Psychology & Health, 25(10), pp. 1229–1245. doi:10.1080/08870440903194015.
- Freshwater, D. (2007) 'Reading mixed methods research: Contexts for criticism', *Journal of Mixed Methods Research*, 1(2), pp. 134–146. doi:10.1177/1558689806298578.
- Fu, J.-R. et al. (2020) 'Investigating consumers' online social shopping intention: An information processing perspective', *International Journal of Information Management*, 54, p. 102189. doi:10.1016/j.ijinfomgt.2020.102189.
- Fu, S., Xu, Y. and Yan, Q. (2017) 'Enhancing the parasocial interaction relationship between consumers through similarity effects in the context of Social Commerce', *Journal of Strategic Marketing*, 27(2), pp. 100–118. doi:10.1080/0965254x.2017.1384045.

- Fuller, C.M. et al. (2016) 'Common methods variance detection in Business Research', Journal of Business Research, 69(8), pp. 3192–3198. doi:10.1016/j.jbusres.2015.12.008.
- Gabriel, S. et al. (2018) 'From apprentice to president', Social Psychological and Personality Science, 9(3), pp. 299–307. doi:10.1177/1948550617722835.
- Gamble, J. *et al.* (2011) 'The marketing concept in the 21st Century: A review of how marketing has been defined since the 1960s', *The Marketing Review*, 11(3), pp. 227–248. doi:10.1362/146934711x589444.
- Gannon, M.J., Taheri, B. and Azer, J. (2022) 'Contemporary research paradigms and Philosophies', *Contemporary Research Methods in Hospitality and Tourism*, pp. 5– 19. doi:10.1108/978-1-80117-546-320221002.
- Gelinas-Chebat, C. and Chebat, J.-C. (1992) 'Effects of two voice characteristics on the attitudes toward advertising messages', *The Journal of Social Psychology*, 132(4), pp. 447–459. doi:10.1080/00224545.1992.9924724.
- Gendall, P. and Healey, B. (2010) 'Effect of a promised donation to charity on survey response', *International Journal of Market Research*, 52(5), pp. 565–577. doi:10.2501/s147078531020148x.
- Gerbing, D.W. and Anderson, J.C. (1988) 'An updated paradigm for scale development incorporating unidimensionality and its assessment', *Journal of Marketing Research*, 25(2), p. 186. doi:10.2307/3172650.
- Gibson, C.B. (2016) 'Elaboration, generalization, triangulation, and interpretation', Organizational Research Methods, 20(2), pp. 193–223. doi:10.1177/1094428116639133.
- Gilani, S.O. *et al.* (2013) 'Impact of image appeal on visual attention during photo triaging', *2013 IEEE International Conference on Image Processing* [Preprint]. doi:10.1109/icip.2013.6738048.
- Giles, D.C. (2002) 'Parasocial Interaction: A review of the literature and a model for future research', *Media Psychology*, 4(3), pp. 279–305. doi:10.1207/s1532785xmep0403 04.
- Gilliam, D.A. and Voss, K. (2013) 'A proposed procedure for construct definition in marketing', *European Journal of Marketing*, 47(1/2), pp. 5–26. doi:10.1108/03090561311285439.
- Goldberg, C.B. and Allen, D.G. (2008) 'Black and white and read all over: Race differences in reactions to recruitment web sites', *Human Resource Management*, 47(2), pp. 217–236. doi:10.1002/hrm.20209.
- Gomes, M.A., Marques, S. and Dias, Á. (2022) 'The impact of Digital Influencers' characteristics on purchase intention of fashion products', *Journal of Global Fashion Marketing*, 13(3), pp. 187–204. doi:10.1080/20932685.2022.2039263.

- Gong, W. and Li, X. (2017) 'Engaging fans on microblog: The synthetic influence of parasocial interaction and source characteristics on celebrity endorsement', *Psychology & Marketing*, 34(7), pp. 720–732. doi:10.1002/mar.21018.
- Gong, Z.H. and Holiday, S. (2023) 'Parasocial interaction message elements and disclosure timing in nano- and Microinfluencers' sponsored content as alternative explanations for follower count's influence on engagement', *Journal of Interactive Advertising*, pp. 1–14. doi:10.1080/15252019.2023.2236093.
- Goulding, C. (1999) 'Consumer research, interpretive paradigms and methodological ambiguities', *European Journal of Marketing*, 33(9/10), pp. 859–873. doi:10.1108/03090569910285805.
- Grand View Research (2023) *Influencer Marketing Platform Market Size Report, 2030*. Available at: https://www.grandviewresearch.com/industry-analysis/influencermarketing-platform-market (Accessed: 15 February 2024).
- Granot, E., Brashear, T.G. and Cesar Motta, P. (2012) 'A structural guide to in-depth interviewing in business and Industrial Marketing Research', *Journal of Business & Industrial Marketing*, 27(7), pp. 547–553. doi:10.1108/08858621211257310.
- Grant, A.E., GUTHRIE, K.K. and BALL-ROKEACH, S.J. (1991) 'Television shopping', *Communication Research*, 18(6), pp. 773–798. doi:10.1177/009365091018006004.
- Greenwood, D.N. (2008) 'Television as escape from self: Psychological predictors of media involvement', *Personality and Individual Differences*, 44(2), pp. 414–424. doi:10.1016/j.paid.2007.09.001.
- Greenwood, D.N. and Long, C.R. (2009) 'Psychological predictors of media involvement', *Communication Research*, 36(5), pp. 637–654. doi:10.1177/0093650209338906.
- Grossmann, A., Mooney, L. and Dugan, M. (2019) 'Inclusion fairness in accounting, finance, and management: An investigation of A-star publications on the ABDC Journal list', *Journal of Business Research*, 95, pp. 232–241. doi:10.1016/j.jbusres.2018.10.035.
- Groyecka, A. *et al.* (2017) 'Attractiveness is multimodal: Beauty is also in the nose and ear of the beholder', *Frontiers in Psychology*, 8. doi:10.3389/fpsyg.2017.00778.
- Guest, G., Bunce, A. and Johnson, L. (2006) 'How many interviews are enough? An experiment with data saturation and variability', *Field Methods*, 18(1), pp. 59–82. doi:10.1177/1525822x05279903.
- Gunbayi, I. (2020) 'Knowledge-constitutive interests and social paradigms in Guiding Mixed Methods Research (MMR)', *Journal of Mixed Methods Studies*, 1(1), pp. 1– 13. doi:10.14689/jomes.2020.1.3.
- Gunes, H. (2011) 'A survey of perception and computation of human beauty', *Proceedings* of the 2011 joint ACM workshop on Human gesture and behavior understanding [Preprint]. doi:10.1145/2072572.2072580.

- Guo, Y., Zhang, K. and Wang, C. (2021) 'Way to success: Understanding top streamer's popularity and influence from the perspective of source characteristics', *Journal of Retailing and Consumer Services*, 64, p. 102786. doi:10.1016/j.jretconser.2021.102786.
- Gupta, H., Singh, S. and Sinha, P. (2016) 'Multimedia tool as a predictor for social media advertising- A youtube way', *Multimedia Tools and Applications*, 76(18), pp. 18557–18568. doi:10.1007/s11042-016-4249-6.
- Gupta, P., Burton, J.L. and Costa Barros, L. (2022) 'Gender of the online influencer and follower: The differential persuasive impact of homophily, attractiveness and product-match', *Internet Research*, 33(2), pp. 720–740. doi:10.1108/intr-04-2021-0229.
- Hair, J.F. et al. (2014) Multivariate Data Analysis. Upper Saddle River, NJ: Prentice Hall.
- Hair, J.F., Ortinau, D.J. and Harrison, D.E. (2021) *Essentials of Marketing Research*. New York, NY: McGraw-Hill.
- Hall, J.A. (2016) 'When is social media use social interaction? defining mediated social interaction', *New Media & Society*, 20(1), pp. 162–179. doi:10.1177/1461444816660782.
- Han, S. and Yang, H. (2018) 'Understanding adoption of intelligent personal assistants', *Industrial Management & Data Systems*, 118(3), pp. 618–636. doi:10.1108/imds-05-2017-0214.
- Handarkho, Y.D. (2021) 'Understanding mobile payment continuance usage in physical store through Social Impact Theory and Trust Transfer', *Asia Pacific Journal of Marketing and Logistics*, 33(4), pp. 1071–1087. doi:10.1108/apjml-01-2020-0018.
- Haobin Ye, B., Fong, L.H. and Luo, J.M. (2021) 'Parasocial interaction on tourism companies' social media sites: Antecedents and consequences', *Current Issues in Tourism*, 24(8), pp. 1093–1108. doi:10.1080/13683500.2020.1764915.
- Hardesty, D.M. and Bearden, W.O. (2004) "The use of expert judges in scale development," *Journal of Business Research*, 57(2), pp. 98–107. Available at: https://doi.org/10.1016/s0148-2963(01)00295-8.
- Harrison, R.L. and Reilly, T.M. (2011) 'Mixed methods designs in marketing research', *Qualitative Market Research: An International Journal*, 14(1), pp. 7–26. doi:10.1108/13522751111099300.
- Hartmann, P. and Apaolaza-Ibáñez, V. (2012) 'Consumer attitude and purchase intention toward Green Energy Brands: The roles of Psychological Benefits and Environmental concern', *Journal of Business Research*, 65(9), pp. 1254–1263. doi:10.1016/j.jbusres.2011.11.001.
- Hartmann, T. and Goldhoorn, C. (2011) 'Horton and Wohl revisited: Exploring viewers' experience of Parasocial Interaction', *Journal of Communication*, 61(6), pp. 1104–1121. doi:10.1111/j.1460-2466.2011.01595.x.

- Hartmann, T., Stuke, D. and Daschmann, G. (2008) 'Positive parasocial relationships with drivers affect suspense in racing sport spectators', *Journal of Media Psychology*, 20(1), pp. 24–34. doi:10.1027/1864-1105.20.1.24.
- Hassan, S., Nadzim, S.Z. and Shiratuddin, N. (2015) 'Strategic use of social media for small business based on the aida model', *Procedia - Social and Behavioral Sciences*, 172, pp. 262–269. doi:10.1016/j.sbspro.2015.01.363.
- Hassan, S.M. and Rahman, Z. (2021) 'The evolving passage of Consumer Ethics Research: A systematic literature review', *International Journal of Emerging Markets* [Preprint]. doi:10.1108/ijoem-04-2021-0504.
- Haws, K.L., Sample, K.L. and Hulland, J. (2022) 'Scale use and abuse: Towards best practices in the deployment of scales', *Journal of Consumer Psychology*, 33(1), pp. 226–243. doi:10.1002/jcpy.1320.
- Healy, M. and Perry, C. (2000) 'Comprehensive criteria to judge validity and reliability of qualitative research within the realism paradigm', *Qualitative Market Research: An International Journal*, 3(3), pp. 118–126. doi:10.1108/13522750010333861.
- Heinonen, K. (2011) 'Consumer activity in social media: Managerial approaches to consumers' social media behavior', *Journal of Consumer Behaviour*, 10(6), pp. 356– 364. doi:10.1002/cb.376.
- Henseler, J., Ringle, C.M. and Sarstedt, M. (2014) 'A new criterion for assessing discriminant validity in variance-based structural equation modeling', *Journal of the Academy of Marketing Science*, 43(1), pp. 115–135. doi:10.1007/s11747-014-0403-8.
- Herche, J. and Engelland, B. (1996) 'Reversed-polarity items and scale unidimensionality', Journal of the Academy of Marketing Science, 24(4), pp. 366–374. doi:10.1177/0092070396244007.
- Herling, R.W. (2000) 'Operational definitions of expertise and competence', *Advances in Developing Human Resources*, 2(1), pp. 8–21. doi:10.1177/152342230000200103.
- Hernandez-Fernandez, A. and Lewis, M.C. (2019) 'Brand authenticity leads to perceived value and Brand Trust', *European Journal of Management and Business Economics*, 28(3), pp. 222–238. doi:10.1108/ejmbe-10-2017-0027.
- Hinkin, T.R. (1995) 'A review of scale development practices in the study of Organizations', *Journal of Management*, 21(5), pp. 967–988. doi:10.1177/014920639502100509.
- Hoffner, C.A. and Cohen, E.L. (2014) 'Portrayal of mental illness on the TV seriesmonk: Presumed influence and consequences of exposure', *Health Communication*, 30(10), pp. 1046–1054. doi:10.1080/10410236.2014.917840.
- Hollenbeck, J.R. and Wright, P.M. (2016) 'Harking, sharking, and Tharking', *Journal of Management*, 43(1), pp. 5–18. doi:10.1177/0149206316679487.

- Hong, J. et al. (2021) 'Sharing consumers' brand storytelling: Influence of consumers' storytelling on brand attitude via emotions and Cognitions', *Journal of Product & Brand Management*, 31(2), pp. 265–278. doi:10.1108/jpbm-07-2019-2485.
- Hoover, S.M. *et al.* (2018) 'Teaching qualitative research interviewer skills: A developmental framework for social justice psychological research teams.', *Qualitative Psychology*, 5(2), pp. 300–318. doi:10.1037/qup0000101.
- Horton, D. and Wohl, R.R. (1956) 'Mass communication and para-social interaction', *Psychiatry*, 19(3), pp. 215–229. doi:10.1080/00332747.1956.11023049.
- Hou, F. *et al.* (2019) 'Factors influencing people's continuous watching intention and consumption intention in live streaming', *Internet Research*, 30(1), pp. 141–163. doi:10.1108/intr-04-2018-0177.
- Houben, M., Van Den Noortgate, W. and Kuppens, P. (2015) 'The relation between shortterm emotion dynamics and psychological well-being: A meta-analysis.', *Psychological Bulletin*, 141(4), pp. 901–930. doi:10.1037/a0038822.
- Hsieh, J.-K., Kunz, W.H. and Wu, A.-Y. (2022) 'Virtual gifting behavior on new social media: The perspectives of the community gift-giving model and face-negotiation theory', *Internet Research*, 33(4), pp. 1597–1632. doi:10.1108/intr-09-2021-0691.
- Hsieh, S.H. and Lee, C.T. (2021) 'Hey Alexa: Examining the effect of perceived socialness in usage intentions of AI assistant-enabled smart speaker', *Journal of Research in Interactive Marketing*, 15(2), pp. 267–294. doi:10.1108/jrim-11-2019-0179.
- Hu, L. *et al.* (2020) 'Understanding followers' stickiness to digital influencers: The effect of psychological responses', *International Journal of Information Management*, 54, p. 102169. doi:10.1016/j.ijinfomgt.2020.102169.
- Hu, M. (2016) 'The influence of a scandal on parasocial relationship, parasocial interaction, and parasocial breakup.', *Psychology of Popular Media Culture*, 5(3), pp. 217–231. doi:10.1037/ppm0000068.
- Hu, M., Zhang, M. and Wang, Y. (2017) 'Why do audiences choose to keep watching on live video streaming platforms? an explanation of dual identification framework', *Computers in Human Behavior*, 75, pp. 594–606. doi:10.1016/j.chb.2017.06.006.
- Hu, Q. et al. (2023) 'Heterophily or homophily of social media influencers: The role of dual parasocial relationships in impulsive buying', *International Journal of Electronic Commerce*, 27(4), pp. 558–589. doi:10.1080/10864415.2023.2255112.
- Hudders, L. and De Jans, S. (2022) 'Gender effects in influencer marketing: An experimental study on the efficacy of endorsements by same- vs. other-gender social media influencers on Instagram', *International Journal of Advertising*, 41(1), pp. 128–149. doi:10.1080/02650487.2021.1997455.
- Hudders, L., Jans, S.D. and Veirman, M.D. (2021) 'The commercialization of Social Media Stars: A Literature Review and conceptual framework on the strategic use of Social Media influencers', *Social Media Influencers in Strategic Communication*, pp. 24–67. doi:10.4324/9781003181286-3.

- Huettl, V. and Gierl, H. (2012) 'Visual art in advertising: The effects of utilitarian vs. hedonic product positioning and price information', *Marketing Letters*, 23(3), pp. 893–904. doi:10.1007/s11002-012-9196-z.
- Hughes, C., Swaminathan, V. and Brooks, G. (2019) 'Driving brand engagement through online social influencers: An empirical investigation of sponsored blogging campaigns', *Journal of Marketing*, 83(5), pp. 78–96. doi:10.1177/0022242919854374.
- Hung, S.-M., Nieh, C.-H. and Hsieh, P.-J. (2016) 'Unconscious processing of facial attractiveness: Invisible attractive faces Orient Visual Attention', *Scientific Reports*, 6(1). doi:10.1038/srep37117.
- Hunt, S.D. (1993) 'Objectivity in marketing theory and research', *Journal of Marketing*, 57(2), p. 76. doi:10.2307/1252028.
- Hunt, S.D. (2011) 'On the intersection of marketing history and marketing theory', *Marketing Theory*, 11(4), pp. 483–489. doi:10.1177/1470593111418802.
- Hwang, H. et al. (2020) 'A gratification model of sport team mobile application usage', Sport Marketing Quarterly, 29(3), pp. 163–176. doi:10.32731/smq.293.092020.01.
- Hwang, K. and Zhang, Q. (2018) 'Influence of parasocial relationship between digital celebrities and their followers on followers' purchase and electronic word-of-mouth intentions, and persuasion knowledge', *Computers in Human Behavior*, 87, pp. 155– 173. doi:10.1016/j.chb.2018.05.029.
- Hyman, M.R. and Sierra, J.J. (2012) 'Adjusting self-reported attitudinal data for mischievous respondents', *International Journal of Market Research*, 54(1), pp. 129–145. doi:10.2501/ijmr-54-1-129-145.
- Iannone, N.E. *et al.* (2018) 'Connecting in the twitterverse: Using Twitter to satisfy unmet belonging needs', *The Journal of Social Psychology*, 158(4), pp. 491–495. doi:10.1080/00224545.2017.1385445.
- Ingram, J. and Luckett, Z. (2019) 'My friend Harry's A wizard: Predicting parasocial interaction with characters from fiction.', *Psychology of Popular Media Culture*, 8(2), pp. 148–158. doi:10.1037/ppm0000169.
- Jacobson, J., Hodson, J. and Mittelman, R. (2022) 'PUP-ularity contest: The advertising practices of popular animal influencers on Instagram', *Technological Forecasting and Social Change*, 174, p. 121226. doi:10.1016/j.techfore.2021.121226.
- Jeganathan, K. and Szymkowiak, A. (2020) 'Social media content headlinesand their impact on attracting attention', *Journal of Marketing and Consumer Behaviour in Emerging Markets*, 1/2020(10), pp. 49–59. doi:10.7172/2449-6634.jmcbem.2020.1.3.
- Jiménez, F.R. and Voss, K.E. (2014) 'An alternative approach to the measurement of emotional attachment', *Psychology & Marketing*, 31(5), pp. 360–370. doi:10.1002/mar.20700.

- Jin, S.-A.A. and Phua, J. (2014) 'Following celebrities' tweets about brands: The impact of Twitter-based electronic word-of-mouth on consumers' source credibility perception, buying intention, and social identification with celebrities', *Journal of Advertising*, 43(2), pp. 181–195. doi:10.1080/00913367.2013.827606.
- Jin, S.V. (2018) "Celebrity 2.0 and beyond!" effects of facebook profile sources on social networking advertising', *Computers in Human Behavior*, 79, pp. 154–168. doi:10.1016/j.chb.2017.10.033.
- Jin, S.V. and Muqaddam, A. (2019) 'Product placement 2.0: "Do brands need influencers, or do influencers need brands?", *Journal of Brand Management*, 26(5), pp. 522– 537. doi:10.1057/s41262-019-00151-z.
- Jin, S.V. and Ryu, E. (2020) "I'll buy what she's #wearing": The roles of envy toward and Parasocial interaction with influencers in Instagram celebrity-based brand endorsement and Social Commerce', *Journal of Retailing and Consumer Services*, 55, p. 102121. doi:10.1016/j.jretconser.2020.102121.
- Jin, S.V., Ryu, E. and Muqaddam, A. (2021) 'I trust what she's #endorsing on Instagram: Moderating effects of parasocial interaction and social presence in fashion influencer marketing', *Journal of Fashion Marketing and Management: An International Journal*, 25(4), pp. 665–681. doi:10.1108/jfmm-04-2020-0059.
- Jobber, D., Saunders, J. and Mitchell, V.-W. (2004) 'Prepaid monetary incentive effects on mail survey response', *Journal of Business Research*, 57(1), pp. 21–25. doi:10.1016/s0148-2963(02)00280-1.
- Johnson, R.B., Onwuegbuzie, A.J. and Turner, L.A. (2007) 'Toward a definition of mixed methods research', *Journal of Mixed Methods Research*, 1(2), pp. 112–133. doi:10.1177/1558689806298224.
- Joshi, Y. *et al.* (2023) 'Social Media Influencer Marketing: Foundations, trends, and Ways Forward', *Electronic Commerce Research* [Preprint]. doi:10.1007/s10660-023-09719-z.
- Jun, S. and Yi, J. (2020) 'What makes followers loyal? the role of influencer interactivity in building influencer brand equity', *Journal of Product & Brand Management*, 29(6), pp. 803–814. doi:10.1108/jpbm-02-2019-2280.
- Kabadayi, S. and Price, K. (2014) 'Consumer brand engagement on Facebook: Liking and commenting behaviors', *Journal of Research in Interactive Marketing*, 8(3), pp. 203–223. doi:10.1108/jrim-12-2013-0081.
- Kallio, H. *et al.* (2016) 'Systematic methodological review: Developing a framework for a qualitative semi-structured interview guide', *Journal of Advanced Nursing*, 72(12), pp. 2954–2965. doi:10.1111/jan.13031.
- Kane, G.C. (2012) 'Psychosocial stages of symbolic action in social media', SSRN Electronic Journal [Preprint]. doi:10.2139/ssrn.2041940.
- Kang, S. (2022) 'Travelers' pro-environmental behaviors in the Hyperloop context: Integrating norm activation and AIDA models', *International Journal of Tourism Research*, 24(6), pp. 813–826. doi:10.1002/jtr.2547.

- Kang, S. *et al.* (2021) 'Does virtual reality affect behavioral intention? testing engagement processes in a K-pop video on YouTube', *Computers in Human Behavior*, 123, p. 106875. doi:10.1016/j.chb.2021.106875.
- Kapitan, S. *et al.* (2021) 'Influencer marketing and authenticity in content creation', *Australasian Marketing Journal*, 30(4), pp. 342–351. doi:10.1177/18393349211011171.
- Kaplan, A.M. and Haenlein, M. (2010) 'Users of the world, unite! the challenges and opportunities of social media', *Business Horizons*, 53(1), pp. 59–68. doi:10.1016/j.bushor.2009.09.003.
- Kashif, M. and Udunuwara, M. (2020) 'Twenty Years of research in Brand Globalness/Localness: A systematic literature review and future research agenda', *Journal of International Consumer Marketing*, 33(2), pp. 178–193. doi:10.1080/08961530.2020.1743805.
- Kassarjian, H.H. (1977) 'Content analysis in consumer research', Journal of Consumer Research, 4(1), p. 8. doi:10.1086/208674.
- Kassing, J.W. and Sanderson, J. (2012) 'Playing in the new media game or riding the virtual bench', *Journal of Sport and Social Issues*, 39(1), pp. 3–18. doi:10.1177/0193723512458931.
- Katsuki, F. and Constantinidis, C. (2013) 'Bottom-up and top-down attention', *The Neuroscientist*, 20(5), pp. 509–521. doi:10.1177/1073858413514136.
- Kelton, A.S. and Pennington, R.R. (2020) 'If you tweet, they will follow: CEO tweets, social capital, and investor say-on-pay judgments', *Journal of Information Systems*, 34(1), pp. 105–122. doi:10.2308/isys-52449.
- Kemp, S. (2023) Digital 2023: The United Kingdom DataReportal global digital insights, DataReportal. Available at: https://datareportal.com/reports/digital-2023unitedkingdom#:~:text=More%20broadly%2C%2086.4%20percent%20of,while%2048.4% 20percent%20were%20male. (Accessed: 17 March 2024).
- Keng, C., Ting, H. and Chen, Y. (2011) 'Effects of virtual-experience combinations on consumer-related "sense of virtual community", *Internet Research*, 21(4), pp. 408– 434. doi:10.1108/10662241111158308.
- Kerr, N.L. (1998) 'Harking: Hypothesizing after the results are known', *Personality and Social Psychology Review*, 2(3), pp. 196–217. doi:10.1207/s15327957pspr0203\_4.
- Ki, C. 'Chloe' and Kim, Y. (2020) 'The mechanism by which social media influencers persuade consumers: The role of consumers' desire to mimic', *Psychology & Marketing*, 36(10), pp. 905–922. doi:10.1002/mar.21244.
- Ki, C.-W. (Chloe) et al. (2020) 'Influencer Marketing: Social media influencers as human brands attaching to followers and yielding positive marketing results by fulfilling needs', *Journal of Retailing and Consumer Services*, 55, p. 102133. doi:10.1016/j.jretconser.2020.102133.

- Kim, C. and Lee, J.K. (2016) 'Social media type matters: Investigating the relationship between motivation and online social network heterogeneity', *Journal of Broadcasting & Electronic Media*, 60(4), pp. 676–693. doi:10.1080/08838151.2016.1234481.
- Kim, D.Y. and Kim, H.-Y. (2021) 'Trust me, Trust Me not: A nuanced view of influencer marketing on social media', *Journal of Business Research*, 134, pp. 223–232. doi:10.1016/j.jbusres.2021.05.024.
- Kim, E. and Drumwright, M. (2016) 'Engaging consumers and building relationships in social media: How social relatedness influences intrinsic vs. extrinsic consumer motivation', *Computers in Human Behavior*, 63, pp. 970–979. doi:10.1016/j.chb.2016.06.025.
- Kim, E.-H., Yoo, D. and Doh, S.-J. (2021) 'Self-construal on brand fan pages: The mediating effect of para-social interaction and consumer engagement on Brand Loyalty', *Journal of Brand Management*, 28(3), pp. 254–271. doi:10.1057/s41262-020-00211-9.
- Kim, H. (2020) 'Unpacking unboxing video-viewing motivations: The uses and gratifications perspective and the mediating role of Parasocial Interaction on purchase intent', *Journal of Interactive Advertising*, 20(3), pp. 196–208. doi:10.1080/15252019.2020.1828202.
- Kim, H. (2022) 'Keeping up with influencers: Exploring the impact of social presence and parasocial interactions on instagram', *International Journal of Advertising*, 41(3), pp. 414–434. doi:10.1080/02650487.2021.1886477.
- Kim, H. (2022) 'Patient influencers' promotion of prescription drugs on Instagram: Effects of illness disclosure on persuasion knowledge through narrative transportation', *International Journal of Advertising*, 41(8), pp. 1541–1566. doi:10.1080/02650487.2022.2054614.
- Kim, H. and Park, M. (2023) 'Virtual influencers' attractiveness effect on purchase intention: A moderated mediation model of the product–endorser fit with the brand', *Computers in Human Behavior*, 143, p. 107703. doi:10.1016/j.chb.2023.107703.
- Kim, I. and Kim, J.J. (2017) 'Older adults' parasocial interaction formation process in the context of travel websites: The moderating role of parent-child geographic proximity', *Tourism Management*, 63, pp. 399–416. doi:10.1016/j.tourman.2017.07.012.
- Kim, I. and Kim, J.J. (2021) 'Emotional attachment, age and online travel community behaviour: The role of Parasocial Interaction', *Current Issues in Tourism*, 24(24), pp. 3466–3488. doi:10.1080/13683500.2021.1952942.
- Kim, J. and Song, H. (2016) 'Celebrities self-disclosure on Twitter and Parasocial relationships: A mediating role of Social Presence', *Computers in Human Behavior*. doi:10.2139/ssrn.2756744.
- Kim, J., Kang, S. and Lee, K.H. (2020) 'How social capital impacts the purchase intention of Sustainable Fashion Products', *Journal of Business Research*, 117, pp. 596–603. doi:10.1016/j.jbusres.2018.10.010.

- Kim, M. and Kim, J. (2020) 'How does a celebrity make fans happy? interaction between celebrities and fans in the social media context', *Computers in Human Behavior*, 111, p. 106419. doi:10.1016/j.chb.2020.106419.
- Kim, S., Zhang, X.A. and Zhang, B.W. (2016) 'Self-mocking crisis strategy on social media: Focusing on Alibaba chairman Jack Ma in China', *Public Relations Review*, 42(5), pp. 903–912. doi:10.1016/j.pubrev.2016.10.004.
- Kim, S. and Choudhury, A. (2021) 'Exploring older adults' perception and use of smart speaker-based voice assistants: A longitudinal study', *Computers in Human Behavior*, 124, p. 106914. doi:10.1016/j.chb.2021.106914.
- Kim, S.H. *et al.* (2018) 'Visual attention during the evaluation of facial attractiveness is influenced by facial angles and smile', *The Angle Orthodontist*, 88(3), pp. 329–337. doi:10.2319/080717-528.1.
- Kleinginna, P.R. and Kleinginna, A.M. (1981) 'A categorized list of motivation definitions, with a suggestion for a consensual definition', *Motivation and Emotion*, 5(3), pp. 263–291. doi:10.1007/bf00993889.
- Knapp, Thomas R. (2002), The Reliability of Measuring Instruments (2nd ed.). Vancouver, BC: Edgeworth Laboratory for Quantitative Educational and Behavioral Science Series.
- Knoll, J. et al. (2015) 'Good Guy vs. Bad Guy: The influence of parasocial interactions with media characters on brand placement effects', *International Journal of Advertising*, 34(5), pp. 720–743. doi:10.1080/02650487.2015.1009350.
- Knupfer, H., Neureiter, A. and Jörg Matthes (2023) 'From Social Media Diet to public riot? engagement with "Greenfluencers" and Young Social Media Users' environmental activism', *Computers in Human Behavior*, 139, p. 107527. doi:10.1016/j.chb.2022.107527.
- Kostyk, A. and Huhmann, B.A. (2021) 'Perfect social media image posts: Symmetry and contrast influence consumer response', *European Journal of Marketing*, 55(6), pp. 1747–1779. doi:10.1108/ejm-09-2018-0629.
- Kostyk, A. *et al.* (2021) 'Securing higher-quality data from self-administered questionnaires', *International Journal of Market Research*, 63(6), pp. 685–692. doi:10.1177/14707853211057172.
- Kostyk, A., Zhou, W. and Hyman, M.R. (2019) 'Using surveytainment to counter declining survey data quality', *Journal of Business Research*, 95, pp. 211–219. doi:10.1016/j.jbusres.2018.10.024.
- Krebs, D. and Adinolfi, A.A. (1975) 'Physical attractiveness, social relations, and personality style.', *Journal of Personality and Social Psychology*, 31(2), pp. 245– 253. doi:10.1037/h0076286.
- Kudeshia, C. and Kumar, A. (2017) 'Social Ewom: Does it affect the brand attitude and purchase intention of brands?', *Management Research Review*, 40(3), pp. 310–330. doi:10.1108/mrr-07-2015-0161.

- Kühne, R. and Opree, S.J. (2020) 'From admiration to devotion? the longitudinal relation between adolescents' involvement with and viewing frequency of reality TV', *Journal of Broadcasting & Electronic Media*, 64(2), pp. 111–130. doi:10.1080/08838151.2020.1728688.
- Kwon, S. (2020) 'Understanding user participation from the perspective of psychological ownership: The moderating role of social distance', *Computers in Human Behavior*, 105, p. 106207. doi:10.1016/j.chb.2019.106207.
- Labrecque, L.I. (2014) 'Fostering consumer–brand relationships in social media environments: The role of Parasocial Interaction', *Journal of Interactive Marketing*, 28(2), pp. 134–148. doi:10.1016/j.intmar.2013.12.003.
- Ladhari, R., Massa, E. and Skandrani, H. (2020) 'YouTube vloggers' popularity and influence: The roles of homophily, emotional attachment, and expertise', *Journal of Retailing and Consumer Services*, 54, p. 102027. doi:10.1016/j.jretconser.2019.102027.
- Lagrosen, S. (2005) 'Effects of the internet on the marketing communication of service companies', *Journal of Services Marketing*, 19(2), pp. 63–69. doi:10.1108/08876040510591376.
- Ledbetter, A.M. and Meisner, C. (2021) 'Extending the personal branding affordances typology to parasocial interaction with public figures on social media: Social Presence and media multiplexity as mediators', *Computers in Human Behavior*, 115, p. 106610. doi:10.1016/j.chb.2020.106610.
- Lee, E.-J. and Jang, J. (2011) 'Not so imaginary interpersonal contact with public figures on social network sites', *Communication Research*, 40(1), pp. 27–51. doi:10.1177/0093650211431579.
- Lee, E.-Jiang. (2013) 'Effectiveness of politicians' soft campaign on Twitter versus TV: Cognitive and experiential routes', *Journal of Communication* [Preprint]. doi:10.1111/jcom.12049.
- Lee, H. and Cho, C.-H. (2020) 'Uses and gratifications of smart speakers: Modelling the effectiveness of smart speaker advertising', *International Journal of Advertising*, 39(7), pp. 1150–1171. doi:10.1080/02650487.2020.1765657.
- Lee, J.A. and Eastin, M.S. (2021) 'Perceived authenticity of social media influencers: Scale development and validation', *Journal of Research in Interactive Marketing*, 15(4), pp. 822–841. doi:10.1108/jrim-12-2020-0253.
- Lee, J.A. *et al.* (2021) 'Why are consumers following social media influencers on Instagram? exploration of consumers' motives for following influencers and the role of materialism', *International Journal of Advertising*, 41(1), pp. 78–100. doi:10.1080/02650487.2021.1964226.
- Lee, J.E. and Watkins, B. (2016) 'YouTube vloggers' influence on consumer luxury brand perceptions and intentions', *Journal of Business Research*, 69(12), pp. 5753–5760. doi:10.1016/j.jbusres.2016.04.171.

- Lee, M. and Lee, H. (2021) 'Do parasocial interactions and vicarious experiences in the beauty YouTube Channels Promote Consumer Purchase Intention?', *International Journal of Consumer Studies*, 46(1), pp. 235–248. doi:10.1111/ijcs.12667.
- Lee, M. and Park, J. (2022) 'Do parasocial relationships and the quality of communication with AI shopping chatbots determine middle-aged women consumers' continuance usage intentions?', *Journal of Consumer Behaviour*, 21(4), pp. 842–854. doi:10.1002/cb.2043.
- Lee, M.T. and Theokary, C. (2021) 'The Superstar Social Media Influencer: Exploiting linguistic style and emotional contagion over content?', *Journal of Business Research*, 132, pp. 860–871. doi:10.1016/j.jbusres.2020.11.014.
- Lee, S. (Ally) (2018) 'Enhancing customers' continued mobile app use in the service industry', *Journal of Services Marketing*, 32(6), pp. 680–691. doi:10.1108/jsm-01-2017-0015.
- Lee, S.A. and Lee, M. (2017) 'Effects of relationship types on customers' parasocial interactions', *Journal of Hospitality and Tourism Technology*, 8(1), pp. 133–147. doi:10.1108/jhtt-09-2016-0053.
- Lee, S.H. and Hoffman, K.D. (2015) 'Learning the SHAMWOW: Creating infomercials to teach the Aida Model', *Marketing Education Review*, 25(1), pp. 9–14. doi:10.1080/10528008.2015.999586.
- Lee, S.S. and Johnson, B.K. (2021) 'Are they being authentic? the effects of self-disclosure and message sidedness on sponsored post effectiveness', *International Journal of Advertising*, 41(1), pp. 30–53. doi:10.1080/02650487.2021.1986257.
- Lee, Y.-H., Yuan, C.W. and Wohn, D.Y. (2021) 'How video streamers' mental health disclosures affect viewers' risk perceptions', *Health Communication*, 36(14), pp. 1931–1941. doi:10.1080/10410236.2020.1808405.
- Leite, F.P. and Baptista, P. de (2022) 'The effects of social media influencers' selfdisclosure on behavioral intentions: The role of source credibility, Parasocial Relationships, and Brand Trust', *Journal of Marketing Theory and Practice*, 30(3), pp. 295–311. doi:10.1080/10696679.2021.1935275.
- Leonhardt, J.M., Pezzuti, T. and Namkoong, J.-E. (2020) 'We're not so different: Collectivism increases perceived homophily, trust, and seeking user-generated product information', *Journal of Business Research*, 112, pp. 160–169. doi:10.1016/j.jbusres.2020.03.017.
- Leung, L. and Wei, R. (1998) 'The gratifications of pager use: Sociability, informationseeking, entertainment, utility, and fashion and status', *Telematics and Informatics*, 15(4), pp. 253–264. doi:10.1016/s0736-5853(98)00016-1.
- Leung, X.Y. and Bai, B. (2013) 'How motivation, opportunity, and ability impact travelers' social media involvement and revisit intention', *Journal of Travel & Tourism Marketing*, 30(1–2), pp. 58–77. doi:10.1080/10548408.2013.751211.

- Leys, C. *et al.* (2013) 'Detecting outliers: Do not use standard deviation around the mean, use absolute deviation around the median', *Journal of Experimental Social Psychology*, 49(4), pp. 764–766. doi:10.1016/j.jesp.2013.03.013.
- Li, M.-W., Teng, H.-Y. and Chen, C.-Y. (2020) 'Unlocking the customer engagementbrand loyalty relationship in tourism social media: The roles of Brand Attachment and customer trust', *Journal of Hospitality and Tourism Management*, 44, pp. 184– 192. doi:10.1016/j.jhtm.2020.06.015.
- Liao, J., Li, C. and Filieri, R. (2022) 'The role of humor in management response to positive consumer reviews', *Journal of Interactive Marketing*, 57(2), pp. 323–342. doi:10.1177/10949968221083095.
- Liberati, A. (2009) 'The prisma statement for reporting systematic reviews and metaanalyses of studies that evaluate health care interventions: Explanation and elaboration', *Annals of Internal Medicine*, 151(4). doi:10.7326/0003-4819-151-4-200908180-00136.
- Lichtenstein, D.R., Netemeyer, R.G. and Burton, S. (1990) 'Distinguishing coupon proneness from value consciousness: An acquisition-transaction utility theory perspective', *Journal of Marketing*, 54(3), p. 54. doi:10.2307/1251816.
- Liebers, N. and Schramm, H. (2019). 'Parasocial interactions and relationships with media characters–an inventory of 60 years of research', *Communication Research Trends*, 38(2), pp.4-31.
- Lietz, P. (2010) 'Research into questionnaire design: A summary of the literature', *International Journal of Market Research*, 52(2), pp. 249–272. doi:10.2501/s147078530920120x.
- Lim, C.M. and Kim, Y.-K. (2011) 'Older consumers' TV home shopping: Loneliness, parasocial interaction, and perceived convenience', *Psychology and Marketing*, 28(8), pp. 763–780. doi:10.1002/mar.20411.
- Lim, J.S. *et al.* (2020) 'The role of wishful identification, emotional engagement, and parasocial relationships in repeated viewing of live-streaming games: A Social Cognitive Theory Perspective', *Computers in Human Behavior*, 108, p. 106327. doi:10.1016/j.chb.2020.106327.
- Lim, M.S. and Kim, J. (2018) 'Facebook users' loneliness based on different types of interpersonal relationships', *Information Technology & People*, 31(3), pp. 646–665. doi:10.1108/itp-04-2016-0095.
- Lindgaard, G. et al. (2006) 'Attention web designers: You have 50 milliseconds to make a good first impression!', *Behaviour & Information Technology*, 25(2), pp. 115–126. doi:10.1080/01449290500330448.
- Liu, M.T., Liu, Y. and Zhang, L.L. (2019) 'Vlog and brand evaluations: The Influence of Parasocial Interaction', *Asia Pacific Journal of Marketing and Logistics*, 31(2), pp. 419–436. doi:10.1108/apjml-01-2018-0021.
- Liu, Q., Shao, Z. and Fan, W. (2018) 'The impact of users' sense of belonging on social media habit formation: Empirical evidence from social networking and

microblogging websites in China', *International Journal of Information Management*, 43, pp. 209–223. doi:10.1016/j.ijinfomgt.2018.08.005.

- Lo, P.-S. *et al.* (2022) 'Why do consumers buy impulsively during live streaming? A deep learning-based dual-stage SEM-Ann Analysis', *Journal of Business Research*, 147, pp. 325–337. doi:10.1016/j.jbusres.2022.04.013.
- López, M., Sicilia, M. and Verlegh, P.W.J. (2021) 'How to motivate opinion leaders to spread e-wom on social media: Monetary vs non-monetary Incentives', *Journal of Research in Interactive Marketing*, 16(1), pp. 154–171. doi:10.1108/jrim-03-2020-0059.
- Lou, C. and Yuan, S. (2019) 'Influencer marketing: How message value and credibility affect consumer trust of branded content on social media', *Journal of Interactive Advertising*, 19(1), pp. 58–73. doi:10.1080/15252019.2018.1533501.
- Lu, B. and Chen, Z. (2021) 'Live streaming commerce and consumers' purchase intention: An uncertainty reduction perspective', *Information & Management*, 58(7), p. 103509. doi:10.1016/j.im.2021.103509.
- Lu, Y. et al. (2023) 'Influence of livestreamers' intimate self-disclosure on tourist responses: The lens of parasocial interaction theory', *Journal of Hospitality and Tourism Management*, 57, pp. 170–178. doi:10.1016/j.jhtm.2023.10.003.
- Lu, Z. et al. (2018) 'You watch, you give, and you engage', Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems [Preprint]. doi:10.1145/3173574.3174040.
- Lueck, J.A. (2012) 'Friend-zone with benefits: The Parasocial Advertising of Kim Kardashian', *Journal of Marketing Communications*, 21(2), pp. 91–109. doi:10.1080/13527266.2012.726235.
- Luo, Y., Liu, Y. and Xue, J. (2009) 'Relationship investment and channel performance: An analysis of mediating forces', *Journal of Management Studies*, 46(7), pp. 1113–1137. doi:10.1111/j.1467-6486.2009.00843.x.
- MacKenzie, S.B. (2003) 'The dangers of poor construct conceptualization', *Journal of the Academy of Marketing Science*, 31(3), pp. 323–326. doi:10.1177/0092070303031003011.
- Magnusson, E. and Marecek, J. (2015) 'Chapter 5: Designing the interview guide', in *Doing interview-based qualitative research: A learner's guide*. New York: Cambridge University Press.
- Majid, M.A. *et al.* (2017) 'Piloting for interviews in qualitative research: Operationalization and lessons learnt', *International Journal of Academic Research in Business and Social Sciences*, 7(4). doi:10.6007/ijarbss/v7-i4/2916.
- Malhotra, N.K. *et al.* (2012) 'One, few or many?: An integrated framework for identifying the items in measurement scales', *International Journal of Market Research*, 54(6), pp. 835–862. doi:10.2501/ijmr-54-6-835-862.

- Malle, B. F. (2005). Three puzzles of mindreading. In B. F. Malle & S. D. Hodges (Eds.), Other minds: *How humans bridge the divide between self and other* (pp. 26 – 43). New York, NY: Guilford Press.
- Malmqvist, J. *et al.* (2019) 'Conducting the pilot study: A neglected part of the research process? methodological findings supporting the importance of piloting in qualitative research studies', *International Journal of Qualitative Methods*, 18, p. 160940691987834. doi:10.1177/1609406919878341.
- Manchanda, P., Arora, N. and Sethi, V. (2022) 'Impact of beauty vlogger's credibility and popularity on ewom sharing intention: The mediating role of Parasocial Interaction', *Journal of Promotion Management*, 28(3), pp. 379–412. doi:10.1080/10496491.2021.1989542.
- Mandler, T. *et al.* (2021) 'Performance consequences of marketing standardization/adaptation: A systematic literature review and future research agenda', *Journal of Business Research*, 125, pp. 416–435. doi:10.1016/j.jbusres.2020.12.023.
- Mariani, M.M., Perez-Vega, R. and Wirtz, J. (2021) 'AI in marketing, consumer research and psychology: A systematic literature review and research agenda', *Psychology & Marketing*, 39(4), pp. 755–776. doi:10.1002/mar.21619.
- Marsden, D. and Littler, D. (1996) 'Evaluating alternative research paradigms: A marketoriented framework', *Journal of Marketing Management*, 12(7), pp. 645–655. doi:10.1080/0267257x.1996.9964442.
- Marshall, B. *et al.* (2013) 'Does sample size matter in qualitative research?: A review of qualitative interviews in is research', *Journal of Computer Information Systems*, 54(1), pp. 11–22. doi:10.1080/08874417.2013.11645667.
- Martínez-López, F.J. *et al.* (2020) 'Behind influencer marketing: Key marketing decisions and their effects on followers' responses', *Journal of Marketing Management*, 36(7– 8), pp. 579–607. doi:10.1080/0267257x.2020.1738525.
- Masse, B., Ba, S. and Horaud, R. (2018) 'Tracking gaze and visual focus of attention of people involved in social interaction', *IEEE Transactions on Pattern Analysis and Machine Intelligence*, 40(11), pp. 2711–2724. doi:10.1109/tpami.2017.2782819.
- Masuda, H., Han, S.H. and Lee, J. (2022) 'Impacts of influencer attributes on purchase intentions in social media influencer marketing: Mediating roles of characterizations', *Technological Forecasting and Social Change*, 174, p. 121246. doi:10.1016/j.techfore.2021.121246.
- Mathwick, C., Malhotra, N. and Rigdon, E. (2001) 'Experiential value: conceptualization, measurement and application in the catalog and Internet shopping environment', *Journal of Retailing*, 77(1), pp. 39–56. doi:10.1016/s0022-4359(00)00045-2.
- Maulana, M.I. (2022) 'Leveraging zoom video-conferencing features in interview data generation during the COVID-19 pandemic', *Research and Teaching in a Pandemic World*, pp. 391–407. doi:10.1007/978-981-19-7757-2\_26.

- Mayring, P. (2014) 'Qualitative content analysis: Theoretical background and procedures', *Advances in Mathematics Education*, pp. 365–380. doi:10.1007/978-94-017-9181-6\_13.
- McQuitty, S. (2004) 'Statistical Power and structural equation models in business research', *Journal of Business Research*, 57(2), pp. 175–183. doi:10.1016/s0148-2963(01)00301-0.
- Meade, A.W. and Craig, S.B. (2012) 'Identifying careless responses in survey data.', *Psychological Methods*, 17(3), pp. 437–455. doi:10.1037/a0028085.
- Medina-Molina, C., Rey-Moreno, M. and Periáñez-Cristóbal, R. (2021) 'Analysis of the moderating effect of front-of-pack labelling on the relation between brand attitude and purchasing intention', *Journal of Business Research*, 122, pp. 304–310. doi:10.1016/j.jbusres.2020.08.062.
- Men, L.R. and Tsai, W.-H.S. (2013a) 'Beyond liking or following: Understanding public engagement on social networking sites in China', *Public Relations Review*, 39(1), pp. 13–22. doi:10.1016/j.pubrev.2012.09.013.
- Men, L.R. and Tsai, W.-H.S. (2013b) 'Toward an integrated model of public engagement on corporate social networking sites: Antecedents, the process, and relational outcomes', *International Journal of Strategic Communication*, 7(4), pp. 257–273. doi:10.1080/1553118x.2013.822373.
- Men, L.R. and Tsai, W.-H.S. (2015) 'Infusing Social Media with humanity: Corporate character, public engagement, and relational outcomes', *Public Relations Review*, 41(3), pp. 395–403. doi:10.1016/j.pubrev.2015.02.005.
- Meske, C., Wilms, K. and Stieglitz, S. (2019) 'Enterprise social networks as digital infrastructures - understanding the utilitarian value of social media at the Workplace', *Information Systems Management*, 36(4), pp. 350–367. doi:10.1080/10580530.2019.1652448.
- Miell, D. and Dallos, R., 1996. *Social interaction and personal relationships*. 1st ed. London: Sage.
- Mitchell, A.A. and Olson, J.C. (1981) 'Are product attribute beliefs the only mediator of advertising effects on brand attitude?', *Journal of Marketing Research*, 18(3), p. 318. doi:10.2307/3150973.
- Mitchell, T.R. and James, L.R. (2001) 'Building better theory: Time and the specification of when things happen', *Academy of Management Review*, 26(4), pp. 530–547. doi:10.5465/amr.2001.5393889.
- Moeller, R.W., Seehuus, M. and Peisch, V. (2020) 'Emotional intelligence, belongingness, and Mental Health in college students', *Frontiers in Psychology*, 11. doi:10.3389/fpsyg.2020.00093.
- Mondal, J. and Chakrabarti, S. (2019) 'Emerging phenomena of the branded app: A Systematic Literature Review, strategies, and Future Research Directions', *Journal* of Interactive Advertising, 19(2), pp. 148–167. doi:10.1080/15252019.2019.1647580.
- Mongeon, P. and Paul-Hus, A. (2015) 'The journal coverage of web of science and scopus: A comparative analysis', *Scientometrics*, 106(1), pp. 213–228. doi:10.1007/s11192-015-1765-5.
- Morgan, R.M. and Hunt, S.D. (1994) 'The commitment-trust theory of Relationship Marketing', *Journal of Marketing*, 58(3), p. 20. doi:10.2307/1252308.
- Morhart, F. *et al.* (2014) 'Brand authenticity: An integrative framework and measurement scale', *Journal of Consumer Psychology*, 25(2), pp. 200–218. doi:10.1016/j.jcps.2014.11.006.
- Moulard, J.G., Garrity, C.P. and Rice, D.H. (2015) 'What makes a human brand authentic? identifying the antecedents of celebrity authenticity', *Psychology & Marketing*, 32(2), pp. 173–186. doi:10.1002/mar.20771.
- Moyer-Guse, E. and Nabi, R.L. (2010) 'Explaining the effects of narrative in an entertainment television program: Overcoming resistance to persuasion', *Human Communication Research*, 36(1), pp. 26–52. doi:10.1111/j.1468-2958.2009.01367.x.
- Mueller, Daniel J. (1986), *Measuring Social Attitudes: A Handbook for Researchers and Practitioners*. New York, NY: Teachers College Press, 8-19, pp.52-56.
- Munnukka, J. et al. (2019) "thanks for watching". the effectiveness of YouTube vlogendorsements', Computers in Human Behavior, 93, pp. 226–234. doi:10.1016/j.chb.2018.12.014.
- Murrow, J.L. and Hyman, M.R. (1994) 'Direct marketing: Passages, definitions, and Déjà Vu', *Journal of Direct Marketing*, 8(3), pp. 46–56. doi:10.1002/dir.4000080307.
- Naderer, B., Matthes, J. and Schäfer, S. (2021) 'Effects of disclosing ads on Instagram: The moderating impact of similarity to the influencer', *International Journal of Advertising*, 40(5), pp. 686–707. doi:10.1080/02650487.2021.1930939.
- Nah, H.S. (2022) 'The appeal of "real" in Parasocial Interaction: The effect of selfdisclosure on message acceptance via perceived authenticity and liking', *Computers in Human Behavior*, 134, p. 107330. doi:10.1016/j.chb.2022.107330.
- Nanda, A.P. and Banerjee, R. (2020) 'Binge watching: An exploration of the role of Technology', *Psychology & Marketing*, 37(9), pp. 1212–1230. doi:10.1002/mar.21353.
- Newberry, C. (2023) 109 social media demographics marketers need to know in 2024, Social Media Marketing & Management Dashboard. Available at: https://blog.hootsuite.com/social-media-demographics/ (Accessed: 17 March 2024).
- Newell, S.J. *et al.* (2011) 'The effect of personal relationship and consultative task behaviors on buyer perceptions of salesperson trust, expertise, and Loyalty', *Journal of Marketing Theory and Practice*, 19(3), pp. 307–316. doi:10.2753/mtp1069-6679190304.
- NFI (2023) *Fictional character everything you need to know*, *NFI*. Available at: https://www.nfi.edu/fictional-

character/#:~:text=What%20is%20a%20fictional%20character,it%20is%20the%20a uthor's%20imagination. (Accessed: 03 October 2023).

- Noguti, V. and Waller, D.S. (2020) 'Motivations to use social media: Effects on the perceived informativeness, entertainment, and intrusiveness of paid mobile advertising', *Journal of Marketing Management*, 36(15–16), pp. 1527–1555. doi:10.1080/0267257x.2020.1799062.
- Noor, N., Rao Hill, S. and Troshani, I. (2022) 'Artificial Intelligence Service agents: Role of Parasocial Relationship', *Journal of Computer Information Systems*, 62(5), pp. 1009–1023. doi:10.1080/08874417.2021.1962213.
- Nordlund, J.-E. (1978) 'Media interaction', *Communication Research*, 5(2), pp. 150–175. doi:10.1177/009365027800500202.
- Nowell, L.S. *et al.* (2017) 'Thematic Analysis: Striving to Meet the Trustworthiness Criteria', *International Journal of Qualitative Methods*, 16(1), p. 160940691773384. doi:10.1177/1609406917733847.
- Oh, S. and Syn, S.Y. (2015) 'Motivations for sharing information and social support in social media: A comparative analysis of Facebook, Twitter, Delicious, YouTube, and Flickr', *Journal of the Association for Information Science and Technology*, 66(10), pp. 2045–2060. doi:10.1002/asi.23320.
- Ohlwein, M. and Bruno, P. (2021) 'A brand like a friend—how brand likeability influences brand perception', *International Journal of Market Research*, 64(3), pp. 295–305. doi:10.1177/14707853211039190.
- Oleson, M. (2003) 'Exploring the relationship between money attitudes and Maslow's hierarchy of needs', *International Journal of Consumer Studies*, 28(1), pp. 83–92. doi:10.1111/j.1470-6431.2004.00338.x.
- Olfat, M. *et al.* (2023) 'Bloggers' interactive practices and their followers' purchase intentions: The mediating roles of perceived credibility and followers' para-social interactions', *Journal of Promotion Management*, 30(3), pp. 390–415. doi:10.1080/10496491.2023.2279771.
- Oliffe, J.L. et al. (2021) 'Zoom interviews: Benefits and concessions', International Journal of Qualitative Methods, 20, p. 160940692110535. doi:10.1177/16094069211053522.
- Oliver, M.B. *et al.* (2019) 'A penchant for the immoral: Implications of parasocial interaction, perceived complicity, and identification on liking of anti-heroes', *Human Communication Research*, 45(2), pp. 169–201. doi:10.1093/hcr/hqy019.
- Oppermann, M. (2000) 'Triangulation A methodological discussion', *International Journal of Tourism Research*, 2(2), pp. 141–145. doi:10.1002/(sici)1522-1970(200003/04)2:2<141::aid-jtr217&gt;3.0.co;2-u.
- Osborne, J. W. and Costello, A. B., (2005) 'Sample size and subject to item ratio in principal components analysis', *Practical Assessment, Research, and Evaluation*, 9(1): 11. doi: org/10.7275/ktzq-jq66

- Otter (2023) Otter for personal use real-time notes & transcription, Otter for Personal Use - Real-time Notes & Transcription | Otter.ai. Available at: https://otter.ai/individuals (Accessed: 23 October 2023).
- Overby, J.W. and Lee, E.-J. (2006) 'The effects of utilitarian and hedonic online shopping value on consumer preference and intentions', *Journal of Business Research*, 59(10– 11), pp. 1160–1166. doi:10.1016/j.jbusres.2006.03.008.
- Paas, L.J. and Morren, M. (2018) 'Please do not answer if you are reading this: Respondent attention in online panels', *Marketing Letters*, 29(1), pp. 13–21. doi:10.1007/s11002-018-9448-7.
- Page, M.J. *et al.* (2021) 'The Prisma 2020 statement: An updated guideline for reporting systematic reviews', *BMJ*[Preprint]. doi:10.1136/bmj.n71.
- Pahlevan-Sharif, S., Mura, P. and Wijesinghe, S.N.R. (2019) 'A systematic review of systematic reviews in tourism', *Journal of Hospitality and Tourism Management*, 39, pp. 158–165. doi:10.1016/j.jhtm.2019.04.001.
- Palan, S. and Schitter, C. (2018) 'Prolific.ac—a subject pool for online experiments', *Journal of Behavioral and Experimental Finance*, 17, pp. 22–27. doi:10.1016/j.jbef.2017.12.004.
- Pallant, J. (2020) SPSS survival manual: A step-by-step guide to data analysis using SPSS. Abingdon: Routledge.
- Papa, M.J. *et al.* (2000) 'Entertainment-education and Social Change: An analysis of parasocial interaction, social learning, collective efficacy, and paradoxical communication', *Journal of Communication*, 50(4), pp. 31–55. doi:10.1111/j.1460-2466.2000.tb02862.x.
- Parhankangas, A. and Renko, M. (2017) 'Linguistic style and crowdfunding success among social and commercial entrepreneurs', *Journal of Business Venturing*, 32(2), pp. 215–236. doi:10.1016/j.jbusvent.2016.11.001.
- Park, J., Hyun, H. and Thavisay, T. (2021) 'A study of antecedents and outcomes of social media wom towards luxury brand purchase intention', *Journal of Retailing and Consumer Services*, 58, p. 102272. doi:10.1016/j.jretconser.2020.102272.
- Park, J.H. and Lennon, S.J. (2004) 'Television Apparel Shopping: Impulse Buying and Parasocial Interaction', *Clothing and Textiles Research Journal*, 22(3), pp. 135–144. doi:10.1177/0887302x0402200304.
- Park, N., Kee, K.F. and Valenzuela, S. (2009) 'Being immersed in social networking environment: Facebook groups, uses and gratifications, and social outcomes', *CyberPsychology & Behavior*, 12(6), pp. 729–733. doi:10.1089/cpb.2009.0003.
- Paruthi, M. and Kaur, H. (2017) 'Scale development and validation for measuring online engagement', *Journal of Internet Commerce*, 16(2), pp. 127–147. doi:10.1080/15332861.2017.1299497.
- Pashootanizadeh, M. and Khalilian, S. (2018) 'Application of the aida model', *Information and Learning Science*, 119(11), pp. 635–651. doi:10.1108/ils-04-2018-0028.

- Patzer, G.L. (1983) 'Source credibility as a function of communicator physical attractiveness', *Journal of Business Research*, 11(2), pp. 229–241. doi:10.1016/0148-2963(83)90030-9.
- Paul, J. and Criado, A.R. (2020) 'The Art of Writing Literature Review: What do we know and what do we need to know?', *International Business Review*, 29(4), p. 101717. doi:10.1016/j.ibusrev.2020.101717.
- Paul, J. et al. (2020) 'Flagship Special Issue Call for Papers- "Systematic Literature Reviews"', International Journal of Consumer Studies [Preprint].
- Paul, J. et al. (2021) 'Scientific procedures and rationales for systematic literature reviews (spar-4-slr)', International Journal of Consumer Studies, 45(4). doi:10.1111/ijcs.12695.
- Paul, J., Khatri, P. and Kaur Duggal, H. (2023) 'Frameworks for developing impactful systematic literature reviews and theory building: What, why and how?', *Journal of Decision Systems*, pp. 1–14. doi:10.1080/12460125.2023.2197700.
- Paunonen, S.V. (2006) 'You are honest, therefore I like you and find you attractive', *Journal of Research in Personality*, 40(3), pp. 237–249. doi:10.1016/j.jrp.2004.12.003.
- Peer, E. *et al.* (2021) 'Data quality of platforms and panels for online behavioral research', *Behavior Research Methods*, 54(4), pp. 1643–1662. doi:10.3758/s13428-021-01694-3.
- Pennington, N. and Hall, J.A. (2020) 'Does facebook-enabled communication influence weak-tie relationships over time? A longitudinal investigation into mediated relationship maintenance', *Communication Monographs*, 88(1), pp. 48–70. doi:10.1080/03637751.2020.1854476.
- Penttinen, V., Ciuchita, R. and Čaić, M. (2022) 'YouTube it before you buy it: The role of Parasocial Interaction in consumer-to-consumer video reviews', *Journal of Interactive Marketing*, 57(4), pp. 561–582. doi:10.1177/10949968221102825.
- Perse, E.M. and Rubin, R.B. (1989) 'Attribution in social and Parasocial relationships', *Communication Research*, 16(1), pp. 59–77. doi:10.1177/009365089016001003.
- Peter, J.P. (1981) 'Construct validity: A review of basic issues and marketing practices', *Journal of Marketing Research*, 18(2), pp. 133–145. doi:10.1177/002224378101800201.
- Peterson, R., 2000. *Constructing effective questionnaires*. Thousand Oaks: CA: Sage Publications.
- Phua, J., Lin, J.-S. (Elaine) and Lim, D.J. (2018) 'Understanding consumer engagement with celebrity-endorsed e-cigarette advertising on Instagram', *Computers in Human Behavior*, 84, pp. 93–102. doi:10.1016/j.chb.2018.02.031.
- Podsakoff, P.M., MacKenzie, S.B. and Podsakoff, N.P. (2012) 'Sources of method bias in social science research and recommendations on how to control it', *Annual Review of Psychology*, 63(1), pp. 539–569. doi:10.1146/annurev-psych-120710-100452.

- Prathapan, M., Sahadevan, S. and K.A, Z. (2018) 'Effectiveness of Digital Marketing: Tourism Websites Comparative Analytics Based on AIDA Model', *Pramana Research Journal*, 8(5), pp. 15–26.
- Pressrove, G. and Pardun, C.J. (2016) 'Relationship between personal technology use and the donor/volunteer: A parasocial approach', *Journal of Promotion Management*, 22(1), pp. 137–150. doi:10.1080/10496491.2015.1107012.
- Qin, Y.S. (2020) 'Fostering brand–consumer interactions in social media: The role of social media uses and Gratifications', *Journal of Research in Interactive Marketing*, 14(3), pp. 337–354. doi:10.1108/jrim-08-2019-0138.
- Qualtrics (2024) *Strategic Market Research Software*. Available at: https://www.qualtrics.com/uk/strategy/research/ (Accessed: 17 March 2024).
- Quelhas-Brito, P. et al. (2020) 'Diffusing fashion information by Social Media Fashion influencers: Understanding antecedents and consequences', *Journal of Fashion Marketing and Management: An International Journal*, 24(2), pp. 137–152. doi:10.1108/jfmm-09-2019-0214.
- Razzaq, A., Shao, W. and Quach, S. (2023) 'Towards an understanding of meme marketing: Conceptualisation and empirical evidence', *Journal of Marketing Management*, pp. 1–32. doi:10.1080/0267257x.2022.2158906.
- Reinikainen, H. *et al.* (2020) "you really are a great big sister" parasocial relationships, credibility, and the moderating role of audience comments in Influencer marketing, *Journal of Marketing Management*, 36(3–4), pp. 279–298. doi:10.1080/0267257x.2019.1708781.
- Reinikainen, H. et al. (2021) 'Making and breaking relationships on social media: The impacts of Brand and influencer betrayals', *Technological Forecasting and Social Change*, 171, p. 120990. doi:10.1016/j.techfore.2021.120990.
- Richards, M.N. and Calvert, S.L. (2015) 'Toddlers' judgments of media character source credibility on Touchscreens', *American Behavioral Scientist*, 59(14), pp. 1755–1775. doi:10.1177/0002764215596551.
- Rietveld, R. *et al.* (2020) 'What you feel, is what you like influence of message appeals on customer engagement on Instagram', *Journal of Interactive Marketing*, 49(1), pp. 20–53. doi:10.1016/j.intmar.2019.06.003.
- Robinson, O.C. (2013) 'Sampling in interview-based qualitative research: A theoretical and practical guide', *Qualitative Research in Psychology*, 11(1), pp. 25–41. doi:10.1080/14780887.2013.801543.
- Robinson, O.C. (2023) 'Probing in qualitative research interviews: Theory and practice', *Qualitative Research in Psychology*, 20(3), pp. 382–397. doi:10.1080/14780887.2023.2238625.
- Rogers, E.M. and Bhowmik, D.K. (1970) 'Homophily-heterophily: Relational concepts for communication research', *Public Opinion Quarterly*, 34(4), p. 523. doi:10.1086/267838.

- Rossiter, J.R. (2002) 'The C-OAR-SE procedure for scale development in marketing', *International Journal of Research in Marketing*, 19(4), pp. 305–335. doi:10.1016/s0167-8116(02)00097-6.
- Rubin, A.M. and Perse, E.M. (1987) 'Audience activity and soap opera involvement a uses and effects investigation', *Human Communication Research*, 14(2), pp. 246–268. doi:10.1111/j.1468-2958.1987.tb00129.x.
- Rubin, A.M., Perse, E.M. and Powell, R.A. (1985) 'Loneliness, parasocial interaction, and local television news viewing', *Human Communication Research*, 12(2), pp. 155– 180. doi:10.1111/j.1468-2958.1985.tb00071.x.
- Ruiz-Mafe, C., Chatzipanagiotou, K. and Curras-Perez, R. (2018) 'The role of emotions and conflicting online reviews on consumers' purchase intentions', *Journal of Business Research*, 89, pp. 336–344. doi:10.1016/j.jbusres.2018.01.027.
- Ryan, A.B. (2006) 'Post-Positivist Approaches to Research', in *Researching and writing your thesis: A guide for postgraduate students*. Maynooth: NUI Maynooth.
- Ryan, R.M. and Deci, E.L. (2000) 'Self-determination theory and the facilitation of intrinsic motivation, social development, and well-being.', *American Psychologist*, 55(1), pp. 68–78. doi:10.1037//0003-066x.55.1.68.
- Ryu, G. and Feick, L. (2007) 'A penny for your thoughts: Referral reward programs and referral likelihood', *Journal of Marketing*, 71(1), pp. 84–94. doi:10.1509/jmkg.71.1.84.
- Sahaym, A., (Avi) Datta, A. and Brooks, S. (2021) 'Crowdfunding success through social media: Going beyond entrepreneurial orientation in the context of small and medium-sized enterprises', *Journal of Business Research*, 125, pp. 483–494. doi:10.1016/j.jbusres.2019.09.026.
- Sahin, S. and Baloglu, S. (2014) 'City branding: Investigating a brand advocacy model for distinct segments', *Journal of Hospitality Marketing & Management*, 23(3), pp. 239– 265. doi:10.1080/19368623.2013.779562.
- Sakib, M.N., Zolfagharian, M. and Yazdanparast, A. (2020) 'Does parasocial interaction with weight loss vloggers affect compliance? the role of vlogger characteristics, consumer readiness, and Health Consciousness', *Journal of Retailing and Consumer Services*, 52, p. 101733. doi:10.1016/j.jretconser.2019.01.002.
- Sangrador, J.L. and Yela, C. (2000) "what is beautiful is loved": Physical attractiveness in love relationships in a representative sample', *Social Behavior and Personality: an international journal*, 28(3), pp. 207–218. doi:10.2224/sbp.2000.28.3.207.
- Santos, Z.R. et al. (2022) 'Consumer engagement in Social Media Brand Communities: A literature review', International Journal of Information Management, 63, p. 102457. doi:10.1016/j.ijinfomgt.2021.102457.
- Santoso, I. *et al.* (2020) 'Is digital advertising effective under conditions of low attention?', *Journal of Marketing Management*, 36(17–18), pp. 1707–1730. doi:10.1080/0267257x.2020.1801801.

- Sanz-Blas, S., Bigné, E. and Buzova, D. (2017) 'M-WOM in a brand's Facebook fan page', Online Information Review, 41(7), pp. 936–953. doi:10.1108/oir-08-2016-0237.
- Sanz-Blas, S., Buzova, D. and Pérez-Ruiz, P. (2021) 'Building relational worth in an online social community through virtual structural embeddedness and relational embeddedness', *Technological Forecasting and Social Change*, 162, p. 120350. doi:10.1016/j.techfore.2020.120350.
- Sappleton, N. and Lourenço, F. (2015) 'Email subject lines and response rates to invitations to participate in a web survey and a face-to-face interview: The sound of silence', *International Journal of Social Research Methodology*, 19(5), pp. 611–622. doi:10.1080/13645579.2015.1078596.
- Saunders, M., Lewis, P. and Thornhill, A. (2023) 'Chapter 4: Understanding research philosophy and approaches to theory development', in *Research methods for business students*. Harlow: Pearson, pp. 128–170.
- Schwarz, N. (1999) 'Self-reports: How the questions shape the answers.', *American Psychologist*, 54(2), pp. 93–105. doi:10.1037/0003-066x.54.2.93.
- Seidman, G. and Miller, O.S. (2013) 'Effects of gender and physical attractiveness on visual attention to Facebook profiles', *Cyberpsychology, Behavior, and Social Networking*, 16(1), pp. 20–24. doi:10.1089/cyber.2012.0305.
- Shan, Y., Chen, K.-J. and Lin, J.-S. (Elaine) (2019) 'When Social Media influencers endorse brands: The effects of self-influencer congruence, parasocial identification, and perceived endorser motive', *International Journal of Advertising*, 39(5), pp. 590–610. doi:10.1080/02650487.2019.1678322.
- Sharma, V.M. and Klein, A. (2020) 'Consumer perceived value, involvement, trust, susceptibility to interpersonal influence, and intention to participate in online group buying', *Journal of Retailing and Consumer Services*, 52, p. 101946. doi:10.1016/j.jretconser.2019.101946.
- Shen, H. et al. (2022) 'The effect of hotel livestreaming on viewers' purchase intention: Exploring the role of parasocial interaction and emotional engagement', *International Journal of Hospitality Management*, 107, p. 103348. doi:10.1016/j.ijhm.2022.103348.
- Sheng, J., Lee, Y.H. and Lan, H. (2023) 'Parasocial relationships with micro-influencers: Do sponsorship disclosure and electronic word-of-mouth disrupt?', *Internet Research* [Preprint]. doi:10.1108/intr-12-2021-0903.
- Sheth, J.N. and Sisodia, R.S. (1999) 'Revisiting marketing's lawlike generalizations', *Journal of the Academy of Marketing Science*, 27(1), pp. 71–87. doi:10.1177/0092070399271006.
- Shimul, A.S., Phau, I. and Lwin, M. (2019) 'Conceptualising luxury brand attachment: Scale Development and validation', *Journal of Brand Management*, 26(6), pp. 675– 690. doi:10.1057/s41262-019-00158-6.

- Shoenberger, H. and Kim, E. (Anna) (2022) 'Explaining purchase intent via expressed reasons to follow an influencer, perceived homophily, and perceived authenticity', *International Journal of Advertising*, 42(2), pp. 368–383. doi:10.1080/02650487.2022.2075636.
- Siddaway, A.P., Wood, A.M. and Hedges, L.V. (2019) 'How to do a systematic review: A best practice guide for conducting and reporting narrative reviews, meta-analyses, and meta-syntheses', *Annual Review of Psychology*, 70(1), pp. 747–770. doi:10.1146/annurev-psych-010418-102803.
- Silva, J. et al. (2019) 'Antecedents of online purchase intention and behaviour: Uncovering unobserved heterogeneity', Journal of Business Economics and Management, 20(1), pp. 131–148. doi:10.3846/jbem.2019.7060.
- Skumanich, S.A. and Kintsfather, D.P. (1998) 'Individual Media Dependency Relations Within Television Shopping Programming', *Communication Research*, 25(2), pp. 200–219. doi:10.1177/009365098025002004.
- Snyder, H. (2019) 'Literature review as a research methodology: An overview and guidelines', *Journal of Business Research*, 104, pp. 333–339. doi:10.1016/j.jbusres.2019.07.039.
- So, J. and Nabi, R. (2013) 'Reduction of perceived social distance as an explanation for media's influence on personal risk perceptions: A test of the Risk Convergence Model', *Human Communication Research*, 39(3), pp. 317–338. doi:10.1111/hcre.12005.
- So, J. and Shen, L. (2016) 'Personalization of risk through convergence of self- and character-risk', *Communication Research*, 43(8), pp. 1094–1115. doi:10.1177/0093650215570656.
- Sokolova, K. and Kefi, H. (2020) 'Instagram and YouTube bloggers promote it, why should I buy? how credibility and Parasocial Interaction Influence Purchase Intentions', *Journal of Retailing and Consumer Services*, 53, p. 101742. doi:10.1016/j.jretconser.2019.01.011.
- Sokolova, K. and Perez, C. (2021) 'You follow fitness influencers on YouTube. but do you actually exercise? how parasocial relationships, and watching fitness influencers, relate to intentions to exercise', *Journal of Retailing and Consumer Services*, 58, p. 102276. doi:10.1016/j.jretconser.2020.102276.
- Song, H., Ruan, W.J. and Jeon, Y.J. (2021) 'An integrated approach to the purchase decision making process of food-delivery apps: Focusing on the tam and aida models', *International Journal of Hospitality Management*, 95, p. 102943. doi:10.1016/j.ijhm.2021.102943.
- Spiller, S.A. et al. (2013) 'Spotlights, floodlights, and the magic number zero: Simple effects tests in moderated regression', *Journal of Marketing Research*, 50(2), pp. 277–288. doi:10.1509/jmr.12.0420.
- Stephens, D.L., Hill, R.P. and Bergman, K. (1996) 'Enhancing the consumer-product relationship: Lessons from the QVC Home Shopping Channel', *Journal of Business Research*, 37(3), pp. 193–200. doi:10.1016/s0148-2963(96)00069-0.

- Stern, B., Zinkhan, G.M. and Jaju, A. (2001) 'Marketing images: Construct Definition, Measurement Issues, and Theory Development', *Marketing Theory*, 1(2), pp. 201– 224. doi:10.1177/147059310100100203.
- Stever, G. S., & Lawson, K. (2013). Twitter as a way for celebrities to communicate with fans: Implications for the study of parasocial interaction. *North American Journal of Psychology*, 15(2), pp.339–354.
- Stokes, D. and Bergin, R. (2006) 'Methodology or "methodolatry"? an evaluation of focus groups and depth interviews', *Qualitative Market Research: An International Journal*, 9(1), pp. 26–37. doi:10.1108/13522750610640530.
- Storm, L.K. et al. (2014) 'Influential relationships as contexts of learning and becoming elite: Athletes' retrospective interpretations', *International Journal of Sports Science* & Coaching, 9(6), pp. 1341–1356. doi:10.1260/1747-9541.9.6.1341.
- Su, N., John Mariadoss, B. and Reynolds, D. (2015) 'Friendship on social networking sites: Improving relationships between hotel brands and consumers', *International Journal of Hospitality Management*, 51, pp. 76–86. doi:10.1016/j.ijhm.2015.08.009.
- Sui, J. and Liu, C.H. (2009) 'Can beauty be ignored? effects of facial attractiveness on covert attention', *Psychonomic Bulletin & Review*, 16(2), pp. 276–281. doi:10.3758/pbr.16.2.276.
- Summers, J.O. (2001) 'Guidelines for conducting research and publishing in marketing: From conceptualization through the review process', *Journal of the Academy of Marketing Science*, 29(4), pp. 405–415. doi:10.1177/03079450094243.
- Sun, S., Rubin, A.M. and Haridakis, P.M. (2008) 'The role of motivation and media involvement in explaining internet dependency', *Journal of Broadcasting & Electronic Media*, 52(3), pp. 408–431. doi:10.1080/08838150802205595.
- Sun, T. (2010) 'Antecedents and Consequences of Parasocial Interaction with Sport Athletes and Identification with Sport Teams', *Journal of Sport Behavior*, 33(2), pp. 194–217.
- Sun, T. and Wu, G. (2012) 'Influence of personality traits on parasocial relationship with sports celebrities: A hierarchical approach', *Journal of Consumer Behaviour*, 11(2), pp. 136–146. doi:10.1002/cb.1378.
- Sundermann, G. and Munnukka, J. (2022) 'Hope you're not totally commercial! toward a better understanding of advertising recognition's impact on influencer marketing effectiveness', *Journal of Interactive Marketing*, 57(2), pp. 237–254. doi:10.1177/10949968221087256.
- Sung, Y. and Kim, J. (2010) 'Effects of brand personality on brand trust and brand affect', *Psychology & Marketing*, 27(7), pp. 639–661. doi:10.1002/mar.20349.
- Swani, K. and Labrecque, L.I. (2020) 'Like, comment, or share? self-presentation vs. Brand Relationships as drivers of social media engagement choices', *Marketing Letters*, 31(2–3), pp. 279–298. doi:10.1007/s11002-020-09518-8.

- Taillon, B.J. et al. (2020) 'Understanding the relationships between social media influencers and their followers: The moderating role of closeness', *Journal of Product & Brand Management*, 29(6), pp. 767–782. doi:10.1108/jpbm-03-2019-2292.
- Tasci, A.D.A. and Ko, Y.J. (2015) 'A fun-scale for understanding the hedonic value of a product: The destination context', *Journal of Travel & Tourism Marketing*, 33(2), pp. 162–183. doi:10.1080/10548408.2015.1038421.
- Teijlingen, van and Hundley, V. (2001) 'The importance of pilot studies', *Social research update* [Preprint], (35).
- Teng, L. *et al.* (2014) 'Effects of culturally verbal and visual congruency/incongruency across cultures in a competitive advertising context', *Journal of Business Research*, 67(3), pp. 288–294. doi:10.1016/j.jbusres.2013.05.015.
- Thorson, K.S. and Rodgers, S. (2006) 'Relationships between blogs as EWOM and interactivity, perceived interactivity, and parasocial interaction', *Journal of Interactive Advertising*, 6(2), pp. 5–44. doi:10.1080/15252019.2006.10722117.
- Tian, Y. and Yoo, J.H. (2015) 'Connecting with *the biggest loser*: An extended model of parasocial interaction and identification in health-related reality TV shows', *Health Communication*, 30(1), pp. 1–7. doi:10.1080/10410236.2013.836733.
- Torraco, R.J. (2005) 'Writing integrative literature reviews: Guidelines and examples', *Human Resource Development Review*, 4(3), pp. 356–367. doi:10.1177/1534484305278283.
- Torres, P., Augusto, M. and Matos, M. (2019) 'Antecedents and outcomes of Digital Influencer Endorsement: An exploratory study', *Psychology & Marketing*, 36(12), pp. 1267–1276. doi:10.1002/mar.21274.
- Traackr, D.V. (2024) *Global Influencer Market Size 2024, Statista.* Available at: https://www.statista.com/statistics/1092819/global-influencer-market-size/ (Accessed: 15 February 2024).
- Trafimow, D. (2018) 'An a priori solution to the replication crisis', *Philosophical Psychology*, 31(8), pp. 1188–1214. doi:10.1080/09515089.2018.1490707.
- Trafimow, D. (2019) 'A frequentist alternative to significance testing, p-values, and confidence intervals', *Econometrics*, 7(2), p. 26. doi:10.3390/econometrics7020026.
- Trafimow, D. *et al.* (2021) 'The harmful effect of null hypothesis significance testing on marketing research: An example', *Journal of Business Research*, 125, pp. 39–44. doi:10.1016/j.jbusres.2020.11.069.
- Trafimow, D., Hyman, M.R. and Kostyk, A. (2020) 'The (im)precision of scholarly consumer behavior research', *Journal of Business Research*, 114, pp. 93–101. doi:10.1016/j.jbusres.2020.04.008.
- Tranfield, D., Denyer, D. and Smart, P. (2003) 'Towards a methodology for developing evidence-informed management knowledge by means of systematic review', *British Journal of Management*, 14(3), pp. 207–222. doi:10.1111/1467-8551.00375.

- Tsai, W.-H.S. and Men, L.R. (2013) 'Motivations and antecedents of consumer engagement with Brand Pages on social networking sites', *Journal of Interactive Advertising*, 13(2), pp. 76–87. doi:10.1080/15252019.2013.826549.
- Tsai, W.-H.S. and Men, L.R. (2017) 'Consumer engagement with brands on social network sites: A cross-cultural Comparison of China and the USA', *Journal of Marketing Communications*, 23(1), pp. 2–21. doi:10.1080/13527266.2014.942678.
- Tsai, W.-H.S., Liu, Y. and Chuan, C.-H. (2021) 'How chatbots' social presence communication enhances consumer engagement: The mediating role of Parasocial Interaction and dialogue', *Journal of Research in Interactive Marketing*, 15(3), pp. 460–482. doi:10.1108/jrim-12-2019-0200.
- Tseng, T.H. and Wang, H.-Y. (2023) 'Consumer attitudes and purchase intentions towards internet celebrity self-brands: An integrated view of congruence, Brand Authenticity and internet celebrity involvement', *Journal of Product & Brand Management*, 32(6), pp. 863–877. doi:10.1108/jpbm-03-2022-3892.
- Tseng, T.H. et al. (2021) 'Success factors driving consumer reuse intention of Mobile Shopping Application Channel', *International Journal of Retail & Distribution* Management, 50(1), pp. 76–99. doi:10.1108/ijrdm-08-2020-0309.
- Tseng, T.H. et al. (2022) 'Success factors driving consumer reuse intention of Mobile Shopping Application Channel', International Journal of Retail & Distribution Management, 50(1), pp. 76–99. doi:10.1108/ijrdm-08-2020-0309.
- Tsfati, Y. *et al.* (2021) 'Political para-social relationship as a predictor of voting preferences in the Israeli 2019 elections', *Communication Research*, 49(8), pp. 1118–1147. doi:10.1177/00936502211032822.
- Tsiotsou, R.H. (2015) 'The role of social and parasocial relationships on social networking sites loyalty', *Computers in Human Behavior*, 48, pp. 401–414. doi:10.1016/j.chb.2015.01.064.
- Tsiotsou, R.H. (2016) 'The social aspects of consumption as predictors of consumer loyalty', *Journal of Service Management*, 27(2), pp. 91–116. doi:10.1108/josm-04-2015-0117.
- Tukachinsky, R. and Stever, G. (2018) 'Theorizing development of parasocial engagement', *Communication Theory*, 29(3), pp. 209–230. doi:10.1093/ct/qty032.
- Tukachinsky, R., Walter, N. and Saucier, C.J. (2020) 'Antecedents and effects of parasocial relationships: A meta-analysis', *Journal of Communication*, 70(6), pp. 868–894. doi:10.1093/joc/jqaa034.
- Ugwu, C.I., Ekere, J.N. and Onoh, C. (2021) 'RESEARCH PARADIGMS AND METHODOLOGICAL CHOICES IN THE RESEARCH PROCESS', Journal of applied Information Science and Technology, 14(2).
- Utz, S., Otto, F. and Pawlowski, T. (2021) "Germany crashes out of World Cup": A mixed-method study on the effects of crisis communication on Facebook', *Journal of Sport Management*, 35(1), pp. 44–54. doi:10.1123/jsm.2019-0430.

- Vazquez, D., Dennis, C. and Zhang, Y. (2017) 'Understanding the effect of Smart Retail Brand – Consumer Communications via Mobile Instant Messaging (MIM) – an empirical study in the Chinese context', *Computers in Human Behavior*, 77, pp. 425– 436. doi:10.1016/j.chb.2017.08.018.
- Vieira, V.A., Rafael, D.N. and Agnihotri, R. (2022) 'Augmented reality generalizations: A meta-analytical review on consumer-related outcomes and the mediating role of hedonic and utilitarian values', *Journal of Business Research*, 151, pp. 170–184. doi:10.1016/j.jbusres.2022.06.030.
- Vilnai-Yavetz, I. and Levina, O. (2018) 'Motivating social sharing of e-business content: Intrinsic motivation, extrinsic motivation, or crowding-out effect?', *Computers in Human Behavior*, 79, pp. 181–191. doi:10.1016/j.chb.2017.10.034.
- Vriens, M., Wedel, M. & Sandor, Z. (2001) 'Split-Questionnaire Designs: A New Tool in Survey Design and Panel Management', *Marketing research*, 13(2), pp. 14-19.
- Vriens, Marco and Eric Melton (2002) 'Managing Missing Data', *Marketing Research*, 14 (Fall), pp.12-17.
- Vrontis, D. et al. (2021) 'Social Media Influencer Marketing: A systematic review, Integrative Framework and future research agenda', *International Journal of Consumer Studies*, 45(4), pp. 617–644. doi:10.1111/ijcs.12647.
- Wacker, J.G. (2004) 'A theory of formal conceptual definitions: Developing theorybuilding Measurement Instruments', *Journal of Operations Management*, 22(6), pp. 629–650. doi:10.1016/j.jom.2004.08.002.
- Wahab, H.K. et al. (2022) 'Social media celebrities and New World Order. what drives purchasing behavior among social media followers?', *Journal of Retailing and Consumer Services*, 68, p. 103076. doi:10.1016/j.jretconser.2022.103076.
- Wallace, E., Buil, I. and de Chernatony, L. (2014) 'Consumer engagement with selfexpressive brands: Brand Love and WOM outcomes', *Journal of Product & Brand Management*, 23(1), pp. 33–42. doi:10.1108/jpbm-06-2013-0326.
- Walther, J.B. *et al.* (2008) 'The role of friends' appearance and behavior on evaluations of individuals on Facebook: Are we known by the company we keep?', *Human Communication Research*, 34(1), pp. 28–49. doi:10.1111/j.1468-2958.2007.00312.x.
- Wang, E.S.-T. and Hu, F.-T. (2021) 'Influence of self-disclosure of internet celebrities on normative commitment: The mediating role of para-social interaction', *Journal of Research in Interactive Marketing*, 16(2), pp. 292–309. doi:10.1108/jrim-09-2020-0194.
- Wang, S. (2023) 'Why do female audiences subscribe to these types of streamers? an empirical study on the motivations of Chinese huya users', *Frontiers in Psychology*, 14. doi:10.3389/fpsyg.2023.1247451.
- Wang, S. et al. (2016) 'Identifying search keywords for finding relevant social media posts', Proceedings of the AAAI Conference on Artificial Intelligence, 30(1). doi:10.1609/aaai.v30i1.10387.

- Wei, X. *et al.* (2022) 'Influencers as endorsers and followers as consumers: Exploring the role of parasocial relationship, congruence, and followers' identifications on consumer–brand engagement', *Journal of Interactive Advertising*, 22(3), pp. 269– 288. doi:10.1080/15252019.2022.2116963.
- Weijters, B., Cabooter, E. and Schillewaert, N. (2010) 'The effect of rating scale format on response styles: The number of response categories and response category labels', *International Journal of Research in Marketing*, 27(3), pp. 236–247. doi:10.1016/j.ijresmar.2010.02.004.
- Weng, L., Huang, Z. and Bao, J. (2021) 'A model of tourism advertising effects', *Tourism Management*, 85, p. 104278. doi:10.1016/j.tourman.2020.104278.
- Whang, C. and Im, H. (2021) "I like your suggestion!" the role of humanlikeness and parasocial relationship on the website Versus Voice Shopper's perception of recommendations', *Psychology & Marketing*, 38(4), pp. 581–595. doi:10.1002/mar.21437.
- Wicherts, J.M. *et al.* (2016) *Degrees of freedom in planning, running, analyzing, and reporting psychological studies a checklist to avoid P-hacking* [Preprint]. doi:10.31219/osf.io/umq8d.
- Wiedmann, K.-P. and von Mettenheim, W. (2020) 'Attractiveness, trustworthiness and expertise – social influencers' winning formula?', *Journal of Product & Brand Management*, 30(5), pp. 707–725. doi:10.1108/jpbm-06-2019-2442.
- Williams, A.L. and Merten, M.J. (2011) 'IFamily: Internet and social media technology in the family context', *Family and Consumer Sciences Research Journal*, 40(2), pp. 150–170. doi:10.1111/j.1552-3934.2011.02101.x.
- Wilson, R.T., Baack, D.W. and Till, B.D. (2015) 'Creativity, attention and the memory for brands: An outdoor advertising field study', *International Journal of Advertising*, 34(2), pp. 232–261. doi:10.1080/02650487.2014.996117.
- Wolfe, J.M. (2021) 'Guided search 6.0: An updated model of visual search', *Psychonomic Bulletin & Review*, 28(4), pp. 1060–1092. doi:10.3758/s13423-020-01859-9.
- Wong, I.H. et al. (2023) 'Social media celebrities' influence on youths' diet behaviors: A gender study based on the Aida Marketing Communication Model', Aslib Journal of Information Management [Preprint]. doi:10.1108/ajim-11-2022-0495.
- Wongkitrungrueng, A. and Assarut, N. (2020) 'The role of live streaming in Building Consumer Trust and engagement with Social Commerce Sellers', *Journal of Business Research*, 117, pp. 543–556. doi:10.1016/j.jbusres.2018.08.032.
- Wongkitrungrueng, A., Dehouche, N. and Assarut, N. (2020) 'Live Streaming Commerce from the sellers' perspective: Implications for online relationship marketing', *Journal of Marketing Management*, 36(5–6), pp. 488–518. doi:10.1080/0267257x.2020.1748895.
- Woods, M. *et al.* (2016) 'Advancing qualitative research using qualitative data analysis software (QDAS)? reviewing potential versus practice in published studies using

atlas.ti and NVIVO, 1994–2013', *Social Science Computer Review*, 34(5), pp. 597–617. doi:10.1177/0894439315596311.

- Wu, L.-Y. *et al.* (2014) 'Perceived value, transaction cost, and repurchase-intention in online shopping: A Relational Exchange perspective', *Journal of Business Research*, 67(1), pp. 2768–2776. doi:10.1016/j.jbusres.2012.09.007.
- Wu, X. and Lai, I.K. (2023) 'How the creativity and authenticity of destination short videos influence audiences' attitudes toward videos and destinations: The mediating role of emotions and the moderating role of parasocial interaction with internet celebrities', *Current Issues in Tourism*, pp. 1–20. doi:10.1080/13683500.2023.2229483.
- Xiang, L. et al. (2016) 'Exploring consumers' impulse buying behavior on Social Commerce Platform: The role of parasocial interaction', *International Journal of Information Management*, 36(3), pp. 333–347. doi:10.1016/j.ijinfomgt.2015.11.002.
- Xiao, M., Wang, R. and Chan-Olmsted, S. (2018) 'Factors affecting YouTube influencer marketing credibility: A heuristic-systematic model', *Journal of Media Business Studies*, 15(3), pp. 188–213. doi:10.1080/16522354.2018.1501146.
- Xiao, Y. and Watson, M. (2017) 'Guidance on conducting a systematic literature review', *Journal of Planning Education and Research*, 39(1), pp. 93–112. doi:10.1177/0739456x17723971.
- Xie, Q. and Feng, Y. (2023) 'How to strategically disclose sponsored content on Instagram? the synergy effects of two types of sponsorship disclosures in influencer marketing', *International Journal of Advertising*, 42(2), pp. 317–343. doi:10.1080/02650487.2022.2071393.
- Xin, B., Hao, Y. and Xie, L. (2023) 'Strategic product showcasing mode of e-commerce live streaming', *Journal of Retailing and Consumer Services*, 73, p. 103360. doi:10.1016/j.jretconser.2023.103360.
- Xu, Z. *et al.* (2020) "'I'm like you, and I like what you like" sustainable food purchase influenced by vloggers: A moderated serial-mediation model', *Journal of Retailing and Consumer Services*, 63, p. 102737. doi:10.1016/j.jretconser.2021.102737.
- Yang, S. *et al.* (2018) 'Perceived values on mobile GMS continuance: A perspective from perceived integration and interactivity', *Computers in Human Behavior*, 89, pp. 16– 26. doi:10.1016/j.chb.2018.07.032.
- Yang, Y. and Ha, L. (2021) 'Why people use TikTok (Douyin) and how their purchase intentions are affected by social media influencers in China: A uses and gratifications and Parasocial Relationship Perspective', *Journal of Interactive Advertising*, 21(3), pp. 297–305. doi:10.1080/15252019.2021.1995544.
- Yasa, N.N. et al. (2020) 'Service strategy based on Tri Kaya Parisudha, social media promotion, Business Values and business performance', *Management Science Letters*, pp. 2961–2972. doi:10.5267/j.msl.2020.5.029.

- Ye, F. et al. (2024) 'Influencer selection and strategic analysis for live streaming selling', Journal of Retailing and Consumer Services, 77, p. 103673. doi:10.1016/j.jretconser.2023.103673.
- Yılmazdoğan, O.C., Doğan, R.Ş. and Altıntaş, E. (2021) 'The impact of the source credibility of Instagram influencers on travel intention: The mediating role of Parasocial Interaction', *Journal of Vacation Marketing*, 27(3), pp. 299–313. doi:10.1177/1356766721995973.
- Yoo, K. et al. (2015) 'Trends in the use of statistics in major advertising journals over four decades', International Journal of Advertising, 34(3), pp. 549–572. doi:10.1080/02650487.2015.1005513.
- Youn, S. and Jin, S.V. (2021) "In A.I. we trust?" the effects of parasocial interaction and technopian versus luddite ideological views on chatbot-based customer relationship management in the emerging "Feeling economy", *Computers in Human Behavior*, 119, p. 106721. doi:10.1016/j.chb.2021.106721.
- Young, A.F., Gabriel, S. and Hollar, J.L. (2013) 'Batman to the rescue! the protective effects of parasocial relationships with muscular superheroes on men's Body image', *Journal of Experimental Social Psychology*, 49(1), pp. 173–177. doi:10.1016/j.jesp.2012.08.003.
- Yousaf, S. (2022) 'Food vloggers and Parasocial Interactions: A comparative study of local and foreign food vlogs using the S-O-R paradigm', *International Journal of Contemporary Hospitality Management*, 34(9), pp. 3525–3549. doi:10.1108/ijchm-09-2021-1090.
- Yuan, C. et al. (2021) 'Study on the influencing of B2B parasocial relationship on repeat purchase intention in the online purchasing environment: An empirical study of B2B e-commerce platform', *Industrial Marketing Management*, 92, pp. 101–110. doi:10.1016/j.indmarman.2020.11.008.
- Yuan, C. et al. (2021) 'The influence of flow experience in the augmented reality context on psychological ownership', *International Journal of Advertising*, 40(6), pp. 922– 944. doi:10.1080/02650487.2020.1869387.
- Yuan, C.L. et al. (2021) 'The influence of parasocial relationship in Fashion Web on Customer Equity', Journal of Business Research, 130, pp. 610–617. doi:10.1016/j.jbusres.2019.08.039.
- Yuan, C.L., Kim, J. and Kim, S.J. (2016) 'Parasocial relationship effects on customer equity in the Social Media Context', *Journal of Business Research*, 69(9), pp. 3795– 3803. doi:10.1016/j.jbusres.2015.12.071.
- Yuan, S. and Lou, C. (2020) 'How social media Influencers Foster relationships with followers: The roles of source credibility and fairness in parasocial relationship and product interest', *Journal of Interactive Advertising*, 20(2), pp. 133–147.
- Yuksel, M. and Labrecque, L.I. (2016) "Digital Buddies": Parasocial Interactions in social media', *Journal of Research in Interactive Marketing*, 10(4), pp. 305–320. doi:10.1108/jrim-03-2016-0023.

- Yusuf, A.S., Che Hussin, A.R. and Busalim, A.H. (2018) 'Influence of E-wom engagement on consumer purchase intention in Social Commerce', *Journal of Services Marketing*, 32(4), pp. 493–504. doi:10.1108/jsm-01-2017-0031.
- Zafar, A.U., Qiu, J. and Shahzad, M. (2020) 'Do digital celebrities' relationships and social climate matter? impulse buying in F-Commerce', *Internet Research*, 30(6), pp. 1731–1762. doi:10.1108/intr-04-2019-0142.
- Zaichkowsky, J.L. (1985) 'Measuring the involvement construct', *Journal of Consumer Research*, 12(3), p. 341. doi:10.1086/208520.
- Zaichkowsky, J.L. (1986) 'Conceptualizing involvement', *Journal of Advertising*, 15(2), pp. 4–34. doi:10.1080/00913367.1986.10672999.
- Zaltman, G., LeMasters, K. and Heffring, M. (1982) 'Chapter 5: Deductive and inductive thinking', in *Theory construction in marketing: Some thoughts on thinking*. New York: Wiley.
- Zeithaml, V.A. (1988) 'Consumer perceptions of price, quality, and value: A means-end model and synthesis of evidence', *Journal of Marketing*, 52(3), p. 2. doi:10.2307/1251446.
- Zeng, F., Huang, L. and Dou, W. (2009) 'Social factors in user perceptions and responses to advertising in online social networking communities', *Journal of Interactive Advertising*, 10(1), pp. 1–13. doi:10.1080/15252019.2009.10722159.
- Zhang, C. and Conrad, F. (2014) 'Speeding in Web Surveys: The tendency to answer very fast and its association with straightlining'. *Survey Research Methods*, 8(2), pp.127– 135. doi:10.18148/srm/2014.v8i2.5453
- Zhang, C.-B. et al. (2022) 'Effect of WeChat interaction on Brand Evaluation: A moderated mediation model of para-social interaction and affiliative tendency', *Journal of Retailing and Consumer Services*, 64, p. 102812. doi:10.1016/j.jretconser.2021.102812.
- Zhang, H. et al. (2015) 'The impacts of technological environments and co-creation experiences on customer participation', *Information & Management*, 52(4), pp. 468– 482. doi:10.1016/j.im.2015.01.008.
- Zhang, H., Yang, Y. and Zhao, J. (2023) 'Does game-irrelevant chatting stimulate highvalue gifting in live streaming? A session-level perspective', *Computers in Human Behavior*, 138, p. 107467. doi:10.1016/j.chb.2022.107467.
- Zhang, S. et al. (2020) 'Expressive brand relationship, Brand Love, and brand loyalty for Tablet pcs: Building a sustainable brand', Frontiers in Psychology, 11. doi:10.3389/fpsyg.2020.00231.
- Zhang, W. (2022) 'Understanding gift-giving in game live streaming on Douyu: An evaluation of PSR/Social Presence', *Frontiers in Psychology*, 13. doi:10.3389/fpsyg.2022.953484.

- Zhang, X. and Choi, J. (2022) 'The importance of social influencer-generated contents for user cognition and emotional attachment: An information relevance perspective', *Sustainability*, 14(11), p. 6676. doi:10.3390/su14116676.
- Zhao, X., Lynch, J.G. and Chen, Q. (2010) 'Reconsidering baron and Kenny: Myths and truths about mediation analysis', *Journal of Consumer Research*, 37(2), pp. 197–206. doi:10.1086/651257.
- Zheng, L. *et al.* (2023) 'The role of social media followers' agency in Influencer Marketing: A study based on the heuristic–systematic model of information processing', *International Journal of Advertising*, 43(3), pp. 554–579. doi:10.1080/02650487.2023.2229148.
- Zheng, X. et al. (2020) 'Role of technology attraction and Parasocial Interaction in social shopping websites', *International Journal of Information Management*, 51, p. 102043. doi:10.1016/j.ijinfomgt.2019.102043.
- Zhong, Y., Shapoval, V. and Busser, J. (2021) 'The role of parasocial relationship in social media marketing: Testing a model among Baby Boomers', *International Journal of Contemporary Hospitality Management*, 33(5), pp. 1870–1891. doi:10.1108/ijchm-08-2020-0873.
- Zhou, F. and Jia, W. (2018) 'How a retailer's website Quality Fosters Relationship Quality: The mediating effects of Parasocial Interaction and psychological distance', *International Journal of Human–Computer Interaction*, 34(1), pp. 73–83. doi:10.1080/10447318.2017.1328006.
- Zhu, J. and Liu, W. (2020) 'A tale of two databases: The use of web of science and scopus in academic papers', *Scientometrics*, 123(1), pp. 321–335. doi:10.1007/s11192-020-03387-8.
- Zhu, J. et al. (2023) 'Richness and dynamics: How to improve virtual reality tourism adoption with virtual social clues', *Journal of Research in Interactive Marketing*, 18(1), pp. 142–158. doi:10.1108/jrim-09-2022-0298.
- Zimmer, F., Scheibe, K. and Stock, W.G. (2018) 'A model for Information Behavior Research on Social Live Streaming Services (slsss)', *Lecture Notes in Computer Science*, pp. 429–448. doi:10.1007/978-3-319-91485-5\_33.
- Zinkhan, G.M. and Hirschheim, R. (1992) 'Truth in marketing theory and research: An alternative perspective', *Journal of Marketing*, 56(2), p. 80. doi:10.2307/1252044.
- Zoom (2023) *Meet face-to-face from any device, Zoom*. Available at: https://www.zoom.com/en/products/virtual-meetings/ (Accessed: 23 October 2023).